The semiconductor beta gallium oxide $(\beta-Ga_2O_3)$ combines a 4.6–4.9 eV wide band gap with the availability of melt-grown wafers and could help meet the growing demand for high-efficiency and low-cost power electronics. However, research mostly focuses on basic device structures, and the low thermal conductivity raises concerns about potential thermal management problems. The objective of this work is to transition from fundamental device structure research to power electronics applications by experimentally and through simulation investigating the electrical and thermal characteristics of novel β -Ga₂O₃ diodes from a planned production line. A change of the conduction mechanism leading to an initial decrease and then increase of the conduction losses with rising temperature is observed for multiple but not all diodes. This seems to originate from the device processing rather than the intrinsic properties of β -Ga₂O₃. Despite a strong variation of the material properties between diodes of the same type. a lower increase in differential on-resistance with rising temperature is observed compared to silicon carbide (SiC) diodes, and measurements of the temperaturedependent ideality factors and Schottky barrier heights indicate stable junction properties. The heat dissipation in Ga_2O_3 diodes can be improved by thinning the currently 600 µm thick standard devices to thicknesses of 200 µm. In contrast to SiC, however, cooling the devices from the junction side is found to be significantly more effective in reducing the junction temperature despite the smaller cooling area, if the entire anode area is covered with die-attach material. Combined with the potentially low conduction losses, it seems realistic for future Ga_2O_3 diodes to achieve similar junction temperatures as modern commercial SiC diodes at the same forward current. Even 600 µm thick diodes are successfully implemented in a 400 V buck converter operated at frequencies up to 350 kHz. Peak voltage slew rates exceeding 100 V/ns are achieved, but in continuous operation the diodes exhibit a higher temperature rise than their SiC counterparts. Nevertheless, the efficiencies with a state-of-the-art silicon diode of similar size can be surpassed with the Ga_2O_3 diodes despite the still higher on-resistance, owing to the absence of recombination losses.

Florian Wilhelmi: Gallium oxide (Ga2O3) Schottky diodes for power electronics

Res Electricae Magdeburgenses Magdeburger Forum zur Elektrotechnik



Florian Wilhelmi

On the characteristic electrical and thermal properties of beta-phase gallium oxide (β -Ga₂O₃) Schottky diodes in view of their potential application in power electronics



ISSN 1612-2526 ISBN 978-3-948749-49-1 DOI 10.24352/UB.OVGU-2024-087 MAFO 101

Magdeburg 2024

MAFO Band 101 On the characteristic electrical and thermal properties of beta-phase gallium oxide (β -Ga₂O₃) Schottky diodes in view of their potential application in power electronics

Dissertation

zur Erlangung des akademischen Grades

Doktoringenieur

(Dr.-Ing.)

von M.Sc. Florian Wilhelmi

geb. am 04.01.1994 in Trier

genehmigt durch die Fakultät für Elektrotechnik und Informationstechnik der Otto-von-Guericke-Universität Magdeburg

Gutachter:

Prof. Dr.-Ing. Andreas Lindemann

Prof. Dr.-Ing. Sibylle Dieckerhoff

Promotionskolloquium am 02. Juli 2024

Res Electricae Magdeburgenses

Magdeburger Forum zur Elektrotechnik, Jg. 22, Band 101, 2024 http://www.mafo.ovgu.de/

IMPRESSUM

Herausgeber:

- Prof. Dr.-Ing. Andreas Lindemann, Lehrstuhl für Leistungselektronik, Institut für Elektrische Energiesysteme
- Prof. Dr.-Ing. habil. Martin Wolter, Lehrstuhl für Elektrische Netze und Erneuerbare Energie, Institut für Elektrische Energiesysteme
- Prof. Dr. rer. nat. Georg Rose, Lehrstuhl für Medizinische Telematik/Medizintechnik, Institut für Medizintechnik
- Prof. Dr.-Ing. Ralf Vick, Lehrstuhl für Elektromagnetische Verträglichkeit, Institut für Medizintechnik

Gründungsherausgeber:

- Prof. Dr. rer. nat. habil. Jürgen Nitsch
- Prof. Dr.-Ing. habil. Zbigniew Antoni Styczynski

alle: Otto-von-Guericke-Universität Magdeburg Postfach 4120, 39016 Magdeburg

V. i. S. d. P.:

Dr.-Ing. Florian Wilhelmi Otto-von-Guericke-Universität Magdeburg, Postfach 4120, 39016 Magdeburg

1. Auflage, Magdeburg, Otto-von-Guericke-Universität, 2024 Zugl.: Magdeburg, Univ., Diss., 2024

Auflage: 50

Redaktionsschluss: August 2024

ISSN: 1612-2526 ISBN: 978-3-948749-49-1 DOI: 10.24352/UB.OVGU-2024-087

© Copyright 2024 Florian Wilhelmi

Bezug über die Herausgeber

Druck: docupoint GmbH Otto-von-Guericke-Allee 14, 39179 Barleben

Declaration of Authorship

I hereby declare that I produced this thesis without prohibited external assistance and that none other than the listed references and tools have been used. I did not make use of any commercial consultant concerning graduation. A third party did not receive any nonmonetary perquisites neither directly nor indirectly for activities which are connected with the contents of the presented thesis.

All sources of information are clearly marked, including my own publications.

In particular I have not consciously:

- Fabricated data or rejected undesired results
- Misused statistical methods with the aim of drawing other conclusions than those warranted by the available data
- Plagiarized data or publications
- Presented the results of other researchers in a distorted way

I do know that violations of copyright may lead to injunction and damage claims of the author and also to prosecution by the law enforcement authorities. I hereby agree that the thesis may need to be reviewed with an electronic data processing for plagiarism.

This work has not yet been submitted as a doctoral thesis in the same or a similar form in Germany or in any other country. It has not yet been published as a whol

Friedrichshafen, 05.12.2023, _____

Florian Wilhelmi

Abstract

The semiconductor beta gallium oxide (β -Ga₂O₃) combines a 4.6-4.9 eV wide band gap with the availability of melt-grown wafers and could help meet the growing demand for high-efficiency and low-cost power electronics. However, research still mostly focuses on basic device structures, and the low thermal conductivity raises concerns about potential thermal management problems. The objective of this work is to transition from fundamental device structure research to power electronics applications by experimentally and through simulation investigating the electrical and thermal characteristics of novel β -Ga₂O₃ diodes from a planned production line. A change of the conduction mechanism leading to an initial decrease and then increase of the conduction losses with rising temperature is observed for multiple but not all diodes. This seems to originate from the device processing rather than the intrinsic properties of β -Ga₂O₃. Despite a strong variation of the material properties between diodes of the same type, a lower increase in differential on-resistance with rising temperature is observed compared to silicon carbide (SiC) diodes, and measurements of the temperature-dependent ideality factors and Schottky barrier heights indicate stable junction properties. The heat dissipation in Ga₂O₃ diodes can be improved by thinning the currently $600 \,\mu\text{m}$ thick standard devices to thicknesses of $200 \,\mu\text{m}$. In contrast to SiC, however, cooling the devices from the junction side is found to be significantly more effective in reducing the junction temperature despite the smaller cooling area, if the entire anode area is covered with die-attach material. Combined with the potentially low conduction losses, it seems realistic for future Ga₂O₃ diodes to achieve similar junction temperatures as modern commercial SiC diodes at the same forward current. Even 600 µm thick diodes are successfully implemented in a 400 V buck converter operated at frequencies up to 350 kHz. Peak voltage slew rates exceeding $100 \,\mathrm{V/ns}$ are achieved, but in continuous operation the diodes exhibit a higher temperature rise than their SiC counterparts. Nevertheless, the efficiencies with a state-of-theart silicon diode of similar size can be surpassed with the Ga₂O₃ diodes despite the still higher on-resistance, owing to the absence of recombination losses.

Zusammenfassung

Der Halbleiter beta-Galliumoxid (β -Ga₂O₃) vereint eine 4.6–4.9 eV breite Bandlücke mit der Verfügbarkeit schmelzgewachsener Wafer und könnte dazu beitragen, den steigenden Bedarf an hocheffizienter und kostengünstiger Leistungselektronik zu decken. Jedoch konzentriert sich die Forschung derzeit noch hauptsächlich auf die grundlegenden Strukturen der Bauelemente. Außerdem weckt die geringe thermische Leitfähigkeit von Ga₂O₃ Bedenken hinsichtlich möglicher Probleme bei der Wärmeabfuhr. Ziel dieser Arbeit ist, einen Übergang von der bisherigen Forschung auf Bauelementestruktur-Ebene hin zur Anwendung in der Leistungselektronik zu schaffen, indem elektrische und thermische Eigenschaften neuartiger β -Ga₂O₃ Dioden aus einer geplanten Fertigungslinie experimentell und simulativ untersucht werden. Bei mehreren β -Ga₂O₃ Dioden ist eine Änderung des Leitmechanismus zu beobachten, die zu einer anfänglichen Abnahme und dann zu einem Anstieg der Leitungsverluste mit steigender Temperatur führt. Dies scheint jedoch auf die Herstellung der Chips zurückzuführen zu sein und nicht direkt auf die intrinsischen Eigenschaften von β -Ga₂O₃. Trotz einer starken Variation der Materialeigenschaften zwischen Dioden desselben Typs wird ein geringerer Anstieg des differentiellen Durchlasswiderstands mit steigender Temperatur im Vergleich zu Siliziumkarbid (SiC) Dioden beobachtet, und Messungen der temperaturabhängigen Idealitätsfaktoren und Schottky-Barrierenhöhen weisen auf stabile Sperrschicht- bzw. Grenzflächeneigenschaften hin. Die Entwärmung von Ga₂O₃ Dioden kann durch Abdünnen der derzeit 600 µm dicken Standardbauteile auf Dicken von 200 µm signifikant verbessert werden. Im Gegensatz zu SiC erweist sich jedoch die sperrschichtseitige Kühlung trotz der kleineren Kühlfläche als wesentlich effektiver bei der Senkung der Bauteiltemperatur, wenn die gesamte Anodenfläche mit Lot- bzw. Sinterpaste bedeckt ist. In Verbindung mit den potenziell geringen Leitungsverlusten scheint es realistisch, dass künftige Ga₂O₃ Dioden bei gleichem Durchlassstrom ähnliche Sperrschichttemperaturen erreichen wie moderne kommerzielle SiC Dioden. Selbst 600 µm dicke Dioden werden erfolgreich in einem 400 V Abwärtswandler eingesetzt, der bei Schaltfrequenzen von bis zu 350 kHz betrieben wird. Es werden maximale Anstiegsgeschwindigkeiten der Spannung von über $100 \,\mathrm{V/ns}$ erreicht, aber im Dauerbetrieb weisen die Ga₂O₃ Dioden einen höheren Temperaturanstieg auf als die SiC Vergleichsbauteile. Dennoch können mit den Ga2O3 Freilaufdioden trotz des aktuell höheren Durchlasswiderstandes aufgrund der fehlenden Rekombinationsverluste die Effizienzen mit einer modernen Silizium-Diode ähnlicher Größe übertroffen werden.

Contents

Nomenclature

| No | Nomenclature | | | | |
|----|---|--|--|----|--|
| 1 | Introduction | | | | |
| | 1.1 | Motiva | ation and research goals | 1 | |
| | 1.2 | Struct | are of this thesis | 3 | |
| 2 | Basics and state-of-the-art of gallium oxide power electronics | | | | |
| | 2.1 | Chapte | er overview | 5 | |
| | 2.2 | Basics | of power electronics | 5 | |
| | | 2.2.1 | Basic principles of power conversion | 5 | |
| | | 2.2.2 | Semiconductor power devices | 6 | |
| | | 2.2.3 | Physical basics of Schottky diodes | 9 | |
| | 2.3 | Galliu | m oxide as power semiconductor | 11 | |
| | | 2.3.1 | Important figures of merit | 11 | |
| | | 2.3.2 | Material properties | 13 | |
| | | 2.3.3 | Growth and metal-semiconductor contacts | 16 | |
| | | 2.3.4 | Diode concepts | 18 | |
| | | 2.3.5 | Transistor concepts | 21 | |
| | 2.4 | Summ | ary and basic research questions arising | 22 | |
| 3 | Experimental methods: sample preparation, measurement, and evaluation | | | 25 | |
| | 3.1 | Chapte | er overview | 25 | |
| | 3.2 | Packag | ging and overview of the investigated samples | 25 | |
| | 3.3 | Evalua | ation of static electrical properties | 29 | |
| | 3.4 | Captur | ring nanosecond switching transients | 32 | |
| | | 3.4.1 | General basics of double pulse testing | 32 | |
| | | 3.4.2 | Experimental procedure to assess novel current sensors | 33 | |
| | | 3.4.3 | Exemplary analysis of switching waveforms | 35 | |
| | | 3.4.4 | Simulative validation of measured waveforms | 40 | |
| | | 3.4.5 | Comparison of characteristic quantities | 42 | |
| | | 3.4.6 | Sensor selection and fast switching with custom package | 45 | |
| | 3.5 | Evalua | ation of transient thermal properties | 46 | |
| | 3.6 | Chapte | er conclusion | 49 | |
| 4 | Basi | Basic characteristics of gallium oxide Schottky diodes | | | |
| | 4.1 | Chapte | er overview | 51 | |
| | 4.2 | Basic | electrical properties | 51 | |
| | | 4.2.1 | Forward conduction characteristics and junction properties | 51 | |
| | | 4.2.2 | Reverse blocking characteristics | 57 | |
| | | 4.2.3 | In-plane variation of electric properties across the wafer | 62 | |
| | | 4.2.4 | Low increase in on-resistance with rising temperature | 65 | |
| | 4.3 | Basic | thermal considerations | 67 | |
| | | | | 60 | |

| | 4.5 | Summary and questions arising | . 71 |
|------------------|---------|---|-------|
| 5 | The | rmal properties and methods to improve the heat dissipation | 73 |
| | 5.1 | Chapter overview | . 73 |
| | 5.2 | Heat dissipation in TO247 packaged diodes with thinned substrate | . 73 |
| | | 5.2.1 Setup and investigated samples | . 73 |
| | | 5.2.2 Transient temperature response and potential measurement issues . | . 75 |
| | | 5.2.3 Thermal structure functions, impedance profiles, and their variation | . 76 |
| | | 5.2.4 Parametric studies to identify critical parameters | . 78 |
| | 5.3 | Direct comparison of substrate thinning and junction-side cooling | . 82 |
| | | 5.3.1 Overview and investigated samples | . 82 |
| | | 5.3.2 Differences in the thermal structure | . 84 |
| | | 5.3.3 Impact on power and current rating | . 86 |
| | | 5.3.4 Simulative validation and methods to improve junction-side cooling | . 91 |
| | | 5.3.5 How conduction losses can affect the assembly strategy | . 97 |
| | 5.4 | Chapter conclusion | . 100 |
| 6 | Арр | lication of gallium oxide power diodes in a buck converter | 103 |
| | 6.1 | Chapter overview | . 103 |
| | 6.2 | Experimental setup and investigated samples | . 103 |
| | 6.3 | Investigating 600 µm thick TO247 packaged diodes with different chip sizes | . 105 |
| | | 6.3.1 Static electrical properties relevant for switching operation | . 105 |
| | | 6.3.2 Analysis of the basic switching waveforms | . 107 |
| | | 6.3.3 Buck converter operation under varying conditions | . 114 |
| | 6.4 | Investigating 200 µm thin diodes with different chip sizes assembled on ce- | |
| | | ramic substrates | . 117 |
| | | 6.4.1 Electrical properties relevant for converter operation | . 118 |
| | | 6.4.2 Buck converter operation under varying conditions | . 119 |
| | 6.5 | Chapter conclusion | . 122 |
| 7 | Sum | mary and outlook | 123 |
| Appendix | | 125 | |
| | А | Overview of material properties of different semiconductors | . 125 |
| | В | List of measurement equipment | . 125 |
| | С | List of used software | . 127 |
| Li | st of I | igures | 127 |
| List of Tables 1 | | | 131 |
| References 1 | | | 135 |

Acronyms and abbreviations

| 2DEG | two-dimendional electron gas |
|-----------|--|
| 3D | three-dimensional |
| Ga_2O_3 | gallium oxide |
| AC | alternating current |
| Al_2O_3 | aluminium oxide |
| AlN | aluminum nitride |
| B1B4 | batch 1batch 4 of the studied diodes |
| C200 | medium large c athode-side cooled $200 \mu m$ thin Ga_2O_3 die |
| C600 | medium large c athode-side cooled $600 \mu m$ thin Ga_2O_3 die |
| C600LL | largest cathode-side cooled $600 \mu m$ thin Ga_2O_3 die |
| CBL | current blocking layer |
| CSC | cathode-side cooled, cathode-side cooling |
| CuMo | copper molybdenum |
| CVD | chemical vapor deposition |
| DC | direct current |
| DPC | direct plated copper |
| EFG | edge-defined film-fed growth |
| FE | field emission |
| FET | field-effect transistor |
| FOM | figure of merit |
| FP | field plate |
| GaN | gallium nitride |
| GR | guard ring |
| HEMT | high-electron-mobility transistor |
| HfO_2 | hafnium(IV) oxide |
| HV | high voltage |
| HVPE | halide vapor phase epitaxy |
| IGBT | insulated-gate bipolar transistor |
| J200 | medium large junction-side cooled $200\mu\mathrm{m}$ thin $\mathrm{Ga_2O_3}$ die |
| JSC | junction-side cooled, junction-side cooling |
| L | second largest anode size of the diodes under investigation |
| LL | largest anode size of the diodes under investigation |
| MBE | molecular beam epitaxy |
| ML | medium large anode size of the diodes under investigation |
| MOCVD | metalorganic chemical vapor deposition |

| MOS | metal-oxide-semiconductor |
|-----------------------------|---|
| MOSFET | metal-oxide-semiconductor field-effect transistor |
| MPS | merged PiN Schottky |
| MS | metal-semiconductor |
| NiO | nickel oxide |
| PCB | printed circuit board |
| PFC | power factor correction |
| PFE | Poole–Frenkel emission |
| PFOM | power figure of merit |
| RMS | root mean square |
| S377 | cathode-side cooled $377\mu\mathrm{m}$ thin SiC reference die |
| SBD | Schottky barrier diode |
| Si | silicon |
| $\mathrm{Si}_3\mathrm{N}_4$ | silicon nitride |
| SiC | silicon carbide |
| SiO_2 | silicon(IV) oxide |
| SMD | surface-mount device |
| SMT | surface-mount technology |
| TAT | trap assisted tunneling |
| TE | thermionic emission |
| TEOS | te traeth o xy s ilane |
| TFE | thermionic field emission |
| TIM | thermal interface material |
| TSP | temperature sensitive parameter |
| XS | smallest anode size of the diodes under investigation |

Greek symbols

| α | alpha polymorph of a semiconductor |
|----------------------------|---|
| α_{ref} | temperature coefficient of the differential on-resistance in relation to a certain reference temperature |
| $\alpha_{\rm Z}$ | fitting variable to determine the junction-to-case thermal resistance |
| $lpha_{300}$ | temperature coefficient of the differential on-resistance related to $300 \mathrm{K}$ |
| $lpha_{373}$ | temperature coefficient of the differential on-resistance related to $373\mathrm{K}$ |
| β | beta polymorph of a semiconductor |
| β_{Z} | fitting variable to determine the junction-to-case thermal resistance |
| Δ | change of a quantity |
| $\delta_{ m Z}(Z_{ m th})$ | exponential trendline to determine the junction-to-case thermal resistance |

| $\epsilon_{\rm s}$ | relative dielectric constant (low-frequency) | |
|-----------------------|---|--|
| ϵ_0 | vacuum permittivity | F/m |
| ϵ_{∞} | relative dielectric constant (high-frequency) | |
| $\eta_{ m p}$ | power-loop efficiency | % |
| $\lambda_{ m th}$ | thermal conductivity | $\mathrm{Wm^{-1}K^{-1}}$ |
| $\mu_{ m e}$ | electron mobility | ${\rm cm}^2 {\rm V}^{-1} {\rm s}^{-1}$ |
| $\phi_{\mathrm{b},0}$ | Schottky barrier height | eV |
| $\phi_{ m if,0}$ | image force lowering | eV |
| $\Phi_{\rm M}$ | work function of a metal | eV |
| ψ_{bi} | built-in potential | V |
| ρ | mass density | $ m kg/m^3$ |
| $	au_{\mathrm{i}}$ | time constant of the i-th component | S |
| $\zeta_{ m fs}$ | forward model shift parameter | |
| $\zeta_{ m rb}$ | reverse model bending parameter | |
| $\zeta_{ m ri}$ | reverse model inclination parameter | |
| $\zeta_{ m rs}$ | reverse model shift parameter | |
| | | |

Latin symbols

| A | active area | m^2 |
|---------------------|--|--|
| $A_{\rm ch, eff}$ | effective chip size | m^2 |
| $a_{\rm i}(t)$ | step-function response of the i-th component | |
| a(t) | step-function response | |
| A^{**} | Richardson constant | $\mathrm{A}\mathrm{m}^{-2}\mathrm{K}^{-2}$ |
| $B_{\rm i}$ | bandwidth of the i-th measurement instrument | Hz |
| $B_{\rm osci}$ | bandwidth of an oscilloscope | Hz |
| C | capacitance | \mathbf{F} |
| $C_{\rm ds}$ | drain-source capacitance | \mathbf{F} |
| $C_{\rm gd}$ | gate-drain capacitance | \mathbf{F} |
| $C_{\rm gs}$ | gate-source capacitance | \mathbf{F} |
| $C_{\mathbf{j}}$ | junction capacitance of a diode | \mathbf{F} |
| $C_{\rm j0}$ | junction capacitance of a diode at zero bias voltage | \mathbf{F} |
| C_{th} | thermal capacitance | Ws/K |
| $C_{\mathrm{th,i}}$ | thermal capacitance of the i-th component | Ws/K |
| $c_{\rm th}$ | specific heat capacity | $\rm Jkg^{-1}K^{-1}$ |
| D_1 | diode | |
| $E_{\rm br}$ | critical electric field or breakdown field | V/m |
| $E_{\mathbf{C}}$ | energy of the conduction band minimum | eV |
| $E_{\rm c,a}$ | energy associated with the initial charging phase | |
| | of the junction capacitance | J |
| $E_{c,b}$ | energy associated with the second charging phase | - |
| | of the junction capacitance | J |

| $E_{\rm c,tot}$ | total capacitance energy associated with diode turn off | J |
|-------------------------|--|--------------------------------------|
| $E_{\rm F}$ | Fermi energy | eV |
| $E_{\rm off}$ | turn-off switching energy | J |
| $E_{\rm Z}(Z_{\rm th})$ | linear trendline to determine the junction-to-case thermal resistance | |
| $E_{\rm on}$ | turn-on switching energy | J |
| E_{00} | characteristic tunneling energy | eV |
| $E_{\mathbf{V}}$ | energy of the valence band maximum | eV |
| γ_{T} | required drop of the on-resistance of Ga_2O_3 diodes below the SiC level | |
| $I_{\rm D}, i_{\rm D}$ | current through a diode | А |
| I _{D,ave} | average current through a diode | А |
| $i_{\rm D,rm}$ | maximum reverse current during diode turn off | А |
| I _{D,rms} | root-mean-square current through a diode | А |
| I_{f} | forward current | А |
| $I_{\rm L}$ | load current | А |
| I _r | reverse leakage current | А |
| $I_{\rm r,TFE}$ | reverse thermionic-field emission current | А |
| Is | saturation current | А |
| i _s | source current | А |
| J_0 | reference current density | A/m^2 |
| $J_{\rm r,TFE}$ | reverse thermionic-field emission current density | A/m^2 |
| $J_{\rm s}$ | saturation current density | A/m^2 |
| k | Boltzmann constant | J/K |
| $K_{\rm bw}$ | measurement error due to bandwidth limitations | % |
| L | inductance | Н |
| $L_{p,loop}$ | parasitic commutation-loop inductance | Н |
| $m_{\rm e}^*$ | electron effective mass | kg |
| m_{T} | slope of a linear fit of the transient temperature response | K/s |
| $\mu_{ m e,sub}$ | electron mobility in the substrate | ${\rm cm}^2{\rm V}^{-1}{\rm s}^{-1}$ |
| $N_{\rm C}$ | density of states in the conduction band | m^{-3} |
| $N_{\rm D}$ | effective donor concentration | m^{-3} |
| $N_{\mathrm{D,sub}}$ | effective donor concentration in the substrate | m^{-3} |
| n | ideality factor | |
| $n_{ m if}$ | ideality factor with image force lowering | |
| n_{tun} | ideality factor with tunneling | |
| $P_{\rm cond}$ | conduction loss | W |
| $P_{\rm H}$ | heating power | W |
| h | Planck constant | ${ m m}^2{ m kg/s}$ |
| e | elementary charge | Ċ |
| Q_{c} | capacitive charge | \mathbf{C} |

| $Q_{\rm c,a}$ | charge associated with the initial charging phase of the junction capacitance | С |
|------------------------------|--|-------------------------|
| $Q_{\rm c,b}$ | charge associated with the second charging phase | C |
| 0 | of the junction capacitance | C |
| $Q_{c,tot}$ | total capacitive charge associated with diode turn off | C |
| R | resistance | Ω |
| \hbar | reduced Planck constant | m² kg/s |
| r _{epi,sp} | specific resistance of the epitaxial layer | $\Omega \mathrm{m^2}$ |
| $R_{\rm g,on}$ | external gate turn-on resistance | Ω |
| ro | differential on-resistance | Ω |
| $r'_{\rm o,sp}$ | fitted effective specific differential on-resistance | $\Omega\mathrm{m}^2$ |
| $r_{o,sp}$ | specific differential on-resistance | $\Omega{ m m}^2$ |
| $r_{\rm sc}$ | ohmic resistance of a semiconductor layer | Ω |
| $r_{\rm sub,sp}$ | specific resistance of the substrate | $\Omega{ m m}^2$ |
| R_{th} | thermal resistance | K/W |
| $R_{\rm th,die}$ | thermal resistance of the die | K/W |
| $R_{\mathrm{th,i}}$ | thermal resistance of the i-th component | K/W |
| $R_{\mathrm{th,ja}}$ | thermal resistance from junction to ambient | K/W |
| $R_{\mathrm{th,jc}}$ | thermal resistance from junction to case | K/W |
| $\mathbf{S}_1, \mathbf{S}_2$ | bidirectional switch | |
| $T_{\rm amb}$ | ambient temperature | К |
| $T_{\rm a,exp}$ | experimentally determined rise time | S |
| T _{a,sense,i} | inherent rise time of the i-th measurement instrument | S |
| $T_{a,true}$ | true impulse rise time of a signal | S |
| $T_{\rm epi}$ | temperature of the epitaxial layer | Κ |
| t | time | S |
| $T_{\rm j}$ | junction temperature | Κ |
| $T_{j,ave}$ | average junction temperature | Κ |
| $T_{j,max}$ | maximum junction temperature | Κ |
| $T_{i,\min}$ | minimum junction temperature | Κ |
| T_{i0} | initial junction temperature | Κ |
| T_1 | unidirectional switch | |
| $T_{\rm ref}$ | reference temperature | Κ |
| $t_{\rm ref}$ | reference time | $1\mathrm{s}$ |
| V | voltage | V |
| $V_{\rm br}$ | breakdown voltage | V |
| $V_{\rm D}, v_{\rm D}$ | voltage across a diode | V |
| $V_{\rm ds}$ | drain-source voltage | V |
| V _{f@10mA} | forward voltage at a sensing current of $10 \mathrm{mA}$ | V |
| V_{gs} | gate-source voltage | V |
| V_0 | knee voltage | V |
| | | |

| V'_0 | fitted effective knee voltage | V |
|-------------------|-------------------------------------|-----|
| $V_{\rm ref}$ | reference voltage | V |
| V_{th} | threshold voltage | V |
| $z_{\rm att}$ | thickness of the die-attach layer | m |
| $z_{ m die}$ | thickness of the die | m |
| $z_{ m epi}$ | thickness of the epitaxial layer | m |
| $z_{ m sc}$ | thickness of a semiconductor layer | m |
| $z_{ m sub}$ | thickness of the substrate of a die | m |
| Z_{th} | thermal impedance | K/W |

Chapter 1

Introduction

1.1 Motivation and research goals

The demand for global electricity is expected to continuously grow from about 25 000 TWh in the year 2021 to between 44 000 TWh and 62 000 TWh in the year 2050 [1]. The associated efficient transmission of electrical energy from the energy source to the consumer clearly determines daily life now and in the future. However, the electrical input quantities, such as current and voltage amplitude or frequency supplied by the source (e.g., a car battery), are generally different from the waveforms required at the electrical load (e.g., an electric motor). Power electronics concerns the corresponding transmission and conversion of electrical energy with the maximum possible efficiency. The underlying processes are based on semiconductor devices that can be switched either actively or passively between a conducting and a non-conducting state [2, 3].

While bipolar silicon (Si) semiconductor devices have been standard in all power electronics applications for years, recently power devices based on the wide-bandgap materials silicon carbide (SiC) and gallium nitride (GaN) have entered the market. Their wide bandgap and associated high breakdown fields allow the fabrication of unipolar high-voltage devices with lower on-resistances and switching losses than their Si counterparts [2–5]. This enables high switching frequencies and allows a reduction of the size and weight of passive components used to store the energy – a trend that is especially important also in the automotive industry [3, 4].

Vertical SiC power devices and modules up to several kV are commercially available already, and first SiC-based inverters have found their way into series production [3, 5, 6]. However, the growth of SiC wafers from the vapor phase is cost-intensive and hampers a widespread commercial implementation of SiC-based modules [5, 7, 8]. The less mature GaN technology, where the currently most promising devices are lateral GaN transistors grown on Si wafers, struggles to realize transistors with rated blocking voltages above 650 V [9, 10]. Nevertheless, it excels with fast switching transients and high switching frequencies at lower voltages [9, 10].

Recently, the transparent semiconductor gallium oxide (Ga_2O_3) has raised interest regarding a potential application in power electronics. While it occurs in different crystal structures, beta-phase gallium oxide $(\beta$ -Ga₂O₃), which is the most stable of all polymorphs [11, 12], differs from Si, SiC, and GaN in that it combines two decisive properties: First, it exhibits an even wider bandgap than SiC or GaN, leading to a more than three times larger theoretical breakdown field than that of SiC [12]. Second, large β -Ga₂O₃ wafers can be grown from the melt using standard techniques such as the Czochralski method known from Si wafer technology [13–17]. As such, it holds potential for unipolar and vertical high-voltage power devices at a lower cost than comparable SiC devices [8, 12, 18].

Based on these properties, the on-resistance of a vertical 1.2 kV Ga₂O₃ transistor could theoretically be as low as one-tenth that of a SiC and about one-third that of a GaN counterpart [18]. At the same time, the charge × resistance figure of merit could theoretically be improved by a factor of 3 and 1.2 compared to SiC and GaN, respectively [18]. A Ga₂O₃ based 500 kW bi-directional charger was proposed in [19] through simulation and, from a purely electrical perspective, suspected to exceed the efficiencies of current SiC-based chargers by 3 % [19]. However, one major drawback of Ga₂O₃ that is often not considered in such studies is its low thermal conductivity – about one order of magnitude lower than that of SiC [12]. This limits heat dissipation and raises concerns about potential thermal management problems, but a "disproportionately small amount of work is addressing this sensitive issue" [8].

Today, the research into Ga_2O_3 substrate growth and device processing is steadily increasing [12]. The first research milestone was achieved in 2012 with the demonstration of a lateral Ga_2O_3 transistor cell grown on a native substrate [20]. Shortly after, in 2013, a vertical Ga_2O_3 Schottky diode was demonstrated [21]. Since then, research has intensified and the growth conditions and device structures have continuously improved. This resulted in another milestone in 2022, when first 6 kV and 8.3 kV vertical Ga_2O_3 diode cells on native substrates exceeding the theoretical unipolar on-resistance versus breakdown voltage limit of SiC were realized [22, 23]. However, the reported diodes are only several hundreds of micrometers large and thus still need to be scaled up to larger sizes. Furthermore, especially vertical transistors are significantly less developed although cells with several kV breakdown voltage have been developed already [24–26]. Hence, research still mainly focuses on the understanding and improvement of basic device structures.

One reason for this is that β –Ga₂O₃ cannot be p-doped and that it is highly anisotropic in its crystal structure and electrical and thermal properties [12, 27–29]. Therefore, intense research effort on a device level is still required. As a result, there are no established device concepts so far, power devices are not yet commercially available, and corresponding suitable packaging concepts are yet to be developed. In short: Ga₂O₃ power devices are still far from being integrable into real-world applications.

Taking this initial situation as a starting point, this work aims at building a bridge towards the use of Ga_2O_3 devices in power electronics. Although not yet commercially available, here the very first large-area Ga_2O_3 diodes from a planned manufacturing line are available for research of their basic properties and development towards potential applications. Through various electrical and thermal measurements, simulations, and theoretical considerations, the work addresses basic questions such as: What are characteristic similarities and differences between these novel Ga_2O_3 and commercial SiC diodes? Is the device quality sufficient to take the next steps towards an application in power electronics? Is an implementation in fast-switching power converters feasible despite the low thermal conductivity? What are the difficulties and potential methods to improve the application-relevant properties? By gradually examining the basic properties of novel large-area Ga_2O_3 diodes in an application-oriented manner, this thesis contributes to paving the way to Ga_2O_3 power electronics.

1.2 Structure of this thesis

In order to successively address the previously motivated research questions, the thesis is organized as follows:

Chapter 2 briefly introduces the basic concepts of power electronics, including the most important topologies and established semiconductor power devices. This forms the basis for understanding the theoretical advantages of β -Ga₂O₃ as a power semiconductor. Finally, the material properties of Ga₂O₃ are explained in more detail, and an overview of current state-of-the-art device structures and achievements is given. From this, the current research gaps are identified, and more detailed guiding questions are derived.

Chapter 3 covers the experimental methods used in this work. First, the packaging procedure of the devices under investigation is explained. Furthermore, the basic electric and thermal measurement techniques are introduced, and the associated measurement uncertainties are discussed. Since Ga_2O_3 devices are expected to exhibit fast-switching capabilities, it is crucial to capture the associated waveforms accurately. Therefore, novel current sensors are assessed and comparatively evaluated to find the most suitable sensors for further tests.

Before novel Ga_2O_3 devices can be tested in an application environment, their fundamental characteristics must be examined. **Chapter 4** therefore first studies the basic temperature-dependent electrical characteristics and junction properties that, as will be shown, can differ significantly from those of standard SiC diodes. Furthermore, the variation of characteristic device properties due to the early development stage is discussed, and Spice simulation models that allow a simple but effective replication of these characteristics are introduced.

After having gained insight into the basic properties, the heat dissipation in Ga_2O_3 diodes, which is identified as one of the major limitations, is studied in more detail in **Chapter 5**. The effectiveness of different thermal management strategies, such as substrate thinning, cooling from the junction-side, or the use of underfill materials, is studied experimentally and simulatively. In view of the potentially low on-resistance of Ga_2O_3 devices, it is discussed if junction temperatures similar to those of SiC counterparts can realistically be achieved in the future despite the low thermal conductivity.

Chapter 6 uses the newest generations of Ga_2O_3 diodes to assess their switching properties in an application-oriented environment. To this end, the samples are subjected to double pulse tests and, for the first time, also studied in a 400 V buck converter application. The switching waveforms, power-loop efficiencies, and thermal behavior are compared to that of Si and SiC reference diodes and connected to the findings of the previous chapters.

Chapter 7 summarizes key findings of this work and provides a brief outlook on potential future research activities.

Note that several results obtained as a part of this work have been published previously in [253–260]. Furthermore, the student work [249–252] has been supervised by the author during the course of this thesis.

Chapter 2

Basics and state-of-the-art of gallium oxide power electronics

2.1 Chapter overview

This chapter summarizes the state-of-the-art of Ga_2O_3 power electronics. Section 2.2 briefly introduces the general principles of power electronics, including the buck converter as a central topology and a basic overview of the structure of common commercialized semiconductor power devices. Section 2.3 then introduces Ga_2O_3 as a material for power electronics. Section 2.3.1 first motivates the use of Ga_2O_3 by comparing basic figures of merit of different power semiconductors. Building on this, Section 2.3.2 describes the most important structural, electrical, and thermal properties and challenges of gallium oxide in more detail. Section 2.3.3 discusses the different methods to grow β -Ga₂O₃ power devices, whose structures and current state of research are covered in Section 2.3.4 (diodes) and Section 2.3.5 (transistors). Section 2.4 summarizes key aspects and formulates the corresponding research questions.

2.2 Basics of power electronics

2.2.1 Basic principles of power conversion

The buck converter topology shown in Fig. 2.1a is a central element that can be found in various types of power converters. In its simplest form, it consists of a unidirectional active switch T_1 that can be periodically opened and closed through a control circuit, a diode D_1 , and an inductance L [30].

 D_1 blocks the current in the reverse direction but provides a freewheeling path when T_1 is open, and the magnetized inductance L acts as a current source. By controlling the ratio of the switch-on and switch-off time, the so-called duty cycle, the input voltage V_1 is consequently stepped down to a voltage $V_2 \leq V_1$ at the output. The input capacitance C_1 decouples the circuit from the input voltage supply and the output capacitance C_2 can be added to stabilize the output waveforms [30].

When T_1 and D_1 are replaced by two bidirectional switches S_1 and S_2 as shown in Fig. 2.1b (either by combining a unidirectional switch and an antiparallel diode, or by using intrinsically bidirectional switches), stepping up the voltage in the opposite direction is also possible. First, the voltage source V_2 magnetizes the inductance L when S_2 is closed. When opening S_2 and closing S_1 , the stored energy is transferred to the capacitor C_1 , leading to an output voltage $V_1 > V_2$. The resulting topology is called a half bridge [30].



Figure 2.1: (a) The buck converter in its simplest form consists of a unidirectional switch T_1 , a diode D_1 , and an inductance *L*. (b) When T_1 and D_1 are replaced by two bidirectional switches S_1 and S_2 , a half-bridge topology is obtained. Based on [30].

The current commutates between the different switches with a high frequency of typically several kHz to several hundreds of kHz. The corresponding current loop, the so-called commutation loop, is shaded in Fig. 2.1. In reality, parasitic inductances that exist in this loop (e.g., caused by the current traces, package leads, or bond wires) can lead to strong oscillations or voltage and current overshoots and limit the commutation speed. The commutation-loop inductance must therefore be kept as low as possible [31].

Generally, increasing the switching frequency is targeted as it translates to a smaller amount of energy that needs to be stored in the passive components per cycle, thus enabling to use smaller and lighter inductances and capacitors. In reality, however, the switching frequency is limited. The main reason (apart from additional losses such as core losses in the passive components) is that real semiconductor devices exhibit switching losses and generate heat during switching events [30, 31]. The basic properties of these semiconductor devices will be covered in the following section.

2.2.2 Semiconductor power devices

Power diodes

One of the most common diode structures found in power electronics is the bipolar silicon PiN diode shown in Fig. 2.2a. It consists of a lightly (n^-) doped drift region that is surrounded by a p-region and a heavily doped n^+ -region. The p-region and n^+ -region are metallized to allow an electrical connection to the circuit and constitute the anode and cathode of the device [32].

In forward direction, when the anode potential is higher than the cathode potential (defined as $V_{\rm D} > 0$ V in the following), the lightly doped drift region is flooded by charge carriers from the outer zones. Consequently, both holes and electrons contribute to the current flow that can be described by the Shockley equation [32, 33]. In blocking direction ($V_{\rm D} < 0$ V), the drift region is depleted from charge carriers and only a small leakage current (typically in the order of μ A at the specified reverse voltage) flows in reverse direction [32].

The strength of the electric field that is built up in the drift zone increases with rising reverse voltage. At a certain reverse voltage, the so-called breakdown voltage $V_{\rm br}$, the electric field is strong enough to accelerate charge carriers to such an extent that they can create new electron-hole pairs by impact ionization, which themself gain enough kinetic energy to create additional electron-hole pairs. This process is called avalanche multiplication or avalanche breakdown. The critical field strength $E_{\rm br}$ at which this happens is one of the most important parameters for semiconductor devices [32].



Figure 2.2: Schematic structure of (a) a bipolar PiN diode and (b) a merged PiN Schottky (MPS) diode composed of a Schottky junction with merged p-areas to improve the blocking characteristics and the surge-current capability. Based on [32, 34].

Wide-bandgap semiconductors such as SiC, GaN, or Ga_2O_3 exhibit a significantly higher critical electric field than Si (see Section 2.3.1). This allows the thickness of the drift zone to be reduced and the doping concentration to be increased, which reduces the differential on-resistance at a given breakdown voltage. In addition, purely unipolar diodes with a direct metal–semiconductor (MS) contact, also called Schottky contact, become available for high blocking voltages [34].

A substantial advantage of such unipolar devices is apparent when the device is turned off, i.e., switched from the forward to the blocking state. The charge removal and recombination of minority carriers in the case of bipolar diodes causes a high reverse current before the full blocking state is reached. This reverse current and the associated reverse recovery charge lead to power losses that increase significantly with rising current, current slope, and temperature. In the case of unipolar diodes, however, only majority carriers are involved in the charge transport. Therefore, only their removal is necessary to build up a blocking voltage in the drift region, which is associated with a certain displacement current. This process is temperature-independent and causes negligible power losses in the diode, making unipolar devices favorable for fast-switching applications [34].

Commercial Schottky diodes for high-voltage applications are typically made from SiC. One of the most common structures is the merged PiN Schottky (MPS) diode shown in Fig. 2.2b. Instead of a pure MS contact it consists of additional p-type areas distributed across the entire junction area. The exact diode properties depend critically on the density and doping level of these merged p-regions. Generally speaking, the structure can bring two main improvements: In blocking direction, the p-areas can effectively shield the Schottky junction from high electric fields. The reduced electric field lowers the reverse leakage current and prevents premature breakdown close to the junction [34–36]. In forward direction, the p-areas can be activated and inject holes at sufficiently large forward voltages above the normal operation area. This effect can reduce the resistance and improve the surge-current capability significantly [34, 37]. Both Si PiN diodes and SiC Schottky diodes are used as benchmarks throughout this thesis.

Power transistors

The metal–oxide–semiconductor field-effect transistor (MOSFET) is a voltage-controlled switch that is either normally-off (enhancement mode) or normally-on (depletion mode) when no control voltage is applied [38]. In power electronics, normally-off MOSFETs are preferred and will also be used in this thesis as active switches.



Figure 2.3: Basic structure and circuit symbol of (a) a vertical n-channel power MOSFET and (b) a lateral GaN HEMT with p-GaN gate to achieve enhancement-mode operation. Based on [38, 39].

As shown in Fig. 2.3, the drift region of the MOSFET is typically formed by a lightly n-doped epitaxial layer grown on a heavily n-doped substrate. It blocks the entire voltage when the MOSFET is turned off and a positive voltage is applied between drain and source $(V_{ds} > 0)$. The source contact is formed by a metallized heavily doped n-region that is surrounded by a p-area. To avoid a turn-on of the resultant parasitic npn-transistor, the p-area is also connected to the source metallization. The gate, usually heavily doped poly-silicon, is electrically insulated from the source pad and the drift region by a thin oxide layer such as SiO₂. When the gate–source voltage V_{gs} is larger than a certain threshold voltage V_{th} ($V_{gs} > V_{th} > 0$ V for normally-off transistors), an inversion layer with an accumulation of electrons is created in the p-type area below the gate. A forward current can now flow through this n-channel when a drain-source voltage $V_{ds} > 0$ V is applied. A parasitic diode, the so-called body diode, is formed between the p-area, the n⁻ drift layer, and the n⁺ substrate. When the MOSFET is reverse biased ($V_{ds} < 0$ V), the diode is forward biased and can thus serve as an internal freewheeling diode [38].

An improvement of the MOSFET technology was achieved by moving from planar structures to trench designs. In SiC MOSFETs, e.g., the gate is often formed at a trench that is etched into the lightly doped drift region between the source contacts [38, 40]. This structure offers an improved channel mobility, which reduces the on-resistance. In addition, a higher number of trench gates can be formed in a given area in comparison to the planar gate structure, which increases the current density [38, 40]. For other commonly used power devices such as insulated-gate-bipolar transistors (IGBTs), where the n⁺ layer at the drain side is basically replaced by a p-type layer that can inject holes into the drift region, it is referred to [41] for more details as they will not be used in this work.

The device structures are associated with parasitic internal capacitances that have to be charged and discharged during the switching events. These are the gate–source capacitance C_{gs} , the gate–drain capacitance C_{gd} , and the drain–source capacitance C_{ds} . Minimizing these capacitances is crucial to enable fast switching [41]. SiC power MOSFETs will be used in this thesis to build a hybrid buck converter with Ga₂O₃ Schottky diodes in Chapter 6.

HEMT

Transistors based on the wide-bandgap semiconductor gallium nitride (GaN) have recently been commercialized. A thorough technology review can be found in [39]. As shown in Fig. 2.3b, a low-cost Si wafer typically provides the basis for the following fabrication steps.

First, a heteroepitaxial AlN nucleation layer is grown on the substrate, which mainly prevents the diffusion of gallium into the wafer. To avoid excessive mechanical strain resulting from the difference in lattice parameters and thermal expansion between Si and GaN, several $Al_xGa_{(1-x)}N$ buffer layers are grown on the nucleation layer. The central part of the device is formed at the interface between a GaN/AlGaN heterostructure. Strong polarization and piezoelectric effects at the interface associated with a strong bending of the conduction band below the Fermi level lead to the formation of a two-dimensional electron gas (2DEG). This enables a lateral current flow with high carrier mobility, which is why the device is called a high-electron-mobility transistor (HEMT). As a consequence, the device is inherently normally-on. Several methods exist to create an enhancement-mode operation. The GaN HEMTs that will be used in this thesis have a p-GaN or p-AlGaN gate structure. The p-layer increases the channel potential below the gate and depletes it from charge carriers at zero bias voltage. Other common ways to achieve normally-off behavior are cascode structures with normally-off Si MOSFETs [42], a recessed gate metal–insulator–semiconductor structure to interrupt the 2DEG [43], or HEMTs with fluorine implanted AlGaN layer [44].

Due to the symmetrical structure, the GaN HEMT can conduct current both in forward and reverse direction, if the potential at the gate in relation to drain or source is high enough. Compared to power MOSFETs, the parasitic capacitances $C_{\rm gs}$, $C_{\rm gd}$, and $C_{\rm ds}$ are significantly smaller. Combined with the high electron mobility, this enables faster switching transients [39]. Therefore, GaN HEMTs will be used to study new high-bandwidth current sensor concepts in Section 3.4. Due to the lateral structure, however, it is difficult to achieve voltage ratings above 650 V [39].

2.2.3 Physical basics of Schottky diodes

Figure 2.4a shows the band structure of a Schottky contact with an n-type semiconductor, which is the central part of the diodes studied in this work. In thermal equilibrium, when the gradient in the electrochemical potential is zero, the conduction band at the metal– semiconductor (MS) junction bends upwards and a region that is depleted from charge carriers is formed. The energy difference between the conduction band minimum $E_{\rm C}$ and the Fermi energy $E_{\rm F}$ at the junction is called Schottky barrier height $\phi_{\rm b,0}$. It depends critically on the band-gap energy $E_{\rm G}$ and doping concentration $N_{\rm D}$ of the semiconductor, the work function $\Phi_{\rm M}$ of the metal, and defect states at the interface. The bending in the conduction band is described by the built-in potential $\psi_{\rm bi}$ [45].

The thermionic emission (TE) model is widely used to describe the current flow in forward direction for Si Schottky diodes [45], but also for MS contacts with wide-bandgap semiconductors such as SiC [46, 47], GaN [48], or Ga_2O_3 [21]. From the Fermi-Dirac statistics it follows that for temperatures above 0 K there will always be a number of electrons above the Schottky barrier giving rise to a saturation current I_s and the associated current density [45]

$$J_{\rm s} = \frac{4\pi e m_{\rm e}^* k^2}{h^3} \cdot T_{\rm j}^2 \exp\left(-\frac{e\phi_{\rm b,0}}{kT_{\rm j}}\right) = A^{**} \cdot T_{\rm j}^2 \exp\left(-\frac{e\phi_{\rm b,0}}{kT_{\rm j}}\right)$$
(2.1)

with e the elementary charge, k the Boltzmann constant, h the Planck constant, m_e^* the effective electron mass, T_j the junction temperature, and A^{**} the Richardson constant. When the external voltage is increased (potential at Schottky metal positive in relation to potential at semiconductor), the electrons gain additional energy and a current with the current density [45]



Figure 2.4: Band structure of a Schottky contact in (a) thermal equilibrium and (b) when a reverse voltage is applied. The forward current can often be described by the thermionic emission (TE) model. In reverse direction, different conduction mechanisms such as thermionic-field emission (TFE), field emission (FE), or a combination of different processes can dominate depending on the semiconductor properties. Based on [45].

$$J_{\rm D} = J_{\rm s} \left[\exp\left(\frac{eV_{\rm D}}{nkT_{\rm j}}\right) - 1 \right]$$
(2.2)

can flow in forward direction. The dimensionless ideality coefficient n is 1 for an ideal diode and > 1 when tunneling and recombination processes are present at the junction. Equation 2.2 holds true only for low currents when the series resistance of the diode can be neglected. The series resistance is given by the sum of contact resistances the resistances of the auxiliary parts (bondwires, contact resistance, etc.), the ohmic contact, the substrate, and the drift zone, i.e., the epitaxial layer [31, 34]. The resistance of a semiconductor layer with thickness z_{sc} can be calculated directly from its electron mobility μ_e , effective doping concentration N_D , and current-conducting area A according to [34]

$$r_{\rm sc} = \frac{z_{\rm sc}}{e\mu_{\rm e}N_{\rm D}A}.$$
(2.3)

The substrate is usually heavily doped, leading to a rather small specific resistance. It can be reduced, e.g., by using thinner substrates and improving the electron mobility at high doping levels. The resistance of the lightly doped drift layer usually makes up the major part of the device resistance. As the drift layer is responsible for blocking the voltage during reverse operation, it needs to have a minimum thickness of $z_{epi} = 2V_{br}/E_{br}$ when a triangular field is assumed. Materials with high E_{br} such as Ga₂O₃ thus allow the reduction of the drift layer thickness and a reduction of the specific differential on-resistance $r_{o,sp}$ for a given breakdown voltage [34].

In reverse direction, the drift zone is depleted from charge carriers. With rising reverse voltage, the depletion region extends further into the drift layer, giving rise to a space charge Q_c and an associated parasitic junction capacitance [49]

$$C_{\rm j}(V_{\rm D}) = \frac{\partial Q_{\rm c}}{\partial V_{\rm D}} = A \sqrt{\frac{e\epsilon_0 \epsilon_{\rm s} N_{\rm D}}{2(\psi_{\rm bi} - V_{\rm D})}}, \qquad (2.4)$$

where ϵ_0 is the vacuum permittivity and ϵ_s is the relative dielectric constant of the semiconductor. This parasitic junction capacitance is (dis-)charged during every switching event in a converter operation. Despite the absence of bipolar recombination this results in a non-negligible current undershoot during diode turn-off [34].

In blocking direction, high electric fields can occur at the Schottky junction. This can lead to a substantial image force lowering $\Delta \phi_{if,0}$ of the Schottky barrier, and the (at least partial) tunneling of charge carriers through the barrier can become increasingly dominant [45]. This is shown schematically in Fig. 2.4b. Here, field emission (FE) is a pure tunneling process of charge carriers through the Schottky barrier, which can occur especially at very high doping concentrations. Thermionic-field emission (TFE) describes the tunneling of thermally excited electrons through the barrier, which appears thinner at higher energies [45]. This theory has been found to accurately describe the reverse conduction mechanism of many MS contacts, including wide-bandgap semiconductors such as SiC [50], GaN [51], and Ga_2O_3 [52]. Often, however, especially defects and traps at the interfaces can lead to a variety and even combination of different conduction mechanisms, such as Poole-Frenkel emission (PFE), i.e., the thermal emission of electrons trapped at the MS interface into the conduction band [53, 54], or the tunneling of electrons through trap states at the MS interface, called trap-assisted tunneling (TAT) [53, 55]. Both conduction mechanisms have been identified previously in wide-bandgap devices such as GaN [56] or $Al_x(Ga_{1-x})_2O_3$ [57] diodes. Clearly, the properties of the semiconductor and the MS interface critically determine the behavior of the resulting power devices. The following section will therefore cover the material properties of Ga_2O_3 and the related power devices in more detail.

2.3 Gallium oxide as power semiconductor

Gallium oxide is a transparent oxide semiconductor with five different known modifications of the crystal structure. These five polymorphs (or phases), designated as α -, β -, γ -, δ -, and ϵ -Ga₂O₃ differ in mechanical, electronic, and optical properties. The most important polymorphs regarding an application in power electronics are β -Ga₂O₃ and α -Ga₂O₃. Relevant material properties are discussed in the following.

2.3.1 Important figures of merit

As explained in Section 2.2.3, the band-gap energy $E_{\rm G}$ and the associated critical electric field strength $E_{\rm br}$ belong to the essential material parameters for power semiconductor devices. The critical electric field can empirically be expressed directly as a function of the bandgap energy according to [58]. Figure 2.5a shows this relationship for semiconductors with direct and indirect bandgap, as well as the generalized relationship for all semiconductors. β -Ga₂O₃ can be considered as a material with a quasi-direct bandgap of about 4.8 eV leading to a theoretical critical electric field of about 8 MV/cm, which compares to 2.5 MV/cm for 4H-SiC and 3.3 MV/cm for GaN [12]. In the unipolar limit, the specific resistance of the drift layer with thickness $z_{\rm epi}$ can be calculated from [45]

$$r_{\rm o,sp} = \frac{z_{\rm epi}}{e\mu_{\rm e}N_{\rm D}A} = \frac{4V_{\rm br}^2}{\epsilon_0\epsilon_{\rm s}\mu_{\rm e}E_{\rm br}^3},$$
(2.5)

where $N_{\rm D} = \epsilon_0 \epsilon_{\rm s} E_{\rm br}^2 / (2eV_{\rm br})$ is the maximum allowed doping concentration necessary to achieve the breakdown voltage $V_{\rm br}$. Note that $r_{\rm o,sp}$ scales inversely with the third power of



Figure 2.5: (a) Based on the relationship between bandgap energy and critical electric field $E_{\rm br}$ for semiconductors with direct or indirect band gap, a high $E_{\rm br}$ of about 8 MV/cm is typically assumed for Ga₂O₃. In the unipolar limit, this theoretically leads to (b) a specific on-resistance of one-seventh of that of SiC and (b) a charge × resistance FOM similar to that of GaN. Based on [20, 34, 58] and Tab. A.1.

 $E_{\rm br}$. Critical other material parameters are the electron mobility and the relative dielectric constant of the semiconductor, which are summarized in Tab. A.1. Note that Eq. 2.5 does not consider the electric field dependence of $\mu_{\rm e}$, the dependence of $E_{\rm br}$ on $N_{\rm D}$, temperature dependences, or more practical concerns such as necessary junction terminations and current spreading or crowding [34, 59]. Bearing this in mind, Fig. 2.5b illustrates that the $r_{\rm o,sp}$ of Ga₂O₃ devices can potentially be as low as a seventh of the $r_{\rm o,sp}$ of SiC, and almost one-third of the $r_{\rm o,sp}$ of GaN. A deeper physical comparison deviating from this rather generalized view was carried out in [60]. By analyzing the impact-ionization coefficients and dielectric properties of different semiconductors, it was shown that a wide bandgap does not necessarily result in a in a high critical electric field and breakdown voltage. Regarding Si, SiC, GaN, and Ga₂O₃, however, the same trend as shown in Fig. 2.5b was found [60]. The ratio $V_{\rm br}^2/r_{\rm o,sp}$ will be referred to as power figure-of-merit (PFOM) in the following.

As discussed in Section 2.2.3, the charge Q_c stored in the drift region affects the turn-off behavior of the diode. The theoretical trade-off between r_o and Q_c can be expressed by [34]

$$Q_{\rm c}r_{\rm o} = \frac{4V_{\rm br}^{2}}{\mu_{\rm e}E_{\rm br}^{2}}\,.$$
(2.6)

Due to the rather low electron mobility of $200 \text{ cm}^2 \text{ V}^{-1} \text{ s}^{-1}$ [61] compared to $1000 \text{ cm}^2 \text{ V}^{-1} \text{ s}^{-1}$ for 4H-SiC or $1200 \text{ cm}^2 \text{ V}^{-1} \text{ s}^{-1}$ for GaN [12], the theoretical advantages are less pronounced than expected only from the on-resistance. As illustrated in Fig. 2.5c, however, the $r_0 \times Q_c$ FOM for Ga₂O₃ can theoretically still be approximately half that of SiC and similar to that of GaN. It must be noted that slightly varying values for μ_e , ϵ_s , and E_{br} are used for Ga₂O₃ in literature. This is partially related to varying physical properties based on the anisotropy of the β -Ga₂O₃ crystal structure which, together with other important material properties, will be discussed in the following section.

2.3.2 Material properties

Crystal structure

The α phase of Ga₂O₃ has a corundum structure with six Ga₂O₃ formula units per crystallographic cell and all Ga³⁺ ions being coordinated to the O²⁻ ions in an octahedral geometry [28, 62, 63]. This intrinsic crystal symmetry leads to rather isotropic electronic and thermal properties [64–66]. However, α –Ga₂O₃ is metastable and undergoes a phase transition to β –Ga₂O₃ at temperatures above 600 °C in dry air and 300 °C under hydrothermal conditions [11]. An important consequence is that it is not possible to grow α –Ga₂O₃ wafers from the melt. Therefore, α –Ga₂O₃ devices must be fabricated on non-native substrates such as α –Al₂O₃ [67–69] and will not be investigated in this thesis.

The β -phase is the thermodynamically most stable one of all Ga₂O₃ polymorphs and is obtained when heating any other phase of gallium oxide above a certain critical temperature between 300 °C and 1000 °C [11]. This also enables the fabrication of melt-grown β –Ga₂O₃ bulk substrates and corresponding homoepitaxially grown vertical devices. β –Ga₂O₃ has a monoclinic crystal structure with four Ga₂O₃ formula units in each crystallographic cell. The Ga³⁺ ions are coordinated both in an octahedral and in a tetrahedral geometry to the O²⁻ ions, which means there are two types of nonequivalent gallium ions and three types of non-equivalent oxygen ions. The interionic Ga–O and O–O distances differ for the respective coordinations [28, 62]. This intrinsic anisotropy in the crystal structure causes the optical, electronic, and thermal properties to vary along different crystal directions [27–29] and ultimately also affects the fabrication and processing of power devices as well as the resulting device characteristics significantly.

Electronic structure and doping

For β -Ga₂O₃, multiple theoretical and experimental studies find a fundamental indirect band gap in the range of 4.66 to 4.86 eV [28, 70, 71] as well as a some 10 meV larger direct band gap of 4.69 to 4.91 eV [28, 70–73]. Due to the anisotropy of the crystal structure, the bandgap energy also depends on the crystal direction [74].

Similarly, a fundamental indirect bandgap of 5.03 to 5.19 eV [28, 64, 70] and a several 10 meV larger direct bandgap of 5.08 to 5.40 eV [28, 64, 66, 70] has been determined for α -Ga₂O₃. This makes both α - and β -Ga₂O₃ semiconductors with a quasi-direct band gap, where corundum Ga₂O₃ seems to exhibit a slightly larger bandgap than its monoclinic counterpart. According to Fig. 2.5a, this leads to a higher theoretical breakdown field of up to 9 to 10 MV/cm for α -Ga₂O₃.

The structures of both α -Ga₂O₃ and β -Ga₂O₃ allow for further adjustment of the bandgap energy. The corundum structure of α -Ga₂O₃ makes the combination with other corundum structured materials of similar lattice constant such as α -Al₂O₃ or α -In₂O₃ possible. This, in principle, allows tuning the bandgap in almost any range between 3.8 eV and 8.8 eV, whereas the experimental realization is difficult due to the meta-stability of α -Ga₂O₃ and α -In₂O₃ [67]. Alloys with Al₂O₃ have opened the possibility to further increase the bandgap also for the β phase of Ga₂O₃. In [75], e.g., a bandgap energy of 5.7 eV was experimentally determined for β -(Al_{0.47}Ga_{0.53})₂O₃ films.

The valence band of both α -Ga₂O₃ and β -Ga₂O₃ is rather flat, which results in a generally low hole mobility. In addition, polarons (quasi-particles describing the interaction between

electrons and atoms in a solid) can significantly decrease the electron mobility in semiconductors. In the specific case of Ga_2O_3 , holes tend to form localized polarons, i.e., they tend to be trapped by localized lattice distortions [76]. This phenomenon is often referred to as the self-trapping of holes, and it was concluded from a theoretical perspective that for this reason no p-type conductivity should occur in Ga_2O_3 [76]. Combined with the fact that potential acceptor states in Ga_2O_3 are located deep within the bandgap, i.e., no shallow acceptors have been found so far, Ga_2O_3 is generally considered as a material that cannot be p-doped [18, 76, 77].

Despite these considerations, NiO in p-NiO/n-Ga₂O₃ heterostructures has recently been experimentally shown to be capable of injecting holes into the Ga₂O₃ layers at sufficiently large forward voltages, which is referred to as conductivity modulation [23]. However, the associated hole mobility was estimated to be rather low $(1.9 \text{ to } 8.3 \text{ cm}^2 \text{ V}^{-1} \text{ s}^{-1})$ [23]. On the other hand, n-type doping in Ga₂O₃ can easily be achieved with effective donor concentrations ranging from 10^{15} cm^{-3} [78] to 10^{20} cm^{-3} [79]. Typical impurities that act as shallow donors in Ga₂O₃ are Sn [80], Si [81], or Ge [77].

According to Eq. 2.5, the electron mobility in a semiconductor material directly determines its specific resistance. The electron mobility in β -Ga₂O₃ at 300 K and a doping concentration below 10^{18} cm⁻³ is generally limited to 200 cm² V⁻¹ s⁻¹ due to phonon scattering [61]. High values of 194 cm² V⁻¹ s⁻¹ at room temperature approaching this theoretical limit have already been achieved experimentally [82]. In this respect, it is interesting to note that the formation of a 2DEG can be achieved at (Al_xGa_{1-x})₂O₃/Ga₂O₃ heterojunctions [83]. The electron mobility is expected to be much higher in these structures, which could enable the fabrication of (Al_xGa_{1-x})₂O₃/Ga₂O₃ HEMTs for high-frequency applications [61].

The drift velocity of charge carriers in a semiconductor is another central property. At low electric fields, it increases linearly with rising electric field strength. At high electric fields, electrons lose energy by the emission of optical phonons. Therefore, the drift velocity approaches a maximum value, the so-called saturation drift velocity. When devices are blocking, high fields are present in the space-charge region. Thus, it can be assumed that charge carriers in the space-charge region move with the saturation velocity, which therefore affects the switching behavior of the devices [84]. In β -Ga₂O₃, the saturation velocity was calculated to reach a peak of 2.0×10^7 cm/s at an electric field of 0.2 MV/cm, and then slightly decrease to about 1.5×10^7 cm/s at higher electric fields, where minor variations resulting from the crystal anisotropy are present [85]. This value is slightly lower than that of 4H-SiC, where about 2.0 MV/cm are approached at high electric fields [12, 86].

Thermal properties and cooling methods

The thermal conductivity in β -Ga₂O₃ was found to vary between (10.9 ± 1.0) W m⁻¹ K⁻¹ in the [100], (14.7 ± 1.5) W m⁻¹ K⁻¹ in the [001], and up to (27.0 ± 2.0) W m⁻¹ K⁻¹ in the [010] crystal direction [29]. Other studies report slightly varying values [87–89]. Based on these values, the thermal conductivity is typically assumed to be in the range of 10 to 30 W m⁻¹ K⁻¹ [12], which is at least one order of magnitude lower than that of SiC (about 270 W m⁻¹ K⁻¹ at room temperature [12]).

In addition, it was shown that oxygen vacancy defects can increase phonon scattering, which ultimately leads to a reduction of the thermal conductivity [90, 91]. Simulations in [92] suggest that the effect of Ga vacancies might be even stronger, with a reduction of the thermal conductivity of 15 % for 1 % oxygen vacancies and 28 % for 1 % gallium vacancies in Ga_2O_3



Figure 2.6: The thermal conductivity of β -Ga₂O₃ depends on the crystal direction, which is also expected to affect the heat dissipation in corresponding power devices. Diodes studied in this thesis are based on (001) oriented wafers. Therefore, the direct (vertical) cooling path, referred to as [001]^{*}, is about 14° off of the [001] crystal direction. Image of the crystal structure adapted from [93].

thin films. A similarly strong effect on the thermal conductivity was found when inducing linear defects into the crystal [92]. Note that these results are based on β -Ga₂O₃ thin films, whose thermal conductivity is generally lower than that of corresponding bulk crystals [92]. Nevertheless, the crystal quality could potentially also affect the thermal conductivity of thick Ga₂O₃ layers, which is important for the thermal studies conducted on power devices in this thesis.

The low thermal conductivity of Ga_2O_3 limits the heat dissipation through the substrate significantly and leads to a heat concentration at the anode side (i.e., the junction side) of the devices [94]. Furthermore, due to the strong anisotropy of the thermal conductivity in β -Ga₂O₃, the crystal orientation of the corresponding power devices can be expected to have a direct effect on the thermal management. As illustrated in Fig. 2.6, the devices studied in this thesis are based on (001) oriented wafers. Due to the slightly tilted crystal structure, the direct cooling path is not in [001] direction but about 14° off of this crystal direction, which will be referred to as [001]* in the following. Therefore, the thermal conductivity along the direct cooling path may differ from the above summarized literature values.

Some studies focusing on the issue of low thermal conductivity have been published recently. By transferring β -Ga₂O₃ thin films onto a SiC wafer, e.g., Ga₂O₃-on-SiC diodes were realized in [95]. Compared to diodes fabricated on bulk Ga₂O₃ wafers, the increase in junction temperature was reduced by one-quarter [95]. However, a procedure based on SiC substrates may contradict the potential cost and quality advantage of Ga₂O₃ resulting from the availability of native substrates.

Instead, cooling the devices from the junction side was suspected to be potentially effective for Ga_2O_3 [94]. During the course of this thesis, junction-side cooling of Ga_2O_3 diodes was reported to enable a junction-to-case thermal resistance of 1.48 K/W for a die with an anode area of 9 mm^2 sintered to a copper-molybdenum-copper substrate [96], and 0.5 K/W for a diode with the same anode area that was sintered to a silver plate [97]. Clearly, the reported values vary significantly. The different packaging methods and the fact that no reference devices with similar chip area were assembled in the same way in these studies prevents a thorough understanding of the effectivity of different cooling methods, let alone a comparison with other technologies such as SiC. Generally, as pointed out in [8], the heat dissipation in large-area Ga_2O_3 power devices is critical both from a performance and cost point-of-view, but rarely studied in the literature. Therefore, parts of this thesis target a comparable analysis of different thermal management methods for Ga_2O_3 power diodes.

In the case of α -Ga₂O₃, the thermal conductivity seems to be rather isotropic according to first principle calculations [65]. However, the calculated values of 11.6 W m⁻¹ K⁻¹ in [100] direction, 9.4 W m⁻¹ K⁻¹ in [010] direction, and 8.9 W m⁻¹ K⁻¹ in [001] direction are slightly lower than for β -Ga₂O₃. This might result from the bond strength of Ga and O atoms in α -Ga₂O₃, leading to stronger phonon scattering processes [65].

Thermomechanical properties

The coefficient of thermal expansion (CTE) was also found to be highly anisotropic for β -Ga₂O₃. In the range of 300 to 700 K, CTEs of 1.5×10^{-6} K⁻¹ in [100] up to 3.2×10^{-6} K⁻¹ in [001] and 3.4×10^{-6} K⁻¹ in [010] direction were reported in [98]. Slightly higher values were determined in [99]. Due to the limited number of available large-area devices, it has not yet been studied if the anisotropy of the CTE affects, e.g., the assembly and packaging procedure or reliability of the devices. As a comparison, the CTE of GaN increases from about 2.8 to 6.1×10^{-6} K⁻¹ [100] and the CTE of 4H-SiC from 2.2 to 4.5×10^{-6} K⁻¹ [101] in the same temperature range, respectively.

The mechanical properties of the semiconductor are critical not only from a reliability point-ofview but also regarding the device fabrication and wafer handling. One important mechanical property, the microhardness as measured by the Vickers indentation method also indicates a certain anisotropy in β -Ga₂O₃. The reported values are in the range of about 6.4 GPa for (010) oriented substrates [102], 8.3 to 8.9 GPa for (100) oriented substrates [102, 103], 10.1 GPa for (001) oriented substrates [102], and up to a maximum hardness of 12.5 GPa as specified by a substrate manufacturer for ($\overline{2}$ 01) oriented substrates [231]. It should be noted that the determined hardness depends on the force applied during the test as cracks and cleavages in certain crystal planes can form even at rather low test forces [104]. The values are similar to those typically reported for Si (about 9 to 10 GPa [105]) and significantly lower than those of 4H-SiC (about 32 GPa [106]). A similar observation is made for the Young modulus, which characterizes the stiffness of a material. It is about 230 GPa in β -Ga₂O₃ substrates [103, 231], which compares to about 160 GPa for Si [105, 107] and about 500 GPa for 4H-SiC [107, 108].

The comparison of Young's modulus and the microhardness shows that Ga_2O_3 is a rather soft material that, in contrast to 4H-SiC, can in principle be processed with standard tools known from Si wafer technology (such as cutting with a diamond saw). However, Ga_2O_3 is rather brittle and fragile in certain crystal directions. Associated cleavage planes with weak atomic bonds make the handling and dicing of Ga_2O_3 wafers with certain orientations such as (010) difficult [104], which is a new challenge for Ga_2O_3 device technology. Nevertheless, β -Ga₂O₃ wafers with or without high-quality epitaxial layer have been fabricated using different methods that will be summarized in the next section.

2.3.3 Growth and metal-semiconductor contacts

While SiC wafers need to be grown from the vapor phase at high temperatures above 2000 °C [109], and for the same reason GaN devices are mostly based on hetero-epitaxially grown GaN layers on Si or sapphire substrates (see Section 2.2.2) [39], β –Ga₂O₃ wafers can be grown from the melt with standard methods known from Si technology [18]. In this section, fundamental steps to fabricate basic vertical power devices, i.e., bulk growth, epitaxial growth, and the formation of ohmic and Schottky contacts, are discussed.

Wafer growth

The Czochralski (CZ) method is a well-understood standard method for the fabrication of largearea Si, sapphire, Ge, or GaAs wafers, and has been successfully adapted to grow β –Ga₂O₃ substrates [13, 14]. Due to high growth temperatures of 1800 °C, an Ir crucible is required to contain the Ga₂O₃ melt. This is associated with high cost [8] and the possibility of introducing impurities to the melt [110]. Furthermore, to scale up CZ-grown bulk crystals to sizes beyond 4 inches, a current challenge is to achieve a sufficiently high oxygen concentration in the entire melt [110]. Nevertheless, high-quality crystals have been fabricated at typical growth rates of 1 to 2.5 mm/h [13].

Edge-defined film-fed growth (EFG) is a standard method to grow large sapphire ribbons. For Ga₂O₃, it is especially appealing due to fast growth rates of currently 6 to 15 mm/h [15, 18]. EFG-based Ga₂O₃ substrates were the first ones to be commercialized [231] and device-quality wafers with a diameter of up to 6 inches have been realized [111, 232]. All Ga₂O₃ devices studied in this thesis are fabricated on EFG-grown wafers. Although an Ir crucible is used to contain the Ga₂O₃ melt, it seems to be less susceptible to oxidation and melt volatilization than in the case of CZ growth [18].

The two previously described methods are currently the most commonly applied ones to fabricate β -Ga₂O₃ wafers. Nevertheless, β -Ga₂O₃ substrates have also been grown using other techniques such as vertical Bridgman (VB) [16, 112] or optical floating zone (FZ) [17, 113]. While FZ-growth is feasible in principle as no crucible is required and a high oxygen atmosphere is possible, scaling up the wafer size beyond 1 inch seems to be hampered by difficulties in achieving sufficient heating of the material [18, 113]. The number of studies on VB-grown substrates is limited, although it was concluded in [18] that this method might hold a certain potential. So far, substrates with a diameter of 2 inches have been achieved, where the growth took place in ambient air without a seed crystal [112]. However, with 0.5 mm/h the growth rate was rather low [16]. The availability of β -Ga₂O₃ substrates fabricated by one of these methods constitutes the basis for the subsequent growth of homo-epitaxial Ga₂O₃ drift layers.

Epitaxial growth

Device-quality epitaxial Ga₂O₃ layers can be grown by a number of standard methods such as molecular beam epitaxy (MBE) [80], halide-vapor-phase epitaxy (HVPE) [114], metalorganic chemical vapor deposition (MOCVD) [115], low-pressure chemical vapor deposition [116], pulsed-laser deposition [117], physical vapor deposition [118], and mist chemical vapor deposition (mist-CVD) [119]. A thorough overview and comparison of the different techniques can be found, e.g., in [18] and [120]. Here, only the most important aspects will be covered.

The first gallium oxide power devices were fabricated by MBE [80]. Since then, the layer quality was improved significantly, and MBE was also employed to grow high-quality $(Al_xGa_{1-x})_2O_3/Ga_2O_3$ heterostructures forming a 2DEG [83]. Due to low growth rates below $0.5 \,\mu\text{m/h}$ [121], MBE is rather interesting for lateral Ga_2O_3 devices that do not require a thick drift layer [18]. Although an ultra-high vacuum is required, MBE is expected to enable uniform and large Ga_2O_3 epitaxial layers that are scalable for production [18].

Epitaxial layers grown by MOCVD have achieved very low unintentional doping, leading to record electron mobilities exceeding $190 \text{ cm}^2 \text{ V}^{-1} \text{ s}^{-1}$ at room temperature [82]. With up to $10 \,\mu\text{m/h}$ [122], the growth rates significantly exceed those achieved with MBE. Therefore,

MOCVD of Ga_2O_3 thin films could be scalable for production of both lateral and vertical β -Ga₂O₃ power devices [18].

Even higher growth rates of $25 \,\mu\text{m/h}$ on β -Ga₂O₃ substrates [123] and up to $250 \,\mu\text{m/h}$ on sapphire substrates [124] are possible with HVPE while achieving reasonable electron mobilities [18]. Therefore, HVPE is also a potential candidate for large-volume production of vertical Ga₂O₃ power devices with thick drift layers. A challenge for HVPE is high deposition temperatures that can trigger a diffusion of atoms between the epitaxial layer and the substrate or lead to the incorporation of Si or Cl impurities originating from the quartz reactor and precursor [18, 123, 125, 126]. It was suspected that this could affect the electrical properties of HVPE-grown films [126], as Cl is assumed to be a shallow donor in β -Ga₂O₃ [127]. These considerations are important because all devices studied in this thesis were fabricated by HVPE on EFG-grown substrates.

Ohmic and Schottky contacts to Ga₂O₃

Taking the availability of epitaxial wafers as a starting point, the next step towards the fabrication of gallium oxide power devices is the formation of ohmic and Schottky contacts to Ga_2O_3 as an integral part of every power device (see also Section 2.2.3). The properties of these contacts depend on the metal work function but also on the doping concentration of the semiconductor and the interface quality of the MS contact [128]. It was shown that defect states at the Ga_2O_3 surface can actually have a stronger impact on the contact properties than the metal work function itself [129, 130]. Furthermore, owing to the anisotropy of the β -Ga₂O₃ crystal structure, basic properties such as the barrier height depend strongly on the Ga₂O₃ wafer orientation [130].

Ohmic contacts to β -Ga₂O₃ are almost exclusively formed using Ti/Au, where Ti forms the ohmic contact and Au acts as a capping and protection layer [22, 23, 129, 131–135]. Compared to other metals (such as In, Ag, Sn, or Zr) it shows a high temperature stability and low specific contact resistances [129]. In [136], e.g., a specific contact resistance as low as $8.3 \times 10^{-7} \Omega \text{ cm}^2$ was achieved on highly n-doped Ga₂O₃.

The formation of Schottky contacts to Ga_2O_3 is possible with a variety of metals, including W, Cu, Ni, Ir, Pt [130], Au [137], or Mo [138]. The barrier height is then typically in the order of 1 to 1.5 eV [120, 130, 131, 134, 139]. In most Ga_2O_3 devices Ni [22, 23, 134, 140] or Pt [134, 140–142] is used as a Schottky metal. This lays the foundation for the fabrication of Ga_2O_3 power diodes, the concepts of which are discussed below.

2.3.4 Diode concepts

 β -Ga₂O₃ diodes based on native substrates have been fabricated on a lab scale with various lateral and vertical structures. Currently, β -Ga₂O₃ power devices are not commercially available, and research is mostly restricted to the on-wafer characterization and analysis of basic structures with lateral dimensions of several hundreds of µm [18, 134]. In this thesis, first diced and partly standard molded diode chips from a planned production line that significantly exceed these dimensions and thus enable more application-oriented tests will be investigated. The most promising diode concepts, including the structures of the diodes under investigation in this thesis, are summarized in the following.



(b) Vertical p-NiO/ β -Ga₂O₃ heterojunction diodes.

Figure 2.7: Overview of promising vertical β -Ga₂O₃ diode structures based on lightly doped epitaxial layers deposited on heavily doped native substrates. Both (a) fully unipolar and (b) heterojunction devices have been fabricated with ion-implanted guard rings (GR), field plates (FP), and beveled structures. Some of the diodes studied in this thesis have a structure as shown in (a)-(1). Adapted from [22, 23, 131–133, 149].

Vertical planar diodes and edge terminations

The first β -Ga₂O₃ diode introduced in 2013 consisted of simple planar structures without edge termination [21]. The diode was based on an unintentionally doped FZ-grown substrate without an additional epitaxial layer and showed a basic rectifying behavior [21]. As shown in Fig. 2.7, most diodes developed since then consist of a heavily doped melt-grown β -Ga₂O₃ substrate, typically with an effective donor concentration in the range of 10^{18} cm⁻³ to more than 10^{19} cm⁻³ that is followed by a high-quality epitaxial layer which is usually around 10 µm thick and exhibits an effective donor concentration in the order of 10^{16} cm⁻³ [22, 131, 132, 134, 135, 143]. Ohmic and Schottky contacts are typically formed with Ti/Au and Ni/Au or Pt/Au, respectively, as explained in Section 2.3.3, which leads to knee voltages typically in the range of around 0.8 to 1 V [52, 131, 142, 144–148] but also up to 1.8 V have been reported [149].

Without edge termination, the peak electric field at the edge of the anode is substantially higher than in the volume of the drift layer, which leads to a premature reverse breakdown of the device. To fully exploit the high theoretical breakdown field of gallium oxide, different edge termination structures such as field plates [150], guard rings [149], floating potential rings [132, 151], or beveled structures [131] that lead to a reduction of the peak electric field and an enhancement of the breakdown voltage have been implemented successfully.

Figure 2.7a–(1) shows a planar Schottky diode with field plates and guard rings. In this thesis, planar diodes without and with such kind of edge termination are studied. Field plates are often based on SiO₂ [254, 134, 149, 150] or Al₂O₃ [134, 139, 152]. Guard rings have been fabricated by N-ion [134, 149] or Mg-ion implantation [134, 153], but p-AlGaN structures [154] or the thermal oxidation of the anode edge region [146] have also been reported to be effective methods. As shown in Fig. 2.7b, also beveled structures with spin-on-glass and SiO₂ field
plates have been successfully realized [131]. Figure 2.7a–(3) shows a rather simple structure, with deep SiO₂ trenches [22]. With all of these structures, breakdown voltages exceeding 1 kV and PFOMs of several hundreds of MW/cm² have been achieved. The structure with deep SiO₂ trenches (anode diameter of 180 µm) enabled a low $r_{o,sp}$ of $3.4 \text{ m}\Omega \text{ cm}^2$ at a V_{br} of 6 kV. The corresponding PFOM of 10.6 GW/cm² exceeds the unipolar limit of SiC and GaN [22]. Recently, a Ga₂O₃ diode with a V_{br} of 1720 V and a high PFOM of 1.32 GW/cm² was realized even without edge termination. This was achieved by plasma etching to remove donor-like impurities at the surface of the drift layer [144].

Due to the absence of p-type doping in Ga₂O₃ (see Section 2.3.2), other p-type semiconductors can be implemented to form edge terminations or heterojunctions. In this respect, p-type NiO has recently gained the most interest. The diode shown in Fig. 2.7b–(1), e.g., consists of alternating Ni/n-Ga₂O₃ Schottky contacts and p-NiO/n-Ga₂O₃ heterojunctions, as well as field-limiting p-NiO rings as edge termination. At a still reasonable knee voltage of 1.1 V a $V_{\rm br}$ of 1.89 kV and an $r_{\rm o,sp}$ of 7.7 m Ω cm² were achieved [132]. For full p-NiO/n-Ga₂O₃ heterojunction diodes, the knee voltage increases to typically about 1.8 to 1.9 V [23, 133, 155]. However, such structures have achieved high PFOMs. The heterojunction diode in Fig. 2.7b–(3) with two differently doped p-NiO layers, Mg-implanted guard rings and SiO₂ double field plates, and an anode diameter of 150 µm achieved a record PFOM of 13.2 GW/cm² [23]. The corresponding $r_{\rm o,sp}$ of 5.24 m Ω cm² at a $V_{\rm br}$ of 8.32 kV is about one third of the unipolar SiC limit [23].

Recently, first rudimentary switching tests have been conducted for Ga_2O_3 Schottky and heterojunction diodes [96, 133, 155–158]. The studies proved the basic benefits of unipolar Ga_2O_3 Schottky diodes, such as the absence of reverse recovery. All devices that could sustain application-relevant reverse voltages of at least 400 V were limited to lateral dimensions of only some hundreds of μ m up to a maximum anode area of 1 mm² for a heterojunction diode in [133, 155]. Although a first diode operation in a PFC circuit was demonstrated, the achieved current slew rates were well below 1 A/ns [133, 155]. Furthermore, other application-critical parameters, such as the voltage slew rates and waveforms, were not studied and a comparison to mature technologies such as SiC is missing. This motivates the thorough and comparable investigation of switching properties of novel 600 V rated Schottky diodes with anode areas up to 10 mm² performed in this thesis.

Vertical trench diodes

In Schottky barrier diodes there is an intrinsic trade-off between the reverse leakage current and the forward knee voltage, both of which are determined by the Schottky barrier height in an opposite manner. Furthermore, the maximum electric field $E_{\rm max}$ occurs at the Schottky junction and directly determines the leakage current. Since basic MPS structures such as in the case of SiC (see Section 2.2.2) cannot be implemented to resolve this issue due to the lack of p-Ga₂O₃, fabricating trench-MOS structures such as illustrated in Fig. 2.8 is another promising concept for Ga₂O₃ diodes [159–161].

Basically, the additional vertical PN junctions found in MPS diodes are replaced by MOS junctions at the sidewalls of the vertical trenches. Because of that, the depletion region formed during blocking operation does not only extend vertically but also horizontally. As a consequence, the electric field at the surface E_{surf} is reduced and the peak electric field is shifted towards the volume of the chip. Therefore, the effect is often referred to as reduced surface field (RESURF) effect [160, 162] and can be quantified by the field reduction factor $\mathcal{R} = E_{surf} / E_{max}$ [160]. The exact field distribution and diode properties depend critically on



Figure 2.8: (Left) Basic structure of a field-plated β -Ga₂O₃ trench MOS diode. (Middle) The topview SEM and X-ray images of a trench MOS diode studied in this thesis show the field plates and the about 3 µm wide trenches of a diced chip that is bonded by soldering and wire-bonding in a TO247 package. (Right) Two Al bond wires can be seen after removing the mold mass with a laser. Adapted from [254].

the trench depth, width, distance, crystal orientation, the doping profile, and the dielectric that needs to withstand high electric fields [160, 163, 164].

In [152] a Ga₂O₃ trench diode with field plates and Al₂O₃ dielectric achieved a breakdown voltage of 2.89 kV and an $r_{0,sp}$ of 8.8 m Ω cm², resulting in a promising PFOM of 0.95 GW/cm². So far, however, the small number of Ga₂O₃ trench diodes that have been subjected to temperature-dependent analysis in the literature showed strong junction inhomogeneities as characterized by a large change of ideality factor and barrier height with temperature [165, 166].

Therefore, a temperature-dependent characterization of a trench diode with HfO_2 instead of Al_2O_3 dielectric is performed in this thesis. The scanning electron microscopy (SEM) image in Fig. 2.8 clearly shows the trench and field-plate structure of this type of diode. As can be seen from the X-ray images, the chip is located in the center of a fully molded TO247 package. More information on the structure is provided in the respective chapters.

Lateral diodes

Several lateral β -Ga₂O₃ diodes have also been demonstrated. In [139] the existence of certain cleavage plane in β -Ga₂O₃ (see Section 2.3.3) was used to exfoliate 600 to 650 nm thin nano-membranes simply by using scotch tape and transfer them to a sapphire substrate, motivated by their low cost but better thermal conductivity. The resulting diode with Al₂O₃ field plates achieved a PFOM of 500 MW/cm² at a breakdown voltage of 2.25 kV. In [167] native substrates were used to grow lateral NiO/(Al_xGa_{1-x})₂O₃/Ga₂O₃ heterojunction diodes. The corresponding devices achieved high breakdown voltages above 7 kV, however, at low PFOMs below 1 MW/cm². A significantly higher PFOM of 1.34 GW/cm² at a breakdown voltage of 1.5 kV was achieved by fabricating lateral diodes with trench-MOS structures on native substrates [168]. Given the focus of this thesis is on vertical diodes, which are typically preferred in power electronics, please refer to the respective publications for more details.

2.3.5 Transistor concepts

Transistors based on Ga_2O_3 are currently at a significantly earlier stage of development than Ga_2O_3 diodes. Nevertheless, both vertical and lateral β – Ga_2O_3 transistor cells with lateral dimensions of some tens to some hundreds of μ m have been fabricated. Two vertical transistor structures that are currently considered promising candidates are planar FETs with a current blocking layer (CBL) and FETs with a trench structure.



Figure 2.9: Schematic structures of promising vertical β–Ga₂O₃ enhancement mode transistor concepts. (a) Planar FET with N-ion or Mg-ion implanted current blocking layer (CBL). (b) FinFETs with optimized trench structure have achieved the highest PFOMs and breakdown voltages so far. Adapted from [24, 169, 170].

Fig. 2.9a shows the basic structure of a planar MOSFET that was introduced in [169]. Since Ga_2O_3 cannot be effectively p-doped, N-ion or Mg-ion implantation is used instead to create a high-resistive CBL below the heavily n-doped source contacts. It electrically isolates the drain from the source contact, thus limiting the drain–source leakage current, and forms the conductive aperture through which the current flows. The N⁺⁺-region an the n-doped Si⁺ implanted channel region basically form a PN junction that fully depletes the channel from charge carriers if the donor concentration in the channel is sufficiently low, thus resulting in a normally-off behavior. Due to the simple planar structure with implantation steps, the device is attractive from a mass production point-of-view. However, it only reached a V_{br} of 263 V at a $r_{o,sp}$ of 135 m Ω cm² [169]. A recent news release claimed a V_{br} exceeding 1 kV for a similar structure [233].

Higher breakdown voltages and PFOMs have been achieved with trench or FinFET structures exemplarily shown in Fig. 2.9b. Initial FinFETs with normally-off behavior showed breakdown voltages exceeding 1 kV and PFOMs of 125 MW/cm^2 [170, 171]. By introducing a post-deposition annealing step, the channel mobility was significantly increased to about $130 \text{ cm}^2 \text{ V}^{-1} \text{ s}^{-1}$, leading to PFOM of 280 MW/cm^2 at a breakdown voltage of 2.66 kV [24]. Although other promising structures such as the vertical trench U-gate FET [172] or lateral FETs with trench gate [25] were demonstrated, the PFOMs of diodes are not yet achieved, and only one group has demonstrated a large-area (gate width 10 mm) MOSFET with lateral structure [173, 174]. First switching tests indicated a dynamic increase in on-resistance preventing experiments above DC link voltages of 300 V [173] and 400 V [174], respectively. Since Ga₂O₃ transistors are not available in this thesis, the reader is referred to the corresponding publications for more information.

2.4 Summary and basic research questions arising

The current state of knowledge and the main issues that form the guiding questions for this work can be summarized as follows:

1. β -Ga₂O₃ is a highly anisotropic material that combines a high critical electric field with the availability of melt-grown and comparatively soft wafers. So far, no particular device structures have prevailed.

- 2. Research mostly concerns basic structures with lateral dimensions of some hundred μm and lacks application-oriented analyses. Especially the switching properties and the impact of low thermal conductivity are essential, but have hardly been examined so far.
- 3. The following aspects are thus identified as critical for the path towards application:
 - a) understanding the basic properties of novel device structures with respect to their potential application in power electronics
 - b) examination of the thermal properties and the effectiveness of different heat dissipation strategies
 - c) transition to large-area devices that allow first application-oriented studies of the switching properties

In this work, the first β -Ga₂O₃ diodes from a planned industrial wafer and device fabrication line are analyzed. This allows the formulated problems to be addressed gradually.

Chapter 3

Experimental methods: sample preparation, measurement, and evaluation

3.1 Chapter overview

This chapter introduces the main experimental methods used to study a wide range of characteristics of Ga_2O_3 diodes, from basic physical properties to application-oriented parameters. To this end, Section 3.2 first introduces the various Ga_2O_3 diodes under investigation and their different packaging and assembly processes. Section 3.3 briefly discusses the measurement of forward and reverse current–voltage profiles and the junction capacitance, as well as the corresponding evaluation procedures. Diodes that can sustain sufficiently large reverse voltages can be subjected to double pulse tests in order to study their switching behavior. Capturing the associated short switching transients is challenging but vital to draw sound conclusions about the semiconductor properties. In Section 3.4, novel current sensor technologies are therefore evaluated and the most suitable ones are selected for further studies. For any continuous operation, the thermal resistance of the power devices is a central aspect. Section 3.5 therefore examines the forward voltage drop as a temperature-sensitive parameter to estimate the junction temperature and introduces the transient thermal measurement methods applied to Ga_2O_3 diodes. Parts of this chapter have been published in [255, 260] by the author during the course of this thesis.

3.2 Packaging and overview of the investigated samples

Assembly on custom substrates

Gallium oxide devices are not yet commercially available and as such cannot be purchased in standardized packages. While some diodes were already assembled on TO247 lead frames by the device manufacturer, other samples were received as bare dies that require further packaging. Therefore, a suitable custom substrate was designed that enables all types of planned measurements, and in particular fulfills the following requirements: First, it should enable the assembly of bare dies with different thicknesses, lateral dimensions and surface metallizations. Second, the package should enable effective cooling via different methods such as standard cathode-side cooling but also junction-side cooling, and thermal imaging of the bare die should be possible for these setups. Third, the package must allow a low-inductive assembly on a printed circuit board to enable fast switching tests. Fourth, the package should be easy to handle (e.g., with regard to mounting on a circuit board, connection to test fixtures, or mounting on heat sink) and at the same time protect the die sufficiently.



Figure 3.1: Custom package used to enable a wide range of different experiments. It is based on an aluminum nitride ceramic substrate with silver-plated copper pads on both sides. Plated through contacts at the sides of the substrate allow for a low-inductive SMD assembly on a printed circuit board. The silver plating facilitates different assemblies with either the cathode or the anode facing downwards. The terminal that is facing upwards can either be connected with bond wires in case of bottom-side cooling, or a metal plate can be attached to enable top-side or double-side cooling. A solder stop layer and 3D-printed plastic frame protect the die, while a mounting hole can be used to attach the package to a heat sink.

To meet the requirements, a package based on an aluminum nitride ceramic is designed. The manufacturing of the substrate is done externally by a contractor. The basic structure and two examples with an assembled Ga_2O_3 die are shown in Fig. 3.1. Aluminum nitride is the ceramic of choice for high thermal conductivity with about $170 \text{ W m}^{-1} \text{ K}^{-1}$ compared to about 30 to $70 \text{ W m}^{-1} \text{ K}^{-1}$ for Si_3N_4 or 20 to $30 \text{ W m}^{-1} \text{ K}^{-1}$ for Al_2O_3 , combined with high electrical insulation capabilities (dielectric strength about 14 to 20 kV/m) [175–177]. The high thermal conductivity reduces the effect the package has on the junction-to-case thermal resistance, which is helpful when studying the thermal properties of different dies.

The total package exhibits lateral dimensions of $10.7 \text{ mm} \times 16 \text{ mm}$. The ceramic is 0.3 mm thick and has $70 \,\mu\text{m}$ thick direct-plated copper tracks on both sides. The copper tracks are silver metallized, which prevents oxidation and simplifies silver sintering while also enabling standard wire bonding with aluminum or gold bond wires. In this way, a large degree of freedom for different surface metallizations of the die is achieved.

The front side consists of three different copper areas. The upper pad provides mechanical support and contains an M2 mounting hole to attach the substrate to a heat sink with a screw. The bottom right pad constitutes the die bonding area. Here, one electrode of the die can be bonded directly to the substrate, e.g., using soldering or sintering methods. In this way, an electrical and thermal coupling with the substrate is established. The bottom left hand pad serves as electrical connection to the second electrode of the diode, e.g., by wire-bonding or by attaching copper clips via soldering or sintering.

A solder stop layer is deposited on the front side to prevent solder from flowing to the die and potentially harming it. Optionally, a 3D-printed plastic frame is glued around the chip to facilitate safe handling and to enable silicone potting if required. The polarity of the die bonding pad changes depending on whether the cathode or anode is bonded to it. This in turn also changes the polarity of the package terminals. Therefore, an almost identical second version of the substrate was designed. It only differs in the fact that the two electrode pads on



Figure 3.2: The custom package can be mounted either horizontally or vertically on a printed circuit board. Simulations in Ansys Q3D for the assembly method with a metal plate show that a parasitic inductance as low as 0.9 to 1.4 nH for frequencies between 1 GHz and 1 kHz can be achieved with the horizontal assembly, which enables fast switching experiments.

the front side are mirrored. As a result, when choosing the appropriate package for a certain assembly, the polarity of the package terminals remains the same.

The back side of the ceramic substrate also consists of three different copper pads. The large copper pad in the middle is electrically isolated and can serve as a connection to a heat sink in order to dissipate the heat created in the chip. Two narrow copper pads at the sides are connected to the two diode pads by copper plated through contacts through the ceramic. These two pads enable a low-inductive surface mounting of the package on a printed circuit board to enable fast switching. In such a surface mount assembly, the die can either be cooled from the bottom side, or from the top side when a metal plate is attached to the top-side of the die as shown on the right in Fig. 3.1. Alternatively, the package can be assembled on a circuit board in a vertical way, similar to a TO220 package without leads.

The parasitic inductance of the package is estimated using the Ansys Q3D Extractor [234]. To this end, a $3 \text{ mm} \times 3 \text{ mm}$ large and 0.6 mm thick die that is assembled with a 0.5 mmthick copper plate according to Fig. 3.1 is assumed. As shown in Fig. 3.2, the electrical loop is closed 200 μ m below the backside of the substrate by a 70 μ m thin copper plate to model the horizontal assembly. In this case, the area that is enclosed by the loop is rather small and thus a comparably low inductance is expected. In case of the vertical assembly the loop is closed at the front side of the substrate, which should increase the parasitic inductance. In both cases, a drop in inductance for increasing frequencies is expected as the proximity effect leads to a concentration of the current distribution at the inward-facing surface of the conductor [178, 179]. In agreement with these considerations, the simulated stray inductance for frequencies between 1 kHz and 1 GHz is 1.4 to 0.9 nH for the horizontal, and 4.0 to 2.0 nH for the vertical assembly. Note that the exact assembly method (such as assembly with bond wires instead of a metal plate, or size, thickness and position of the die) can affect the calculated parasitic inductance. Yet, the achieved inductance is several times lower than that of a standard TO220 package, which is in the range of 5 to 8 nH per lead [180]. Low-inductive switching tests are therefore possible with the introduced assembly method.

The functionality of the package is verified by assembling commercial 650 V/10 A rated SiC diode chips (Cree CPW2-0650-S010B [235]) using either soft soldering or pressureless silver sintering to attach the die to the substrate, and aluminum wire bonding to electrically connect the anode. The resulting temperature-dependent forward and reverse current–voltage characteristics and capacitance–voltage profiles are in agreement with the datasheet specifications

Table 3.1: Overview of β -Ga₂O₃ devices studied in this thesis. The samples differ in junction structure, edge termination, i.e., field plates (FP) or guard rings (GR), substrate thickness z_{sub} , the characteristic anode size (XS, ML, L, and LL) and metallization, and the packaging, i.e., either on a TO247 lead frame or the custom direct plated copper (DPC) ceramic substrate.

| Batch | Junction | Edge | $z_{ m sub}$ | Die size | Anode size | Size | Anode | Package |
|-------|-----------|--------|--------------|------------------|------------------|------|-------|---------|
| no. | structure | term. | (µm) | (mm) | (mm) | code | metal | type |
| B1 | trench | FP | 500 | _ | Ø = 0.3 | XS | Al | TO247 |
| B2a | planar | none | 200 | 2.2×3.8 | 1.6 	imes 1.6 | ML | Ni/Au | TO247 |
| B2b | planar | none | 200 | 2.2×2.2 | 1.6 	imes 1.6 | ML | Ni/Au | DPC |
| | planar | none | 200 | 2.2×3.8 | 1.6 	imes 1.6 | ML | Ni/Au | DPC |
| B2c | planar | none | 600 | 2.2×2.2 | 1.6 	imes 1.6 | ML | Al | DPC |
| | planar | none | 600 | 3.8	imes 3.8 | 3.2×3.2 | LL | Al | DPC |
| B3 | planar | FP, GR | 600 | 3.4×4.2 | 1.6 	imes 1.6 | ML | Ni/Au | TO247 |
| B3 | planar | FP, GR | 600 | 3.3 	imes 3.3 | 2.4×2.4 | L | Ni/Au | TO247 |
| B3 | planar | FP, GR | 600 | 4.1×4.1 | 3.2×3.2 | LL | Ni/Au | TO247 |
| B4 | planar | FP, GR | 200 | 3.4×4.2 | 1.6 	imes 1.6 | ML | Al | DPC |
| B4 | planar | FP, GR | 200 | 4.2×4.2 | 3.2×3.2 | LL | Al | DPC |

(see also Section 3.3). In addition, double pulse tests are carried out in Section 3.4.6 to confirm the suitability for low-inductive switching tests.

Overview of the types of investigated samples

Table 3.1 provides an overview of the different types of β -Ga₂O₃ diodes studied in this thesis, starting with the first batch received (B1) and ending with the final batch received (B4). The basic structure of state-of-the-art β -Ga₂O₃ diodes is explained in Section 2.3.4. For all samples that have an edge termination, field plates (FP) are based on SiO₂ and guard rings (GR) are nitrogen-ion implanted. The junction is completely unipolar for all samples, i.e., no p-type structures (such as p-NiO) are implemented.

While all samples are Ni/Au metallized on the cathode side of the chip, the anode metallization is either Ni/Au or Al. This affects the assembly process of the dies: On Ni/Au surfaces, soldering and pressureless silver sintering can be performed easily in addition to gold wire bonding [181]. For aluminum metallized surfaces, a thin oxide layer on the metal surface prevents the formation of a bond with conventional soldering or sintering methods [181, 182]. Therefore, only aluminum bond wires are used to electrically connect aluminum metallized anodes.

The first sample received (B1) has a Mo-Ga₂O₃ Schottky contact and a circular anode with a diameter of 0.3 mm. All other diodes have a Ni-Ga₂O₃ Schottky contact and a square anode with one of the sizes ML ($1.6 \text{ mm} \times 1.6 \text{ mm}$), L ($2.4 \text{ mm} \times 2.4 \text{ mm}$), or LL ($3.2 \text{ mm} \times 3.2 \text{ mm}$). Note that some ML-size samples were diced out in a rectangular instead of a square shape. As will be shown in Section 5.2, this has no significant influence on the electrothermal properties. All samples that were obtained as bare dies were assembled on AlN substrates in one of the previously presented ways, depending on the type of available diode and the measurement goal.

Other samples were received already in TO247 packages. Among these samples, only diode B1-XS was attached directly to the lead frame via soldering (cathode) and wire bonding (anode). For all other TO247 packaged samples, the die is soldered to a 0.5 mm thick CuMo spacer, which itself is soldered to the lead frame. Such spacers can serve as a buffer layer to

Table 3.2: Overview of commercial reference devices used in this thesis. For TO packaged devices, the die and anode size were estimated by optical microscopy after removing the mold mass with a laser. Bare dies were assembled on custom direct plated copper (DPC) ceramic substrates.

| Type/ | Diode | Part | Die size | Anode size | Package | Ref. |
|-------|-------------------------------|---------------|--------------------|--------------------|---------|------------|
| no. | rating | number | (mm) | (mm) | type | |
| SiC-A | $650\mathrm{V/10A}$ | CPW20650S010B | 1.92×1.92 | 1.65×1.65 | DPC | [235] |
| SiC-B | $650\mathrm{V/16A}$ | STPSC16H065A | 2.45×2.00 | 2.25×1.80 | TO247 | [236] |
| Si-A | $600 \mathrm{V/10}\mathrm{A}$ | DHG20C600PB | 3.00 	imes 3.00 | 1.80×1.80 | TO220 | [237, 238] |

reduce thermal strain due to differences in the thermal expansion between die and lead frame. Furthermore, all samples of batch B3 and one sample of batch B2a were sealed with silicone gel instead of being fully molded. More specific information about the diode structure or the assembly process can be found in the respective chapters, if relevant.

In addition to the Ga_2O_3 samples, commercial diodes listed in Tab. 3.2 are used as reference devices to investigate characteristic similarities and differences between them. To determine the die and anode size of the TO packaged diodes, their mold mass was removed by laser ablation and the corresponding dimensions were determined by optical microscopy. The anode sizes of the reference diodes are similar to those of the Ga_2O_3 samples in Tab. 3.1, which enables a fair analysis and comparison of their electrical and thermal properties. This is particularly important for the results derived in Chapters 5 and 6, as will be discussed in more detail in the respective places. Important measurement procedures to examine the basic features of the corresponding devices are explained in the following sections.

3.3 Evaluation of static electrical properties

General remarks

Many characteristic diode properties can be analyzed by measuring their forward and reverse current–voltage characteristics and the capacitance–voltage profiles. In this work, a Keysight B1505A power device analyzer [239] is used for this purpose. As this device is widely known, only the most important measurement and evaluation procedures will be explained in the following by the example of the commercial reference diode SiC-A. This on the one hand validates the measurement settings and on the other hand verifies the functionality of the AlN-DPC package introduced in Section 3.2.

For temperature-dependent measurements, the samples are heated with air in a closed chamber using a Temptronic ATS-545 Thermostream and a k-type thermocouple to determine the temperature. In case of the TO-packaged samples, the thermocouple is placed on the metal lead frame below the presumed chip location. A thin heat conductive sheet is used to electrically isolate the sensor from the lead frame. In the case of bare dies assembled on AlN-DPC substrates, the sensor is placed on the electrically isolated back side cooling pad below the die. After reaching a ± 0.5 K window around the target temperature, a waiting period of 15 min is initiated. This ensures a uniform temperature and allows the assumption that the junction temperature is equal to the temperature measured with the thermocouple.

Current-voltage (I-V) characteristics

Regarding the forward characteristics, two different types of measurements are performed. The first one shown in Fig. 3.3a is performed under dc conditions, which provides the highest

resolution (typically 100 pA) at low currents below the knee voltage. Under the assumption of current transport via thermionic emission, such measurements can be used to gain insight into the basic junction properties. When plotting the data semi-logarithmic, the saturation current I_s and the corresponding saturation current density $J_s = I_s/A$, where A is the active area of the device, can be determined from the intercept of a fit to the linear part of the low-current region below the knee voltage but at voltages larger than $3kT_j/e$ [45]. The Schottky barrier height $\phi_{b,0}$ is then calculated from the saturation current density according to [45]

$$\phi_{\mathbf{b},0} = \frac{kT_{\mathbf{j}}}{e} \cdot \ln\left(\frac{A^{**} \cdot T_{\mathbf{j}}^2}{J_{\mathbf{s}}}\right) \,, \tag{3.1}$$

where k is the Boltzmann constant, T_j is the junction temperature, e is the elementary charge, and A^{**} is the Richardson constant. For β -Ga₂O₃, A^{**} is assumed to be 41 A cm⁻² K⁻², which is calculated from its effective electron mass m_e^* of 0.34 times the electron rest mass [21, 28]. The ideality coefficient n results from the slope [45]

$$n = \frac{e}{kT_{\rm i}} \frac{dV_{\rm D}}{d\ln(J_{\rm D}/J_0)}, \qquad (3.2)$$

of the fit where V_D is the diode voltage, J_D is the current density, and J_0 is a reference current density of 1 A/m^2 . The built-in potential ψ_{bi} across the junction without image-force lowering then follows from [45, 183]

$$e\psi_{\rm bi} = e\phi_{\rm b,0} - (E_{\rm C} - E_{\rm F}) = e\phi_{\rm b,0} - kT_{\rm j}\ln\left(\frac{N_{\rm C}}{N_{\rm D}}\right),$$
(3.3)

where $E_{\rm F}$ is the Fermi energy, $E_{\rm C}$ is the energy of the conduction band minimum, and $N_{\rm C} = 2(2\pi m_{\rm e}^* kT_{\rm j}/h^2)^{3/2}$ is the effective density of states in the conduction band. The effective donor concentration $N_{\rm D}$ can be calculated from capacitance–voltage measurements, as will be discussed in more detail later.

The second type of measurement shown in Fig. 3.3b concerns the high-current region of the forward characteristics. Here, only 200 μ s short voltage pulses are applied within a pulse period of 200 ms. This pulse length, which is also commonly found in commercial datasheets [240], is long enough for oscillations in the current and voltage waveforms to fade but short enough to prevent excessive self-heating of the device. The differential on-resistance is obtained from the slope of a straight line fitted to the profile. For Ga₂O₃ diodes, current ratings or other datasheet specifications are not yet defined by the manufacturer and thus not known. The fitting interval is therefore chosen to be within the linear part of the profile, i.e., above the knee voltage but well below the point where self-heating at high currents causes a rise in differential on-resistance. If not specified otherwise, a fitting range from 2 A to 4 A for ML-size diodes, 3 A to 6 A for L-size diodes, and 6 A to 10 A for LL-size diodes is chosen.

Finally, the reverse current-voltage characteristics are measured via a dc voltage sweep. Figure 3.3c shows a typical profile, indicating that the current resolution for this type of measurement is at least 100 pA. In Fig. 3.3c, the reverse bias was increased until breakdown occurred at a voltage of $V_{\rm br} = -1140$ V. For an ideal planar Schottky diode with a triangular electric field, the average breakdown field can then be calculated from [34]

$$E_{\rm br} = \sqrt{2eN_{\rm D}V_{\rm br}/(\epsilon_0\epsilon_{\rm s})}\,,\tag{3.4}$$



Figure 3.3: (a) Forward current–voltage (*I–V*) measurements at low currents with a resolution of about 100 pA are used to determine the saturation current, ideality factor, and Schottky barrier height. (b) The differential on-resistance r_o is calculated from a linear fit to the pulsed forward *I–V* profiles above the knee voltage but below the onset of substantial self-heating. (c) The current resolution for reverse *I–V* measurements is about 100 pA. (d) The effective donor concentration is proportional to the square of the inverse junction capacitance, which allows conclusions about differences in doping profiles. Comparing the profiles measured with the SiC-A reference diode with the datasheet validates the measurement settings and confirms the functionality of the custom AlN-DPC package.

if the effective donor concentration N_D is known. Here, ϵ_0 is the vacuum permittivity and ϵ_s is the relative permittivity of the semiconductor (10 to 12 for β -Ga₂O₃ [184]).

Capacitance-voltage (C-V) characteristics

For voltage-dependent capacitance measurements, a reverse dc bias voltage is superimposed upon an ac measurement voltage. The change dV in the ac voltage signal leads to a displacement dQ of the charge stored in the depletion region, through which the associated capacitance is determined [45]. If not specified otherwise, the measurement frequency and amplitude are set to 1 MHz and 25 mV. Based on the C-V profile, the associated capacitive charge Q_c and capacitance stored energy E_c can be calculated from

$$Q_{\rm c}(V_{\rm D}) = \int_0^{V_{\rm D}} C_{\rm j}(V_{\rm D}') \,\mathrm{d}V_{\rm D}' \tag{3.5}$$

$$E_{\rm c}(V_{\rm D}) = \int_0^{V_{\rm D}} C_{\rm j}(V_{\rm D}') V_{\rm D}' \,\mathrm{d}V_{\rm D}'\,, \qquad (3.6)$$

where $V_{\rm D}$ takes negative values. Capacitance–voltage measurements can also be used to determine the effective donor concentration $N_{\rm D}$ of the epitaxial layer of the investigated samples. When the reverse bias voltage $-V_{\rm D}$ is increased, the depletion region extends further into the epitaxial layer. From Eq. 2.4 it follows that $N_{\rm D}$ can directly be determined from the slope of a $C_{\rm j}^{-2}$ versus $V_{\rm D}$ plot [183]

$$N_{\rm D} = \frac{-2}{e\epsilon_0\epsilon_{\rm s}A^2} \cdot \left(\frac{dC_{\rm j}^{-2}}{dV_{\rm D}}\right)^{-1}.$$
(3.7)

Figure 3.3d shows the corresponding plot exemplarily for a SiC-A reference diode and a B3-LL type Ga_2O_3 diode. To take into account the difference in the active area, the capacitance is normalized to the respective anode area. For various analyses, only the background doping concentration near the Schottky junction is of interest. If not specified otherwise, the fitting interval is therefore restricted to the low voltage region up to -3 V. In case of the Ga_2O_3 diodes, this is also reasonable as the practically constant slope up to -600 V suggests a rather homogeneous doping concentration through the entire thickness of the epitaxial layer. Taking the SiC-A reference diode as another example, the slope decreases drastically for voltages below -200 V, which suggests an increase in the effective donor concentration.

Once the basic electrical properties of the Ga_2O_3 samples are known and it is confirmed that they can sustain a sufficiently large reverse voltage, they can be subjected to switching tests. As shown in the following section, these tests require advanced sensor technologies to achieve reliable results.

3.4 Capturing nanosecond switching transients

3.4.1 General basics of double pulse testing

Basics and motivation

The dynamic electrical properties of power semiconductor devices can be studied by assembling them in a buck converter or half-bridge topology (see Fig. 2.1a) and performing double pulse tests. The probes used to capture the transients can significantly affect the measurement accuracy. Current sensors are especially critical components, as they have to be placed in the current path of the commutation loop. This affects the commutation-loop inductance and with that also the behavior of the switching cell. To enable a sound analysis of the switching properties of novel Ga_2O_3 diodes, it is therefore vital to select a sensor with an acceptable tradeoff between parasitic inductance and bandwidth. An additional galvanic isolation of the sensor is advantageous to facilitate potential-independent measurements. Since common current sensors do not provide all necessary features to accurately capture switching transients in the range of just some nanoseconds, this section investigates novel current sensors. Parts of Section 3.4 have been published in [257] during the course of this thesis.

Measurement uncertainty considerations

The true impulse rise time $T_{a,true}$ of a signal, usually defined between 10% and 90% of the final value, can only be determined with a certain accuracy. The inherent rise times $T_{a,sense,i}$ of the measurement instruments lead to an experimentally determined rise time of $T_{a,exp} = (T_{a,true}^2 + \Sigma_i T_{a,sense,i}^2)^{1/2}$ [185, 186]. The error associated with the measurement can therefore be expressed by [186]

$$K_{\rm bw} = \frac{T_{\rm a,exp}}{T_{\rm a,true}} - 1 = \sqrt{1 + \frac{\Sigma_i T_{\rm a,sense,i}^2}{T_{\rm a,true}^2}} - 1.$$
(3.8)

Instead of $T_{a,sense,i}$, often the equivalent -3 dB bandwidth B_i of the components, which is related to the rise time by $B_i = 0.35/T_{a,sense,i}$ [185, 186], is specified by the manufacturers. Assuming a bandwidth of the oscilloscope of 1 GHz (which is given for the equipment used in this thesis), the expected measurement uncertainty expected for different sensor bandwidths



Figure 3.4: Estimation of the measurement error K_{bw} associated with the sensor's bandwidth when capturing switching transients with a true rise time between 2 ns and 6 ns using an oscillo-scope with a bandwidth B_{osci} of 1 GHz.

according to Eq. 3.8 is shown in Fig. 3.4. To achieve an uncertainty K_{bw} of less than 5 % for a 3 ns short transient, e.g., the bandwidth of the sensor should be at least 400 MHz. In addition, ringing caused by the *LC* circuit of junction capacitance and loop inductance $L_{p,loop}$ can be in the range of several hundreds of MHz (e.g., 356 MHz for $L_{p,loop} = 5 \text{ nH}$ and $C_j = 40 \text{ pF}$ for SiC-A at 400 V) and needs to be resolved properly.

3.4.2 Experimental procedure to assess novel current sensors

Current sensors under investigation

Table 3.3 provides an overview of the current sensors under investigation. The coaxial shunt is commonly used to capture fast switching transients due to its claimed bandwidth of up to 2 GHz for the SDN-414-05 version used here [241]. Although its inner measurement path is optimized to prevent disturbances by high-frequency effects [187], it was shown that the actual 3 dB bandwidth can be less than one-tenth of the specified one [188, 189]. Furthermore, the external leads or alternative screw connectors that are used to mount the shunt on a PCB add parasitic inductances of several nH to the commutation loop [190], which affects the switching transients significantly. Nevertheless, since it is widely used, the coaxial shunt is chosen as a well-known reference for all other sensors studied here.

The resistive shunt RL7520WT consists of several paralleled resistors that are combined into a footprint that is short in the direction of the current flow and wide perpendicular to it. Since it is surface mounted on the PCB, it is expected to enable a significantly lower parasitic inductance than the coaxial shunt. However, previous measurements indicate that the inner stray inductance of such types of shunts (but not the exact one used here) can cause the measured current to exceed the actual one substantially [191]. Another resistive shunt that targets a low insertion inductance, high signal fidelity, and high maximum energy input by enhanced power dissipation is the M-shunt [187, 192]. An analysis of the 2016 version of the shunt indicated the combination of an inner stray inductance of several 10 pH in combination with parasitic inductances as low as 0.4 nH. Here, the updated version from 2021 is studied, for which an insertion inductance of 1.1 nH was estimated in [193].

Inductive sensors such as the standard Rogowski coil offer an intrinsic galvanic isolation. Although commonly used, Rogowski coils exhibit a maximum bandwidth of only 30 MHz [243]. Depending on the setup, the coil can be placed around the current-carrying trace, which means that no parasitic inductance is added. Another inductive sensor, which will be referred to as IPM sensor [242] in the following, is formed by a rod-shaped coil that is inserted into a brass cylinder bridging a 20 mm wide and 0.5 mm long slit in the current carrying path.

Table 3.3: Characteristic properties (galvanic isolation, insertion inductance, bandwidth) of the different resistive and inductive current sensors under investigation [241, 190, 192–194, 242, 243]. Note that the actual 3 dB bandwidth of the coaxial shunt has been shown to be potentially lower than the one claimed by the manufacturer [188, 189]. Adapted from [257].

| Sensor | Version / Name | Туре | Iso- | Inductance | Bandwidth | Manufacturer |
|-----------------|-----------------|---------|--------|------------------|----------------|--------------------|
| | | | lation | (nH) | (MHz) | |
| Coaxial shunt | SDN-414-05 | resist. | no | 3–8 | 2000 (claimed) | T&M Research |
| SMD shunt | RL7520WT-R050-F | resist. | no | 0.3-0.6 | not specified | Susumu |
| M-shunt | Version 2021 | resist. | no | 0.7-1.1 | not specified | Bremen University |
| Rogowski coil | CWTUM/1x | induct. | yes | \geq 0 (setup- | 30 | PEM |
| | | | | dependent) | | |
| Infinity sensor | Generation 1 | induct. | yes | 0.2 | >225 | Bristol University |
| IPM sensor | Generation 4 | induct. | yes | 0.3 | >475 | Fraunhofer IZM/ |
| | | | | | | ipmdesign |

The signal that is picked up by the coil is pre-processed by a built-in analog integrator but requires further bandpass correction. The correction is performed in MATLAB after the measurement with a script provided by the manufacturer [242]. The specific sensor used here was custom-made and optimized for high bandwidth. A tradeoff, however, is that the associated built-in integrator saturates at currents of about 40 A (the exact value depends on the signal history). In contrast to the IPM sensor, the Infinity sensor [194] allows an inductive current measurement by placing the sensor above the current-carrying trace instead of having to interrupt it. Only the additional space needed on the PCB causes an insertion inductance of about 0.2 nH. The magnetic field is sensed by horizontally aligned coils. The output signal is measured by soldering a coaxial cable directly to the measurement terminals and corresponds to the derivative of the source current [194]. The final waveforms are obtained by integration in MATLAB.

Measurement procedure

To investigate the different current sensors, two double pulse test boards A and B shown in Fig. 3.5a are designed. Each of the two PCBs can fit two current sensors in the commutation loop between DC minus and the transistor. Their layout is identical except for one sensor footprint. By mounting the well-known coaxial reference shunt on either test board A or test board B together with one of the sensors listed in Tab. 3.3, a direct comparison of the captured waveforms is possible.

Two 650 V/60 A GaN HEMTs in half-bridge topology are used as switches to enable fast transients exceeding 100 V/ns. The inductance of the vertical commutation loop when modeling the decoupling capacitors as copper blocks is estimated to be between 1.8 to 1.3 nH in the frequency range of 1 MHz–1 GHz using Ansys Q3D. The gate–source voltage ($V_{\rm gs}$) of the high-side GaN HEMT is kept at -6 V while the low-side GaN HEMT is actively switched between a $V_{\rm gs}$ of -6 V and +6 V using external gate resistors of 1.0Ω (turn-off) and 5.1Ω (turn-on). Load currents $I_{\rm L}$ between 5 A and 60 A are set by varying the duration of the first turn-on time of the low-side GaN HEMT at a constant DC link voltage of 400 V. The sensors are analyzed regarding basic waveforms during turn-on and turn-off, the corresponding current slopes d_{i_s}/dt (measured between 10% and 90% of the load current) as they directly reflect the capability to capture fast switching transients, and the respective switching energies $E_{\rm on}$ and $E_{\rm off}$ as defined in the IEC 60747-9 standard [195]. Differences in the propagation delays



Figure 3.5: (a) Two PCBs with GaN HEMTs in half bridge configuration are designed in a way that each one can fit two current sensors S1 and S2 at the same time. Their layout is identical except for one sensor footprint. (b) This allows to pair each sensor under investigation with the coaxial shunt, which is used as a reference for all measurements. Adapted from [257].

between the current and voltage probes must be considered and deskewed as they significantly affect the determination of the switching energies. Here, this is done by measuring the differences in the propagation delays and precisely aligning the onset of the drop in drain–source voltage (ΔV_{ds}) during turn-on with the onset of the rise in i_s . Nevertheless, uncertainties in the range of 10 % can be expected. Since noise or peaks in the waveforms can additionally influence the calculations, e.g., by shifting the integration limits, 15 turn-on and 15 turn-off pulses are captured for each load current and the corresponding median values with the median absolute deviation as error bars are shown in the respective figures.

The oscilloscope used is a LeCroy Waverunner 104Xi with a bandwidth of 1 GHz and a sample rate of 5 GS/s. The drain-source voltage V_{ds} is measured with a 500 MHz LeCroy PP006-WR passive probe. Note that the passive probe offers high bandwidth but has a defined ground reference. To suppress ground loop effects when performing measurements with two non-isolated current sensors, the ground reference is set between the current sensors for all probes. The current slopes, switching energies, and exemplary waveforms at $I_L = 30$ A are discussed in the following for each sensor combination.

3.4.3 Exemplary analysis of switching waveforms

SMD shunt and coaxial shunt

The SMD shunt and the coaxial shunt are mounted together on test board A. The turn-on and turn-off waveforms in Fig. 3.6a show that short transient times in the order of several nanoseconds associated with high turn-off voltage slew rates of 60 V/ns are achieved at $I_{\rm L} = 30$ A. Generally, up to 110 V/ns are realized at $I_{\rm L} = 60$ A. This enables the targeted analysis of the different sensors.

The parasitic commutation loop inductance $L_{p,loop}$ causes ringing with a frequency of about 135 MHz and a drop ΔV_{ds} in the drain–source voltage during turn-on when the current rises,



Figure 3.6: Coaxial shunt and SMD shunt. (a) The captured waveforms during turn-on and turn-off are similar, but the turn-on peak current and the oscillation amplitude are slightly lower with the SMD shunt. (b) The turn-on and turn-off current slopes are also slightly lower, but (c) the calculated switching energies are almost identical. Adapted from [257].

visible as a plateau at about 10 ns. As confirmed by the subsequent measurements, this drop is mainly caused by the insertion inductance of the coaxial shunt. Using

$$\Delta V_{\rm ds} = L_{\rm p,loop} \cdot \frac{\mathrm{d}i_{\rm s}}{\mathrm{d}t} \tag{3.9}$$

and subtracting the insertion inductance of the SMD shunt as well as the simulated commutation loop inductance yields an insertion inductance of about 5 to 8 nH for the coaxial shunt, which is consistent with literature [190].

The waveforms captured by the two current sensors are similar in general. The peak turn-on current measured with the coaxial shunt is about 5 A higher than determined with the SMD shunt. Both sensors capture the oscillations, but the amplitude is slightly higher with the coaxial shunt. The difference is related to the bandwidth of the sensors, or more precisely their specific frequency response, and will be discussed in more detail throughout this section. Note that in contrast to previous reports with other SMD shunts [191] no current overshoot is observed here. This is most likely due to the wide footprint of the SMD shunt with its multiple paralleled resistors, leading to a sufficiently low inner stray inductance.

Sharp transitions in the current waveforms tend to appear rather rounded for sensors with low bandwidth. Since the current slopes are defined between 10% and 90% of the load current (not the peak turn-on current), this translates to a high sensitivity of the measured di_s/dt to the first hundreds of picoseconds of the transients. Additional peaks or spikes in the waveforms can thus have a significant impact on the calculated di_s/dt . In Fig. 3.6b, where the turn-on and turn-off current slopes are depicted, this leads to median absolute deviations of up to 2.5 A/ns at a median current slope of 9.6 A/ns. This shows that capturing multiple rather than just one switching transient helps to achieve more meaningful results.

The turn-on di_s/dt are slightly lower in case of the SMD shunt for all load currents. During turn-off, the absolute di_s/dt are similar up to 4 A/ns at $I_L = 30 \text{ A}$ but lower than determined with the coaxial shunt at higher load currents. As will be discussed in Section 3.4.5, this is connected to the change in switching time in combination with the sensor bandwidth. Figure 3.6c shows that the turn-on and turn-off switching energies agree well despite the slight difference in the switching waveforms.



Figure 3.7: Coaxial shunt and M-shunt. (a) The turn-on and turn-off waveforms are similar. The slightly lower peak turn-on current in case of the M-shunt, and (b) the slightly lower current slopes lead to (c) a miniscule lower turn-on energy and a minuscule higher turn-off energy. Adapted from [257].

M-shunt and coaxial shunt

Figure 3.7a shows the exemplary turn-on and turn-off waveforms at $I_L = 30$ A as captured with the M-shunt and coaxial shunt assembled on test board A. The waveforms are similar and the ringing (frequency about 150 MHz) is less pronounced than previously observed with the SMD shunt in Fig. 3.6a. This indicates a reduction of the parasitic commutation-loop inductance. Using Eq. 3.9 and the same procedure as before, i.e., subtracting the parasitic insertion inductance of the M-shunt and the simulated commutation-loop inductance, the insertion inductance of the coaxial shunt is now estimated to be about 3 nH. This is significantly lower than the previously estimated 5 to 8 nH, the reason being that the exact way of positioning the coaxial shunt on the PCB affects its insertion inductance and the corresponding waveforms significantly.

The amplitude of the oscillations is slightly lower with the M-shunt than with the coaxial shunt, and with a difference of about 3.5 A this also applies to the peak turn-on current. In [189], measurements of the transfer functions showed that the M-shunt increasingly attenuates signals above about 60 MHz, reaching the -3 dB limit at around 200 MHz. A 400 MHz coaxial shunt version used in this study showed the opposite trend of increasingly amplifying signals, reaching the +3 dB limit already at 45 MHz [189], which agrees with observations made in [188]. This suggests that the true current waveform may be "in between" that of the coaxial shunt and the M-shunt.

This also explains the difference in current slopes shown in Fig. 3.7b. During turn-on, they are about 1 to 1.5 A/ns lower in case of the M-shunt. During turn-off, the absolute current slopes are similar up to 3 A/ns at $I_{\rm L} = 30 \text{ A}$ (i.e., at low $di_{\rm s}/dt$ and thus low frequencies associated with the transients) but slightly lower at higher load currents. Consequently, the turn-on energies are slightly lower and the turn-off energies are slightly higher than determined with the coaxial shunt, but still agree well (see Fig. 3.7c).

Rogowski probe and coaxial shunt

Next, the Rogowski probe and the coaxial shunt are mounted together on test board A. To mount the Rogoswki coil, a small cylinder of copper foil is soldered to the PCB, and the probe is inserted into it. The oscillations after turn-on and turn-off (frequency about 140 MHz) and the drop in $V_{\rm ds}$ during turn-on (Fig. 3.8a) are again mainly caused by the inductance of the coaxial shunt.



Figure 3.8: (a)–(c) Coaxial shunt and Rogowski probe. (a) The Rogowski probe cannot capture the peak current and ringing accurately and (b) the corresponding absolute turn-on and turn-off current slopes are significantly lower. (c) The switching energies are similar only because of the specific deskew setting: During turn-on, e.g., the energy is overestimated during the first, but underestimated during the second half of the transient. (d)–(f) Coaxial shunt and Infinity sensor. (d) The Infinity sensor can resolve the peak current and ringing more accurately. The slightly lower peak turn-on current in case of the Infinity sensor, and (e) the slightly lower absolute current slopes lead to (f) a slightly lower turn-on energy and a slightly higher turn-off energy. Adapted from [257].

In contrast to the previous measurements, the waveforms measured with the two current sensors differ significantly. The turn-on current peak is broadened in case of the Rogowski probe and the peak turn-on current is approximately 12 A lower than with the coaxial shunt. Furthermore, the oscillations triggered by the switching transients are not resolved properly both during turn-on and turn-off. The absolute current slopes during turn-on and turn-off (Fig. 3.8b) are significantly smaller with the Rogowski probe (at $di_s/dt = 10 \text{ A/ns}$, e.g., the difference is about 60 %). It is noticeable that the difference in the measured current slopes increases with rising load current during turn-off, but it decreases during turn-on. This is related to the actual transient time and will be discussed in Section 3.4.5.

Figure 3.8c shows that despite the considerable differences, the calculated switching energies are almost identical. This is solely caused by the specific deskew setting, which is somewhat arbitrary for sensors with low bandwidth: By using the coaxial shunt as a reference, the signal is adjusted in a way that the turn-on peaks and the onset of the oscillations after the turn-on transient (at about 25 ns) are aligned. As the current slope appears too flat due to the low bandwidth, this leads to an overestimation of the switching energy during the first half of the turn-on transient, and an underestimation during the second half. Here, both effects balance each other out. The opposite is the case during turn-off. If in the absence of the reference signal the deskew were set such that the rise in i_s coincided with the drop in V_{ds} , this would result in E_{on} being severely underestimated, and E_{off} being considerably overestimated.

Infinity sensor and coaxial shunt

Figure 3.8d shows the turn-on and turn-off waveforms with the Infinity sensor and the coaxial shunt mounted on test board B at $I_{\rm L} = 30$ A. As observed before, the ringing (frequency about 140 MHz) and drop in $V_{\rm ds}$ are mainly caused by the coaxial shunt. The infinity sensor itself does not affect the commutation-loop inductance, however, the necessary slight enlargement of the commutation cell adds about 0.2 nH [194].

The oscillations after the switching transients are resolved accurately with the amplitude being slightly lower compared to the coaxial shunt, and the peak turn-on current being about 10 A lower with the Infinity sensor. Again, the absolute current slopes captured with the Infinity sensor are lower than those captured with the coaxial shunt. However, with a difference of about 18 % at 10 A/ns, the deviation is substantially lower than observed previously with the Rogowski probe.

The similarity of the waveforms shows that the bandwidth of the Infinity sensor is significantly increased (simulated bandwidth: 225 MHz [194]) compared to the Rogowski probe without having to interrupt the commutation loop. Nevertheless, the slight underestimation of the di_s/dt leads to slightly higher turn-off and slightly lower turn-on energies, with differences in the range of 5 to 15 %.

IPM sensor and coaxial shunt

Lastly, the IPM sensor is compared to the coaxial shunt on test board A. The waveforms at $I_{\rm L} = 30$ A (Fig. 3.9a) are almost identical, including the plateau at 13 ns during turn-off. The peak turn-on current as captured with the IPM sensor is about 1 to 2 A lower than measured with the coaxial shunt, and the amplitude of the oscillations (frequency about 140 MHz) following the switching transients are marginally lower. Again, the drop in $V_{\rm ds}$ and the ringing is mainly caused by the coaxial shunt, since the assembly with wide brass cylinder in case of the IPM sensor is expected to add only about 0.3 nH to the loop inductance [242]. The absolute current slopes (Fig. 3.9b) during turn-off are similar. During turn-on, they are slightly lower with the IPM sensor. Note that the sensor saturates at too high currents and current slopes. For this reason, measurements are performed only up to 30 A for the turn-off, and 40 A for the turn-on. As shown in Fig. 3.9c, the switching energies are practically identical.

Comparing the results with the previous measurements with M-shunt and coaxial shunt in Fig. 3.7a, it can be seen that the waveforms with IPM sensor deviate from the waveforms with the coaxial shunt in the same direction as observed with the M-shunt, albeit to a lesser extent. Bearing in mind that, as discussed earlier, the coaxial shunt might tend to amplify high frequencies while the M-shunt tends to attenuate high frequencies, this suggests that the IPM sensor could be the closest to the true current waveform.

IPM sensor and Infinity sensor

The previous results identify the IPM sensor and Infinity sensor as two promising galvanically isolated current sensors. The turn-on and turn-off waveforms at $I_{\rm L} = 30$ A when mounting the two sensors together on test board B (Fig. 3.9d) differ significantly from the previous ones: First, the drop in $V_{\rm ds}$ during turn-on is only weakly pronounced. Second, only miniscule oscillations are visible after turn-off and no oscillations are induced after turn-on. This confirms the low-inductive design of the commutation loop and demonstrates the combination of high bandwidth and low insertion inductance achieved with the two novel sensors.



Figure 3.9: (a)–(c) Coaxial shunt and IPM shunt. (a) The captured waveforms during turn-on and turn-off are almost identical. The turn-on current slopes are slightly lower with the IPM sensor, but the difference in switching energies is negligible. (d)–(f) When IPM and Infinity sensor are mounted on one board, (d) the previously observed ringing and drop in V_{ds} during turn-on vanish, which demonstrates their low parasitic inductance. In agreement with the previous measurements, the turn-on peak current and (e) the current slopes are slightly lower with the Infinity sensor, leading to (f) slight differences in the switching energies. Adapted from [257].

The captured current waveforms are similar. As expected from the previous measurements, the peak turn-on current with the IPM sensor is about 7 A higher. The current slopes (Fig. 3.9e) with the Infinity sensor are slightly lower (about 1 A/ns difference at $I_{\rm L} = 30$ A), leading to about 9 to 15 µJ lower turn-on and 2 to 4 µJ higher turn-off energies (Fig. 3.9f). In comparison with the previous measurements with IPM sensor and coaxial shunt (Figs. 3.9a to 3.9c), $E_{\rm on}$ is slightly increased (e.g., 156 µJ here versus 137 µJ with coaxial shunt, both at $I_{\rm L} = 30$ A). The reason is that replacing the coaxial shunt with the Infinity sensor leads to a substantial reduction of the commutation-loop inductance. Using Eq. 3.9, it is now estimated to be about 2 nH, which agrees well with the simulated value. During turn-on, this increases the di_s/dt (difference about 0.9 A/ns at $I_{\rm L} = 30$ A), but also reduces the drop in $V_{\rm ds}$. The latter aspect leads to a rise in turn-on energy, which agrees with simulations reported in [196].

3.4.4 Simulative validation of measured waveforms

In order to validate the previous measurement results, the experimental setup is replicated in LTspice. The circuit model of the power loop shown in Fig. 3.10a is based on [244] and considers the simulated commutation-loop inductance as well as the wires leading to the highvoltage power supply. The central element is a level 3 model of the GaN HEMT provided by the manufacturer. Various parameters used an input for the simulation are not known precisely but affect the switching waveforms. As discussed previously, even the exact way the sensors are positioned can affect the results. Therefore, to obtain the unaffected ideal waveforms and switching energies, the sensors are not specifically considered in the simulation. However,



Figure 3.10: (a) The experimental setup is modeled in LTspice using a level 3 model of the GaN HEMT and considering the simulated parasitic inductances in the commutation loop as well as the connections to the high-voltage source according to [244]. The gate driver model includes the output resistances specified in the datasheet. The corresponding simulation and measurement (the data is taken from the measurement with IPM sensor in Figs. 3.9d to 3.9f) are in agreement regarding (a) the waveforms (b) the current slopes and (c) the switching energies. Adapted from [257].

an additional external common source inductance of 0.1 nH is implemented. This common source inductance critically influences the turn-on and turn-off waveforms. The right part of Fig. 3.10a shows the model of the gate drive circuit. As in the experimental setup, an isolated voltage is supplied and divided into +6 V and -6 V. The gate driver is modeled by two switches with pull-up and pull-down resistances taken from its datasheet (Silicon Labs SI8271GB-ISR).

Figure 3.10b shows the simulated turn-on and turn-off waveforms at $I_{\rm L}$ =30 Å. In addition, the data captured with the IPM sensor in Fig. 3.9a is shown. In this measurement, the IPM sensor was mounted together with the Infinity sensor, providing the lowest additional stray inductance of all setups and making it a reasonable reference. The rising (turn-on) and falling (turn-off) edges of the simulation and measurement almost coincide. With about 57.6 Å, the measured peak turn-on current is slightly lower than the simulated one of 60.0 Å. Furthermore, the simulated peak is broader than the measured one. The fact that the Infinity sensor in Fig. 3.9a also shows a rather sharp turn-on peak, and an additional verification measurement with IPM sensor and SMD shunt also exhibits the same shape (not shown here), suggests slight inaccuracies in the simulation model rather than the measurements.

As shown in Fig. 3.9b, the turn-off current slopes agree well. During turn-on and at load currents below 15 A, the simulated slopes are about 1.5 to 2.0 A/ns higher than the measured ones. As discussed before, the data are highly sensitive to roundings of the waveforms at the onset of the rise in i_s . This effect is especially prominent at low load currents and associated short transient times (see also Fig. 3.12). Again, inaccuracies in the LTspice model might cause a certain deviation. As will be discussed in Section 3.4.5, however, it is also likely that the current sensors underestimate the current slopes under these conditions. As shown in Fig. 3.10d, the turn-on and turn-off switching energies are in agreement despite these differences.

Bearing in mind that the simulation results should not be considered as an ideal reference because of the discussed influence of not exactly known circuit parameters, the similarity of measurement and simulation confirms the experimental results. Based on this simulative validation, a comparative analysis of the experimental data is performed in the next section.

3.4.5 Comparison of characteristic quantities

Method of comparing the data

As discussed previously, the current flow in the commutation cell depends on the loop inductance. It is affected by the sensors themselves (even the exact way of positioning the sensor), which complicates the accurate comparison of the sensor properties. With regard to switching energy, even small differences in signal propagation times between the voltage and current probe can affect the results. In addition, the deskew settings are somewhat arbitrary for sensors with low bandwidth, even if the results presented here agree well. Therefore, the switching energies are not suitable to compare the sensor properties.

A comparison based solely on characteristic parameters of the captured current waveforms, such as slope and peak, is more meaningful. Due to the different stray inductances, these quantities also differ considerably between the different setups, even when the same measurement conditions are applied. For example, the current slopes and turn-on peak currents differ even with the same load current and gate resistance. Therefore, the coaxial shunt is used as a general reference in the following. All parameters measured with the different sensors are only considered in relation to the value measured by the coaxial shunt in the respective measurement setup. This allows a fair comparison.

Comparison of captured peak currents

Accurate recording of the peak turn-on current is essential, e.g., when investigating the turn-off behavior of the freewheeling diode or the current sharing between paralleled devices. Being aware of the differences between the individual sensors is therefore critical.

When the load current is increased, the peak turn-on current also increases. Figure 3.11a shows how much the turn-on peak current measured with the different sensors deviates from the value measured with the coaxial shunt. The data is interpreted in the following way: If, e.g., the coaxial shunt measured a peak turn-on current of about 60 A (which is the case at different load currents for the different setups), the M-shunt captured an about 3.8 % lower and the Infinity sensor and about 10.6 % lower peak current.

For all sensors and in all measurements, the peak currents are lower than the reference values. This might be connected to the previously discussed potential tendency of the coaxial shunt to



Figure 3.11: (a) For all current sensors and for all measurements, the captured peak turn-on currents are lower than the ones determined with the reference coaxial shunt. This might be connected to its potential tendency to amplify high frequencies even below the specified bandwidth. The differences in the current slopes between coaxial shunt and sensors under investigation (b) remain mostly constant during turn-on, but (c) increase substantially during turn-off with rising reference di_s/dt (i.e., with rising load current). This is caused by the opposing change of the transient times. Adapted from [257].

amplify high frequencies even below the specified maximum bandwidth [189]. The M-shunt, on the other hand rather attenuates high frequency signals [189]. At low turn-on currents the deviation increases in the order IPM sensor, M-shunt, SMD shunt, Infinity sensor, Rogowski probe. Regarding the IPM sensor, the M-shunt, and the SMD shunt, the difference is almost independent of peak current. For the Infinity sensor it tends to increase, while it decreases for the Rogowski probe.

Comparison of captured current slopes

Next, the deviation between the current slope measured with the reference coaxial shunt and the various sensors under investigation is compared in the same way as the peak currents. During turn-on (Fig. 3.11b), the current slopes captured with the Rogowski probe are up to 3 A/ns lower than the reference value, which corresponds to a deviation of more than 50 %. However, the difference shrinks to about 20 % for the highest di_s/dt , i.e., the highest load currents. For both the SMD shunt and Infinity sensor, the current slopes are about 1 to 1.5 A/ns lower than the reference ones and rather independent of the reference di_s/dt . With a difference of about -0.5 A/ns, the IPM sensor and M-shunt show the lowest deviation.

Since the turn-off current slopes change over a large range from < 0.1 A/ns to > 20 A/ns, they are displayed logarithmically in Fig. 3.11c. In contrast to the turn-on, the differences now increase with rising di_s/dt for all sensors. In case of the Rogowski probe, the differences become even larger, with a 70 % lower slope than the reference when the latter measures 20 A/ns. With about 26 to 27 % below the reference, Infinity and SMD shunt again exhibit similar results, while with about -22 % the M-shunt is again closer to the reference (all at a reference di_s/dt of about 17 to 20 A/ns). The IPM sensor shows the lowest deviation, with a maximum difference of +0.3 A/ns at a reference di_s/dt of 3 A/ns.

Taking the coaxial shunt as a reference, it seems that as di_s/dt rises, the ability to resolve fast transients deteriorates at turn-off, but remains the same or even improves at turn-on. To understand this, the change in actual transient times must be taken into account. Figures 3.12a and 3.12b show the rise (turn-on) and fall (turn-off) time as a function of the current slope and the directly connected load current as determined with the coaxial shunt in the setup with Rogowski probe (see Figs. 3.8a to 3.8c). During turn-on (Fig. 3.12a), the current starts to rise



Figure 3.12: When increasing the load current, (a) the rise time during turn-on increases slightly, but (b) the fall time during turn-off drops substantially. (c)–(d) Deviation of current slew rates measured with the different sensors with respect to the slew rate and transient time determined with the reference coaxial shunt. The deviation decreases with increasing transient time, which corresponds to (c) an increasing current slope during turn-on, but (d) a decreasing current slope during turn-off. Adapted from [257].

after the $V_{\rm gs}$ reaches the threshold voltage and the channel of the low-side becomes conductive. A higher load current results in a steeper rise of the current through the channel. The time required to reach higher load current also grows, resulting in an increase of the current rise time from 1.4 ns at $I_{\rm L}$ =5 A ($di_{\rm s}/dt$ = 3 A/ns) to 6 ns at $I_{\rm L}$ =60 A ($di_{\rm s}/dt$ = 8 A/ns). During turn-off, the output capacitance of the HEMT has to be charged by the load current. Increasing load currents reduce this charging time significantly. Therefore, the current slope increases and the current fall time decreases significantly when raising the load current, in the example in Fig. 3.12b from 55 ns and $di_{\rm s}/dt$ = 3 A/ns at $I_{\rm L}$ =5 A to 2.4 ns and $di_{\rm s}/dt$ = 20.6 A/ns at $I_{\rm L}$ =60 A.

As explained in Section 3.4.1, such changes in the transient time directly affect the measurement error associated with the limited bandwidth of the sensors. To consider this effect, Figs. 3.12c and 3.12d compare the current slopes relative to the reference current slopes as a function of the reference current rise and fall time. This way, when taking the coaxial shunt as a reference, an overall either increase of the capability to capture fast di_s/dt or an approximately constant level is visible when the transient time rises. Variations between the sensors are caused by their different internal structures (and potential signal processing units such as electronic integrators in case of the Rogowski probe and the IPM sensor), leading to deviating signal rise times and bandwidths.

The results illustrate the following: During turn-off the transient time decreases and the current slope increases with rising load current, which poses high requirements on the current sensor. During turn-on, however, higher current slopes at high turn-on currents (i.e., longer rise times)



Figure 3.13: (a) The SiC-A reference diode is packaged on an AIN DPC substrate and mounted on a PCB in a buck converter configuration together with a GaN HEMT. (b) Double pulse tests with short switching transients < 3 ns and current slew rates of 15 A/ns, as well as a commutation-loop inductance of 4.6 nH including sensor, diode, HEMT, and decoupling capacitors (estimated from the oscillation frequency after diode turn-off) validate the sensor choice an demonstrate the functionality of the low-inductive package design and assembly procedure.</p>

may actually be captured more reliably than similar or even lower current slopes at lower turn-on currents (i.e., shorter transient times). When evaluating switching transients, this effect must be considered.

3.4.6 Sensor selection and fast switching with custom package

The comparative analysis shows that both the resistive (SMD and M-shunt) and the inductive (IPM and Infinity sensor) current sensors can capture nanosecond switching transients reliably while at the same time achieving a significantly lower stray inductance than the standard coaxial shunt. The sensor choice depends on the application target. The Infinity sensor offers a small footprint with the lowest parasitic inductance. However, a substantial offset can occur at longer time windows with increased DC components (typically longer than about 60 to 100 ns after the switching transient). The data also show that short transients may not be captured as accurately as, e.g., with the custom-made IPM sensor. Based on the combination of low insertion inductance (0.3 nH) and high bandwidth (475 MHz) without excessively attenuating or amplifying high-frequency components, the sensor is used in the following to capture single switching transients.

Figure 3.13a shows a corresponding exemplary setup of a double pulse test with a 50 m $\Omega/650$ V GaN HEMT (GaN Systems GS66508T) and the SiC-A diode (see Tab. 3.2) packaged on a custom AlN DPC substrate (see Section 3.2) and assembled in a buck converter topology (see Fig. 2.1a). The IPM sensor is placed between the anode contact of the diode and HV–potential, similar to Fig. 3.5a. The measurement at a low $R_{g,on}$ of 2.2 Ω displayed in Fig. 3.13b shows the turn-off behavior of the diode when the HEMT is turned-on at a load current of about 40 A and a DC-link voltage of 400 V. The 370 MHz ringing caused by the *LC* circuit of diode capacitance (40 pF determined with *C*–*V* measurements) and loop inductance can be used to estimate the latter one. This way, a low value of about 4.6 nH is obtained for the loop including the current sensor, the diode, the HEMT, and the decoupling capacitors. Current fall times of 2.6 ns associated with slew rates of 15 A/ns demonstrate that fast transients can be achieved with the setup and confirm the functionality of the low-inductive package design. According to Fig. 3.4, a measurement error K_{bw} below 5% is expected for the IPM

sensor, while a $200\,\rm MHz$ sensor would cause an error of up to $20\,\%.$ This consideration further validates the sensor choice.

The slightly lower bandwidth of the M-shunt compared to the IPM sensor and its tendency to attenuate high-frequency components result in a somewhat flatter representation of the switching transients. However, it offers a DC capability as well as higher energy dissipation and lower insertion inductance than the coaxial shunt. Therefore, the M-shunt is used for all applications with continuous current in the following. The SMD shunt is attractive as an easy-to-use and low-cost alternative with comparatively low insertion inductance. Compared to the M-shunt, however, the heat dissipation capability is significantly reduced, and it seems to be slightly inferior in terms of resolving fast transients accurately. Therefore, it is not used in this thesis.

When the devices are operated continuously instead of being subjected to individual pulses only, their temperature rises and their transient thermal properties must be taken into account. The corresponding measurement and evaluation procedures applied to the Ga_2O_3 diodes under investigation will be explained in the next section.

3.5 Evaluation of transient thermal properties

General remarks

During their operation, semiconductor power devices create a considerable amount of heat that must be dissipated to a heatsink. The entire thermal path between the junction, where a major part of the heat is created, and the ambient causes a certain thermal resistance called junction-to-ambient thermal resistance ($R_{th,ja}$). It affects the thermal management strategy of power converters and determines the maximum possible current and power rating. Measurements of the thermal properties are of special interest for devices made from Ga₂O₃ due to its low thermal conductivity (see Section 2.3.2). All transient thermal measurements are carried out in compliance with the JEDEC JESD51-14 standard [197]. Parts of this section are based on [255].

Temperature-sensitive parameter for Ga₂O₃ diodes

The measurement method is based on a temperature-sensitive parameter (TSP) to determine the junction temperature of the device. In the case of diodes, the forward voltage drop at a certain suitable sensing current is usually used [197]. The temperature T_j determined in this way corresponds to the average temperature of the junction area [198].

In order to examine whether the forward voltage drop is a suitable TSP also for Ga_2O_3 diodes, different sensing currents between 0.1 mA and 100 mA are applied and the corresponding forward voltage drop is measured for temperatures between $0 \,^{\circ}C$ and $150 \,^{\circ}C$. Figure 3.14a shows the results exemplarily for a B3-ML type Ga_2O_3 diode. A linear relationship between the forward voltage drop and the junction temperature is observed for all sensing currents. The corresponding slope, the so-called K-factor, increases with decreasing sensing current. While a high K-factor is desirable in principle, the associated low sensing currents lead to a low signal-to-noise ratio. On the other hand, a too-high sensing current must be avoided to prevent self-heating of the device and ensure a sufficiently high sensitivity. For all further measurements a sensing current of 10 mA is chosen, leading to a sensitivity of about 1.7 mV/K for all investigated Ga_2O_3 diodes. This value is similar to that of commercial SiC diodes (e.g., K-factors of 1.61 mV/K and 1.65 mV/K at 10 mA are determined for the reference samples SiC-A and SiC-B, respectively) and allows to use the forward voltage drop as a TSP for thermal resistance measurements.



Figure 3.14: (a) A linear relationship between forward voltage and junction temperature is observed at different sensing currents between 0.1 mA and 100 mA. The forward voltage at a sensing current of 10 mA provides sufficient sensitivity and is used as a temperature sensitive electrical parameter. (b) Setup for thermal resistance measurements: A rubber stud presses the device down onto a thermal interface material (TIM) between case and water-cooled heatsink. (c) Temperature response after heating a Ga₂O₃ B3-L type diode for 100 s with two different TIMs, showing that a measurement duration of 100 s is sufficient to reach thermal equilibrium. The inset shows the determination of the initial junction temperature using the \sqrt{t} -method. Parts adapted from [255].

Transient thermal measurement and evaluation procedure

A T3Ster thermal transient tester from Mentor is used to study the transient thermal properties in accordance with the JEDEC JESD51-14 standard [197]. All related calculations discussed below are performed using Mentor's standard-compliant T3Ster Master software [245]. The experimental setup for all transient thermal measurements is shown in Fig. 3.14b. The samples are mounted in the center of a water-cooled aluminum heatsink (cooling area 47.5 mm × 47.5 mm) with a coolant temperature of 25 °C and a flow rate of 101/min. A rubber stud is screwed down onto the diode with a torque of 60 cN m, pressing it down onto a thermal interface material (either a thermally conductive sheet or thermal grease) between casing and heat sink.

During the actual measurement phase, a constant forward current of several amperes corresponding to a heating power $P_{\rm H}$ is applied and heats up the diode. Then, the heating power is switched off abruptly, and the time-dependent junction temperature $T_{\rm j}(t)$ during the subsequent cooling phase is recorded using the previously calibrated TSP. The duration of the cooling phase is equal to the duration of the heating phase. Figure 3.14c shows the temperature response exemplarily for two measurements with a B3-L type Ga₂O₃ diode: one with thermal grease at a heating current of 6.0 A, and one with a heat conductive sheet at a heating current of 5.3 A. The power loss leads to a rise in junction temperature of about 90 to 100 K for both samples. Since the transients associated with switching off the heating current disturb the measurement signal during the first 50 to 100 µs of the cooling phase, the initial junction temperature $T_{\rm j0}$ cannot be determined directly. Instead, it is estimated with the \sqrt{t} -method [197]. To this end, the temperature response is plotted against \sqrt{t} and a straight line is fitted to the profile (see inset in Fig. 3.14c). The initial junction temperature then follows from an extrapolation to t = 0 s.



Figure 3.15: (a) Thermal impedance profiles and (b) structure functions obtained from transient thermal measurements with a B3-L type Ga_2O_3 diode and two different TIMs. The splitting point of the two curves marks the $R_{th,jc}$. It is calculated either from (c) the difference of the derivatives of the thermal impedance profiles or (d) the difference of the thermal structure functions according to the JEDEC JESD51-14 standard [197].

It can also be seen that it takes several seconds longer for the setup with heat conductive sheet to reach a steady temperature. However, in both cases the temperature remains nearly constant between 30 s and 100 s (temperature difference < 0.2 K), indicating that a thermal equilibrium is reached within this time. Thus, a heating and cooling time of 100 s is used for all further transient thermal measurements unless explicitly stated otherwise. With that, the thermal impedance $Z_{th}(t)$ can be calculated directly from the temperature response [197]:

$$Z_{\rm th}(t) = \frac{T_{\rm j0} - T_{\rm j}(t)}{\Delta P_{\rm H}}$$
(3.10)

Figure 3.15a shows the corresponding thermal impedance profiles with the two different TIMs exemplarily for the B3-L type Ga₂O₃ diode. As explained in detail in [199, 200], the recorded temperature profile basically represents a step-function response a(t) that can be regarded as a sum of single response functions $a_i(t) \propto (1 - \exp(t/\tau_i))$, each with an individual time constant τ_i . The individual time constants mathematically correspond to those of individual parallel RC elements, and thus the whole system can be understood as a series connection of individual parallel RC circuit elements. This purely mathematical model based on a onedimensional heat flow is called Foster model. A transformation to the equivalent Cauer model is helpful to obtain a one-dimensional thermal network with an actual physical meaning. In this model, all capacitances are connected to ground, and each RC section corresponds to a certain element in the experimental setup with its respective thermal capacitance $C_{\text{th,i}}$ and thermal resistance $R_{\text{th,i}}$ (see also Fig. 4.12 in Section 4.4). A common representation of the corresponding heat flow path from the heat source to the heat sink is the so-called structure function, where the cumulative thermal capacitance is plotted against the cumulative thermal resistance [199, 200]. Figure 3.15b shows the corresponding structure functions with two different TIMs exemplarily for the B3-L type Ga₂O₃ diode.

When two measurements with different thermal interface materials are performed, the junctionto-case thermal resistance ($R_{\rm th,jc}$) can be determined from the splitting point of the two thermal impedance profiles or structure functions following the JEDEC standard [197]. The first procedure relies on the difference of the derivatives of the two thermal impedance curves and is shown for the previously discussed measurement in Fig. 3.15c. When defining $a(z) = Z_{th}(t_{ref} \cdot exp(z))$ with $z = \ln(t/t_{ref})$ and $t_{ref}=1$ s, the difference of the derivatives follows as $\Delta(da/dz) = (da_1/dz)-(da_2/dz)$. In the next step, an exponential curve of the form $\delta_Z = \alpha_Z \cdot exp(\beta_Z Z_{th})$ is fitted to the data, where α_Z and β_Z are free variables. The $R_{th,jc}$ then follows from the intersection of this curve and a linear trendline $E_Z(Z_{th}) = 0.0045 \cdot Z_{th} + 0.003$. The second procedure is based on the difference of the structure functions and illustrated in Fig. 3.15d. The $R_{th,jc}$ is defined as the point where the difference ΔC_{th} "clearly starts to rise" [197]. In some cases, the difference rises monotonically. The critical difference used to specify the $R_{th,jc}$ is then defined as 0.05 mW s/K. In other cases, the difference fluctuates around zero before it is followed by a distinct rise (e.g., in [255]). In this case, the last zero-crossing point of ΔC_{th} is defined as $R_{th,jc}$. The general accuracy of the measurement procedure was verified by measuring the $R_{th,jc}$ of commercial SiC diodes and comparing them with the datasheet value.

As the two thermal impedance profiles and structure functions split gradually over a certain range, the definition of the separation point is still somewhat arbitrary and the junction-to-case resistances determined by both methods usually differ by a few percent (see also Fig. 3.15) [197]. In addition, the geometry of the die and package strongly affects the structure function and thermal impedance profile [201]. In combination with uncertainties in the exact determination of the initial junction temperature and potential numerical artifacts arising from the various analysis steps, this can result in the two profiles not splitting up precisely at the actual case-to-TIM interface and lead to an inherent uncertainty of up to 15 % associated with the determination of the $R_{\rm th,jc}$ [201]. Nevertheless, the method helps to match different sections of the structure function with the corresponding part in the experimental setup, especially when combined with thermal simulations. This constitutes the basis for the investigations in Section 4.3 and Chapter 5.

Summary

The section establishes that the forward voltage drop at a certain sensing current is a suitable temperature-sensitive parameter for Ga_2O_3 diodes. Transient thermal measurements are feasible for the samples under investigation within the general non-negligible inherent measurement uncertainties. Further measurement issues that may result from specific types of Ga_2O_3 diodes will be discussed in the respective sections.

3.6 Chapter conclusion

This chapter introduces the experimental methods used to study Ga_2O_3 diodes. The main aspects can be summarized as follows:

- 1. While some Ga_2O_3 devices under investigation are TO247 packaged by the manufacturer, other diodes were received as bare dies. A low-inductive package has been designed and an assembly procedure has been developed to allow a variety of different experiments for bare chips with different sizes and surface metallizations.
- Static electrical measurement procedures to investigate basic junction properties of Ga₂O₃ diodes have been introduced, and typical measurement resolutions have been discussed.

- 3. To enable a sound analysis of the switching properties of Ga_2O_3 diodes, novel current sensors are evaluated. The IPM sensor is selected to resolve fast transients due to its combination of low insertion inductance and high bandwidth. A measurement error $\leq 4\%$ is expected for transients ≥ 3 ns.
- 4. The forward voltage at a certain sensing current was found to be a suitable temperaturesensitive parameter to determine the junction temperature of Ga_2O_3 diodes. An experimental setup to measure transient thermal properties is established, but inherent uncertainties of up to 15 % associated with the determination of the junction-to-case thermal resistance have to be taken into account.

The results lay the foundation for further detailed studies on Ga_2O_3 diodes, with a focus on their general characteristic properties in Chapter 4, their thermal properties in Chapter 5, and their switching properties in Chapter 6.

Chapter 4

Basic characteristics of gallium oxide Schottky diodes

4.1 Chapter overview

Despite noticeable progress in the fabrication of β -Ga₂O₃ devices, many of their basic characteristics are still not well understood, and especially research on application-relevant large-area devices is lacking. As a first step towards the application of Ga_2O_3 power diodes, their fundamental properties must be investigated to provide the necessary understanding and basis for more application-oriented experiments. In Section 4.2, physical properties of the Schottky junction are derived from forward and reverse current-voltage and capacitancevoltage profiles. In addition, it is discussed how the diode characteristics can differ between different devices — either because of their intrinsically different structure, or due to a variation of material properties across the wafer as a result of the early level of maturity. Many of the electrical properties are discussed by the example of two different types of TO247 packaged β -Ga₂O₃ diodes (see overview of devices in Tab. 3.1): an early generation small-area trench MOS diode (B1-XS), and one of the more recently obtained large-area planar diodes (B3-ML). The TO247 package is one of the most commonly used ones in power electronics and serves as a vehicle to enable first measurements. Note that it is known to deteriorate the switching behavior due to its large parasitic inductance of up to about 10 nH [31, 180], but it does not influence the results presented here. Section 4.3 provides basic considerations on the thermal resistance of Ga₂O₃ diodes and Section 4.4 introduces a simple method to precisely reproduce unusual electrical characteristics observed for some diodes in the common simulation tool LTspice. Based on the results, further research questions that will be investigated in Chapters 5 and 6 are derived. Parts of the findings presented in this chapter have been published by the author in [253–255, 258] during the course of this thesis.

4.2 Basic electrical properties

4.2.1 Forward conduction characteristics and junction properties

Initial investigations concern the forward current–voltage characteristics. Measurements in the high-current regime show how the standard forward conduction profiles can vary between different diodes, while measurements in the low-current regime allow statements about the junction properties (see Section 3.3).



Figure 4.1: The forward current–voltage characteristics at junction temperatures T_j between $-50 \,^{\circ}\text{C}$ and $150 \,^{\circ}\text{C}$ can show significant differences. A change in temperature-coefficient of the forward voltage and a non-constant differential on-resistance at low temperatures is observed for various diodes, as shown exemplarily for a trench MOS diode in (a) and (b). Other diodes exhibit the typical characteristics known from SiC Schottky diodes, as shown exemplarily for a field-plated planar diode in (c). Parts adapted from [253, 254].

Varying forward conduction profiles

First, the forward characteristics of the small-area β -Ga₂O₃ trench MOS diode (B1-XS) are discussed [254]. The basic structure and benefits of this type of diode are explained in Section 2.3.4. For this specific diode, the EFG-grown Ga₂O₃ substrate is 500 µm thick and the HVPE-grown epitaxial layer is approximately 7 µm thin with an effective donor concentration of about 5 to 6×10^{16} cm⁻³ after doping with Si. The trenches are approximately 1.5 µm deep and 3 µm wide, the mesas between the trenches are about 2 µm wide. The junction is terminated by a 50 nm thin HfO₂ layer and 300 µm thick and 75 µm long TEOS-SiO₂ field plates serve as edge termination. The aluminum metallized anode with a diameter of 300 µm yields an active area of about 0.071 mm², which is used to calculated all area-related quantities shown in the following.

Figure 4.1a shows the forward current–voltage profiles at junction temperatures between $-50 \,^{\circ}\text{C}$ and $150 \,^{\circ}\text{C}$. The basic functionality at all temperatures confirms that undergoing a standard packaging procedure seems to be feasible for β –Ga₂O₃ devices despite the brittleness. Nevertheless, the forward characteristics show an interesting behavior. Above a junction temperature of $50 \,^{\circ}\text{C}$, the typical characteristics known from SiC Schottky diodes are observed: At low currents the forward voltage decreases with rising temperature (negative temperature coefficient of the forward voltage) because the electron density in the conduction band increases and thus more electrons can gain an energy sufficient to overcome the Schottky barrier [32]. At high currents the forward voltage) because enhanced phonon scattering decreases the electron mobility in Ga₂O₃ [61] and thus increases the differential on-resistance. As a result, a crossing point with a temperature coefficient of zero is obtained at a current density of about $120 \,\text{A/cm}^2$.

Between -50 °C and 50 °C, however, the forward voltage and with that also the conduction losses decrease with rising temperature over the entire voltage and current range. Furthermore, the profile is considerably curved at junction temperatures below 0 °C, which can also be

seen from the non-constant differential on-resistance in Fig. 4.1b. For temperatures above about 50 °C it reaches a constant value above about 150 A/cm². For low temperatures, it continues to decrease with higher currents. Such curved current–voltage profiles with negative temperature-coefficient of the forward voltage are typical for bipolar diodes [202, 203]. As explained in Section 2.3.2, however, β –Ga₂O₃ cannot effectively be p-type doped and the investigated device does not contain a different p-type semiconductor (such as p-NiO) in its structure.

Such curvature and change of sign of the temperature coefficient are observed for multiple (but not all) Ga_2O_3 diodes under investigation in this thesis that differ in substrate thickness (e.g., $200 \,\mu\text{m}$ or $600 \,\mu\text{m}$), size (i.e., small-area and large-area diodes), structure (e.g., trench diode or planar diode), and packages (e.g., molded TO package or assembly on AlN substrate without further sealing, wire-bonded or double-side sintered). The junction temperature at which the temperature coefficient changes can differ but is usually found to be between $50 \,^{\circ}\text{C}$ and $100 \,^{\circ}\text{C}$. SiC reference diodes packaged in the same way do not exhibit this behavior (see also Fig. 3.3).

A decrease in on-resistance and a negative coefficient of the forward voltage with rising temperature was also observed for a planar β -Ga₂O₃ diode in [141] and a β -Ga₂O₃ trench diode in [166] for junction temperature up to 200 °C. It was supposed that this could result from thermal lattice vibrations: Since the vibrations increase with rising temperature, electrons could gain a sufficient amount of energy to overcome the Schottky barrier [141, 166]. In both studies, however, no change in the direction of the shift, i.e., the sign of the temperature coefficient is reported, and no curvature of the forward current–voltage profile (i.e., higher than expected differential on-resistance at low temperatures and low forward voltages) is mentioned.

For other Ga₂O₃ diodes in this thesis, the temperature coefficient of the forward voltage at high current densities is positive for all temperatures. An example of a large-area planar β -Ga₂O₃ diode with field plates (B3, ML type) is shown in Fig. 4.1c. The diode was fabricated on a 600 µm thick EFG-grown substrate and has a 1.6 mm × 1.6 mm large gold metallized anode. This corresponds to an active area of about 2.4 mm². The HVPE-grown epitaxial layer exhibits an effective donor concentration of about 9×10^{15} cm⁻³, as determined by capacitance–voltage measurements according to the procedures described in Section 3.3. The maximum current of 10 A corresponds to a current density of about 400 A/cm², which means that the scaling is similar to that in Fig. 4.1a. With 11.5 mΩ cm², the specific on-resistance at 25 °C is about twice as much as that of the trench diode, but the typical characteristics known from SiC Schottky diodes [202, 203] are observed. It should be noted that the parasitic resistance of the package is in the order of 0.3 mΩ [31], which is ≪1% of the differential on-resistance and does therefore not affect the measurements.

The results suggest that the changing temperature coefficient and curvature are not caused by the packaging or other assembly-induced issues. Instead, there seems to be a resistancemodulating effect in the Ga_2O_3 diode that counteracts the increase in electron mobility at lower temperatures and low forward currents under specific conditions. To clarify the origin, a further combined analysis of the impact of growth conditions and device processing techniques on the device characteristics must be performed by the device manufacturers. For an application in automotive power electronics, this issue can be of importance, e.g., regarding the paralleling of devices or the design of power modules as the lowest power loss and thus optimal operating point may be reached at a certain junction temperature at which the temperature coefficient



Figure 4.2: Below the knee voltage and at junction temperatures between −50 °C and 150 °C, both the (a) B1-XS trench MOS and (b) B3-ML planar diode show a decrease of the forward voltage with rising temperature. Modeling with the thermionic emission (TE) model shows a good agreement in both cases. (c) By using the TE model with a constant differential on-resistance according to Eq. 4.1, the forward characteristics of the planar B3-ML diode from Fig. 4.1c can be modeled precisely, whereas large deviations above the knee voltage are obvious for the trench diode (not shown here). Parts adapted from [254].

changes directions. To gain further insight into the junction properties of both the trench and planar Ga_2O_3 diode type, the forward conduction at low currents is studied in the following.

Junction properties at low and high temperatures

The forward characteristics of Schottky diodes can usually be described by the thermionic emission model [45], and it is often used to extract basic junction properties [135, 152, 159, 165, 166]. First, it will be examined in the following whether applying this model to the diodes under investigation is reasonable, considering the unusual forward characteristics above the knee voltage.

The solid lines in Fig. 4.2 show the measured temperature-dependent forward characteristics between $-50 \,^{\circ}\text{C}$ and $150 \,^{\circ}\text{C}$ below the knee voltage exemplarily for the trench-MOS diode B1-XS (Fig. 4.2a) and the planar diode B3-ML (Fig. 4.2b). The saturation current I_s can be determined from the intercept of a fit to the linear part of the low-current region, and the ideality coefficient *n* follows from the slope of the fit according to Eq. 3.2. The results are shown in Fig. 4.3a and will be discussed later in detail but used for the following consideration already. The ohmic voltage drop across the diode that is mostly caused by the lightly doped drift region dominates the current-voltage profile at high current densities above the knee voltage and leads to a transition from an exponential increase in current with rising voltage to a linear behavior. By including this additional ohmic voltage drop in the thermionic emission model, the forward characteristics can described by [34, 45]

$$I_{\rm D} = I_{\rm s} \left[\exp\left(\frac{e(V_{\rm D} - I_{\rm D} r_{\rm o})}{nkT_{\rm j}}\right) - 1 \right].$$
(4.1)

The resultant data are shown as dashed lines in Figs. 4.2a and 4.2b for the trench-MOS and the planar diode, respectively. The agreement between measurement and model below the knee



Figure 4.3: Measured (*n*) and calculated (n_{if}) ideality factors, Schottky barrier height $\phi_{b,0}$, and built-in potential ψ_{bi} at junction temperatures between $-50 \,^{\circ}\text{C}$ and $150 \,^{\circ}\text{C}$ for (a) a trench MOS diode with Mo as a Schottky metal and (b) a planar diode with a Ni Schottky contact. The barrier height and built-in potential differ in their absolute values but show the same temperature behavior. The ideality factors show opposite trends but are close to the ideal values, indicating a rather homogeneous junction and high temperature stability. Parts adapted from [254].

voltage, where the on-resistance is neglectable, confirms that the thermionic emission model can describe the current transport across the junction accurately for *both* diodes. Figure 4.2c shows the modeled forward current above the knee voltage for the planar diode when the differential on-resistance r_0 is determined by a linear fit to the current–voltage profiles between 2 A and 4 A. It agrees with the measured data in Fig. 4.1c at all temperatures and currents, suggesting that standard forward current models can be applied in common simulation tools such as LTspice. For diodes with non-constant differential on-resistance and change of temperature coefficient, the model is accurate at low currents (see Fig. 4.2a) but will evidently cause significant deviations above the knee voltage [254]. A simple method to improve the accuracy of the model is discussed in Section 4.4.

The previous considerations show that the thermionic emission model describes the basic junction properties despite deviations at high currents. Based on that, the Schottky barrier height $\phi_{b,0}$ and the built-in potential ψ_{bi} across the junction without image-force lowering are derived for different temperatures according to Eqs. 3.1 and 3.3.

Figures 4.3a and 4.3b show the corresponding results exemplarily for the trench MOS diode and the field-plated planar diode. The built-in potential decreases for both diodes with rising junction temperature, which is also apparent from the decrease of the knee voltage in the forward current–voltage profiles, as more charge carriers are thermally lifted into the conduction band. The Schottky barrier height remains almost constant with only a miniscule decrease of less than 0.02 eV over the entire temperature range for both diodes.

The reason is that the density of states in the conduction band and with that also the energy difference between the conduction band and the Fermi energy $(E_{\rm C} - E_{\rm F})$ increases at higher temperatures (see Eq. 3.3) whereas the decreasing built-in potential counteracts this effect. As a result, and as also observed in [52] and [141] for Ga₂O₃ diodes with Pt as a Schottky metal, the Schottky barrier height $\phi_{\rm b,0} = \psi_{\rm bi} + (E_{\rm C} - E_{\rm F})$ remains approximately constant. For both diodes, the ideality factors are only slightly higher than the ideal value of n = 1.00, and they show only a weak temperature dependence. Interestingly, however, the direction of change
with temperature is opposite. The temperature dependency and slight deviation from the ideal value can have several reasons known from other semiconductor devices [45, 204].

Tunneling currents can dominate the conduction mechanisms especially at high doping concentrations and low temperatures [45]. The characteristic tunneling energy E_{00} and accordingly expected ideality coefficient n_{tun} [45, 204]

$$E_{00} = (\hbar/2) \cdot \sqrt{N_{\rm D}/(m_{\rm e}^* \epsilon_0 \epsilon_{\rm s})} \quad \text{and} \tag{4.2}$$

$$n_{\rm tun} = eE_{00}/(kT_{\rm j}) \cdot \coth\left(eE_{00}/(kT_{\rm j})\right)$$
(4.3)

can be calculated for both samples. When decreasing the temperature from $150 \,^{\circ}\text{C}$ to $-50 \,^{\circ}\text{C}$ the ideality factors should increase from 1.001 to 1.005 for the trench diode and 1.0002 to 1.0008 for the planar diode according to this theory. While it agrees with the basic trend of increasing ideality factor with decreasing temperature in case of the planar diode in Fig. 4.3b, the effect is too small to explain the measurement data and even within the measurement uncertainty.

The second commonly observed contribution is based on the image-force effect leading to an effective lowering of the Schottky barrier height. The barrier lowering at zero bias voltage $\Delta \phi_{if,0}$ and the corresponding ideality factors n_{if} can be derived from [205–207]

$$\Delta\phi_{\rm if,0} = e \cdot \sqrt[4]{e^3 N_{\rm D} \psi_{\rm bi} / (8\pi^2 \epsilon_{\rm s} \epsilon_0{}^3 \epsilon_{\infty}{}^2)} \quad \text{and} \tag{4.4}$$

$$n_{\rm if} = \left[1 - \Delta \phi_{\rm if,0} / (4e\psi_{\rm bi})\right]^{-1} \,, \tag{4.5}$$

where ϵ_{∞} is the high-frequency and ϵ_s is the low-frequency dielectric constant (approximately 3.6 [208, 209] and 11 [12, 209] for β -Ga₂O₃, respectively). The ideality factors considering the image-force lowering are higher than the ones expected from tunneling only, with about 1.011–1.012 for the planar diode and 1.024–1.027 for the trench diode for temperatures between $-50 \,^{\circ}$ C and $150 \,^{\circ}$ C. In case of the trench diode (Fig. 4.3a), the calculated ideality factors almost coincide with the measured values for temperatures $\leq 25 \,^{\circ}$ C, while there is a slight deviation at higher temperatures. In case of the planar diode (Fig. 4.3b), the deviations are slightly larger and increase at lower temperatures.

This indicates the commonly observed [165, 166, 210] influence of lateral inhomogeneities of the Schottky barrier caused by potential fluctuations around defects of not atomically flat interfaces [205]. For the Ga₂O₃ diode in [210], e.g., *n* increases from about 1.1 at about 75 °C to about 1.4 at -50 °C, while the Schottky barrier height decreases from about 1.0 eV to 0.9 eV. In comparison, the barrier height of the diodes under test remains almost constant and the ideality factor changes by less than 0.01 in the same temperature range, showing that the effect is miniscule and the barrier is fairly homogeneous for both diodes under test. The slight increase in ideality factor from about 1.03 to 1.06 between 25 °C and 150 °C in case of the trench diode might, e.g., be caused by trap-assisted tunneling of thermally activated charge carriers across the Schottky barrier [165].

The findings illustrate that although Ga_2O_3 diodes can show different forward characteristics, the devices under investigation exhibit a high temperature stability with ideality coefficients close to the ideal values expected from the thermionic emission model. Whether similar statements can be made about the reverse blocking characteristics will be examined next.

4.2.2 Reverse blocking characteristics

The reverse blocking behavior of a diode is characterized in particular by two properties: the reverse current–voltage profiles and breakdown voltage, and the parasitic junction capacitance. Both aspects are covered in this section.

Varying reverse current-voltage profiles

Figure 4.4 shows the reverse characteristics at junction temperatures between 25 °C and 150 °C exemplarily for the two previously discussed types of β –Ga₂O₃ diodes. In case of the trench-MOS diode (Fig. 4.4a), the reverse current increases with rising temperature over the entire voltage and current range, which is also typical for SiC Schottky diodes [235, 211]. The reverse current density $J_{r,TFE}$ of the metal–semiconductor contact with wide-bandgap materials such as SiC or Ga₂O₃ can be described by the thermionic field emission (TFE) model [45, 50, 52]:

$$J_{\rm r,TFE} = \frac{A^{**}T_{\rm j}e\hbar E}{k} \cdot \sqrt{\frac{\pi}{2m_{\rm e}^*kT_{\rm j}}}$$

$$\cdot \exp\left(-\frac{1}{kT_{\rm j}}\left[e\phi_{\rm b,0} + e\Delta\phi_{\rm if,0} - \frac{(e\hbar E)^2}{24m_{\rm e}^*(kT_{\rm j})^2}\right]\right)$$
(4.6)

All important diode parameters that are necessary for the calculation, such as the built-in potential, the Schottky barrier height, or the image-force lowering, are known already from Section 4.2.1. As explained in Section 2.3.4, the MOS junction is expected to decrease the electric field at the junction and reduce the reverse leakage current by decoupling it from the Schottky barrier height [160]. In comparison, the reverse leakage current should therefore be smaller than for a pure metal–semiconductor junction. Indeed, this behavior is observed in Fig. 4.4a: the reverse current calculated exemplarily for a temperature of 25 °C is several orders of magnitude higher than the measured one. To further illustrate this, the electric field *E* in Eq. 4.6 is effectively reduced by multiplying it with the RESURF factor introduced in Section 2.3.4. With $\mathcal{R} = 0.3$, the resulting reverse current (marked as TFE* in Fig. 4.4a) reproduces the measurement much more accurately. This indicates that the intended function of the trench MOS structure is effective and that a simple empirical parameter can be used to adjust the profile, which can be helpful to improve corresponding simulation models (see also Section 4.4).

For many other Ga_2O_3 diodes, however, rather unusual reverse profiles are observed. Figure 4.4b shows this exemplarily for the field-plated planar diode B3-ML. Between 25 °C and 100 °C, the reverse current decreases with rising junction temperature. At junction temperatures above 100 °C, the temperature coefficient changes its sign and the reverse current increases with rising temperature. As evident also from Fig. 4.4a, this is not expected from the thermionic field emission model. It is interesting to note that the trench diode exhibits an unexpected change of the sign of temperature coefficient in forward direction, but a typical reverse current–voltage profile. The opposite is true for the field-plated diode type. In general, all possible combinations were observed for different diodes.

The results indicate contributions to the reverse leakage current other than the typical thermionic field emission. Two possible conduction mechanisms also observed previously, e.g., in GaN [56] and $Al_x(Ga_{1-x})_2O_3$ [57] diodes, are the tunneling of electrons from the metal to the semiconductor through trap states at the interface (trap-assisted tunneling, TAT) [53, 55],



Figure 4.4: (a) The reverse leakage current of the trench MOS diode increases with rising junction temperature. While the classic thermionic field emission (TFE) model overestimates the reverse leakage current, including an empirical factor that reduces the electric field in the model (referred to as TFE*) results in a better match, showing that the trench MOS junction and field plates are highly effective. (b) For multiple other diodes the leakage current initially decreases and then increases with rising temperature, as exemplarily shown here for the field-plated planar diode type. This unusual behavior is not expected from the classic models such as TFE and suggests additional temperature-dependent contributions such as trap-assisted tunneling (TAT). Parts adapted from [254, 258].

and the Poole–Frenkel emission (PFE) where trapped electrons are emitted into the conduction band through thermal excitation [53, 54].

Indeed, a linear relationship is observed at $25 \,^{\circ}$ C when plotting the logarithmic reverse current density against the square root of the inverse reverse voltage, which is characteristic for a TAT process [56, 57]. At higher temperatures, however, the linear relationship vanishes. This suggests a change of the conduction mechanism with increasing temperature, possibly from dominating tunneling processes through interface states at low temperatures to TFE and/or PFE processes at high temperatures.

Since the unusual change of temperature coefficient is observed for multiple different types of diodes but not all devices under investigation, the effects do not seem to be intrinsic to Ga_2O_3 but rather depend on the device growth and processing parameters. Given the analysis of the Schottky junction properties in Section 4.2.1, which indicates a largely ideal TE behavior at low currents, the additional contribution to the reverse (and forward) current may, at least partly, result from issues with the ohmic contact at the cathode. A more detailed analysis allowing a definite conclusion would require knowledge of the exact growth conditions and material properties, which, however, are not known to the author. Nevertheless, this observation is important for the Spice models developed in Section 4.4.

Reverse breakdown and $V_{br}^2/r_{o,sp}$ figure-of-merit

Apart from the junction properties, one of the most important figures-of-merit is the tradeoff between on-resistance and breakdown voltage. In Fig. 4.5 the breakdown voltage at room temperature is measured for the two diodes discussed above, i.e., the early generation smallarea trench diode (B1, XS size) and the newer planar diode (B3, ML size), as well as a second planar diode with the largest anode size (B3, LL size). The profiles are compared to two



Figure 4.5: (a) Reverse breakdown characteristics of three Ga₂O₃ (B1-XS, B3-ML, and B3-LL) and two commercial 650 V rated SiC diodes in a linear and semilogarithmic representation. Within the rated voltage, the Ga₂O₃ diodes exhibit a several orders of magnitude higher leakage current than the SiC reference diodes, showing the need for a further optimization of the diode structure. Nevertheless, breakdown voltages > 1.1 kV can be achieved even for the largest diode. (b) Forward conduction profiles before and after breakdown, exemplarily for the SiC-A reference and the B3-LL Ga₂O₃ diode. The change at low forward voltages in case of the Ga₂O₃ diode reveals a partial destruction of the junction, resulting in an effective barrier lowering and degradation of the blocking capabilities.

commercial SiC diodes (see also Tab. 3.2): a 650 V/10 A rated SiC diode [235] assembled on custom AlN-DPC substrates referred to as SiC-A, and a 650 V/16 A rated SiC diode [236] referred to as SiC-B. Note that all Ga₂O₃ devices underwent multiple measurements at junction temperatures between -50 °C and 150 °C prior to the determination of the breakdown voltage.

In case of the trench diode (B1-XS), the die was fully functional after multiple measurement cycles, but a disconnection of the bond wires at the package lead was revealed by removing parts of the mold mass via laser ablation (see [254] for details). A breakdown voltage of $V_{\rm br} = 484$ V was determined by reestablishing an electrical connection to the bond wires. Using the previously determined specific on-resistance $r_{\rm o,sp}$ of $3.45 \,\mathrm{m\Omega}\,\mathrm{cm}^2$ at $25\,^{\circ}\mathrm{C}$, this yields a $V_{\rm br}^2/r_{\rm o,sp}$ figure-of-merit of $68 \,\mathrm{MW/cm}^2$. This value is significantly lower than the theoretical limit of about 10 to $40 \,\mathrm{GW/cm}^2$ for β -Ga₂O₃ and $3 \,\mathrm{GW/cm}^2$ for SiC, but exceeds the unipolar silicon limit by a factor seven (the exact value depends on the theoretical values used for the calculation, see Section 2.3.1).

From another perspective, the relatively high doping concentration of the epitaxial layer leads to a rather high electric field at the junction. Calculating the average breakdown field according to Eq. 3.4 yields 2.8 to 3.2 MV/cm for the device under test, depending on the exact values of ϵ_s and N_D . In [135] it was simulated for a β -Ga₂O₃ trench MOS diode of similar structure that the peak electric field in the epitaxial layer should be about 1 MV/cm lower than the one expected for a regular Schottky diode. Taking this difference into account, the achieved breakdown field is about seven times higher than the theoretical breakdown field of Si, and close to the theoretical SiC limit of 2.5 MV/cm [12].

The batch 3 planar diode of size ML broke down at a reverse voltage of 1290 V, corresponding to a breakdown field of about 2.0 MV/cm, which is 80% of the theoretical SiC limit and

more than a factor 6 higher than the Si limit. The resulting $V_{\rm br}^2/r_{\rm o,sp}$ figure-of-merit of $145 \,{\rm MW/cm^2}$ is about a factor of two higher than that of the early generation small-area trench diode. During the course of this thesis, similar and even higher breakdown fields exceeding $6 \,{\rm MV/cm^2} \, (V_{\rm br}^2/r_{\rm o,sp}$ figure-of-merit > $10 \,{\rm GW/cm^2}$) have been achieved for small-area (anode diameter < $200 \,\mu$ m) Ga₂O₃ diodes [18, 22, 23], showing the promising features of gallium oxide for high-voltage power devices. However, such high values still need to be demonstrated for large-area devices.

Figure 4.5 also shows that the current–voltage profiles up to the breakdown voltage exhibit distinct differences. The trench diode shows an almost constant leakage current up to a voltage of -410 V, at which point the current rises steeply from 3 nA to 30 µA. A second rather flat region then terminates in a hard and destructive breakdown at -484 V. The first step presumably indicates a non-reversible soft breakdown caused by an additional leakage path through the oxide layer, which was also observed in other Ga₂O₃ MOS capacitor and trench MOS structures [212]. On the other hand, the planar Ga₂O₃ diodes exhibit a rather uniform exponential growth, followed by a hard and non-reversible avalanche breakdown at 1145 V (LL) and 1290 V (ML). Within the specified reverse blocking voltage and up to breakdown voltages of about 1140 V (SiC-A) and 920 V (SiC-B), the leakage current of the SiC diodes is several orders of magnitude lower that that of the Ga₂O₃ diodes. In contrast to the Ga₂O₃ diodes, the SiC diodes remain fully functional even after significantly exceeding the onset of avalanche breakdown.

To further study this aspect, Fig. 4.5b compares the forward currents below the knee voltage before and after breakdown exemplarily for the 650 V/10 A rated SiC diode and the planar LL size Ga₂O₃ diode. While entering the breakdown region has no significant effect on the current–voltage characteristics of the SiC diode, a significant increase in current at forward voltages below 0.7 V can be seen in case of the Ga₂O₃ diode. The corresponding region is associated with a high ideality factor *n* of about 13 (compared to n = 1.03 before breakdown) and a low Schottky barrier height of 0.6 eV (1.22 eV before breakdown). At the same time, a high leakage current of 1 mA at -100 V that increases to the measurement limit of 8 mA at -850 V is observed when repeating the breakdown measurement.

This suggests that the avalanche breakdown in the Ga_2O_3 diodes starts at the edge termination or junction region, leading to substantial damage to the junction structure and degradation of the blocking capabilities. On the other hand, the p-type areas in SiC MPS (Merged Pin Schottky) diodes shield the junction from high electric fields and move the location of the breakdown away from it, which leads to a high avalanche ruggedness [34–36]. Therefore, adopting similar structures, e.g., based on p-type NiO, might be a feasible option for Ga_2O_3 diodes. At this point, however, a more extensive investigation of the breakdown properties was not performed due to the small number of available samples and the associated degradation of the device properties.

With $V_{br}^2/r_{o,sp}$ figures-of-merit of about 870 MW/cm² and 600 MW/cm², both the 10 A and 16 A rated commercial SiC diodes outperform the Ga₂O₃ devices under test by at least a factor of four. Due to the early stage of development of large-area Ga₂O₃ diodes, it is therefore currently not possible to design power converters exceeding the efficiencies with SiC. However, the most recent progress allows to assess the device properties in first application-relevant high-voltage converter circuits, which is the foundation for Chapter 6.



Figure 4.6: The capacitance–voltage profiles of (a) the small-area Ga_2O_3 trench MOS diode and (b) the large-area planar Ga_2O_3 as well as the SiC-B reference diode increase slightly when the junction temperature is raised from 25 °C to 150 °C. As shown in (c) exemplarily for the SiC reference and ML-size Ga_2O_3 diode, the increase in junction capacitance has a negligible effect on the capacitive charge. Therefore, the switching behavior can be expected to be independent of temperature. Parts adapted from [254].

Stable capacitance-voltage characteristics

The preceding sections show that both the forward and reverse current–voltage profiles can deviate significantly from those of typical SiC diodes, and also between the different types of Ga₂O₃ diodes. This section examines the parasitic junction capacitance of the previously discussed diodes. It is one of the central properties that determines the switching behavior of the devices. Typical capacitance–voltage (C-V) profiles are shown for the B1-XS trench diode in Fig. 4.6a, and for the B3-ML type diode and the SiC-B reference diode in Fig. 4.6b, all at a measurement frequency of 1 MHz and at junction temperatures of 25 °C and 150 °C. The basic profile is consistent with those typically observed for SiC diodes, where the capacitance decreases rapidly with increasing reverse bias as the space charge region extends further into the epitaxial layer and depletes it from charge carriers [183].

The junction capacitance at low reverse voltages (above -10 V) increases slightly at 150 °C, i.e., by about 15% for the XS-size Ga₂O₃ trench diode, 8% the ML-size field-plated Ga₂O₃ diode, and 6% for the SiC-B reference diode. Considering that at low reverse voltages the space charge region is only formed close to the Schottky junction, this may indicate a thermally activated release of charge carriers captured in defects at the metal–semiconductor (or metal–oxide–semiconductor) interface.

Fig. 4.6c shows the capacitive charge resulting from an integration of the C-V characteristics according to Eq. 3.5, exemplarily for the field-plated Ga₂O₃ diode and the SiC reference diode. The slight increase in capacitance at low reverse bias has a negligible effect on the stored charge. Therefore, the switching behavior of the device is not expected to change at high temperatures. The specific (i.e., related to the anode area) charge at voltage of -400 V (typical DC link voltage of an automotive inverter) is similar for both diodes, with about 11.4 nC/mm^2 in case of the SiC and 10.4 nC/mm^2 in case of the Ga₂O₃ diode. This value is affected by the doping concentration of the epitaxial layer and determines the switching losses, as will be discussed in detail in Chapter 6 for different types of Ga₂O₃ diodes.



Figure 4.7: The forward current–voltage profiles of four identical types of Ga₂O₃ diodes of one of the early device generations (batch B2a) differ significantly. (a) At a junction temperature of 25 °C sample S3 shows a strong curvature, samples S1 and S2 show a slight curvature, and sample S4 shows a straight line above the knee voltage. (b) At 150 °C all samples show a constant differential on-resistance. (c) The temperature-dependent differential on-resistance varies by up to 80 % between the samples. The results reveal a strong in-plane variation of electrical properties across the wafer. Adapted from [255].

However, the following section shows that these basic properties can vary significantly between different devices, even when they stem from different positions of the same wafer.

4.2.3 In-plane variation of electric properties across the wafer

The fact that the characteristic features of different Ga_2O_3 diodes can differ substantially is also caused by the early stage of device processing technology. This section investigates how large these variations can be. The issue is discussed for an early batch of Ga_2O_3 diodes (B2a) in comparison to a second batch from one of the most recent wafers (B3).

To illustrate the problem, the forward characteristics of four TO247 packaged Ga_2O_3 diodes (B2a) with identical structure are examined. All devices stem from different positions of the same wafer and have a $1.6 \text{ mm} \times 1.6 \text{ mm}$ large Ni/Au anode while the cathode is formed by Ti/Ni/Au layers. Figure 4.7a shows the corresponding forward current–voltage profiles at a junction temperature of $25 \,^{\circ}$ C. Samples S1, S2, and S3 show knee voltages of about 0.9 V while sample S4 shows a significantly lower knee voltage of about 0.7 V. In the case of sample S3, the profile is considerably curved, whereas the profiles of S1 and S2 are only slightly curved, and S4 exhibits a constant differential on-resistance above the knee voltage. At a junction temperature of $150 \,^{\circ}$ C, all samples exhibit a constant differential on-resistance, but the actual values differ significantly.

This is shown in more detail in Fig. 4.7c, where the differential on-resistance is obtained for temperatures between $25 \,^{\circ}\text{C}$ and $150 \,^{\circ}\text{C}$ by fitting a straight line to the data between forward currents of 3 A and 6 A. For sample S3, a significant decrease in r_0 is observed when increasing the temperature from $25 \,^{\circ}\text{C}$ to $125 \,^{\circ}\text{C}$. After that, the differential on-resistance increases with rising junction temperature. For samples S1, S2, and S4, the on-resistances remain approximately constant up to about 75 to $100 \,^{\circ}\text{C}$ and increase after that. The variation in differential on-resistance between the samples is as high as $80 \,\%$ at $150 \,^{\circ}\text{C}$. When increasing



Figure 4.8: The forward current–voltage characteristics of a more recent generation of diodes (batch 3) with three different anode sizes (five of size ML, two of size L, and three of size LL) show a more consistent behavior than the early generation in Fig. 4.7. At a junction temperature of (a) 25 °C and (b) 150 °C, the knee voltage and shape of the profile are similar for each device type. (c) The differential on-resistance varies by up to about 30 % for type ML, and about 20 % for type L and LL. The data still show a variation of device electrical characteristics across the wafer, but suggest an improvement of the processing parameters.

the junction temperature from $25 \,^{\circ}$ C to $150 \,^{\circ}$ C the conduction losses at $10 \,\text{A}$ increase by $11 \,\%$ for sample S1, but decrease by almost $20 \,\%$ for sample S3.

This suggests a strong in-plane variation of material properties across the wafer which can include, e.g., the thickness of the epitaxial layer or its doping concentration. In this respect, it is important to note that the Ga₂O₃ wafer is exposed to oxygen during the growth of the epitaxial layer, which may at least partially reduce the effective donor concentration in the substrate and thus lead to a certain in-plane variation. For the devices under test, a reduction from $5.1 \times 10^{18} \,\mathrm{cm}^{-3}$ to about $1 \times 10^{18} \,\mathrm{cm}^{-3}$ is expected by the manufacturer [255]. Furthermore, the epitaxial layer of the samples under test was grown by halide-vapor phase deposition (HVPE, see Section 2.3.2). This method offers high deposition rates exceeding $20 \,\mu m/h$, but is indeed known for its difficulty in achieving a uniform doping concentration and film thickness, the incorporation of Si or Cl impurities during the growth process, and the potential diffusion of atoms between epitaxial layer and substrate caused by high deposition temperatures [18, 123, 125, 126]. Since Cl is also expected to be a shallow donor in Ga_2O_3 [127], it was previously speculated whether these impurities might affect the electrical properties [126]. From the measurements made here, it cannot be concluded which of these effects contribute to the observed variation in the current-voltage profiles. However, the process control during HVPE growth appears to be a central aspect.

Figure 4.8 shows the forward characteristics of a more recent batch (B3) of Ga_2O_3 Schottky diodes. Again, all diodes are of the identical type and only the anode size differs. Five samples of type ML (anode $1.6 \text{ mm} \times 1.6 \text{ mm}$), two samples of type L (anode $2.4 \text{ mm} \times 2.4 \text{ mm}$) and three samples of type LL (anode $3.2 \text{ mm} \times 3.2 \text{ mm}$) are compared. Again, a certain spreading of the forward characteristics at 25 °C (Fig. 4.8a) and 150 °C (Fig.4.8b) is visible. However, the profile is linear above the knee voltage for all devices, and the variation seems to be much less pronounced than that of the previous batch shown in Fig. 4.7. This becomes more evident when comparing the temperature dependent on-resistances in Fig. 4.8c. First, the



Figure 4.9: Doping concentration of early and recent device generations fabricated on (a) 600 μm thick and (b) 200 μm thin wafers, normalized to the average doping concentration as determined from capacitance–voltage measurements. The spreading of the data, characterized by the difference between the 75th and 25th percentile, decreases significantly between the first and second batches. Although a significant variation of the doping concentration is still present in new generations, this indicates a substantial improvement in the growth and processing conditions of halide vapor phase grown epitaxial wafers.

on-resistance increases linearly between $25 \,^{\circ}$ C and $150 \,^{\circ}$ C for all devices under investigation, with a similar slope for all devices of the same size. Second, with about $31 \,\%$ (type ML, one outlier diode causing the rather high value), $22 \,\%$ (type L) and $19 \,\%$ (type LL), the maximum deviation is significantly lower than that the of the previous batch of diodes. This suggests that, as claimed by the manufacturer [232, 246, 247], the process control was improved for the newer batches.

A more quantitative image can be obtained when comparing the variation in doping concentration of different device generations. As explained in Section 2.2.3, the inverse doping concentration $N_{\rm D}^{-1}$ of a Schottky barrier diode is proportional to $A \cdot dC_{\rm j}^{-2}/dV_{\rm D}$, where A is the active area, $C_{\rm j}$ is the junction capacitance, and $V_{\rm D}$ is the diode voltage. Therefore, the doping concentration can directly be estimated from capacitance–voltage measurements. Figure 4.9a shows the doping concentration normalized to its average value for an early (batch B2c) and new generation (batch B3) of diodes grown on 600 µm thick wafers. The top and bottom of the box indicate the 75th and 25th percentiles with the respective absolute value as number next to it, and the middle line indicates the median. Figure 4.9b shows the same type of evaluation for diodes fabricated on early (B2b) and novel (B4) 200 µm thin wafers.

Taking the difference between the 75th and 25th percentiles as an indicator for the variation of the data, a reduction from 0.33 (old batch) to 0.09 (new batch) is observed for the 600 μ m thick devices. In comparison, the absolute variation of the doping concentrations is higher for the 200 μ m thin devices. However, a reduction from 0.66 (old batch) to 0.20 (new batch) is observed as well. As the number of devices available for this work is limited, a higher number of samples is needed for a more precise statistical evaluation. Nevertheless, the results suggest a decrease in the variation of electrical properties across the wafers as a result of an improved growth and process control, presumably associated with the epitaxial growth by HVPE. Although the device quality has been improved continuously over the course of this work, the variation of device properties that cannot be predicted precisely constitutes a

challenge for the optimal design of application-relevant circuits. Therefore, it is key for all experiments conducted as part of this work to design setups that allow for an investigation of all necessary device properties for a wide range of different devices.

The data also show that the change in on-resistance with rising junction temperature can vary between different devices. Given its importance in power electronics, the following section examines in detail how the change in on-resistance of Ga_2O_3 devices compares to that of SiC diodes.

4.2.4 Low increase in on-resistance with rising temperature

The on-resistance of a diode is one of the most decisive properties as it directly determines the conduction losses and with that also the thermal management strategy and efficiency of a power converter. As the chip temperature increases during operation, the device properties at junction temperatures above 100 °C are of particular interest. Taking the previously discussed variation of device properties into account, this section assesses the change in on-resistance with rising junction temperature T_j . This can be done through the coefficient α_{ref} describing the increase in differential on-resistance r_o relative to its value at a certain reference temperature T_{ref} [203, 213, 214]:

$$r_{\rm o}\left(T_{\rm j}\right) = r_{\rm o}\left(T_{\rm ref}\right) \cdot \left(\frac{T_{\rm j}}{T_{\rm ref}}\right)^{\alpha_{\rm ref}}.$$
(4.7)

Usually, a reference temperature of 300 K is used for the calculation [213, 214]. The resulting coefficient will be referred to as α_{300} in the following. As outlined in Sections 4.2.1 and 4.2.3, however, the differential on-resistance above the knee voltage can be highly non-constant up to junction temperatures of usually about 50 °C to 75 °C. A definition of a single temperature coefficient α_{300} is not meaningful for these diodes. To enable a further comparison for all types of diodes, Eq. 4.7 is adjusted so that the relative rise in on-resistance is referenced to a temperature $T_{\rm ref}$ of 100 °C or 373 K, and only measurements above 75 °C are taken into account. The rise in differential on-resistance at high temperatures is then described by the corresponding parameter α_{373} .

Figure 4.10 shows the α_{373} coefficients for all types of Ga₂O₃ diodes studied in this thesis, except the ones that do not exhibit a steady increase in on-resistance with rising temperature even above 100 °C (the coefficients are <1 for these devices). In addition, the α_{300} coefficients are shown for all Ga₂O₃ devices with a constant differential on-resistance that increases with temperature even at 25 °C. The last column shows the corresponding coefficients of five commercial SiC reference diodes. The exact fitting interval when determining the on-resistance has a slight influence on the corresponding coefficient α_{ref} . However, the influence is minimized by using the same fitting intervals for all diodes as described in Section 3.3 (corresponding to a range from about 70 % to 130 % of the expected typical operating point) and it is negligible compared to the general trend of the data.

The corresponding α_{300} coefficients vary between 1.55 and 2.01 for the SiC diodes but only between 1.17 and 1.48 for the Ga₂O₃ diodes, showing that the Ga₂O₃ diodes exhibit a lower relative increase in differential on-resistance with rising temperature than their SiC counterparts. Interestingly, the α_{373} coefficients do not differ significantly from the α_{300} coefficients in case of Ga₂O₃. This means that the increase in on-resistance can be described well with Eq. 4.7 for all temperatures. Among all Ga₂O₃ diodes (i.e., also the ones with



Figure 4.10: Increase in differential on-resistance with rising temperature as described by the coefficients α_{300} and α_{373} according to Eq. 4.7 for multiple Ga₂O₃ diodes from different batches (see Tab. 3.2) and SiC reference diodes. α_{300} values are only shown for diodes that exhibit a positive temperature coefficient of the forward voltage at all temperatures. The increase in on-resistance is significantly lower for Ga₂O₃ than for SiC, with an α_{373} average and standard deviation of 1.22 ± 0.16 and 2.24 ± 0.22 , respectively. This difference is connected to the temperature-dependent electron mobility and makes Ga₂O₃ diodes appealing for high-temperature applications.

non-constant differential on-resistance at low temperatures), the α_{373} coefficients vary between about 0.95 and 1.47. For the SiC reference devices, however, they increase to values between 2.02 and 2.61, which indicates that the relative increase in differential on-resistance is stronger than described by Eq. 4.7. Averaging over all devices shown in Fig. 4.10 yields coefficients α_{373} of 1.22 with a standard deviation of 0.16 for Ga₂O₃, and 2.24 with a standard deviation of 0.22 for SiC. In the following, this difference will be connected to the basic material properties of Ga₂O₃ and SiC.

The series resistance of a Schottky diode is mainly caused by the lightly doped drift region in the epitaxial layer. Apart from structural parameters such as the active area A and thickness of the epitaxial layer $z_{\rm epi}$, it is determined by the electron mobility $\mu_{\rm e}$ and doping concentration $N_{\rm D}$ according to $r_{\rm epi} = z_{\rm epi}/(eAN_{\rm D}\mu_{\rm e})$ [34]. In a first approximation, all quantities except for the electron mobility are temperature independent. The relative increase in resistance of the epitaxial layer can therefore simply be calculated by $r_{\rm epi} (T_{\rm epi}) / r_{\rm epi} (T_{\rm ref}) = \mu_{\rm e} (T_{\rm ref}) / \mu_{\rm e} (T_{\rm epi})$, i.e., the change of electron mobility with the temperature of the epitaxial layer $T_{\rm epi}$ related to a certain reference temperature $T_{\rm ref}$. By combining this relationship with Eq. 4.7 (i.e., when assuming that the differential on-resistance is mainly caused by the epitaxial layer) the coefficient $\alpha_{\rm ref}$ can be expressed by

$$\alpha_{\rm ref}\left(T_{\rm epi}\right) = \frac{\log\left(\frac{\mu_{\rm e}\left(T_{\rm ref}\right)}{\mu_{\rm e}\left(T_{\rm epi}\right)}\right)}{\log\left(\frac{T_{\rm epi}}{T_{\rm ref}}\right)}$$
(4.8)

A recent study [215] on the material properties of Ga_2O_3 and 4H-SiC showed that at 300 K the electron mobility of Ga_2O_3 is about four times lower than that of SiC, but the decrease

of electron mobility with rising temperature is significantly less pronounced in Ga₂O₃. The electron mobilities of 4H-SiC ($\mu_{e,SiC}$) and β -Ga₂O₃ ($\mu_{e,GaO}$) at an effective donor concentration of 1×10^{17} cm⁻³ were found to decrease exponentially with temperature *T* according to the following equations [215]:

$$\mu_{e,GaO}(T) = [1147 \cdot \exp(-T/131 \,\mathrm{K}) + 31] \,\mathrm{cm}^2 \,\mathrm{V}^{-1} \,\mathrm{s}^{-1} \tag{4.9}$$

$$\mu_{\rm e,SiC}(T) = [5422 \cdot \exp(-T/128\,\rm K) + 95]\,\rm cm^2\,V^{-1}\,s^{-1}$$
(4.10)

By inserting this dependency of the electron mobilities in Eq. 4.8, the coefficient α_{ref} of Ga₂O₃ and SiC can be estimated and compared also from a material point of view. Calculating the corresponding values exemplarily for $T = T_{epi} = 150 \,^{\circ}\text{C}$ (i.e., the maximum measurement temperature in the experiments) results in coefficients α_{300} of 1.91 (Ga₂O₃) and 2.15 (SiC) as well as α_{373} of 1.94 (Ga₂O₃) and 2.23 (SiC). The value calculated for SiC is exactly within the range displayed in Fig. 4.10, which validates the calculation. However, the value for β -Ga₂O₃ is about 0.5 higher than the highest value observed in Fig. 4.10. First, it should be noted that the previously discussed unusual electrical characteristics indicate an additional influence on the current–voltage profiles that is not considered here. Also, note that the doping concentration of the studied diodes is about one order of magnitude lower than assumed in the calculations using the data from [61, 215] (the availability of reliable data to enable more detailed calculations is still limited for Ga₂O₃). As the doping concentration significantly affects the electron mobility [18, 61, 216], this is expected to affect the calculations. In addition, the influence of the heavily doped substrate that also contributes to the on-resistance is not considered here.

Despite the uncertainties accompanied by these simplifications, the data show that the low increase in on-resistance with rising junction temperature is not only caused by the specific structure of the Ga₂O₃ diodes under investigation in this thesis. Instead, it appears to be generalizable as a direct consequence of the material properties of β -Ga₂O₃. In recent studies, small-area 500 V β -Ga₂O₃ diodes (anode diameter up to 250 µm) remained fully functional up to junction temperatures of 600 K or 327 °C [143], and β -Ga₂O₃ transistor cells with a width of 50 µm were operated successfully at 500 °C [217]. In combination with the results discussed here, this means that junction temperatures higher than the current typical limit of 150 to 175 °C may be feasible for Ga₂O₃ diodes. As a consequence, new packaging concepts with high-temperature materials (especially regarding the maximum temperature of the mold mass or die attach material) may become important for Ga₂O₃ devices. The actual device temperature is then determined by its thermal resistance, which will briefly be discussed in the next section.

4.3 Basic thermal considerations

To gain a basic insight into the thermal management of Ga_2O_3 power devices, the thermal impedance is measured for the previously discussed TO247 packaged Ga_2O_3 B3-ML type diode with a thickness of 600 µm (currently, this is a standard value based on the available substrates [22, 231]) and the commercial TO247 packaged 650 V/16 A SiC-B reference diode according to the procedures described in Section 3.5. Here, thermal grease is used as thermal interface material between case and heatsink for both devices.

Figure 4.11b shows the resulting thermal impedance profiles and structure functions measured during the cooling phase after heating for $100 \,\mathrm{s}$ with a heating current of $3 \,\mathrm{A}$ (heating power



Figure 4.11: (a) Absolute and (b) specific (related to anode area) thermal impedance profiles of a TO247 packaged 600 µm thick B3-ML type Ga₂O₃ diode and the TO247 packaged SiC-B reference diode. The low thermal conductivity of Ga₂O₃ leads to a significant increase of the thermal resistance. Even considering the differences in anode area, the junction-to-ambient thermal resistance is about four times higher than that of the SiC reference diode and limits the power rating substantially.

9.0 W) in case of the Ga₂O₃ diode and 16 A (heating power 25.5 W) in case of the SiC diode. The conduction losses lead to a rise in junction temperature of about 99 K for the Ga₂O₃ and 39 K for the SiC diode. The thermal impedance profiles reach constant values of about 10.60 K/W and 1.55 K/W, respectively. The fact that the two curves separate already after only microseconds indicates that the die is primarily responsible for the difference. Relating the values to the anode area of the devices yields $25.4 \text{ mm}^2 \text{ K/W}$ for the Ga₂O₃ diode, but only $6.2 \text{ mm}^2 \text{ K/W}$ for the SiC diode, which means that the specific junction-to-ambient thermal resistance of the Ga₂O₃ diode is more than four times higher than that of the SiC diode.

It should be noted that the different chip sizes affect the heat flow path also in the die-toambient thermal path. To get a clearer view on the influence of the die itself on the thermal resistance, the die and anode sizes should be similar, and they should be assembled exactly in the same way. In this respect, note that a 0.5 mm thick CuMo spacer is inserted between the die and lead frame in case of the Ga₂O₃ diode (see Section 3.2). However, assuming a thermal conductivity of 200 to $250 \text{ W m}^{-1} \text{ K}^{-1}$ for CuMo [218] and a heat flow area of $2.2 \text{ mm} \times 2.2 \text{ mm}$ based on the effective chip size, the thermal resistance should be around 0.4 to 0.5 K/W and thus make up less than 5% of the junction-to-ambient thermal resistance.

Therefore, the low thermal conductivity of Ga_2O_3 clearly translates to a substantially reduced power rating, which is likely to cause thermal management issues if the die is supposed to be operated under the same conditions as the SiC reference diode. The results motivate the investigation of strategies to improve the power and current rating of Ga_2O_3 diodes discussed in detail in Section 5.

4.4 Modeling atypical current-voltage profiles in Spice

The preceding Sections 4.2 and 4.3 show that the classic models based on thermionic (field) emission can be applied to the investigated Ga_2O_3 devices only within certain operating



Figure 4.12: Basic principle of a diode simulation model in LTspice using behavioral current and voltage sources. The model can be divided into an electrical block modeling the forward and reverse current–voltage characteristics (i_1) as well as the current through the junction capacitance (i_2) , and a thermal block modeling the junction temperature based on the heating power (i_3) that is fed into a thermal Cauer network. Adapted from [258].

ranges. Curved characteristics in the current–voltage profiles and a change in temperature coefficient lead to deviations from these models. Although the focus of this work is not the physical modeling of Ga_2O_3 diodes, accurate Spice models are essential for the development of power converters or simply for the design of experimental setups. Since models from manufacturers are not yet available, simple models that can reproduce the characteristics were created. Such models have been used throughout this work to estimate basic operating limits and thereby avoid the destruction of the limited number of devices.

Since the electrical properties of different diodes can differ significantly even when the diode structure and junction design are identical, a main requirement is that the model must offer enough degrees of freedom but also allow a simple adjustment to reproduce the varying characteristics. However, many physical properties of the gallium oxide diodes, such as the exact active area, the precise thickness of the epitaxial layer, or the accurate electron mobility of the substrate and epitaxial layer with their respective temperature dependence, are not known. Furthermore, without detailed knowledge of the growth process of the devices, the physical cause of the unusual behavior cannot be conclusively determined. Therefore, the model cannot be purely physical. Instead, it is based on semiconductor physics, but overlaid with purely mathematical terms.

The standard Spice model for Schottky diodes consists of a thermal block and an electrical block, as shown schematically in Fig. 4.12 and described in more detail in [219–221]. Behavioral current and voltage sources are used to implement formulas in LTspice. In the electrical block, i_1 models the forward and reverse current–voltage profiles, and i_2 is the current through the junction capacitance. The power dissipation leading to a rise in junction temperature is modeled by the current source i_3 . The implementation of i_2 and i_3 follows standard methods also described, e.g., in [219–221]: The current through the junction capacitance is modeled by defining the capacitance values and calculating the current from $i_2(t) = C_j(v)dv/dt$, and the thermal model consists of a Cauer network that can directly be extracted from measurements of the structure function (see Section 3.5). The more important part for this thesis is the implementation of the reverse leakage current and the forward current with changing temperature coefficients and curvatures.

First, the reverse blocking characteristics are considered. From Section 4.2.2 it can be seen that, starting from the simplified thermionic field emission model, the modeled curve can be shifted by several orders of magnitude, the slope in the logarithmic plot can be different, and there can be a temperature and voltage-dependent bending of the curve. To compensate for these shifts, a correction factor attributed to the diode structure (e.g., influence of trench structure or



Figure 4.13: Simulated and measured (a) reverse and (b) forward current–voltage characteristics of Ga₂O₃ diodes with unusual behavior at junction temperatures between 25 °C and 150 °C. (c) Simulated and measured temperature response of the Ga₂O₃ diode from Fig. 4.11b during the corresponding thermal measurement. The adjusted Spice models can reproduce the electrical and electrothermal properties accurately and can thus serve as an aid during further experiments.

edge termination) and the unknown source of additional leakage current is multiplied to the pure TFE term:

$$I_{\rm r} = I_{\rm r,TFE} \cdot 10^{\zeta_{\rm rs}(T_{\rm j}) + \frac{\zeta_{\rm ri}(T_{\rm j})}{V_{\rm ref}} \times (V_{\rm D} - V_{\rm ref}) + \frac{\zeta_{\rm rb}(T_{\rm j})}{V_{\rm ref}^2} \times (V_{\rm D} - V_{\rm ref})^2}$$
(4.11)

In a typical plot where the y-axis is logarithmic, the parameter ζ_{rs} shifts the curve by ζ_{rs} orders of magnitude upwards or downwards. The parameter ζ_{ri} changes the slope/inclination of the curve with respect to a reference voltage V_{ref} , e.g., -600 V for the diode in Section 4.2.2. The parameter ζ_{rb} induces a parabolic bending of the current–voltage profile. The set of parameters (ζ_{rs} , ζ_{ri} , ζ_{rb}) are unique for each diode and can easily be determined for each temperature through a Matlab fitting tool after measuring the static reverse characteristics.

A similar procedure is performed in forward direction. Using the thermionic emission model as a starting point, the on-resistance $r_o(V_D)$ is defined voltage-dependent using a look-up table, and a temperature-dependent shift parameter $\zeta_{fs}(T_i)$ is added:

$$I_{\rm f} = I_{\rm s} \cdot \exp\left(\frac{e\,\zeta_{\rm fs}\left[V_{\rm D} - I_{\rm f}r_{\rm o}(V_{\rm D})\right]}{nkT_{\rm j}}\right) \tag{4.12}$$

Figure 4.13 compares the corresponding simulated and measured characteristics exemplarily for large-area β -Ga₂O₃ Schottky diodes with significantly varying temperature coefficients between junction temperatures of 25 °C and 150 °C. Clearly, the procedure is a simple but effective method to precisely reproduce the unusual reverse (Fig. 4.13a) and forward (Fig. 4.13b) current–voltage characteristics. Note that the parameters ζ_{rs} , ζ_{ri} , ζ_{rb} continuously approach zero at high temperatures for the previously discussed example of the field-plated batch 3 diodes with unusual temperature coefficient, which agrees with the previous presumption of a transition from trap assisted tunneling at low temperatures to TFE and/or PFE at higher temperatures.

Figure 4.13c shows the temperature response from a transient thermal measurement with a B3-ML diode according to the procedures described in Section 3.5. The thermal Cauer network model was extracted from the measurement by using the T3Ster Master software from Mentor. The measurement was then replicated in Spice with a full electrothermal model according to the previously described methods, showing that also the electrothermal properties can be reproduced accurately.

In summary, the unusual characteristics with unknown cause can be modeled in Spice by adding mathematical terms to physics-based equations. This helps with the planning and design of more advanced and application relevant circuits, or simply to lower the risk of harming the device before choosing certain measurement settings throughout the thesis.

4.5 Summary and questions arising

Chapter 4 investigates basic properties of β -Ga₂O₃ diodes. The key results can be summarized as follows:

- 1. Basic current–voltage profiles can show varying temperature coefficients of the forward voltage and thus differ significantly from standard SiC Schottky diodes. This seems to be caused by the device or wafer processing rather than the intrinsic properties of β –Ga₂O₃. A transition between different conduction mechanisms is also observed in reverse direction for multiple diodes. Spice models that can reproduce these varying characteristics have been created to simplify measurement planning.
- 2. A strong variation of material properties for early device generations is observed, but an improvement of the processing technology is discernible for newer device generations. However, experimental setups need to be flexible to allow an investigation of all types of diodes.
- 3. The increase in differential on-resistance with rising temperature is significantly smaller for Ga_2O_3 than for SiC diodes, which makes them attractive for high-temperature applications.
- 4. Reverse breakdown is hard and destructive (complete loss of blocking capability) and should thus be avoided.
- 5. The junction properties and temperature stability are found to be sufficient to proceed with the next step of research, where two main issues are identified regarding a potential application in power electronics:
 - a) The low thermal conductivity of Ga_2O_3 was shown to translate to a substantial rise in junction temperature that limits the power rating. What are potential solutions to this problem, and how effective are they?
 - b) Switching properties of power devices are key, but there is a lack of studies in the literature. Is a stable converter operation possible despite the high thermal resistance, and what must be done to improve the performance? How does the switching behavior compare to mature technologies?

These constitute the guiding questions that will be covered in the following Chapters 5 (thermal properties) and 6 (switching behavior).

Chapter 5

Thermal properties and methods to improve the heat dissipation

5.1 Chapter overview

Section 4.3 showed that the low thermal conductivity of Ga₂O₃ causes a substantial thermal resistance when standard $600 \,\mu\text{m}$ thick Ga_2O_3 diodes are cooled from the cathode side, as common for Si or SiC. One way to alleviate this issue is to thin down the substrate. Motivated by this, Section 5.2 uses four $200 \,\mu m$ thin diodes to investigate the heat dissipation of thinned chips in TO247 standard packages, which are often used in applications. To this end, measurements of the structure functions are combined with parametric studies and X-ray analyses. Another potentially effective thermal management method is to cool the diodes directly from the junction side. In Section 5.3, it is comparatively examined how effective both substrate thinning and junction-side cooling are to improve the current and power rating of Ga_2O_3 diodes. To this end, 600 µm thick and new 200 µm thin Ga_2O_3 bare dice of identical lateral dimensions are assembled in cathode-side cooled (CSC) and junction-side cooled (JSC) configuration on custom-made ceramic substrates. A commercial SiC bare die of similar size is mounted in the same manner and serves as a benchmark. In addition, a $600 \,\mu\text{m}$ thick Ga₂O₃ with a larger chip size is briefly investigated to discuss the influence of increasing chip size on the junction temperature. This way, a direct and fair comparison is possible that is further supported by thermal simulations. Given the potentially low on-resistance of Ga_2O_3 diodes, it is discussed whether and how similar low chip temperatures as for SiC diodes can be achieved in the future. Section 5.4 summarizes the key results of this chapter. Parts of this chapter have been published by the author in [255, 256, 259] during the course of this thesis.

5.2 Heat dissipation in TO247 packaged diodes with thinned substrate

5.2.1 Setup and investigated samples

As outlined in Section 2.3, Ga_2O_3 devices are commonly fabricated on 500 to 700 µm thick substrates. By thinning the substrates, the heat flow path is effectively shortened. This is expected to reduce the thermal resistance and also the specific on-resistance of the devices. Thinning processes are generally known from silicon thin wafer technology, where thicknesses below 100 µm are common [222]. The mechanical properties of Ga_2O_3 wafers (see Section 2.3.2) allow thinning by similar methods such as mechanical grinding.

Here, four ML-size Schottky diodes (S1, S2, S3, S4, all from batch B2a according to Tab. 3.1) with a structure as shown in Fig. 5.1a were externally fabricated on a (001) β -Ga₂O₃ wafer



Figure 5.1: (a) Schematic structure of the 200 µm thin TO247-packaged diodes used to study the heat dissipation. (b) Photograph of a silicone-gel sealed sample. The chips are diced out in a rectangular shape, but only the ML-size anode is electrically connected with bond wires. (c) A computer tomography (CT) scan reveals the cross-sectional structure of the packaged dice with CuMo spacer between die and lead frame, and confirms the targeted die thickness. Adapted from [255].

that was thinned to a thickness of $200\,\mu\text{m}$ by grinding and polishing. Detailed studies on essential wafer parameters, such as the resulting bow and yield after thinning, have not yet been performed. Currently the process is being adapted to enable thicknesses of $100\,\mu\text{m}$ or below.

The substrate exhibits an initial effective donor concentration of 5.1×10^{18} cm⁻³. As explained in Section 2.3.3, the wafer is exposed to oxygen and high temperatures during the subsequent HVPE-growth of the epitaxial layer. This is expected to reduce the effective donor concentration of the substrate to between 1×10^{18} cm⁻³ and the initial value [255]. As specified by the manufacturer, the epitaxial layer is about 6.6 to 13 µm thick with an effective donor concentration of about 0.9 to 3×10^{16} cm⁻³ [255], which agrees with the capacitance–voltage measurements in Section 4.2.3. A Ti(50 nm)/Ni(350 nm)/Au(200 nm) layered structure forms the ohmic cathode contact, and a Ni(20 nm)/Au(800 nm) layer forms the 1.6 mm × 1.6 mm large Schottky contact. Since the devices are fabricated purely for thermal testing, they do not contain any electric field management methods such as edge terminations or trench junctions. Note, however, that switching tests performed with the newest generation of 200 µm thin 650 V rated diodes with edge termination in Chapter 6 suggest that substrate thinning does not affect the breakdown voltage [260].

As described in Section 3.2, all diodes were assembled on TO247 lead frames by soldering (cathode) and wire bonding with ten 50 μ m thin Au wires (anode). Samples S1, S2, and S3 are fully molded, while sample S4 is sealed with Si gel as shown in Fig. 5.1b. The figure also shows that the chips are diced out in a rectangular rather than square format because of additional test structures on the wafer. However, these test structures do not contribute to any current flow, as they are not contacted with bond wires. As will be shown later in this chapter, the heat spreading into these parts of the die is negligible. Therefore, an effective die size of 2.2 mm × 2.2 mm can be used for a more meaningful comparison to other devices. A computer tomography (CT) scan is performed to reveal the cross-sectional structure of the diode after assembly. The resulting image in Fig. 5.1c shows the Ga₂O₃ die soldered to a 0.5 mm thick CuMo spacer between lead frame and die, and confirms a die thickness of about 200 μ m.

As examined in Section 4.2.3, the variation of material properties across the wafer influences the electrical characteristics substantially. The forward I-V profiles of the samples S1–S4 studied here were already discussed exemplarily in Fig. 4.7. It was found that sample S3

shows a visibly curved I-V profile at 25 °C above the knee voltage, while samples S1, S2, and S4 show a constant specific differential on-resistance $r_{o,sp}$. All samples first exhibit a decrease in $r_{o,sp}$ with rising junction temperature that turns into an increase in $r_{o,sp}$ above 100 °C (125 °C for S3). According to Eq. 2.3, thinning the heavily doped substrate from 600 to 200 µm should lower the $r_{o,sp}$ of the devices under investigation by about 5 to 30 %. The actual reduction critically depends on the unknown exact doping concentration and associated electron mobility. Since the specific on-resistances of samples S1–S4 already differ by up to a factor of 1.8, a clear experimental comparison is not possible here, but will be performed in Section 5.3. Nevertheless, the diodes can carry a current of several amperes, which allows them to be subjected to transient thermal measurements.

5.2.2 Transient temperature response and potential measurement issues

The general procedure to calibrate the temperature sensitive parameter (TSP) of Ga₂O₃ diodes and perform transient thermal measurements is described already in Section 3.5. A K-factor calibration is performed for the samples under investigation before the $R_{\rm th}$ measurements. All devices exhibit a linear relationship between the forward voltage at 10 mA ($V_{\rm f@10mA}$) and the junction temperature T_j associated with a K-factor of about 1.7 mV/K. To exclude an influence on the results caused by a potential change in junction properties of the rather experimental dice, K-factor calibrations were performed again after the transient thermal measurements. The differences are well below 1 % for all samples, which corresponds to an uncertainty in the determination of the junction temperature of well below 0.5 K when the average increase from 25 to 65 °C during the $R_{\rm th}$ measurements is considered.

Applying heating currents around 3 A causes an increase in junction temperature of about 40 K for all samples. Figure 5.2a shows the temperature response exemplarily for sample S1 during the cooling phase after heating for 100 s. The shape of the profile is similar to the one obtained with the B3-L diode in Fig. 3.15. The inset in Figure 5.2a illustrates the determination of the initial junction temperature T_{j0} via the \sqrt{t} -method. The slope m_T of the fitted straight line is connected to the chip area $A_{ch,eff}$ and heating power ΔP_H via [94, 107, 197]

$$A_{\rm ch,eff} = \frac{\Delta P_{\rm H}}{m_{\rm T}} \cdot \frac{2}{\sqrt{\pi \cdot c_{\rm th} \cdot \lambda_{\rm th} \cdot \rho}} \,. \tag{5.1}$$

When assuming standard values of $c_{th} = 560 \,\mathrm{J\,kg^{-1}\,K^{-1}}$ [87] for the specific heat, $\lambda_{th} = 21 \,\mathrm{W\,m^{-1}\,K^{-1}}$ [87] for the thermal conductivity, and $\rho = 5945 \,\mathrm{kg/m^3}$ [102] for the mass density of Ga₂O₃, chip areas ranging from $3.1 \,\mathrm{mm^2}$ (S4), $3.4 \,\mathrm{mm^2}$ (S1), and $3.8 \,\mathrm{mm^2}$ (S2) to $4.3 \,\mathrm{mm^2}$ (S4) are obtained. For all samples, this is between the anode area of $2.4 \,\mathrm{mm^2}$ and the previously presumed effective chip area of $4.8 \,\mathrm{mm^2}$, but significantly lower than the entire chip area of $8.4 \,\mathrm{mm^2}$. This is reasonable since the active area below the anode constitutes the major heat source of the die but a certain heat spreading is expected, and thus validates the measurements. Considering the difference between the calculated area and the entire chip area, it further indicates that the heat does not spread significantly into the additional part of the die that is not electrically connected with bond wires. This is also supported by thermal simulations in Section 5.2.4, and confirms that the additional part of the die should be neglected when comparing area-specific properties with other devices.

The fact that the temperature remains constant between about 20 s and the maximum measurement time of 100 s in Fig. 5.2a suggests that a thermal equilibrium is reached. However, the junction temperature appears lower than before the measurement. With a difference



Figure 5.2: (a) Junction-temperature response and determination of the initial junction temperature exemplarily for sample S1. (b) After the cooling phase, an exponential decrease of $V_{f@10mA}$ back to the initial value with a time constant in the range of minutes is observed. (c) The time constant spectrum shows that the largest time constants associated with the R_{th} measurement are only in the range of seconds. The parasitic drift in (b) is therefore expected to be negligible for the evaluation of the R_{th} measurements. Adapted from [255].

of -4 K, the effect is the strongest in case of sample S4. The reason is a parasitic drift in $V_{\text{f}@10\text{mA}}$ of about +6 mV in case of S4. In Fig. 5.2b, $V_{\text{f}@10\text{mA}}$ is recorded for a period of 60 min immediately after the R_{th} measurement. Fitting a function of the form $V_{\text{f}@10\text{mA}} \propto \exp(t/\tau)$ to the data suggests an exponential decrease of $V_{\text{f}@10\text{mA}}$ back to its initial value with a time constant of (352 ± 24) s. This may result from an accumulation or trapping of charge carriers at defects close to the MS junction during forward current flow, which increases the bending of the conduction band. The charge carriers are released when the external voltage and current flow are cut off. This thought in principle agrees with the findings in Section 4.2 that suggest the presence of some trap states at least in early generation devices.

To further study the influence of this effect on the $R_{\rm th}$ measurements, the continuous distribution of the time constants $\tau_{\rm i}$ (see Section 3.5) is examined in Fig. 5.2c exemplarily for sample S4 for two measurements with different TIMs. Changing the TIM affects the time constant distribution on a scale of seconds. The distinct intensity peaks in the range from μ s to s result mainly from the die, die attach, and package, indicating that heating and cooling of the die happens on a much shorter time scale. The time constant of the parasitic effect described above is in the range of several minutes (see Fig. 5.2b), and thus about one to two orders of magnitude larger than the largest time constant in Fig. 5.2c. Therefore, it is not expected to affect the $R_{\rm th}$ measurements significantly: Although the absolute junction temperature estimated from the K-factor calibration may be several Kelvin too low, the impact on the relative temperature increase as well as the cooling profile $T_{\rm j}(t)$ in the early cooling phase should be negligible. This consideration provides the basis for the discussion of the transient thermal measurements in the following section.

5.2.3 Thermal structure functions, impedance profiles, and their variation

The thermal impedance profiles and structure functions of all samples are shown in Fig. 5.3. The thermal impedance profiles (Fig. 5.3a) all reach a constant value after 100 s, from which junction-to-ambient thermal resistances $R_{\text{th,ja}}$ of 5.1 K/W (S3), 5.5 K/W (S1), 5.8 K/W (S2),



Figure 5.3: Results of transient thermal measurements for all TO247-packaged samples of batch B2a. (a) The thermal impedance profiles deviate already in the range of μ s to ms, indicating that differences in the thermal resistance near the heat source might be responsible for the variations in $R_{\text{th,ja}}$ and $R_{\text{th,jc}}$. (b) The variation of the length of the plateau at about 0.001 W s/K in the structure functions supports this hypothesis. Adapted from [255].

6.2 K/W (S4) can be extracted for the measurements with thin sheet as TIM. The splitting points of the curves with two different TIMs mark the transition from heat flow within the package to the TIM and heatsink. They indicate a variation of the associated junction-to-case thermal resistances $R_{\text{th,jc}}$ between 4.3 K/W (S3) and 5.5 K/W (S4), i.e., a variation by a factor of 1.28. This is significantly higher than the maximum expected measurement uncertainty of up to 15 % (see Section 3.5) and therefore requires a closer investigation. The fact that the profiles start to deviate already in the range of μ s to ms suggests that the difference may result from the die or die attach.

The structure functions in Fig. 5.3b allow a more detailed view. The initial plateau at a thermal capacitance of about 0.001 W s/K corresponds to the Ga₂O₃ dice. The die attach, CuMo spacer, subsequent solder layer, and copper lead frame cause the following section with slightly varying slopes up to about 1 W s/K. The $R_{\text{th,jc}}$ determined from the respective splitting points varies between 4.2 K/W (S3) and 5.2 K/W (S4), i.e., by a factor of 1.23, which is consistent with the thermal impedance profiles as summarized in Tab. 5.1. For all samples, the slope resulting from spacer, solder layers, and lead frame spans over the same range of about 2 K/W and 1 W s/K. This can be verified by assuming vertical heat flow in the 0.5 mm thick CuMo spacer ($\lambda_{th} = 200 \text{ W m}^{-1} \text{ K}^{-1}$ [218]) and solder layers ($\lambda_{th} = 60 \text{ W m}^{-1} \text{ K}^{-1}$ [176],

Table 5.1: Essential electrical and thermal parameters of the samples S1–S4 from batch B2a: differential on-resistance r_0 , K-factor K_f , junction-to-case thermal resistance $R_{th,jc}$ estimated from the thermal impedance profile and the structure function, and junction-to-ambient thermal resistance $R_{th,ja}$, which is identical for both methods. Adapted from [255].

| Unit | S1 | S2 | S3 | S4 |
|-----------|---|--|--|--|
| $m\Omega$ | 408 | 362 | 256 | 349 |
| mV/K | 1.73 | 1.74 | 1.77 | 1.68 |
| K/W | 4.7 | 4.9 | 4.3 | 5.5 |
| K/W | 4.7 | 4.8 | 4.2 | 5.2 |
| K/W | 5.5 | 5.8 | 5.1 | 6.2 |
| | Unit mΩ mV/K K/W K/W K/W | Unit S1 mΩ 408 mV/K 1.73 K/W 4.7 K/W 4.7 K/W 5.5 | Unit S1 S2 mΩ 408 362 mV/K 1.73 1.74 K/W 4.7 4.9 K/W 4.7 4.8 K/W 5.5 5.8 | Unit S1 S2 S3 mΩ 408 362 256 mV/K 1.73 1.74 1.77 K/W 4.7 4.9 4.3 K/W 4.7 4.8 4.2 K/W 5.5 5.8 5.1 |



Figure 5.4: Top-view cross-sectional CT scans to investigate the quality of the solder layers between die and CuMo spacer (top row) and CuMo spacer and lead frame (bottom row) for all batch B2a samples. The similarity suggests negligible differences in the quality of the solder layers (especially the die attach layer) between the samples.

thickness 70 µm) through an area that is equal to the effective chip size of 4.84 mm^2 , and a heat spreading in the 3 mm thick copper lead frame ($\lambda_{\text{th}} = 394 \text{ W m}^{-1} \text{ K}^{-1}$) to twice this area, resulting in a thermal resistance of about 1.8 K/W.

The plateau presumably resulting from the Ga_2O_3 die varies in length. Due to the flat rounded transition between two different materials in the structure function the length cannot be determined with a high precision, but is estimated to be about $2.3 \,\mathrm{K/W}$ for S3, $2.6 \,\mathrm{K/W}$ for S2, 2.7 K/W for S1, and 2.9 to 3.4 K/W for S4. Such differences can have multiple reasons such as variations in thickness or thermal conductivity of the die and the quality of the die attach layer. CT scans are performed for all samples to gain further insight. Side-view cross-sectional images such as in Fig. 5.1c indicate that the thickness of the dice varies by less than $20 \,\mu\text{m}$, which corresponds to differences in the $R_{\text{th,die}}$ of below $0.2 \,\text{K/W}$ and is thus significantly less than the observed $R_{\rm th}$ variations. Although the resolution of such analyses is limited by the contrast of structures smaller than a few µm, this is consistent with the manufacturer's specifications of possible variations between $189 \,\mu\text{m}$ and $204 \,\mu\text{m}$. Top-view cross-sectional CT scans in Fig. 5.4 are performed to investigate the quality of the solder layers. The top row images show the solder layer between die and CuMo spacer, and the bottom row images show the solder layer between CuMo spacer and lead frame. For all samples, a rather normal and similar amount of voids in the solder layers is visible. This suggests that differences in the quality of the solder layers should be negligible here. The detailed analysis of the sample structures allows the design of an accurate diode model that is used to gain further insight into the heat dissipation by static and transient thermal simulations conducted in the next section.

5.2.4 Parametric studies to identify critical parameters

The previous section revealed differences in the thermal structure functions and impedance profiles of the four samples under investigation. To better understand this behavior, the transient thermal response of the diodes is simulated and compared to the experimental data. Based on this, the steady state heat distribution is simulated and discussed.



Figure 5.5: Model used to simulate the transient and steady state thermal properties of the TO247packaged Ga_2O_3 devices of batch B2b. The bottom of the cold plate is defined as $25 \,^{\circ}C$ and a 10 µm thick volume at the chip surface serves as the heat source. For components with unknown thermal conductivity and heat capacity, the values are allowed to vary in a defined range when performing parametric studies.

Model properties and simulation procedure

The CT images of the devices serve as a basis for a precise model shown in Fig. 5.5. The die is $200 \,\mu\text{m}$ thick and exhibits the same rectangular shape and anode location as shown in Fig. 5.1b. Since the major part of the heat is created at the Schottky junction and the drift layer [94, 107], a 10 μ m thick volume with the same lateral dimensions as the anode is defined as the heat source. A heating power equal to the respective experimental value (about 7 W) is assigned to this volume. The heat sink is implemented by defining the bottom of the cold plate as 25 °C as suggested in [248]. The ambient temperature is set to 25 °C and surface radiation (e.g., from the surface of the mold mass) is taken into account, but is negligible.

The goal of parametric studies is to find material parameters that best describe the measured impedance profiles and structure functions. As visible from Fig. 5.5, the large number of different materials in the heat flow path complicates the calculations. Therefore, the thermal properties of well-known components (e.g., the Al heatsink and Cu lead frame) are defined as a constant value. Subsequent thermal simulations of the steady state heat distribution will show that a major part of the heat is transferred through the substrate rather than the bond wires or the mold mass. Variations in the thermal properties of these components can therefore

Table 5.2: Overview of model parameters used to study the transient thermal properties of the TO247packaged samples. The allowed ranges of the thermal conductivity λ_{th} and the thermal capacitance C_{th} specified for parametric studies are based on [29, 31, 87, 88, 231, 176, 218]. Adapted from [255].

| Layer | Туре | $\lambda_{\text{th}} (\text{W} \text{m}^{-1} \text{K}^{-1})$ | $C_{\text{th}} \left(\mathbf{J} \mathbf{kg}^{-1} \mathbf{K}^{-1} \right)$ |
|-------------------|------------|--|---|
| die | Ga_2O_3 | 10 - 35 | 400 - 750 |
| die attach layers | SAC solder | 20 - 905 | 150 - 290 |
| spacer | CuMo | 150 - 300 | 200 - 400 |
| lead frame | Cu | ~ 394 | ~ 385 |
| TIM | sheet | ~ 1.5 | ${\sim}800$ |
| cold plate | Al | ~ 237 | ~ 900 |
| bond wire | Au | ~ 300 | ~ 130 |
| encapsulant | mold | ~ 1 | $\sim \! 1000$ |

be neglected in comparison with the materials in the main heat flow path. For all other components, most importantly the Ga_2O_3 die, solder layers, and CuMo spacer, the thermal conductivity and heat capacity are allowed to vary within a defined range. The corresponding simulation parameters summarized in Tab. 5.2. The parametric study is performed with the T3Ster autocalibration feature in FloEFD. For each free parameter, at least five scenarios are generated in which the respective parameter is varied within the specified ranges according to a Latin hypercube method. By simulating the transient thermal response for all scenarios and iteratively reducing the parameter ranges, the respective best-fit values are calculated.

Parametric studies and steady state thermal simulations

Parametric studies are conducted for the three samples S1, S3, and S4. By performing the previously mentioned steps, the model parameters of all components within the junction-to-case heat flow path are optimized. The resulting simulated and measured thermal impedance profiles (Fig. 5.6a) and thermal structure functions (Fig. 5.6b) agree well when certain parameters in the model are slightly varied. In this respect, it was explained in Section 5.2.3 that the length of the plateau that corresponds to the $R_{\rm th,die}$ seems to vary, but it is difficult to distinguish from the die attach layer. Although a variation in thermal conductivity and heat capacity of the die attach is allowed in the parametric study, the best fit is obtained when the thermal conductivity and specific heat capacity of the Ga_2O_3 die are varied between $24 \text{ W m}^{-1} \text{ K}^{-1}$ and $500 \,\mathrm{J \, kg^{-1} \, K^{-1}}$ for S4, $28 \,\mathrm{W \, m^{-1} \, K^{-1}}$ and $615 \,\mathrm{J \, kg^{-1} \, K^{-1}}$ for S1, and $31 \,\mathrm{W \, m^{-1} \, K^{-1}}$ and $660 \,\mathrm{J \, kg^{-1} \, K^{-1}}$ for S3. The specific heat capacities agree with typical literature values of 490 to $600 \,\mathrm{J \, kg^{-1} \, K^{-1}}$ [87, 231]. Furthermore, the results suggest a variation of the thermal conductivity within the upper third of the known range of about 10 to $30 \text{ W m}^{-1} \text{ K}^{-1}$ [12, 88]. Previously experimentally determined values in the [001] direction (which is about 14° off of the [001]* direction of the devices investigated here, see Fig. 2.6) vary between about $15 \,\mathrm{W \,m^{-1} \,K^{-1}}$ [29] and $(21 \pm 2) \,\mathrm{W \,m^{-1} \,K^{-1}}$ [89].

When interpreting these results, it must be considered that experimental data are used as input for the calibration, which in turn are associated with measurement uncertainties. In addition, previously mentioned differences in chip thickness could affect the results. However, a smaller die thickness and volume should not only reduce the thermal resistance but also the thermal capacitance, provided that the thermal conductivity and specific heat capacity remain unchanged. In contrast to this, the die with the lowest thermal resistance (S3) also exhibits the highest thermal capacitance, and the die with the highest thermal resistance (S4) also exhibits the lowest thermal capacitance. Note that, as outlined in Section 2.3.2, several studies find that defects in the crystal structure, such as oxygen or gallium vacancies and line defects, significantly affect the thermal conductivity of Ga₂O₃ thin films [90–92]. Furthermore, such defects are known to affect the effective doping concentration and electrical properties [87, 102]. Considering that the samples S1–S4 also vary in their forward current–voltage profiles, this could indeed hint at a certain variation in the thermal conductivity of the die caused by a variation of material properties across the wafer — even though potentially to a lesser extent than suggested by the simulation and measurement due to the above-discussed influences. To obtain a more definite answer, it is suggested to combine, e.g., energy dispersive X-ray spectroscopy (EDX) to map different elements in the die with thermoreflectance or laser pulsed methods to determine the precise thermal conductivity. These experimental methods are, however, not available in this work.

The calibrated model enables the study of the steady state heat distribution in the Ga_2O_3 dice in more detail. A thermal conductivity of $28 \,\mathrm{W}\,\mathrm{m}^{-1}\,\mathrm{K}^{-1}$ is defined for the Ga_2O_3



Figure 5.6: Measured and simulated (a) thermal impedance profiles and (b) structure functions of the 200 μ m thin TO247-packaged Ga₂O₃ samples when all parts within the junction-to-case heat flow path are fully calibrated. The best match between measurement and simulation is observed when the thermal conductivity and heat capacity of the Ga₂O₃ dice are varied within the typically observed ranges. Adapted from [255].

die based on the best-fit parameter for sample S1, and the heating power is set to 7 W. Other material parameters correspond the average values summarized in Tab. 5.2. Steady state thermal simulations are performed for both the $2.2 \text{ mm} \times 3.8 \text{ mm}$ rectangular and the $2.2 \text{ mm} \times 2.2 \text{ mm}$ square die. Figure 5.7 displays the corresponding temperature distribution and heat flux through a cross section in the center of the dice. Regarding the rectangular die (top), the heat does not spread significantly into the additional volume of the die that is not electrically connected with bond wires. Therefore, the heat distribution and heat flux remain similar when reducing the die size to the effective square size with an only 300 µm wide rim around the anode. Consequently, an average junction temperature of about 62 °C and $R_{\text{th,die}}$ of 2.8 K/W is obtained in both scenarios. These values also agree with the experimental data in Fig. 5.2a and Fig. 5.3. Next, the chip thickness is increased to the standard value of 500 µm such as in [97]. This increases the $T_{\text{j,ave}}$ and $R_{\text{th,die}}$ to about 81 K and 6.3 K/W. This means that, according to the simulation, the thermal resistance of the die is reduced by about 55 % and the average junction temperature is reduced by about 23 % by substrate thinning to 200 µm in this setup.

Comparison and further steps

In a recent study [97], a 1 mm thick silver plate was bonded to both sides of a 500 µm thick and 4.6 mm × 4.6 mm large die with an anode size of $3 \text{ mm} \times 3 \text{ mm}$ by pressureless silver sintering. Subsequent transient thermal measurements indicated an $R_{\text{th,die}}$ ($R_{\text{th,jc}}$) of 0.8 K/W (1.43 K/W) when cooling the assembly from the cathode side. This corresponds to an areaspecific $R_{\text{th,die}}$ of 17 mm² K/W, and an area-specific $R_{\text{th,jc}}$ of 30 mm² K/W. Taking sample S3 with an estimated $R_{\text{th,die}}$ of 2.3 K/W as a reference, this suggests that an up to 34 % lower specific $R_{\text{th,die}}$ of 11 mm² K/W and an up to 33 % lower specific $R_{\text{th,jc}}$ of 20 mm² K/W is achieved here. Thus, substrate thinning seems to be more effective than sintering metal plates to both sides of the die and cooling it from the cathode side, but possibly still less effective than cooling the die from the junction side [94, 97]. Also, the effect of substrate thinning seems to be smaller than expected from the thermal simulations.



Figure 5.7: Simulated steady state temperature and heat flux profiles through a cross section of the actual rectangular die (top) and the hypothetical effective square die (bottom) of the 200 µm thin TO247-packaged sample. The heat does not spread significantly into the part of the die that is not electrically connected with bond wires, and the major part of the heat is dissipated vertically through the substrate. Consequently, the difference in junction temperature is negligible. Adapted from [255].

The main reason for this is that the comparison is inevitably limited by certain factors: First, the package structures vary considerably (sintering to a Ag plate in [97] versus soldering to a CuMo spacer on a TO247 lead frame here).

Second, also the anode and chip sizes differ substantially and, as discussed, the thermal properties of the substrate may vary to a certain degree. Both effects decisively influence the heat flow in the chip and package. Thus, while this section reveals the details of the heat dissipation in thinned Ga_2O_3 chips, a more comparable setup is required to judge the effectiveness of different cooling methods and compare the heat dissipation to mature technologies such as SiC.

5.3 Direct comparison of substrate thinning and junction-side cooling

5.3.1 Overview and investigated samples

In the previous section, new TO247-packaged 200 μ m thin Ga₂O₃ diodes were introduced. The combination of parametric studies and experimental data indicates a certain variation of material properties, but ultimately suggests that substrate thinning seems to be effective in reducing the junction temperature. Based on this, this section directly compares the heat dissipation in 600 μ m thick cathode-side cooled (CSC), 200 μ m thin CSC, and also 200 μ m thin junction-side cooled (JSC) Ga₂O₃ bare dice with identical lateral dimensions (ML-type) more precisely. Furthermore, a commercial SiC bare die (sample SiC-A) with a similar die and anode size is used as a benchmark. As shown in Fig. 5.8, all samples are assembled in the same way on custom AIN-DPC substrates via pressureless silver sintering and wire bonding according to the methods described in Section 3.2. This setup allows a fair and comparable investigation of the thermal properties, which is missing in literature. In addition, a 600 μ m thick LL-type diode is assembled in the same manner to briefly discuss the effect of increasing chip size on thermal resistance, although this will not be the focus for reasons explained below.

Table 5.3: Overview of Ga_2O_3 and SiC diodes with die thickness z_{die} mounted on DPC substrates in cathode-side cooled (CSC) and junction-side cooled (JSC) configuration. As shown in Section 5.2, the rectangular die shape of sample J200* does not affect the thermal properties significantly. The pressureless silver sintering process and determination of the die-attach thickness z_{att} after sintering was performed externally by a contractor. Adapted from [259].

| Die | Batch | Anode | Die | Zdie | Cooling | Sample | Print | Z _{att} |
|---|-------|----------------------------|----------------------------|------|---------|---------|-------------------------|------------------|
| material | no. | (mm ²) | (mm ²) | (µm) | method | abbrev. | area (mm ²) | (µm) |
| β –Ga ₂ O ₃ | B2c | 1.6 	imes 1.6 | 2.2×2.2 | 600 | CSC | C600 | 2.5×2.5 | 114 |
| β –Ga $_2O_3$ | B2b | 1.6 	imes 1.6 | 2.2×2.2 | 200 | CSC | C200 | 2.5×2.5 | 133 |
| β –Ga $_2O_3$ | B2b | 1.6×1.6 | 2.2×2.2 | 200 | JSC | J200 | 1.5×1.5 | 108 |
| β –Ga $_2O_3$ | B2b | 1.6×1.6 | 2.2×3.8 | 200 | JSC | J200* | 1.5×1.5 | 121 |
| β –Ga $_2O_3$ | B2c | 3.2×3.2 | 3.8 	imes 3.8 | 600 | CSC | C600LL | 4.3×4.3 | 114 |
| SiC | SiC-A | 1.65×1.65 | 1.92×1.92 | 377 | CSC | S377 | 2.0×2.0 | 123 |



Figure 5.8: Photograph of the CSC and JSC ML-size Ga₂O₃ samples and the CSC SiC reference diode after assembly on AlN-DPC substrates. The devices are covered in black paint for temperature measurements with an infrared (IR) camera. Adapted from [259].

Table 5.3 provides an overview of the corresponding devices under investigation. Samples J200 and J200^{*} are identical regarding their junction structure, but sample J200^{*} is diced out in a rectangular shape such as the one shown in Fig. 5.1b, whereas sample J200 is diced out in a standard square shape with a 300 μ m wide rim around the anode. As discussed in Section 5.2.4, the difference in shape does not influence the thermal resistance significantly. Nevertheless, the rectangular sample J200^{*} will not primarily be used to study the heat dissipation. It will, however, be used to assess the influence of varying differential on-resistances resulting from a variation of material properties across the wafer on the current and power rating. For CSC samples, the aperture opening of the sintering stencil can be larger than the actual chip size which ensures a proper thermal coupling to the substrate and heatsink. For JSC samples however, the aperture opening leaves a gap of 50 μ m between sinter paste and edge of the anode. This is to ensure full electrical functionality by avoiding an overflow of the sinter paste to the outer parts of the diode when the die is placed, causing undesired electric fields close to the edge termination region or even damage to the die.

Samples C600 and S377 are gold metallized on the cathode side and aluminum metallized on the anode side of the dice, whereas samples C200, J200, and J200* are gold metallized on both sides of the die. Therefore, to ensure a reliable bond and achieve a similar current capability, samples C600 and S377 are electrically contacted with 30 Al wires with a diameter of $40 \,\mu\text{m}$, whereas 30 gold wires with a diameter of $30 \,\mu\text{m}$ are used for samples C200, J200, and J200*. The effect of these slight differences in the assembly on the results derived in this chapter is negligible, especially considering that the major part of the heat is dissipated

through the substrate to the cold plate (see Section 5.2.4) [255]. The aluminum metallized sample C600LL was contacted with 30 pieces of $100 \,\mu\text{m}$ thick aluminum bond wires.

The devices were subjected to JEDEC JESD51-15 compliant measurements of the thermal impedance profiles and structure functions with two different TIMs (either thermal grease or a thin thermal sheet) according to the procedures explained in Section 3.5. In addition, the junction temperature is measured after heating for 50 s with different heating currents and corresponding heating powers applied to the diode. From this, conclusions about the power and current rating of the devices can be derived. The junction temperature is determined via the \sqrt{t} -method and also with an Optris PI 640i high-resolution infrared (IR) camera (system accuracy 2 K or 2 %, the higher value applies) placed directly above the respective die. For IR measurements the devices were covered in a thin layer of black paint as shown in Fig. 5.8 to raise the emissivity close to unity and enhance the measurement accuracy.

5.3.2 Differences in the thermal structure

Transient thermal measurements

For all samples, both the thermal impedance profiles and structure functions are measured and show consistent results. For the setup with grease as TIM, the junction-to-ambient thermal resistance $R_{\text{th,ja}}$ reduces from 11.7 K/W for C600 over 6.9 K/W for C200 to 4.6 K/W for J200. The lowest $R_{\text{th,ja}}$ of 3.6 K/W is achieved with the reference SiC diode S377. Due to the identical assembly method and experimental setup, the differences must be caused predominantly by the dice. Since the structure functions are more suitable to distinguish the different parts in the entire heat flow path, they are analyzed in more detail in the following.

As shown in Fig. 5.9 the curves with two different TIMs coincide up to a certain point as they result from the heat flow within the package, which helps to identify different parts of the structure functions. The regions from the respective separation points to about 1.3 K/W left of them originate from the AlN-DPC and the sintered silver layer. The rather flat plateau-like regions with varying length extending from there down to 0 K/W are associated with the thermal resistance and capacitance of the respective dies.

From this, the contribution of the various parts within the heat flow path to the total $R_{\text{th,ja}}$ can be divided into those coming from the chip, the package (i.e., the die-attach layer and the AlN substrate, abbreviated as pkg), and the cooling apparatus (i.e., the remaining path from the case to ambient, abbreviated as CA). Considering the results obtained from the measurements with thermal grease as TIM, the contribution of the cooling apparatus is similar for all samples (1.8 to 2.2 K/W), and the same applies for the contribution of the package (1.2 to 1.4 K/W). This is plausible since all devices are mounted in the same way and thus validates the measurements.

Furthermore, the thermal resistance of the die ($R_{\text{th,die}}$) is about 3.6 K/W in case of C200, which is similar to the value observed for the 200 µm thin sample S4 in Section 5.2.3. Compared to a value of 8.1 K/W for C600, this corresponds to a reduction of the $R_{\text{th,die}}$ of about 55 % by substrate thinning from 600 to 200 µm, which is in agreement with the considerations in Section 5.2.4. An additional 60 % reduction down to an $R_{\text{th,die}}$ of 1.4 K/W is achieved for J200, i.e., by cooling the diode from the junction side. However, this is still more than twice the $R_{\text{th,die}}$ of the commercial and nearly twice as thick SiC-A reference diode, which exhibits the lowest $R_{\text{th,die}}$ of about 0.5 to 0.6 K/W. Even in JSC assembly, the low thermal conductivity of Ga₂O₃ seems to limit the heat dissipation. The underlying reasons and methods to improve the effective thermal resistance down to the level of SiC will be studied in Section 5.3.4.



Figure 5.9: (a)–(d) The thermal structure functions with two TIMs show a decrease of the junction-toambient ($R_{th,ja}$) and junction-to-case ($R_{th,jc}$) thermal resistance in the order (a) C600, (b) C200, (c) J200, (b) S377. (e) The thermal resistance can be divided into the (e) absolute and (f) relative contributions of the die, the package (pkg) and the remaining thermal path from case to ambient (CA) to the total $R_{th,ja}$. Substrate thinning is highly effective in reducing the $R_{th,ja}$ and $R_{th,jc}$, but even in case of the 200 µm thin sample C200 the die still dominates the $R_{th,ja}$ with more than 50 %. Junction-side cooling is most effective, but $R_{th,jc}$ and $R_{th,ja}$ are still higher than for the CSC SiC sample S377. Adapted from [259].

The relative contributions of the different parts to the total $R_{\text{th,ja}}$ are displayed in Figs. 5.9e and 5.9f. For the C600 sample, the packaging and cooling apparatus together account for only about 30 % of the $R_{\text{th,ja}}$. For the thinned device C200, about half of the $R_{\text{th,ja}}$ is caused by package and cooling apparatus, and for the JSC device J200 the share increases to about 70 %. Thus, for thick Ga₂O₃ chips, improving the TIM or cold plate has comparatively little effect because the chip dominates the thermal resistance. For JSC assemblies, however, this issue is highly relevant. In case of the SiC-A sample, the die contributes with only about 15 % to the $R_{\text{th,ja}}$ and the heat dissipation is mostly limited by the package, TIM, and cold plate. In the following, the effect of these different thermal structures on the surface temperature of the devices is studied in more detail.

Thermal imaging of the steady state surface temperature

To study the surface temperatures of the devices, all samples are heated with a power of about 10 W by applying forward currents of 3.7 A (C600), 4.2 A (C200), 4.6 A (J200), and 7.2 A (S377). The resulting infrared images after heating for 50 s are shown in Fig. 5.10.

For CSC samples, the maximum temperature is located approximately in the middle of the anode, as it constitutes the major heat source. However, the high number of bond wires



Figure 5.10: Steady state surface temperature after heating with about 10 W. For CSC assemblies, the peak temperature is located approximately in the center of the die as the anode constitutes the main heat source, but it is underestimated due to shading by the bond wires. Among the Ga_2O_3 samples, the lowest peak temperature is achieved with JSC sample J200, although it is still higher than that of the CSC SiC sample S377. Furthermore, it occurs on the side of the chip instead of the center, which presumably results from a non-ideal coverage of the anode surface with die-attach material. Adapted from [259].

results in a shading of the anode area. Therefore, the surface temperature appears somewhat inhomogeneous and the measured maximum surface temperature is expected to be lower than the actual one. In addition, note that the junction temperature and anode surface temperature usually differ by up to a few Kelvin. For the JSC sample J200, the peak surface temperature (i.e., cathode temperature) is located at the side of the chip instead of the center, and there is a pronounced lateral temperature gradient from the upper left corner to the lower right corner. This suggests that an area along the outer edge of the anode is not properly coupled to the heatsink – presumably caused by the printing area leaving a 50 μ m edge between sinter paste and edge of the anode area and potential misalignments of anode and paste area. Thus, while the inner part is effectively cooled, the non-connected outer edge of the die can heat up. This hypothesis will be confirmed by simulations in Section 5.3.4.

Substrate thinning from 600 μ m to 200 μ m results in a decrease of the apparent maximum surface temperature from 125 °C (C600) to 72 °C (C200). For the JSC sample, the maximum surface temperature further decreases to 64 °C despite the previously discussed effect. With that it is closest to the commercial SiC chip, which heats up to only 52 °C. The surface temperature thus decreases in the same order as expected from the thermal resistance measurements. Note that the limited heat dissipation in the thick C600 Ga₂O₃ die also causes the bond wires to heat up more than in case of C200 or J200, although the same power is dissipated in the diode and an even lower current is conducted through them. Therefore, substrate thinning might also be advantageous from a bond reliability point of view, as the temperature swing and temperature drop per current unit across them is decreased. However, a larger number of samples would be required for sound reliability studies.

5.3.3 Impact on power and current rating

To examine the impact of the different assembly methods on the power and current rating, different heating currents and corresponding heating powers are applied to the diodes. The average junction temperature is determined with the \sqrt{t} -method after heating for 50 s. This way, the influence of differences between surface and junction temperature and of parasitic



Figure 5.11: Junction temperature measured with the \sqrt{t} -method when different heating powers are applied. (a) The $R_{\text{th,ja}}$ extracted from the slope of the linear fit agrees with the structure functions. The JSC Ga₂O₃ device J200 exhibits the highest power rating, although slightly lower than that of the CSC SiC reference diode S377. (b) The difference between J200 and S377 decreases slightly, when the results are related to the anode area. Adapted from [259].

shading effects due to the bonding wires is circumvented. Note that no safe operating area or maximum possible current is defined by the manufacturer of the diodes. To avoid potential harm to the dice, a maximum average junction temperature of about $140 \,^{\circ}\text{C}$ at the maximum DC forward current is not exceeded. Generally, the low increase in on-resistance (see Section 4.2.4 [253, 260]) in combination with recent reports of Ga₂O₃ transistor cells being operated successfully at $500 \,^{\circ}\text{C}$ [217] indicates that junction temperatures exceeding the chosen maximum are feasible in general.

Power Rating

In the following, the term "power rating" describes the maximum possible heating power that leads to a defined maximum allowed rise in average junction temperature. Figure 5.11a shows how the average junction temperature rises when the heating power $P_{\rm H}$ is increased.

For CSC assemblies, the junction temperatures are significantly higher than suggested by the IR images because, as discussed, the temperature maximum is located in the center of anode, which is shaded by bond wires. For the JSC assembly, the junction temperatures are only slightly higher because the peak temperatures occur at side of the chip resulting from a gap between anode edge and die attach. This agrees with the considerations in Section 5.3.2.

As expected from $\Delta T_{\rm j} = R_{\rm th,ja} \cdot P_{\rm H}$, a linear relationship is observed and the $R_{\rm th,ja}$ can be calculated from the corresponding linear fit. For the sake of clarity, the data for J200^{*} are not shown here. However, its fitted $R_{\rm th,ja}$ of $(4.20 \pm 0.03) \,\rm K/W$ agrees with a value of $(4.20 \pm 0.09) \,\rm K/W$ obtained for J200. This confirms that the additional inactive part of the die (see Section 5.2.1) has a negligible effect on the heat dissipation also in JSC assemblies.

For all samples, the $R_{\text{th,ja}}$ agrees with the respective value obtained from the transient thermal measurements (Fig. 5.9), although systematically being a few percent lower. The reason is that in transient thermal measurements only one value resulting from a temperature swing of about 90 to 110 K is determined. At high temperatures, the thermal conductivity of the materials in the heat flow path is reduced, which translates to an increased $R_{\text{th,ja}}$. In Fig. 5.11a, however, an average value also over lower temperatures is obtained.



Figure 5.12: (a) Junction temperature versus heating current density related to the anode area. Substrate thinning increases the power rating, but the highest current rating is achieved with junction-side cooling. (b) Forward current–voltage profiles at junction temperatures of 100 °C and 150 °C show that the difference between J200 and J200* in (a) results from a variation of the differential on-resistance. Adapted from [259].

Since the lateral dimensions of SiC and the Ga_2O_3 samples are similar but not identical, the power dissipation is related to the respective anode area in Fig. 5.11b. From this, an increase in power rating by 78 % through substrate thinning from 600 µm to 200 µm is obtained. For the JSC assembly, the power rating increases by another 53 % which, however, is still about 6 % below the SiC-A reference power rating.

Current rating

The previous considerations only reflect the pure thermal properties of the devices as they take into account only the power loss. In order for the electrical properties (i.e., in particular the differential on-resistance) to be taken into account as well, the rise in junction temperature is plotted as a function of the applied current density in Fig. 5.12a. In the following, the term "current rating" describes the maximum possible heating current that leads to a defined maximum allowed rise in average junction temperature.

In contrast to the previous measurements, a significant difference is now visible between samples J200 and J200^{*}. Since the $R_{\text{th,ja}}$ of these samples is almost identical, the reason can only lie in their electrical properties. Figure 5.12b shows the forward characteristics of samples J200, J200^{*} and S377 at junction temperatures of 100 °C and 150 °C. All samples exhibit a similar knee voltage V_0 of about 0.7 V. Due to a variation of the drift layer regarding its effective donor concentration and potentially also its thickness (see Section 4.2.3 [255]), the differential on-resistance of sample J200^{*} is almost half that of sample J200. The S377 reference diode shows the lowest differential on-resistance. Although the $R_{\text{th,ja}}$ is similar for these samples, this translates to substantial differences in the junction temperature. For example, at a current density of 3.7 A/mm^2 , a temperature rise of 58 K and 83 K is achieved for S377 and J200^{*}, respectively, while this current density cannot be applied to sample J200 because the maximum junction temperature would be exceeded. To obtain a more quantitative comparison, the rise in junction temperature can be expressed as

$$\Delta T_{\rm j} = R_{\rm th,ja} \cdot P_{\rm cond} = R_{\rm th,ja} \cdot \left(V_0 \, I_{\rm f} + r_0 \, {I_{\rm f}}^2 \right). \tag{5.2}$$

Table 5.4: The effective knee voltage V'_0 and specific differential on-resistance $r'_{0,sp}$ obtained by fitting the data in Fig. 5.12a with Eq. 5.2 agree well with respective curve-tracer measurements at 125 °C. The ratios γ_T , calculated from Eq. 5.3 and the $R_{th,ja}$ in Fig. 5.11b, show that for the JSC assembly the conduction losses would need to drop to only 6 % below the SiC level to reach the same junction temperature as the SiC reference. Adapted from [259].

| Sample | Fitted | Fitted $r'_{0,sp}$ | <i>r</i> ₀,sp at 125 °C | Needed | Needed P _{cond} drop |
|--------|---------------|-----------------------------------|--|-----------------------------|-------------------------------|
| name | V'_{0} (V) | $(\mathbf{m}\Omega\mathbf{cm}^2)$ | $(\mathbf{m}\Omega\mathbf{c}\mathbf{m}^2)$ | ratio $\gamma_{\mathbf{T}}$ | below SiC ref. |
| C600 | 0.43 ± 0.16 | 14.9 ± 1.4 | 11.7 | 0.34 | 66% |
| C200 | 0.44 ± 0.17 | 11.4 ± 4.0 | 8.5 | 0.61 | 39% |
| J200 | 0.74 ± 0.20 | 7.5 ± 0.7 | 7.8 | 0.94 | 6~% |
| J200* | 0.75 ± 0.07 | 4.1 ± 0.2 | 4.3 | 0.94 | 6~% |
| S377 | 0.42 ± 0.25 | 3.6 ± 0.1 | 2.8 | - | - |

The data in Fig. 5.12a can be fitted with this relationship when setting $R_{\text{th,ja}}$ to the respective value determined in Fig. 5.11 and defining V_0 and r_0 as free variables. Both parameters depend highly on the junction temperature. Therefore, this approach results in effective values V'_0 and $r'_{0,\text{sp}}$ averaged over the entire temperature and current range.

The results are summarized in Tab. 5.4. Taking into account the associated uncertainties, the knee voltages are similar for all samples. Furthermore, the averaged specific differential on-resistances increase in the order S377, J200*, J200, C200, and C600. These values agree with the differential on-resistances measured with a curve tracer at a junction temperature of 125 °C, which are also summarized in Tab. 5.4. The $r_{o,sp}$ of C200 and J200 are similar but almost twice as high as that of the also 200 µm thin sample J200* because of the previously discussed influences. Sample C600 exhibits the largest $r'_{o,sp}$ and $r_{o,sp}$. This is reasonable because a thicker substrate should exhibit a larger electrical resistance and demonstrates the benefits of substrate thinning also in terms of on-resistance. With the fitted lines in Fig. 5.12a as a reference, and when assuming a maximum permitted rise in junction temperature of 100 K, the current rating can be quantified: Starting from 1.41 A/mm² for C600 it increases by 54 % to 2.17 A/mm² for C200. With 3.13 A/mm² (4.05 A/mm²) an additional 44 % (87 %) increase is observed with junction-side cooling for sample J200 (J200*). The highest value of 4.83 A/mm² is obtained for the SiC-A reference sample S377.

Lowering the junction temperature by increasing the chip size

It was suggested in [223] that simply increasing the chip and corresponding die attach area may be effective in reducing the junction temperature of Ga_2O_3 diodes. Here, a short experimental examination of this hypothesis is given.

To this end, a 600 μ m thick LL-type Ga₂O₃ diode was assembled in CSC configuration on AlN-DPC substrates in the same way as the previous diodes. The sample will be referred to as C600LL in the following. It exhibits an about four times larger anode area of $3.2 \text{ mm} \times 3.2 \text{ mm}$ and an about 3 times larger chip size of $3.8 \text{ mm} \times 3.8 \text{ mm}$ than the ML-type diodes. Accordingly, a significant reduction in the thermal resistance is expected.

Figure 5.13a shows the rise in average junction temperature of the sample as determined by the \sqrt{t} -method. In addition, the previous results obtained with ML-size diodes C600 and J200 are shown. By increasing the die size, the average junction temperature is reduced significantly. Thus, the junction-to-ambient thermal resistance $R_{\text{th,ja}}$, calculated from a linear fit to the data, is reduced to 4.4 K/W, making the method almost as effective as junction-side cooling. Compared to sample C600, the power rating increases by a factor of about 2.6.



Figure 5.13: Rise in junction temperature T_j for a 600 µm thick LL-size diode compared to previous assemblies with ML-size diodes. (a) Increasing the die size of a 600 µm thick Ga₂O₃ diode by about a factor of three increases the absolute power rating by a factor of about 2.6. (b) Relating the increase in T_j to the actual anode size reveals a decrease in the maximum possible power density compared to the smaller C600 sample. Adapted from [259].

However, it must be taken into account that an increase in the size of the chip also means a reduction in the specific power dissipation. Figure 5.13b therefore compares the junction temperature in relation to the specific (i.e., with respect to the anode area) power dissipation. With a specific $R_{\text{th,ja}}$ of about 43.3 mm² K/W, the area-related cooling performance of sample C600LL is actually worse than that of the smaller ML-size diode. Therefore, increasing the die size is not in line with the general goal of enhancing the power density. In addition, increasing the die size will also increase the chip cost and inherently contradicts the potential cost advantage of Ga₂O₃. Therefore, the following discussion only focuses on methods that allow a reduction of the junction temperature without changing the chip size.

Discussion of experimental results

To be competitive, Ga_2O_3 diodes should exhibit an equivalent or lower rise in junction temperature than their SiC counterparts under the same operating conditions, i.e., the same forward current and die size. To investigate the implications for Ga_2O_3 diodes, the following discussion assumes an application in which conduction losses dominate. Accordingly, $\Delta T_{j,GaO}/\Delta T_{j,SiC} \leq 1$ should be satisfied at the same forward current. Using Eq. 5.2 and assuming the limit where the quotient is equal to one, this requirement can be expressed as

$$\frac{P_{\text{cond,GaO}}}{P_{\text{cond,SiC}}} \stackrel{!}{=} \frac{R_{\text{thJA,SiC}}}{R_{\text{thJA,GaO}}} := \gamma_{\text{T}},$$
(5.3)

where γ_T basically describes the maximum allowed conduction loss in Ga₂O₃ diodes with a given junction-to-ambient thermal resistance $R_{\text{th,ja}}$ that will lead to the same rise in junction temperature as in a SiC counterpart of identical chip size at the same forward current. The junction-to-ambient thermal resistance are already known for all devices from Fig. 5.11. Therefore, γ_T can directly be calculated for each sample. Since the active area constitutes the main heat source, and the on-resistance and thermal resistance depend heavily on it, specific values related to the respective anode area are used for the evaluation.

The results are summarized in Tab. 5.4. Due to the high thermal resistance of the $600 \,\mu m$ thick substrate, the conduction losses of the C600 sample would have to drop to one-third of the

conduction losses of the SiC diode in order to achieve the same junction temperature. For the thinned sample C200, conduction losses 39% below the SiC level would be required, and for the JSC samples J200 and J200^{*} the difference reduces to only 6% below the SiC reference. Therefore, JSC assemblies appear to be the most effective and require the least progress in lowering the on-resistance. Note that the results may be influenced to a certain extent by a variation in thermal properties, as discussed in Section 5.2. To validate the data and gain a deeper understanding of the differences in heat dissipation, the results are backed up with thermal simulations in the next section.

5.3.4 Simulative validation and methods to improve junction-side cooling

The previous section showed that even a slight difference in die size and thickness or a variation of the material properties can affect the results of the thermal measurements. Therefore, the experimental setup is replicated in FloEFD and thermal simulation data are compared to the measurements. This will also allow to study how the cooling methods can be improved.

Simulation model

The results in Section 5.2.4 indicate that only a minor part of the heat is dissipated through the bond wires. Therefore, they are omitted in the following. Fig. 5.14 shows the model for CSC and JSC assemblies, and Tab. 5.5 summarizes the corresponding simulation parameters. For CSC assemblies, the entire cathode is thermally coupled with the ceramic substrate and the cold plate via the die attach. For JSC assemblies, either the entire anode is covered with die attach, or a 50 μ m gap between anode edge and die attach is inserted. This value corresponds to the experimental aperture size of $1.5 \text{ mm} \times 1.5 \text{ mm}$ for the $1.6 \text{ mm} \times 1.6 \text{ mm}$ ML-type diode (see Tab. 5.3).

All devices (i.e., both the Ga_2O_3 and SiC model) are $2.2 \text{ mm} \times 2.2 \text{ mm}$ large and have a $1.6 \text{ mm} \times 1.6 \text{ mm}$ large anode. This means that the SiC model is slightly larger than the actual die which, however, makes the simulation more comparable. Based on the dimensions of the actual drift layer, the anode area is thickened to $10 \mu \text{m}$ and defined as a volume heat source. Even in the scenario with gap, the entire active area is considered to have a diode function. Therefore, the definition of the heat source is not changed, but the heat flow into the die-attach material and ceramic substrate is affected. The heat sink is implemented by defining the bottom of the cold plate as $25 \,^{\circ}\text{C}$.

The thickness of the sintered silver layer is set to $120 \,\mu\text{m}$, which is equal to the average experimental die-attach thickness (see Tab. 5.3). Given this rather high value, the effect of a reduced thickness of the die-attach material will also be studied.

| Part | Material | Th. Conductivity (W/mK) | Thickness (µm) |
|----------------------------|---------------------|-------------------------|----------------|
| Die | β –Ga $_2O_3$ | 20 | 50, 200, 600 |
| | SiC | 270 | 377 |
| Die attach | sintered Ag | 130 | 120 |
| Underfill material | unspecified | 0 - 100 | 40, 120 |
| Substrate top/bottom | Cu | 393 | 70 |
| Substrate ceramic | AlN | 170 | 300 |
| Thermal interface material | grease | 2 | 100 |
| Heat sink with channels | Al | 237 | 7000 |

Table 5.5: Simulation parameters used to study the heat dissipation in Ga_2O_3 and SiC diodes sinteredto AlN-DPC substrates as shown in Fig. 5.14. Adapted from [259].


Figure 5.14: Model of Ga_2O_3 and SiC diodes mounted on AlN-DPC substrates based on the actual experimental setup used to simulate the heat dissipation in (a) cathode-side cooled (CSC) and (b) junction-side cooled (JSC) assemblies. In case of JSC assemblies, either the whole anode area is covered with die-attach material, or a 50 µm large gap is inserted between anode edge and die attach. The effect of underfill materials on the junction temperature is studied for both scenarios. Adapted from [259].

JSC and CSC assemblies: key differences in heat dissipation

The first step in evaluating the effectiveness of the various cooling methods and exploring potential ways to improve them is to understand the difference in heat dissipation between CSC and JSC setups. To this end, 200 µm thin Ga₂O₃ dice in CSC assembly, ideal JSC assembly, and JSC assembly with 50 µm wide gap between anode edge and die attach area are modeled, all at heating power of 30 W. Figure 5.15 shows temperature distribution through a cross section in the center of the dice, as well as a view on the anode side of the die when hiding all other components. Parameters of interest are the average ($T_{j,ave}$) and local maximum ($T_{j,max}$) temperature of the junction area.

In the case of sample C200, the heat is created near the top surface and can be dissipated symmetrically over the whole area of the die, but it has to flow through the entire substrate with low thermal conductivity. Therefore, $T_{j,max}$ is located in the center at the top side of the anode surface, which causes a high temperature drop through the substrate, i.e., a high thermal resistance. Consequently, with 246 °C it exhibits the highest $T_{j,ave}$ and 28 K higher $T_{j,max}$ of all samples.

For the ideal JSC assembly without gap, the entire active area is thermally coupled to the die attach. Therefore, $T_{j,max}$ is again located in the center of the anode. Compared to sample C200, $T_{j,ave}$ decreases by about 97 K. With an only 9 K higher $T_{j,max}$ also the temperature difference across the active area is reduced. The peak temperature shifts slightly away from the anode surface and into the volume of the chip, which translates to a certain remaining thermal resistance of the die. Note that the exact simulation conditions such as the definition of the heat source will influence the exact location of the peak temperature, but not change this trend. Furthermore, the reduced cooling area results in a primarily lateral temperature gradient through the entire die, in contrast to the rather vertical gradient in case of C200.

When a gap is introduced in JSC configuration, a part of the anode that acts as the heat source is not directly cooled. This results in the maximum temperature shifting away from the center of the anode area towards its edge. Consequently, $T_{j,ave}$ increases by 16 K. With a plus of 35 K, the increase in $T_{j,max}$ is even more pronounced. The temperature difference across anode is now similar to C200, but $T_{j,max}$ is still 81 K lower than in case of sample C200.

Chapter 5 Thermal properties and methods to improve the heat dissipation



Figure 5.15: Simulated cross-sectional surface temperature (view on the anode when hiding all other parts) of the 200 µm thin Ga₂O₃ diode at a heating power of 30 W. Assembling the die in CSC configuration (C200) leads to the highest average $(T_{j,max})$ and maximum $(T_{j,ave})$ junction temperature. In JSC assembly, the heat dissipation is enhanced by shifting $T_{j,max}$ closer to the die attach, although it moves slightly towards the volume of the chip. A gap in JSC assembly shifts the $T_{j,max}$ to the edge of the anode and increases $T_{j,ave}$ and $T_{j,max}$ significantly. Adapted from [259].

In summary, junction-side cooling shifts the main heat source and $T_{j,max}$ closer to the die attach layer through which the heat is dissipated, which reduces the thermal resistance of the die significantly and makes JSC highly effective for Ga₂O₃. The fact that even small gaps in JSC setups lead to an enhanced temperature gradient across the die and a significant rise in $T_{j,max}$ but a less severe increase in $T_{j,ave}$ has consequences also for the measurement procedures: Using the forward voltage as a sensor yields the junction temperature averaged over the entire active area [198]. Therefore, the high local peak temperatures may not be detected, posing a potential risk to the chip. For this reason, average junction temperatures of about 140 °C as determined with the \sqrt{t} -method were not exceeded in the measurements presented here.

Simulated power rating

With the improved understanding of the heat dissipation, the simulation is repeated for different heating powers, and the average junction temperature $T_{j,ave}$ is determined. As shown in Fig. 5.16a, the $R_{th,ja}$ is again calculated from a linear fit to the data and summarized in Tab. 5.6 for each setup.

Bearing in mind that some deviations even between the experimental setups are expected (such as variations in thickness and thermal conductivity of dice, the sintered silver layer, or the thickness of the thermal grease), the simulated rise in average junction temperature $\Delta T_{j,ave}$ and the $R_{th,ja}$ are in agreement with the measurements in Fig. 5.11 in case of the cathode-side cooled samples C600 and C200. This verifies the experimental and simulative procedures.



Figure 5.16: The simulated rise in average junction temperature when different heating powers are applied agrees with the experimental results. (a) The 200 µm thin ML-size JSC diode (J200) still shows a higher thermal resistance than its CSC SiC counterpart (S377), but a further decrease in T_j can be achieved by optimizing the die-attach thickness. Even a hypothetical 50 µm thin CSC diode (C50) cannot achieve the JSC limit. (b) Increasing the lateral dimensions of the die to LL-size reduces T_j significantly for a 600 µm thick diode (C600LL). When the substrate is additionally thinned to 200 µm (C200LL), T_j is similar to the SiC diode S377. (b) However, increasing the chip size reduces the maximum possible power density. Parts adapted from [259].

In the case of sample J200, the simulated $\Delta T_{j,ave}$ is significantly lower than the measured one, with about 86 K compared to 112 K in Fig. 5.11, both at a heating power of 27 W. This difference is mainly caused by the impact of the gap between die attach and anode edge, as examined in Fig. 5.15.

In case of the CSC assembled SiC reference sample S377, the simulated $\Delta T_{j,ave}$ is also lower than the measured one (about 68 K and 83 K, respectively, both at a heating power of 23.4 W). A major reason is that the anode sizes of the SiC and Ga₂O₃ dice are similar, but the lateral size of the modeled SiC die is larger than that of the actual one (see Tab. 5.3). As shown in Fig. 5.17a, this enables an effective heat spreading as a result of the high thermal conductivity of SiC and reduces the average and maximum junction temperature. Besides, variations in the die thickness of $\pm 10\%$ according to the datasheet [235] can affect its thermal resistance.

When the SiC die is cooled ideally from the junction side (see also Fig. 5.17a), the temperature difference across the active area is reduced from 18.5 K to 7.5 K, similar to the setup with Ga₂O₃. However, $T_{j,ave}$ increases by about 7.5 K. In this setup, the effective heat spreading

Table 5.6: Simulation results for various Ga_2O_3 samples: The thermal resistances and drop of the conduction losses below the SiC level required to achieve the same T_j as an identical SiC counterpart according to Eq. 5.3 agree with the experimental results. Adapted from [259].

| Sample | Sim. R _{th,ja} | Sim. spec. $R_{\text{th,ja}}$ | Sim. needed | Sim. needed drop of |
|--------|-------------------------|--|-----------------------------|-----------------------------------|
| name | (K/W) | $(\mathbf{mm}^2\mathbf{K}/\mathbf{W})$ | ratio γ_{T} | P_{cond} below SiC level |
| C600 | 11.14 | 27.35 | 0.26 | 74% |
| C200 | 6.46 | 15.85 | 0.45 | 55% |
| C50 | 3.20 | 7.86 | 0.71 | 29% |
| J200 | 2.90 | 7.11 | 0.91 | 9~% |

in SiC and the associated large cooling area seem to outweigh the advantage of a shortened thermal path when cooling directly from the junction side.

This is a decisive difference to Ga_2O_3 , where both the experimental and simulated data confirm that due to the low thermal conductivity of Ga_2O_3 , JSC assemblies can yield lower junction temperatures despite the smaller cooling area — if a sufficiently large portion of the anode area is thermally coupled with the die-attach material. This is further illustrated by simulating the heat dissipation in a hypothetical 50 µm thin CSC die. As shown in Fig. 5.16a, even this device exhibits a higher $T_{j,ave}$ than its 200 µm thick JSC counterpart (both without and with 50 µm gap). This shows that different packaging methods for Ga_2O_3 and SiC devices will be required in order to achieve the maximum possible device performance.

The $\gamma_{\rm T}$ factors, i.e., the necessary reduction in $P_{\rm cond}$ below the SiC level to achieve the same $T_{\rm j}$ at the same forward current, can be calculated from the simulated $R_{\rm th,ja}$ by using Eq. 5.3. The results are summarized in Tab. 5.6. Due to the previously discussed influences, minor differences between the simulated and the experimentally determined $\gamma_{\rm T}$ factors are present, but they show the same trend. While even for the 50 µm thin CSC die the conduction losses would have to drop to 29 % below the SiC level, a difference of only 9 % is necessary for the 200 µm thick JSC device.

Figures 5.16b and 5.16c show the influence of increasing die size on the junction temperature. The simulations confirm the experimentally observed decrease in junction temperature for the 600 μ m thick LL-size diode close to that of the JSC ML-size sample J200, although at the cost of a significantly reduced specific power handling capability. If the LL-size sample is additionally thinned to 200 μ m, the junction temperatures are similar to the CSC SiC sample S377. For the reasons already mentioned, the effect of increasing chip size is not examined in more detail. Nevertheless, the considerations are helpful in interpreting the results of the buck converter tests in Chapter 6.

Influence of die attach and substrate thickness in JSC

At the first glace, the simulative and experimental results suggest that even with JSC assemblies, Ga_2O_3 devices can still not achieve junction temperatures as low as those of CSC assembled SiC counterparts. However, as shown in Fig. 5.17b, the thickness of the die attach layer affects the results.

In case of the Ga₂O₃ sample J200, a decrease of the die-attach thickness from the experimental value of 120 µm to a hypothetical value of 40 µm decreases $T_{j,ave}$ by about 9 K. In case of the CSC diode S377, however, an identical decrease in die-attach thickness does not affect the junction temperature substantially. Thus, by optimizing the thickness of the sintered silver layer, the Ga₂O₃ sample can achieve the approximately same $T_{j,ave}$, yet while still exhibiting an about 6 K higher local $T_{j,max}$ than the SiC reference.

In Fig. 5.17c, the influence of the thickness of the die on the thermal resistance in JSC configuration is studied. Increasing the thickness from $200 \,\mu\text{m}$ to $600 \,\mu\text{m}$ does not affect the junction temperature significantly. From a purely thermal point of view, time-consuming substrate thinning processes do therefore not seem necessary if a JSC assembly is chosen. However, considering the electrical properties, substrate thinning is still recommended to lower the on-resistance and reduce the conduction losses and associated heating power. This aspect will be studied in more detail in Section 5.3.5.

So far, it has been found that optimizing the die-attach thickness in JSC assemblies can reduce the junction temperature, but also that gaps between anode edge and die attach increase the



Figure 5.17: Simulated junction temperature T_j of different samples at a heating power of 30 W.
(a) When the lateral dimensions of SiC diode model are reduced to the actual chip size, the limited heat spreading results in an increase in T_j. Cooling the die from the junction side leads to a further increase in T_j for the same reason. This contrasts with the heat dissipation in Ga₂O₃ dies. (b) The average T_j of JSC assembled Ga₂O₃ dice can be reduced to the CSC SiC level by decreasing the thickness of the sintered silver layer. (c) When JSC assembled, the thickness of the Ga₂O₃ substrate has a negligible effect on T_j. Adapted from [259].

peak temperature and shift it towards the edge of the anode that is not covered with die-attach material. Hereinafter, it is therefore studied if underfill materials can alleviate this issue.

Improving JSC with underfill materials

The effect of underfill materials on the junction temperatures is examined by inserting a part with a thickness identical to that of the die attach layer and filling the entire volume between the ceramic substrate and the surface of the die, as shown in Fig. 5.14b. The thermal conductivity of the underfill material is increased from $0.01 \text{ W m}^{-1} \text{ K}^{-1}$ to $100 \text{ W m}^{-1} \text{ K}^{-1}$ and a heating power of 30 W is applied. The analysis is performed for four different JSC setups: without and with gap as well as with $120 \,\mu\text{m}$ thick and $40 \,\mu\text{m}$ thin underfill. The results are shown in Fig. 5.18.

In the scenario with 120 µm thick underfill and ideal JSC assembly (Fig. 5.18a), no effect on T_j is observed for thermal conductivities below $1 \text{ W m}^{-1} \text{ K}^{-1}$. To increase $T_{j,\text{ave}}$ below the SiC level, a minimum thermal conductivity of $5 \text{ W m}^{-1} \text{ K}^{-1}$ is required. The change in $T_{j,\text{ave}}$ is caused by a decrease of the minimum junction temperature $T_{j,\min}$ while $T_{j,\max}$ remains almost the same. The reason is that $T_{j,\max}$ is located in the center of the anode area (see Fig. 5.15) far away from the underfill, which limits its influence.

When the gap is introduced (Fig. 5.18b), both $T_{j,ave}$ and $T_{j,max}$ increase significantly. An increase in the thermal conductivity of the underfill now reduces both $T_{j,max}$ and $T_{j,min}$. This is because $T_{j,max}$ now occurs at the anode edge (see Fig. 5.15) and thus the underfill has a direct effect on it. However, a thermal conductivity of the underfill of $10 \text{ W m}^{-1} \text{ K}^{-1}$ is required to decrease $T_{j,ave}$ down to the SiC level. Note that underfill materials usually achieve a thermal conductivity below $3 \text{ W m}^{-1} \text{ K}^{-1}$. Intensive research into suitable underfills is therefore critical in this scenario. Yet, even a material with a thermal conductivity $3 \text{ W m}^{-1} \text{ K}^{-1}$ can reduce $T_{j,ave}$ by 16 K and $T_{j,max}$ even more significantly by 22 K.

Next, the thickness of the die-attach material and underfill is reduced to $40 \,\mu m$ (Fig. 5.18d). This enhances the influence of the underfill material. A thermal conductivity of about



Figure 5.18: The junction temperature T_j of JSC assembled Ga₂O₃ diodes can be decreased with underfill materials. The minimum thermal conductivity of the underfill material required to reach the same $T_{j,ave}$ as the SiC counterpart depends on the die-attach thickness, as shown exemplarily in (a) and (b) for the 120 µm die attach layer without and with gap between anode edge and die attach. In (c) and (d) the impact of the underfill material is further increased by reducing the die-attach thickness to 40 µm, both without and with gap. Adapted from [259].

 $1.6 \,\mathrm{W \,m^{-1} \, K^{-1}}$ is now sufficient to reduce $T_{j,ave}$ to the SiC level, demonstrating that also standard underfills can be advantageous in JSC assemblies with Ga₂O₃.

When the gap is removed (ideal JSC assembly, Fig. 5.18c), the $T_{j,ave}$ is already on SiC level even without underfill. Similar to the case in Fig. 5.18a only $T_{j,min}$ decreases with increasing thermal conductivity of the underfill, while the $T_{j,max}$ remains unchanged for the above-discussed reasons. With a thermal conductivity of only $0.3 \text{ W m}^{-1} \text{ K}^{-1}$, $T_{j,ave}$ can be reduced to below the SiC level. In summary, thinner die attach layers not only increase the direct heat dissipation through it, but also make underfill materials more effective. They are thus recommended especially if gaps between anode edge and die attach area cannot be avoided when assembling the dice.

5.3.5 How conduction losses can affect the assembly strategy

Apart from the pure thermal resistance, the junction temperature and safe operating range of a power semiconductor device are also determined by its actual power loss at the respective operating point. If the power loss of the device is low enough, a higher thermal resistance can be accepted. To further discuss this aspect for Ga_2O_3 diodes, an application where conduction losses are dominant is assumed, and a simplified calculative approach that will be justified in the following is pursued.

Table 5.7: Material parameters used to calculate the theoretical on-resistance of Ga_2O_3 diodes in arealistic-optimistic, a moderate, and a pessimistic scenario by using Eq. 5.5. A breakdownvoltage of 1200 V is assumed. Adapted from [259].

| 8 | I I I I I I I I I I I I I I I I I I I | r 1. | | |
|-----------------|---|--------------------|--------------------|--------------------|
| | | optimistic | moderate | pessimistic |
| Substrate | $\mu_{\rm e,sub} \ ({\rm cm}^2/{\rm Vs})$ | 60 | 30 | 30 |
| | $N_{\mathrm{D,sub}}~(\mathrm{cm}^{-3})$ | 1×10^{19} | 1×10^{19} | 1×10^{19} |
| | \Rightarrow Resistivity (m Ω cm) | 10.4 | 20.8 | 20.8 |
| Epitaxial layer | $\mu_{\rm e,epi} \ ({\rm cm^2/Vs})$ | 200 | 100 | 100 |
| | $E_{\rm br}$ (MV/cm) | 8 | 5 | 4 |
| | \Rightarrow Specific resistance (m Ω cm ²) | 0.06 | 0.52 | 1.02 |

According to Eq. 5.2, the forward conduction losses P_{cond} , and with that also the rise in junction temperature ΔT_j are determined by the knee voltage V_0 and the differential onresistance r_0 . In this equation, the contributions of auxiliary parts such as the bond wires but also the Schottky and ohmic contact to the on-resistance are neglected as they accout for less than 1 %, i.e., substrate and drift layer strongly dominate the resistance (see also Section 4.2), and as they do not primarily result from the pure semiconductor material. To facilitate the comparison of different semiconductor technologies regarding their performance limits, it is furthermore common to consider only the differential on-resistance. Before adopting this method, a brief discussion will be given on whether this simplification is reasonable here and what impact it may have on the results. To this end, the conduction losses P_{cond} in Eq. 5.2 are expressed with Eq. 4.7 as

$$P_{\text{cond}} = V_0(T_j) \cdot I_f + \left[r_o(T_{\text{ref}}) \cdot \left(\frac{T_j}{T_{\text{ref}}}\right)^{\alpha_{\text{ref}}} \right] \cdot I_f^2,$$
(5.4)

where α_{ref} describes the increase in differential on-resistance r_o with rising junction temperature T_j related to a reference temperature T_{ref} . The first simplification is to neglect the term V_0I_f in Eq. 5.4. This is reasonable for the following reasons: First, the profiles in Fig. 5.12b and the indications in Tab. 5.4 (bearing in mind their uncertainties) show that the knee voltage is similar for all SiC and Ga₂O₃ diodes under investigation. Second, the term V_0I_f increases linearly with I_f and V_0 decreases with rising junction temperature, whereas the second summand is proportional to I_f^2 and increases for sufficiently high forward currents. Third, as summarized in Sections 2.3.3 and 2.3.4, V_0 is largely determined by the Schottky metal, the quality and type of junction (e.g., pure MS- versus hetero-junction), and the substrate orientation, and can vary almost by a factor of three. This influence is neglected by omitting the term V_0I_f . The second simplification is to assume that the rise in on-resistance with rising junction temperature is identical for Ga₂O₃ and SiC, i.e., $\alpha_{GaO} = \alpha_{SiC}$. As shown in Section 4.2.4, however, α_{GaO} is significantly lower than α_{SiC} , which means that this assumption is rather pessimistic for Ga₂O₃.

With these simplifications, the quotient $P_{\text{cond,GaO}}/P_{\text{cond,SiC}}$ in Eq. 5.3 can be replaced with $r_{\text{o,sp,GaO}}/r_{\text{o,sp,SiC}}$. The remaining theoretical specific differential on-resistance $r_{\text{o,sp}}$ can be calculated from the sum of the resistance of the substrate and the epitaxial layer by combining Eqs. 2.3 and 2.5, which gives

$$r_{\rm o,sp} = r_{\rm sub,sp} + r_{\rm epi,sp} = \frac{z_{\rm sub}}{e\mu_{\rm e,sub}N_{\rm D,sub}} + \frac{4V_{\rm br}^2}{\epsilon_0\epsilon_s\mu_{\rm e,epi}E_{\rm br}^3},$$
(5.5)



Figure 5.19: Specific differential on-resistance $r_{o,sp}$ versus substrate thickness for a Ga₂O₃ Schottky diode in the optimistic, moderate, and pessimistic scenarios defined in Tab. 5.7. The symbols indicate the previously experimentally and simulatively determined target $r_{o,sp}$ required to reach a rise in junction temperature identical to that of the SiC reference diode at the same forward current. With a 600 µm thick CSC diode, the required $r_{o,sp}$ cannot be achieved even in the optimistic case. With a 200 µm thin JSC diode, however, it can be achieved even in the pessimistic case. Adapted from [259].

with z_{sub} , $\mu_{e,sub}$, and $N_{D,sub}$ the thickness, electron mobility and effective donor concentration of the substrate, $\mu_{e,epi}$ the electron mobility of the drift layer, ϵ_0 and ϵ_s the vacuum permittivity and relative dielectric constant, and V_{br} and E_{br} the breakdown voltage and field. The corresponding specific differential on-resistance is calculated for Ga₂O₃ diodes assuming a V_{br} of 1200 V for a 650 V rated diode based on the measurements with batch B3 samples in Fig. 4.5. Electron mobilities and breakdown fields for β –Ga₂O₃ are taken from the experimentally achieved and theoretically derived ones summarized in [216] and [18].

Based on this, three scenarios summarized in Tab. 5.7 with their corresponding $r_{0,sp}(z_{sub})$ shown in Fig. 5.19 are considered: The optimistic but still realistic scenario exhibits the highest electron mobilities and the maximum theoretical breakdown field, leading to the lowest substrate resistivity and specific resistance of the drift layer. In the moderate scenario, the electron mobilities and the breakdown field are reduced to values that have been achieved and even exceeded in different experiments with small-area diodes already [18, 216]. The pessimistic scenario differs from the moderate one only in the fact that the breakdown field is reduced by 1 MV/cm, i.e., to half of the theoretical value. In addition to these theoretical on-resistance limits, Fig. 5.19 also shows the maximum allowed on-resistance for the investigated Ga₂O₃ diodes, which leads to the same rise in junction temperature as for the SiC reference sample S377 at the same forward current. Following the above-discussed simplifications, it is approximated by

$$r_{\text{o,sp,GaO}} \stackrel{!}{=} r_{\text{o,sp,SiC}} \cdot \gamma_{\text{T}}, \qquad (5.6)$$

where both the simulated (Tab. 5.6) and experimentally determined (Tab. 5.4) ratios $\gamma_{\rm T}$ are considered. Based on curve tracer measurements at 25 °C, an $r_{\rm o,sp,SiC}$ of $1.55 \,{\rm m\Omega \, cm^2}$ is used for the calculation.

Figure 5.19 shows that in case of the CSC and $600 \,\mu\text{m}$ thick sample C600, the required $r_{o,sp}$ cannot be achieved even in the optimistic scenario. It can be achieved in the optimistic case,

however, when the CSC substrate is thinned to 200 μ m (sample C200). In the moderate scenario, sample C200 can achieve the required $r_{0,sp}$ according to the measurements, but not according to the simulation data. When the 200 μ m thin substrate is cooled from the junction-side without gap (sample J200), the required $r_{0,sp}$ can be achieved in the optimistic as well as the moderate scenario, and by a very narrow margin also in the pessimistic one. Note that Fig. 5.17c suggests that substrate thickness does not affect the junction temperature significantly in JSC assemblies, provided that the same power is dissipated. However, Fig. 5.19 demonstrates that thinning JSC samples to at least 400 μ m is recommended to ensure a sufficiently low $r_{0,sp}$ in the moderate scenario, and further thinning to at least 200 μ m is recommended in the pessimistic scenario.

The fact that the only difference between the moderate and pessimistic scenario is a difference of 1 MV/cm in E_{br} illustrates that it is crucial to fully exploit this parameter in Ga₂O₃ devices, especially with respect to the medium electron mobility. Therefore, field management methods such as edge terminations or trench junctions and, with regard to potential commercial devices, the safety margin between rated reverse voltage and actual breakdown voltage will affect the mounting method. Here, a fairly large, i.e., rather pessimistic safety margin was chosen.

As summarized in Section 2.3.4 first small-area β -Ga₂O₃ diodes recently achieved record $V_{br}^2/r_{o,sp}$ figures of merit of 6.4 kV/3.4 m Ω cm² for a diode with deep SiO₂ trenches [22], and 8.3 kV/5.2 m Ω cm² for a n-Ga₂O₃/p-NiO heterojunction diode [23]. Both on-resistances fall below the 1D unipolar on-resistance limit of SiC and translate to a ratio γ_T of about 0.3. Note that these results were achieved with test structures with lateral dimensions in the range of some hundreds of µm and thus still need to be demonstrated for chips scaled up to sizes of some mm².

Nevertheless, when comparing these results to the γ_T values in Tabs. 5.4 and 5.6 it thus seems realistic that the on-resistances required to achieve a rise in junction temperature similar to that of commercial SiC diodes may be achieved also for large-area devices that are more relevant for power electronics applications, such as 650 V or 1200 V class Ga₂O₃ diodes. Therefore, from a purely technological point of view (i.e., not considering development effort and cost), it may be preferred to pursue JSC assemblies in the early Ga₂O₃ development phase, while for future generation diodes with thinned substrate and improved on-resistance simple CSC assemblies may be sufficient depending on the application — especially considering that Ga₂O₃ devices could potentially be operated at much higher temperatures than their SiC or GaN counterparts (see Section 4.2.4).

5.4 Chapter conclusion

Chapter 5 deals with the thermal properties of Ga_2O_3 diodes. The main results can be summarized as follows:

- 1. Large-area (001) oriented β -Ga₂O₃ Schottky diodes with 200 µm thin native substrates have been introduced for the first time and shown to reduce the junction temperature significantly compared to previous standard Ga₂O₃ diodes.
- 2. Not only the electrical properties but also the thermal properties have been found to vary between TO247 packaged devices of the same type. Combined X-ray analysis and parametric studies suggest that this may partly be due to a variation in the material properties of the Ga₂O₃ wafer.

- 3. To analyze strategies to reduce the junction temperature, a direct experimental and simulative comparison of the heat dissipation in Ga₂O₃ and SiC bare dice assembled in different ways is performed:
 - a) The heat dissipation in Ga_2O_3 differs significantly from that in SiC. For Ga_2O_3 , junction-side cooling is found to be more effective than substrate thinning.
 - b) Even small gaps between die attach layer and edge of the anode in junction-side cooled assemblies increase the peak and average junction temperature significantly. Underfill materials can alleviate this issue effectively.
 - c) Increasing the die size reduces the junction temperature, but also limits the maximum possible power density.
 - d) The potentially low on-resistances of Ga₂O₃ devices can enable a rise in junction temperature similar to that of SiC counterparts despite the low thermal conductivity, if a suitable assembly method is chosen.

The insight gained into the heat dissipation, including recommendations for different mounting techniques, is essential to design first power converters with Ga_2O_3 devices such as in the following Chapter 6.

Chapter 6

Application of gallium oxide power diodes in a buck converter

6.1 Chapter overview

The previous investigations of the basic current–voltage profiles and thermal properties of Ga_2O_3 diodes in Chapters 4 and 5 as well as novel current sensors in Section 3.4 lay the groundwork to examine the newest Ga_2O_3 Schottky diodes regarding their switching properties in an application-oriented manner. As such, this chapter addresses the lack of studies on the switching application of Ga_2O_3 diodes in the literature, which is pointed out in Section 2.3.4. This is particularly interesting because several diodes already exhibited atypical forward and reverse current-voltage profiles or signs of electron trapping (see Chapters 4 and 5). Novel large-area Ga_2O_3 power diodes and a low-inductive PCB layout enable voltage and current slew rates significantly exceeding those in first reports and allow new insights into the current development status, including the first-ever demonstration of Ga_2O_3 diodes in a 400 V buck converter.

Section 6.2 first introduces the experimental setup in more detail. Section 6.3 studies TO247 packaged 600 µm thick batch B3 diodes of sizes ML, L, and LL. First, the static electric properties that are relevant for buck converter application are briefly discussed. Based on this, the switching properties are examined by means of double pulse tests under varying conditions. This enables the final interpretation of efficiency measurements during buck converter application. Section 6.4 studies new 200 µm thin batch B4 diodes of sizes ML and LL assembled on ceramic substrates but mounted on the same buck converter PCB. Relevant static and dynamic electric properties are first briefly discussed. Then, efficiency and temperature measurements during buck converter operation are analyzed and also compared to the previous results in Chapter 5. Section 6.5 summarizes the main results. Parts of this chapter have been published in [260] by the author during the course of this thesis.

6.2 Experimental setup and investigated samples

A buck converter topology, the basics of which are explained in Section 2.2.1, is chosen to study the switching behavior of Ga_2O_3 diodes. Figure 6.1a shows the detailed measurement circuit schematics and Fig. 6.1b shows a photograph of the corresponding setup. Since the diode is placed on the side of low potential, non-isolated high-bandwidth probes can be used to capture the waveforms. Here, a 400 MHz passive probe (PMK PHV1000) is used to measure the diode voltage. The IPM sensor is used to resolve single fast switching transients with high precision and the M-shunt is used to capture the diode current during buck converter



Figure 6.1: Experimental setup used for double pulse tests and buck converter operation: (a) Circuit schematics and (b) photograph with TO-packaged devices (left) and diode mounted on an AlN-DPC substrate (right). During buck converter operation, the M-shunt is used to measure the diode current. For double pulse tests, the inductive load is connected in parallel to the diode, C_{out} and the electronic load are removed, and the IPM sensor is used to capture single switching transients. Adapted from [260].

operation as it offers enhanced heat dissipation, DC capability and lower insertion inductance than coaxial shunt, as thoroughly investigated in Section 3.4. The drain–source voltage of the high-side transistor is measured with a 120 MHz differential probe (LeCroy HVD3106A) and the drain current is measured with a 30 MHz Rogowski probe (CWTUM 015/B from PEM). Note that the low bandwidth of the Rogowski probe leads to underestimated current slopes and a distorted representation of the oscillations as discussed in Section 3.4.

The circuit is PCB-based with 70 μ m thick copper tracks forming a vertical commutation loop. A 650 V/107 m Ω TO274-4 packaged SiC MOSFET (Infineon IMZA65R107-M1HXKSA1 with an output charge of about 53 nC at 400 V) is used as active switch. The additional Kelvin source pin decreases switching losses and ringing by reducing the coupling between the gate and power loop [224]. By using an isolated Skylabs SI8271 gate driver and varying external gate resistors, the transistor is switched between gate–source voltages of 0 V (off-state) and 18 V (on-state).

For buck converter operation, the diode and transistor are mounted on the same heat sink (Ohmite PV-T22-38E) with an identical electrically insulating thermally conductive pad, and cooled with air using a fan (Delta Electronics AFC0612DB). Note that a certain mutual thermal influence is possible due to the common heat sink. An electronic load (EA ELR 9750-66) controls the output current while the output *LC* filter is formed by a 580 μ H air coil and a 7.5 μ F foil capacitor. However, the input capacitance of the electronic load (>100 μ F) dominates the capacitive part of the output *LC* filter. To avoid associated high inrush currents, the start-up is performed by steadily increasing the DC link voltage from 0 V to its final value, thus charging the output capacitance in a controlled manner. The use of an air coil circumvents saturation effects and keeps the thermal mass low, thus reducing the time to reach a steady state during efficiency measurements with a Zimmer LMG500 power analyzer.

During double pulse tests, a single heating resistor is attached to the diode lead frame instead of using the common heat sink. This allows to heat only the diode without affecting the MOSFET temperature. Furthermore, a $111 \,\mu\text{H}$ coil is connected in parallel to the diode instead of the output *LC* filter and electronic load. As explained in Section 3.4.6, the point of interest is the second turn-on of the MOSFET when the freewheeling current flowing through the diode is abruptly disrupted and the diode turns off.

Note that the buck converter is not designed to achieve maximum efficiency, but to effectively study how the various diodes affect the buck converter switching waveforms and performance. This is achieved by keeping the same MOSFET but swapping the diodes. Diodes under investigation here are the 600 μ m thick TO247 packaged Ga₂O₃ batch B3 samples with field plates and N-ion implanted guard rings of size ML, L, and LL. The TO247-3 packaged SiC-B and state-of-the-art TO220-3 packaged Si-A diode [238] are used as a benchmark as they exhibit similar lateral dimensions. Furthermore, novel 200 μ m thin batch B4 Ga₂O₃ diodes of size ML and LL mounted on AlN-DPC substrates are investigated. The SiC-A diode, which is assembled in the same way and exhibits lateral dimensions similar to the ML type diode, is used as a benchmark for these measurements.

The TO packages cause a rather high stray inductance, which causes ringing and slows down the commutation speed. However, they are still widely used due to their low cost and easy handling, making the results application-relevant. Furthermore, the stray inductance is similar for all samples. Therefore, as will be evident later, differences in waveforms and efficiencies can be clearly attributed to the semiconductor despite the influence of the package.

6.3 Investigating 600 µm thick TO247 packaged diodes with different chip sizes

6.3.1 Static electrical properties relevant for switching operation

Differential on-resistances

The conduction losses caused by the on-resistances dominate the efficiency especially at low switching frequencies and high output currents [225, 226]. The forward characteristics of the three $600 \,\mu\text{m}$ thick Ga₂O₃ diodes and the SiC-B and Si-A reference diodes selected for buck converter operation are shown in Fig. 6.2a at a junction temperature of $100 \,^{\circ}\text{C}$. The SiC and Ga₂O₃ diodes have a similar knee voltage of about 0.8 to 0.9 V but the knee voltage of the Si-A diode is only about 0.5 V. Furthermore, the Si-A diode exhibits a curved forward current–voltage profile as typical for bipolar diodes. The on-resistances of Ga₂O₃ sample B3-LL and reference sample Si-A are similar, which is important for the later comparison of the buck converter efficiencies. Fig. 6.2b.

In Figs. 6.2b and 6.2c the temperature-dependent absolute and specific differential onresistances of the Ga₂O₃ diodes are directly compared to those of the SiC-B and Si-A reference diodes. The values for the Si-A diode with non-constant differential on-resistance are averaged from a fit between 5 A and 8 A, which corresponds to the typical operating range in Section 6.3.3. Since the junction structure of all Ga₂O₃ diodes is identical, the on-resistance increases with decreasing active area in the order $r_o(LL) < r_o(L) < r_o(ML)$, but their specific on-resistances are similar. The SiC-B reference diode reaches the lowest absolute and specific on-resistances. Although the Ga₂O₃ diodes L and LL have the largest active areas, the SiC-B diode is therefore expected to enable higher efficiencies under heavy load conditions.

The increase in on-resistance with rising temperature is discussed in more detail in Section 4.2.4. By using Eq. 4.7, α_{300} and α_{373} are 1.3 to 1.4 for all Ga₂O₃ diodes almost independent of the reference temperature, but 1.6 and 1.9 for the SiC-A diode, respectively. Thus, at high temperatures and under heavy load conditions, Ga₂O₃ is expected to continuously approach the efficiencies with the SiC-B sample – provided the rise in junction temperature of the Ga₂O₃ diode does not reach a level that would compensate the advantage in α_{300} and α_{373} .



Figure 6.2: (a) Forward current–voltage characteristics of the three $600 \,\mu\text{m}$ thick TO247 packaged Ga₂O₃ samples of sizes ML, L, and LL from batch B3 and the SiC-B and Si-A reference diodes at a junction temperature T_j of $100 \,^\circ\text{C}$. (b) The extracted temperature-dependent differential on-resistances increase with decreasing chip size in case of Ga₂O₃, while SiC-B exhibits the lowest values. (c) The specific (related to anode size) differential on-resistance is similar for all Ga₂O₃ diodes and significantly smaller in case of SiC-B. Parts adapted from [260].

Capacitance, capacitive charge, and capacitance stored energy

The junction capacitances C_j shown in Fig. 6.3a for all diodes dominate the efficiency especially at high switching frequencies and low output currents [225, 226]. The capacitive charge Q_c and capacitance stored energy E_c are calculated from the C-V profiles using Eqs. 3.5 and 3.6, and shown in Figs. 6.3b and 6.3c, respectively. Since there is no significant change in junction capacitance with temperature for the unipolar samples (see also Section 4.2.2), only the room temperature profiles are shown for clarity.

The small bandgap and bipolar nature of the Si-A diode, however, cause a significant rise of the junction capacitance at zero bias voltage (C_{j0}) when the temperature is increased to $150 \,^{\circ}$ C. Yet, C_j drops sharply at a voltage of about -10 V, resulting in a minuscule increase in the stored charge and a negligible change of the capacitance energy. Both C_{j0} and Q_c decrease with decreasing anode area in the order $C_{j0}(LL) > C_{j0}(L) > C_{j0}(SiC) > C_{j0}(ML)$. At room temperature, the bipolar Si diode exhibits the lowest C_{j0} and Q_c .

For the Ga₂O₃ diodes, the capacitance energy also decreases with decreasing anode area in the order $E_c(LL) > E_c(L) > E_c(ML)$. Although C_{j0} is lower for SiC-A than for Ga₂O₃ sample L, both samples exhibit a similar E_c at -400 V, and the E_c of Ga₂O₃ sample L even drops below that of sample SiC-B as the reverse voltage increases. This results from differences in the doping profile of the epitaxial layer of the SiC-B and Ga₂O₃ diodes, causing the junction capacitance of sample L to fall beneath that of SiC-A at voltages below -300 V.

The specific (related to the anode area) capacitive charge and energy are shown in Figs. 6.3d and 6.3e, respectively. They are similar for all Ga₂O₃ diodes because of the identical junction design. Below about -200 V the SiC-B diode exhibits a higher specific capacitive charge and energy than the Ga₂O₃ diodes. At the planned operating point of -400 V, the specific Q_c (specific E_c) is about 10 nC/mm^2 ($1.4 \mu \text{J/mm}^2$) for the Ga₂O₃ diodes and about 11.6 nC/mm^2 ($1.9 \mu \text{J/mm}^2$) for the SiC-B diode. Note that the given values are associated



Figure 6.3: For the Ga₂O₃ samples, the absolute (a) junction capacitance C_j (b) capacitive charge Q_c and (c) capacitance stored energy E_c are temperature-independent and increase with increasing chip size. Although Q_c of SiC-B is lower than the Q_c of diode L, the C_j of sample L drops below that of sample SiC-B below about -300 V, leading to similar E_c profiles. The specific (related to anode size) capacitive (d) charge and (e) energy of the Ga₂O₃ diodes are slightly lower than those of the SiC-B reference at voltages below -200 V. (f) However, the high on-resistance leads to a substantially higher charge \times resistance figure of merit. Adapted from [260].

with uncertainties of up to 10% based on the determination of the anode size via optical microscopy after removing the mold mass with a laser (see Section 3.2).

Figure 6.3f shows that despite this difference, the high on-resistance of the Ga₂O₃ diodes causes the $r_0 \times Q_c$ figure of merit to be almost seven times higher than that of the SiC-B reference. From Eq. 3.4 an average breakdown field of 2.0 MV/cm is estimated, which corresponds to about 80 % of the theoretical SiC limit, and 60 % of GaN limit, but only 25 % of Ga₂O₃ limit (see Section 2.3.1) [12]. It is thus vital for future applications to increase the breakdown field. This concerns especially an improved edge termination and process control to enhance the breakdown voltage and allow a reduction of the epitaxial layer thickness to the minimum required value. Also note that with about 1.2 kV for the 650 V rated samples, the safety margin between actual breakdown and rated reverse voltage is rather large here, which should be avoided from an electrothermal point of view according to Section 5.3.5. These basic considerations help to analyze the basic switching properties studied in the following section.

6.3.2 Analysis of the basic switching waveforms

General waveforms

As a first step to study the switching properties, Fig. 6.4 shows the current and voltage waveforms during turn-off for all diode samples at a load current of 15 A and a DC link voltage of 400 V.



Figure 6.4: Current and voltage waveforms during diode turn-off for the three Ga_2O_3 batch B3 samples ML, L, and LL, and the SiC-B and Si-A references at 400 V and 15 A. The current undershoot increases with increasing Ga_2O_3 chip size due to the higher capacitive charge. Even for the largest Ga_2O_3 diode, however, it is still lower than the reverse recovery peak of the bipolar Si-A diode. Adapted from [260].

The current slew rate is mainly controlled by the turn-on of the transistor and is therefore similar for all diodes. The negative current undershoot (sinusoidal half-wave) is caused by the charging of the junction capacitance of the diode. The associated maximum reverse current and undershoot time increase with increasing chip area and junction capacitance (see also Fig. 6.3). More detailed explanations of the waveforms will follow below. In the case of the Si-A diode, the recombination of charge carriers adds to the charging of the comparatively small junction capacitance, resulting in the highest reverse current and longest undershoot time.

Varying gate resistance and distinct charging phases

In the following, the gate turn-on resistance $R_{g,on}$ is varied to study the influence of the MOSFET turn-on speed on the turn-off behavior of the different diodes. The diode current and voltage waveforms at an $R_{g,on}$ of 2.2Ω and 27Ω are shown in Fig. 6.5a exemplarily for the Ga₂O₃ LL type diode. Decreasing the turn-on gate resistance increases both the voltage slew rate dv_D/dt and the current slew rate di_D/dt . The current undershoot, often referred to as "reverse recovery" following the nomenclature for bipolar diodes, has previously been used to assess and compare the switching properties of first Ga₂O₃ diodes from different publications with different experimental setups [133, 156–158]. However, even small changes such as a different gate resistance clearly affect the magnitude and shape of the current undershoot substantially. As will be discussed in the following in more detail, such comparisons can therefore be highly misleading.

At low switching speeds (high $R_{g,on}$) the turn-on speed of the transistor, during whose Miller plateau the junction capacitance of the diode is charged, limits voltage slew rate. At high switching speeds (low $R_{g,on}$), however, the parasitic commutation-loop inductance and associated *LC* circuit increasingly limit the voltage slew rate. This gives rise to the following phenomenon visible in Fig. 6.5a.

A low $R_{g,on}$ (high dv_D/dt and di_D/dt) triggers a high peak reverse current $i_{D,rm}$ of -25 A and an undershoot time of 9 ns. With a correspondingly steep maximum slew rate of 114 V/ns, this current undershoot is already sufficient to charge the junction capacitance to the DC link voltage of about -400 V. A high $R_{g,on}$ lowers the current slew rate di_D/dt and increases



Figure 6.5: (a) The current and voltage slew rate of the Ga_2O_3 diode (here: B3-LL type) during turn-off is reduced by increasing the turn-on gate resistance. This also affects the current undershoot: During the first undershoot associated with the charge $Q_{c,a}$ and energy $E_{c,a}$ the junction capacitance is only partially charged. A second current undershoot supplies the remaining required charge $Q_{c,b}$ and energy $E_{c,b}$ required to reach the DC link voltage. The previous literature comparing the basic properties of different Ga_2O_3 diodes does not take this effect into account. (b) The simulated charging of a voltage-dependent capacitance further illustrates that only the addition of a sufficiently large stray inductance makes the undershoot appear as a sinusoidal half-wave. During this half-wave, however, the charging phase does not necessarily have to be completed. Adapted from [260].

the duration of the first current undershoot to 12 ns but reduces the peak reverse current to -11.5 A. With a correspondingly low peak voltage slew rate dv_D/dt of 40 V/ns, the junction capacitance is only charged to a voltage of about -180 V during the first current undershoot. The charge associated with this process is marked as $Q_{c,a}$ in Fig. 6.5a and must be smaller than in the previous setup with low gate resistance. Therefore, a second charging phase associated with a second current undershoot of longer duration but lower amplitude supplies the remaining charge $Q_{c,b}$ required to charge the junction capacitance to its final value of -400 V.

The occurrence of these two charging phases can be explained using the simulations shown in Fig. 6.5b (diode model created on the basis of the explanations in Section 4.4). The fact that the junction capacitance of the diode and the output capacitance of the MOSFET are voltage-dependent causes the charging profile to intrinsically (i.e., without parasitic inductances in the commutation loop) exhibit a peak current followed by a flattened tail rather than resembling a sinusoidal half-wave. Adding a parasitic inductance slows down di_D/dt and increases the undershoot peak and duration but also induces ringing. A sufficiently high inductance causes the undershoot to appear as a sine half-wave. In agreement with the experimental data, however, the junction capacitance is not yet fully charged after the initial current undershoot.

To get further insight, the gate resistance is varied exemplarily for diode LL at a DC link voltage of 400 V and a load current of 15 A. The initial charge $Q_{c,a}$ and corresponding energy $E_{c,a}$ are calculated by integrating over the duration of the first undershoot, i.e., from t = 0 ns to the first zero-crossing point with positive d_{i_D}/dt . The total charge $Q_{c,tot}$ and corresponding



Figure 6.6: Switching measurements at 400 V and 15 A. (a) At low switching speeds (high gate resistance $R_{g,on}$), the major part of the energy necessary to charge the junction capacitance is supplied in the second charging phase (see Fig. 6.5a). When $R_{g,on}$ is reduced, $E_{c,b}$ decreases and $E_{c,a}$ increases but the total switching energy $E_{c,tot}$ remains constant. The total (b) charge and (c) energy are independent of $R_{g,on}$ in case of Ga₂O₃ and SiC, but increase significantly for the Si-A diode. Due to a saturation of the IPM sensor, the values for Si-A at 2.2 Ω are higher than shown. (d) The absolute peak current slew rate increases with decreasing $R_{g,on}$ but is similar for all diodes. Adapted from [260].

energy $E_{c,tot}$ are obtained by integrating from t = 0 ns up to the time when the diode voltage reaches the DC link voltage of -400 V. The charge $Q_{c,b}$ and energy $E_{c,b}$ associated with the second charging phase are then estimated from the difference of the respective parameters.

The results are shown in Fig. 6.6a. As $R_{g,on}$ increases, both dv_D/dt and $|di_D/dt|$ decrease, reducing the oscillations but increasing the charging time. Therefore, in agreement with the previous considerations in Fig. 6.5, $E_{c,a}$ decreases and $E_{c,b}$ increases, but the total energy $E_{c,tot}$ remains constant. When $R_{g,on}$ is reduced to 2.2 Ω , the junction capacitance is initially charged to even below -400 V because of the voltage overshoot associated with the parasitic loop inductance. Therefore, $E_{c,a}$ is slightly larger than $E_{c,tot}$, which is then compensated by a negative $E_{c,b}$.

Previous statements about the undershoot time, peak current, or charge often only consider the first charging phase [133, 156–158]. This is only meaningful if critical circuit parameters such as loop inductance and switching speed are similar and if the full DC link voltage is reached during the first charging phase. Comparing characteristics of Ga_2O_3 Schottky diodes using these procedures, as often found in literature, is thus not necessarily the most appropriate way. Based on these considerations, only the total charge $Q_{c,tot}$ and total switching energy $E_{c,tot}$ is used to compare the devices under investigation in the following.

Figs. 6.6b and 6.6c show the corresponding results at 400 V and 15 A for different turn-on resistances. When increasing the switching speed by lowering the turn-on gate resistance, both $Q_{c,tot}$ and $E_{c,tot}$ increase significantly for the Si diode due to its bipolar structure. At an $R_{g,on}$ of 2.2 Ω , the built-in integrator of the IPM sensor saturates because of the magnitude of the reverse recovery current (see Section 3.4.2). The associated charge and energy are



Figure 6.7: (a) Current and voltage waveforms and (b) associated switching power and energy of MOSFET and diode during diode turn-off for the smallest (ML) and the largest (LL) Ga₂O₃ diode. The purely capacitive charging creates negligible losses in the Schottky diode. However, the higher capacitance charge of sample B3-LL results in a larger reverse current peak. This increases the turn-on peak current in the transistor and with that also its turn-on switching loss significantly. Adapted from [260].

therefore higher than shown in Figs. 6.6b and 6.6c, which is indicated by the respective arrows. On the contrary, $Q_{c,tot}$ and $E_{c,tot}$ remain constant for both the Ga₂O₃ diodes and the SiC-B diode when the load current is increased.

Among the Ga₂O₃ diodes, both $Q_{c,tot}$ and $E_{c,tot}$ increase in the order as expected from the C-V measurements in Fig. 6.3, i.e., they increase with rising anode area. The different doping profile in the epitaxial layer of Ga₂O₃ sample L and the SiC-B sample results in a lower $Q_{c,tot}$ of the SiC diode, but a similar $E_{c,tot}$ at the selected DC link voltage, which is also in agreement with the previous considerations. Figure 6.6d shows that the absolute value of the maximum current slope increases with decreasing gate resistance. However, it is similar for all diodes because it is mainly controlled by the turn-on speed of the MOSFET and the commutation-loop inductance rather than the diode properties.

While the bipolar recombination in the Si diode comes with a substantial switching energy loss, the purely capacitive charging and discharging of the junction capacitance in case of the Ga_2O_3 samples only causes minimal energy losses related to a hysteresis between the two processes [227]. However, the corresponding charging and discharging current also affects the switching waveforms in the SiC MOSFET.

Figure 6.7 shows the current and voltage waveforms of the MOSFET and the Ga_2O_3 diode ML an LL (Fig. 6.7a), as well as the associated respective power and switching energy (Fig. 6.7b) when the MOSFET turns on. The higher capacitive charge of diode sample LL in comparison to sample ML enhances the current undershoot. This leads to an increased peak in the transistor current and increases its turn-on power loss, which is expected to increase the MOSFET temperature and limit the high-frequency capability.



Figure 6.8: (a) Diode turn-off and MOSFET turn-on switching waveforms exemplarily for Ga₂O₃ sample B3-L at 400 V and load currents of 5 A and 20 A. Increasing the load current increases the time required to commutate the current and charge the junction capacitance. (b) The room-temperature peak voltage slew rate increases with decreasing Ga₂O₃ anode size, is independent of the load current, and (c) does not change when the case temperature is increased to 150 °C. The (d) charge and (e) energy is independent of the load current and the temperature in case of the unipolar diodes but increases with increasing Ga₂O₃ anode size. Parts adapted from [260].

Influence of load current and junction temperature

Next, the load current is varied between 2.5 A and 40 A at a DC link voltage of 400 V and a turn-on gate resistance $R_{g,on}$ of 8.2 Ω by changing the length of the first pulse. Measurements are performed at lead frame temperatures of about 30 °C (room temperature) and 150 °C.

Figure 6.8a shows the room-temperature switching waveforms of the diode and the MOSFET exemplarily with sample B3-L at load currents of 5 A and 20 A. Note that the drain current of the MOSFET was measured with a 30 MHz Rogowski probe, which leads to the problems of underestimated current slope and a distorted representation of the oscillations examined in Section 3.4. Similar to the discussion in Section 3.4.5, the absolute current slew rate di_D/dt increases slightly when the load current is increased, but it is still largely affected by the turn-on speed of the MOSFET, i.e., the continuous opening of its channel while the transistor is in saturation mode. Correspondingly, the time required to commutate higher load currents I_L increases. Taking the interval between $0.9 \cdot I_L$ and 0 A as the basis, it increases from 1.7 ns at $I_L = 2.5$ A to 7.3 ns at $I_L = 30$ A in the exemplary case of sample B3-L.

The peak dv_D/dt depicted in Fig. 6.8b occurs during the initial stage of the charging of the junction capacitance after the current commutation when the reverse current undershoot reaches its maximum. Since the output capacitance of the MOSFET is fairly constant and low when it is still sufficiently charged (about 60 pF for $V_{ds} > 250$ V), the junction capacitance of the diode strongly influences the peak voltage slew rate. Consequently, the peak dv_D/dt decreases with increasing anode size for the Ga₂O₃ diodes as, for the same voltage difference,

the duration of the halfwave-like charging current increases, and so does the associated voltage rise time. However, it does not change significantly with varying load current.

On the other hand, the average voltage slew rate (determined between 10 % to 90 % of the DC link voltage) decreases significantly, e.g., from about 32 V/ns at $I_{\text{L}} = 2.5 \text{ A}$ to 17 V/ns at $I_{\text{L}} = 30 \text{ A}$ in the example case of diode B3-L. The reason is that due to the higher load current, the Miller plateau shifts to higher voltages and its duration (during which the output capacitance is discharged) increases. As a consequence, the duration of the second charging phase as defined in Fig. 6.5a increases significantly (e.g., from about 5 ns at $I_{\text{L}} = 2.5 \text{ A}$ to about 13 ns at $I_{\text{L}} = 30 \text{ A}$), which decreases the overall voltage slew rate.

Comparing the results at room temperature and $150 \,^{\circ}\text{C}$ in Figs. 6.8b and 6.8c shows that it is independent of the junction temperature in case of the Ga₂O₃ and SiC samples, but it decreases significantly with rising temperature for the bipolar Si-A diode.

For the Ga₂O₃ samples, the capacitive charge (Fig. 6.8d) decreases in the same sequence as expected from the C-V measurements, i.e., with decreasing anode area in the order $Q_{c,tot}(LL) > Q_{c,tot}(L) > Q_{c,tot}(SiC) > Q_{c,tot}(ML)$. In agreement with the previously discussed difference in the doping profile of the epitaxial layers, the total capacitance stored energies $E_{c,tot}$ of samples SiC-B and B3-L are similar despite the difference in $Q_{c,tot}$ (see Fig. 6.8e). Apart from that, $E_{c,tot}$ and $Q_{c,tot}$ show the same trend. For both the SiC and the Ga₂O₃ diodes, $E_{c,tot}$ are independent of the junction temperature and load current, proving their purely unipolar character. In case of the Si-A diode, however, the reverse recovery charge and energy increase with both the junction temperature and the load current.

Influence of DC-link voltage

Next, a load current of 10 A is set with an unchanged turn-on gate resistance of 8.2Ω , and the DC link voltage is increased from 100 to 500 V for the SiC and Ga₂O₃ diodes. Measurements are performed only at room temperature, since the preceding experiments confirm the temperature independence of the switching waveforms. Both the capacitive charge $Q_{c,tot}$ (Fig. 6.9a) and energy $E_{c,tot}$ (Fig. 6.9b) increase with rising DC link voltage and with rising anode size as expected from the C-V measurements, while samples SiC-B and Ga₂O₃ sample B3-L again exhibit similar energies.

Both parameters are, however, some 10 % higher than the C-V data suggest. As discussed in Section 3.4.1, note that even slight differences in the deskew between current and voltage signal can lead to deviations up to 20 % in the calculated switching energy. The current signal itself is not affected by this measurement error but two effects lead to additional deviations: First, although the captured current slopes should be accurate within some percent, peak currents might be somewhat overestimated (see also Section 3.4.1) and especially after the transient a slight offset to lower currents is sometimes observed, which increases both $Q_{c,tot}$ and $E_{c,tot}$. Second, the parasitic commutation-loop inductance increases the maximum reverse current and causes oscillations [31, 180, 228], which also means that the junction capacitance is repeatedly slightly discharged during the charging process. The consequences are two-fold: For one, this results in a purely mathematically higher total charging $Q_{c,tot}$ calculated by the analysis algorithm. Furthermore, the (dis-)charging processes can cause additional losses, which should, however, be in the range of well below 1 μ J [227] and thus negligible here. Although low-inductive package designs could further reduce ringing and with that some of the discussed effects, note that the influence of the semiconductor itself is still evident here.



Figure 6.9: Increasing the DC link voltage at a load current of 10 A leads to an increase in (a) capacitive charge (b) capacitance stored energy (c) peak voltage slew rate and (d) absolute peak current slew rate. The parasitic loop inductance and associated distortion of the switching waveforms lead to a slight deviation from the values expected from the capacitance–voltage measurements. Adapted from [260].

The current slew rate di_D/dt (Fig. 6.9d) increases directly with rising DC link voltage according to Eq. 3.9 but is similar for all samples since the loop inductance is comparable for all diodes and the switching speed is determined mainly by the turn-on of the MOSFET. The higher capacitive charge associated with higher DC link voltages increases the initial inrush current into the junction capacitance. Although the voltage transient time increases with rising DC link voltage (e.g., from 6.4 ns at 100 V to 17.2 ns at 500 V for the ML type diode), this leads to a rise in peak dv_D/dt as shown in Fig. 6.9c. In agreement with the previous considerations, the peak dv_D/dt increases with decreasing anode size.

6.3.3 Buck converter operation under varying conditions

The double pulse tests performed in the previous sections provide insight into the switching behavior of Ga_2O_3 diodes and confirm that fast and clean switching is possible with the PCB layout even at voltages exceeding the planned system voltage of 400 V. This provides the necessary basis to proceed with continuous operation tests in a 400 V to 200 V buck converter topology. To this end, first the output current is varied at a constant switching frequency of 100 kHz. Then, a constant output current is set, and the switching frequency is increased up to 350 kHz. It is worth emphasizing that a stable operation of the buck converter at 100 kHz and above for several hours is achieved with the still experimental Ga_2O_3 chips.

Variation of the output current at constant switching frequency

The temperature of the diodes during buck converter operation after reaching a steady state is shown in Fig. 6.11b. It is measured on the lead frame next to the die in case of the Ga_2O_3 diodes, and on the mold mass in case of SiC-B and Si-A. As the output current is increased, the diode temperature rises because of the growing conduction losses. By increasing the chip



Figure 6.10: 400 V to 200 V buck converter operation at a switching frequency of 100 kHz with TOpackaged Ga₂O₃ batch B3 and the Si-A and SiC-B reference diodes. (a) Increasing the output current leads to a significant increase in the lead frame temperature of the Ga₂O₃ diodes due to growing conduction losses. A thermal relief is achieved by increasing the anode area. (b) The transistor temperature increases more with the bipolar Si-A diode than with the unipolar diodes. (c) Under light load conditions, the efficiency decreases with increasing Ga₂O₃ anode size and capacitive charge. The opposite is the case under heavy load conditions, where the on-resistances dominate. Despite the higher on-resistance, a higher efficiency is achieved with the Ga₂O₃ diodes than with the state-of-the-art Si diode. Adapted from [260].

size, a substantial thermal relief can be achieved: At an output current of 4 A it decreases from 53 °C for ML, over $44 ^{\circ}$ C for L to $41 ^{\circ}$ C for LL in agreement with Sections 5.3.3 and 5.3.4.

As shown in Fig. 6.10b, increasing the output current also increases the transistor temperature. The temperature rise is similar for the unipolar Ga_2O_3 and SiC diodes, but slightly higher for the Si-A diode due to the additional reverse recovery current that increases at higher output currents and leads to additional switching losses in the MOSFET in agreement with Fig. 6.7. To avoid possible damage to the devices caused by high junction temperatures related to the high thermal resistance of the 600 µm thick and cathode-side cooled Ga_2O_3 devices (see Chapter 5 and [255, 256, 259]), the maximum output current was limited to 4 A for B3-ML, 6 A for B3-L and 8 A for B3-LL, SiC-B and Si-A. This corresponds to case temperatures of about 50 to $60 \,^{\circ}$ C for the Ga_2O_3 diodes.

Figure 6.10c shows the power loop efficiencies for different output currents. Swapping the freewheeling diode significantly affects the efficiency profiles, which is the basis for the targeted studies.

Under light load conditions, the efficiency is predominantly affected by the parasitic capacitances of the devices. Therefore, the efficiency at 1 A decreases in the order $\eta_p(ML) > \eta_p(SiC, Si) > \eta_p(L) > \eta_p(LL)$, i.e., it decreases with increasing junction capacitance and capacitive charge (increasing anode area) in agreement with the static and dynamic electric characterization in Sections 6.3.1 and 6.3.2. The previously observed thermal relief associated with larger chip sizes is therefore also accompanied by increased switching losses, underscoring the importance of the thermal management strategies explored in Chapter 5. In this respect, the 600 µm thick ML Ga₂O₃ diode broke instantly when the output current was increased from 4 A to 5 A, presumably due to a temperature-induced failure of the blocking capability. However, operating the buck converter with the new 200 µm thin ML type diode was possible at 5 A/100 kHz despite the higher thermal resistance of the AlN-DPC substrate compared to the copper-based TO247 lead frame (see Section 6.4.2).



Figure 6.11: When increasing the switching frequency during buck converter operation, even the largest Ga₂O₃ diode (LL) exhibits a lower efficiency drop than the Si-A diode. Thus, the Ga₂O₃ diodes exceed the efficiencies with Si-A despite their high on-resistance. (b) The lead frame temperature of the Ga₂O₃ diodes increases only slightly with rising switching frequency, while the reverse recovery losses in Si-A lead to a significant rise of the mold temperature. (c) Despite the high capacitive charge of Ga₂O₃ sample LL, the increase in MOSFET temperature is lower than with the Si-A reference diode. Adapted from [260].

Under heavy load conditions, the efficiency is increasingly dominated by the on-resistance of the devices. Although the on-resistances of samples Si-A and Ga₂O₃ LL are similar in the range between 2 A and 8 A (see Fig. 6.2a) and Si-A exhibits an even slightly lower on-resistance at higher temperatures (see Fig. 6.2b), the increasing reverse recovery losses in case of Si-A lead to lower efficiencies. Above about 4 A, the efficiencies therefore decrease in the order $\eta_p(SiC) > \eta_p(LL) > \eta_p(L) > \eta_p(Si) > \eta_p(ML)$. The high on-resistance is clearly a drawback at present, but the fact that small-area diodes have been developed that outperform the theoretical unipolar PFOM of SiC [22, 23] (see also Section 2.3.4) suggests that the efficiencies can be increased in the future. At an output current of 2 A, there appears to be an optimum between the influences of parasitic on-state resistances and capacitances, which is determined largely by the transistor, and leads to a peak efficiency of 97.80 % with Si, 97.75 % with LL, 98.00 % with ML and L, and 98.13 % with SiC.

Variation of the switching frequency at constant output current

Next, a constant output current of 2 A for the ML type Ga_2O_3 diode and 5 A for all other diodes is set during the 400 V to 200 V step-down operation. Then, the switching frequency is increased from 100 to 350 kHz in steps of 50 kHz to assess the high-frequency switching capability of Ga_2O_3 Schottky diodes.

As shown in Fig. 6.11a, a linear decrease in the buck converter efficiency with rising switching frequency is observed for the Ga₂O₃ and SiC Schottky diodes. The efficiency drop increases in the order -0.53% (ML), -0.56% (L), -0.58% (SiC), and -0.67% (LL) per 100 kHz, as estimated from a linear fit to the data with an uncertainty of about 0.01%. This agrees with the results in Sections 6.3.1 and 6.3.3 as the parasitic capacitances dominate the efficiency at high switching frequencies. In the case of the Si-A diode, the absolute slope increases disproportionately with increasing switching frequency. If a linear fit is performed nevertheless, the highest average efficiency drop of -1.11% per 100 kHz is obtained. Compared to the Ga₂O₃ sample B3-L, the efficiency of the converter with Si-A diode is 0.27\% lower at 100 kHz and 1.4\% lower at 300 kHz.

The temperatures measured on the lead frame for the Ga_2O_3 diodes and on the mold mass in case of the Si and SiC diode are shown in Fig. 6.11b. An exact comparison of the absolute temperatures is difficult because of the different positions of the sensors. Nevertheless, significant differences are evident: In case of the unipolar Ga_2O_3 and SiC Schottky diodes, only a slight rise in temperature of about 3 to 4 K over the entire frequency range is observed. In contrast, the switching losses in the bipolar Si-A diode increase significantly, resulting in a temperature rise of 18 K when the switching frequency is increased from 100 kHz to 300 kHz.

Based on the data in [227], a purely capacitive (dis-)charging of the junction capacitance of the SiC-B diode should lead to an energy dissipation of $< 0.2 \,\mu$ J per cycle. Assuming a worst-case $R_{th,ja}$ of $10 \,\text{K/W}$ (see also discussion in Section 6.4.2) the increase in junction temperature should thus be less than 0.5 K when the switching frequency is increased from 100 to $350 \,\text{kHz}$. As shown in Fig. 6.11c, the transistor temperature measured on its mold mass increases significantly with increasing switching frequency for all samples due to its growing switching losses. Measurements with a thermal camera in the setup with 200 μ m thin diodes discussed later (see Fig. 6.11c) show that this increases the temperature of the heat sink by a few Kelvin. The latter effect is therefore (bearing in mind the above estimate) mainly responsible for the slight increase in the temperature of the unipolar diodes.

Since the negative current undershoot during diode turn-off adds to the intrinsic switching losses in the transistor (see Fig. 6.7), the temperature rise is largest when the Si-A diode is used (+33 K), and decreases for the unipolar diodes in the order Ga_2O_3 -LL (+24 K), SiC-B (+22 K) and Ga_2O_3 -L (+19 K). This further illustrates that despite the negligible switching losses in the Ga_2O_3 diodes and the temperature relief that can be achieved by increasing their chip size, other thermal management strategies such as those discussed in Chapter 5 should be preferred. Nevertheless, it is noteworthy that even with 600 µm thick Ga_2O_3 diodes and cathode-side cooling, the efficiencies with a state-of-the-art Si diode of similar chip size (which itself shows a comparatively high efficiency) can be surpassed, coupled with a substantial reduction in the junction temperature of the active high-side switch. Although the devices are still at an early stage of development and have a high on-resistance, the advantages of pure unipolar switching seem to outweigh the problems associated with the high thermal resistances when compared to bipolar Si technology in terms of high-frequency switching operation.

6.4 Investigating 200 µm thin diodes with different chip sizes assembled on ceramic substrates

In this section, the newest available generation of 600 V rated Ga₂O₃ diodes with $200 \,\mu\text{m}$ thin substrate from batch B4, soldered to AlN-DPC substrate, is examined in a buck converter application and compared to the $600 \,\mu\text{m}$ thick TO-packaged batch B3 devices in the previous section. In addition, the SiC-A diode sintered to an AlN-DPC substrate is used as a commercial reference. Since the dice are not sealed with mold mass or silicone gel, an infrared camera can be used to measure chip surface temperature.

The same measurement settings and PCB layout as in the preceding section are used. To fit the AlN-DPC substrate onto the PCB, only the corresponding footprint has been slightly adjusted. Fundamental switching properties of Ga_2O_3 diodes (such as basic waveforms, temperature-independence, or influence of chip size or gate resistance) are already known from Section 6.3. Therefore, only the data that are important for understanding potential changes in buck converter performance are discussed in the following.



Figure 6.12: Preliminary studies for buck converter tests with 200 µm thin batch B4 Ga₂O₃ diodes of sizes ML and LL in comparison to their batch B3 predecessors: (a) differential onresistance, (b) capacitive charge, (c) charge × resistance FOM, and double pulse tests with (d) ML and (e) LL diodes. The significant decrease in on-resistance is found to stem from an increase in the doping concentration of the epitaxial layer, which increases the capacitive charge. Therefore, the $Q_c \times r_o$ FOM is not improved, but the higher charge increases the current undershoot during diode turn-off. Adapted from [260].

6.4.1 Electrical properties relevant for converter operation

The forward current–voltage profiles exhibit a change in the temperature-coefficient of the forward voltage at about 75 °C as discussed in Section 4.2.1, i.e., the conduction losses decrease with rising junction temperature until 75 °C but increase for higher temperatures. The coefficient α_{373} describing the increase in differential on-resistance with rising temperature related to a junction temperature of 100 °C is calculated by using Eq. 4.7. With a value of about 1.0 it is lower than the α_{373} values of 1.3 to 1.4 determined for the previous batch B3 diodes. Furthermore, as shown in Fig. 6.12a, the differential on-resistance is reduced significantly compared to batch B3 diodes. Figure 6.12b shows that at the same time the junction capacitance is a factor 1.3 to 1.5 higher than for the batch B3 diodes.

According to Eqs. 2.3 and 3.7 the differential on-resistance is proportional to the inverse effective doping concentration $N_{\rm D}^{-1}$ while the junction capacitance is proportional $\sqrt{N_{\rm D}}$. This suggests that the reduction in on-resistance is not only achieved by substrate thinning but mainly by increasing the donor concentration of the epitaxial layer. Indeed, the evaluation of capacitance–voltage profiles according to Section 3.3 yields effective donor concentrations of about $0.8 \times 10^{16} \,\mathrm{cm^{-3}}$ and $0.9 \times 10^{16} \,\mathrm{cm^{-3}}$ for the B3-ML and B3-LL Ga₂O₃ diodes, but about $1.5 \times 10^{16} \,\mathrm{cm^{-3}}$ for the B4-ML and $1.8 \times 10^{16} \,\mathrm{cm^{-3}}$ for the B4-LL diode. This



Figure 6.13: The power-loop efficiencies with the (a) ML and (b) LL type diodes of batch B4 are lower under light load conditions but the difference decreases under heavy load conditions and is even reversed in case of the ML diode due to the previously revealed increased doping concentration of the epitaxial layer. (c) The surface temperatures of B4-LL and SiC-A are similar, whereas B4-ML shows a much stronger increase with rising output current. Adapted from [260].

corresponds to an increase of N_D by a factor of 2 and thus constitutes the reason for the rise in junction capacitance by a factor of $\sqrt{2} \approx 1.4$.

Both the substrate and the epitaxial layer contribute to the differential on-resistance, although to a different degree. Assuming an electron mobility of $20 \text{ cm}^2 \text{ V}^{-1} \text{ s}^{-1}$ in the substrate at an effective donor concentration of $1 \times 10^{19} \text{ cm}^{-2}$, which is even lower than the pessimistic scenario in Tab. 5.7, this results in a high resistivity of about $31 \text{ m}\Omega$ cm. In the example case of the ML diode, the substrate should then contribute to the total differential on-resistance with a maximum of $78 \text{ m}\Omega$ for a 600 µm thick diode and $26 \text{ m}\Omega$ in case of the 200 µm thick diode. A comparison to the actual differential on-resistance of about 400 to $450 \text{ m}\Omega$ shows that the diode resistance is dominated by the drift layer despite the increase in doping concentration.

The higher doping concentration of the epitaxial layer translates to a higher capacitive charge and capacitance stored energy. Therefore, as shown in Fig. 6.12c the charge \times resistance figure of merit is not improved with the batch B4 diodes and significantly higher than that of the SiC-A reference diode. The double pulse tests in Figs. 6.12d and 6.12e compare the switching waveforms of the 600 µm thick batch B3 and the 200 µm thin batch B4 diodes at a DC link voltage of 400 V and a load current of 15 A. The higher doping concentration causes the area and duration of the current undershoot to increase. The additional charge released during diode turn-off adds to the turn-on current peak in the transistor and increases its switching losses as shown in Fig. 6.7. This is expected to ultimately impact the efficiency of the buck converters. Signs of atypical behavior, as seen in the forward and reverse current–voltage profiles, are not evident here.

6.4.2 Buck converter operation under varying conditions

Based on the preliminary considerations, the power-loop efficiencies with varying output currents during 400 V to 200 V step-down operation at a switching frequency of 100 kHz are discussed. The results with the $200 \,\mu\text{m}$ thin B4-ML diode, the SiC-A reference diode, and the previous efficiencies with the B3-ML diode are shown in Fig. 6.13a.

Under light load conditions (at 1 A), the efficiency with the B4-ML diode is about 0.3% lower than with the B3-ML diode. Under heavy load conditions (at 4 A), however, the efficiency

the with B4-ML diode is 0.2% higher. Similarly, the B4-LL diode exhibits a 0.7% lower efficiency than its batch B3 counterpart under light load conditions but a flatter decrease towards higher output currents, resulting in the same efficiency at 8 A. At an output current of 10 A the efficiency with diode B4-LL is less than 0.1% below the SiC-A reference.

The maximum surface temperatures of the B4-ML, B4-LL, and SiC-A diode are shown in Fig. 6.13c. The temperatures were measured with an infrared camera from the perspective shown in Fig. 6.15b, which is similar to the angle in the photograph in Fig. 6.1b. Although the SiC and ML diode have similar chip size, the increase in surface temperature is significantly higher in case of ML. As expected from the larger chip size and lower on-resistance of the LL size diode, the surface temperatures are significantly lower. Up to 5 A they are similar to the SiC-A sample, and even lower at higher output currents. A direct comparison of the temperatures of the B3 and B4 samples is not possible due to the different package structures and the fact that only the lead frame temperature was measured for the sealed batch B3 chips. However, the following consideration provides further details.

The junction temperature is determined by the junction-to-ambient thermal resistance $R_{\rm th,ja}$ and the actual power loss. To gain further insight, the temperature dependent knee voltages V_0 and the temperature dependent differential on-resistances r_0 are fitted with polynomial curves. By approximating the junction temperature as the measured chip surface temperature, this allows to estimate the conduction losses during buck converter operation according to

$$P_{\text{cond}}(T_{j}) = V_{0}(T_{j}) \cdot I_{\text{D,ave}} + r_{o}(T_{j}) \cdot I_{\text{D,rms}}^{2}, \qquad (6.1)$$

where $I_{D,ave}$ is the average and $I_{D,rms}$ is the RMS (root-mean-square) current. The respective measured current ripples (about 1.7 A at 100 kHz for all diodes) and duty cycles (about 50 % for all diodes) are used to calculate the RMS current. As shown in Section 6.3.2, the switching losses in the Schottky diodes are negligible. Furthermore, the power loss during blocking operation is roughly two to three orders of magnitude lower than the forward conduction losses. The diode power loss is therefore dominated by the conduction losses and can, in a simplified view, be approximated by them.

On this basis, Figs. 6.14a and 6.14b display the chip surface temperature as a function of the absolute and area-related (with respect to anode area) conduction losses, i.e., the absolute and the specific heating power. Clearly, the same trend as predicted in Sections 5.3.3 and 5.3.4 is observed: The B4-LL size diode shows a similar increase in junction temperature as the SiC-A reference diode, and a significantly lower temperature rise than the ML-size counterpart. Relating the power dissipation to the anode area, however, reveals a significant decrease in the power density capability of the LL-size diode. These considerations directly validate the results in Chapter 5.

To further quantify this aspect, the $\gamma_{\rm T}$ value (see Eq. 5.3) is roughly estimated for the actively operated buck converter. Note that the setup differs from the one in Section 5.3 in several respects. Amongst others, these are the less effective air cooling implemented here, the use of a different thermal interface material than in the previous measurements, and the fact that not the entire DPC substrate is coupled to the heatsink because of the required solder connection to the PCB (see Fig. 6.1b). Furthermore, as discussed in Section 6.3.3, the additional power loss in the transistor also affects the heat sink temperature close to the diode. The apparent $R_{\rm th,ja}$ determined from a linear fit to the data in Fig. 6.14 are therefore expected to be higher than the ones determined with grease in Figs. 5.9b and 5.9d. Indeed, the fit yields about



Figure 6.14: The power dissipation during buck converter operation with different output currents is estimated from the measured surface temperature, the ripple current, and the temperature dependent forward current–voltage profiles. This allows the extraction of the junction-to-ambient thermal resistance. Although increasing the chip size is (a) effective in decreasing the junction temperature, it (b) decreases the maximum possible power density significantly. This validates the result in Section 5.3.

9.6 K/W for Ga₂O₃ sample B4-ML and 6.5 K/W for SiC-A, which is about 3 K/W larger than the previously determined values. Using the apparent $R_{\text{th,ja}}$ values and Eq. 5.3, γ_{T} values of 0.68 and 0.76 are obtained for the B4-ML diode from Figs. 6.14a and 6.14b, respectively, which is slightly higher than the experimentally determined one of 0.61 in Section 5.3.5. This is reasonable since the $R_{\text{th,ja}}$ is larger in the buck converter setup, therefore decreasing the influence of the chip.

In Fig. 6.15a, an output current of 5 A is set for all diodes and the switching frequency is increased in steps of 50 kHz. The B4-ML Ga_2O_3 diode and the SiC-A reference diode exhibit a similar efficiency drop of about -0.6% per 100 kHz due to their similar junction capacitance and capacitive charge. The B4-ML failed after 4 min at 350 kHz and a maximum surface temperature of about 131 °C, which is the last data point shown in Fig. 6.15a. When the failure occurred, the diode had already been in operation for 3 h without interruption, at 400 V and between 100 kHz and 350 kHz.

Since the current undershoot during turn-off of diode B4-LL is enhanced as a result of the higher capacitive charge compared to sample B3-LL (see Fig. 6.12e), the corresponding turnon switching losses in the MOSFET increase. This leads to the highest efficiency drop of about 0.86% per 100 kHz for sample B4-LL. The effect of the additional charge on the MOSFET temperature is so pronounced that the switching frequency has been limited to 250 kHz: As shown in Fig. 6.15b, increasing the switching frequency from 100 kHz to 250 kHz leads to a rise of 7 K in the maximum chip surface temperature of the B4-LL Ga₂O₃ diode, while a 32 K increase in the maximum mold temperature of the MOSFET is observed — twice as much as with the B3-LL diode in the same frequency range. Figure 6.15c shows a thermal image of the rectangular ML-size diode during buck converter operation at 5 A and 100 kHz. The anode clearly constitutes the major heat source and the heat does not spread significantly into the non-active part of the die due to the low thermal conductivity of Ga₂O₃. This further validates the thermal simulations in Section 5.2.4.

The data demonstrate the successful introduction of $200 \,\mu\text{m}$ thin $600 \,\text{V}$ class Ga_2O_3 diodes and validate the thermal considerations in Section 5.3, but also show that the tradeoff between on-resistance and capacitive charge needs to be further improved. It should be noted that as a next step, reliability studies with a larger sample size should also be conducted once available.



Figure 6.15: Buck converter operation at an output current of 5 A with different switching frequencies. (a) When increasing the switching frequency, the drop in efficiency is higher for the batch B4 LL-type diode than for its batch B3 predecessor, due to the higher doping concentration in the epitaxial layer. (b) While the chip surface temperature of the LL-type diode increases by only 7 K for an increase from 100 kHz (top) to 250 kHz (bottom), the mold temperature of the MOSFET increases by 32 K. (c) In agreement with the simulations in Chapter 5, the heat does not spread significantly into the non-active part of rectangular ML-size Ga₂O₃ diodes. Parts adapted from [260].

6.5 Chapter conclusion

This chapter analyzes the switching properties of $600 \,\mu\text{m}$ thick TO247 packaged and new $200 \,\mu\text{m}$ thin Ga₂O₃ diodes mounted on ceramic substrates in a first $400 \,\text{V}$ to $200 \,\text{V}$ buck converter application. The main results can be summarized as follows:

- 1. The new chips enable load currents, DC link voltages and associated current and voltage slew rates significantly exceeding those in previous literature.
- 2. A stable buck converter operation for several hours at switching frequencies up to 350 kHz is observed, demonstrating the successful introduction of $200 \,\mu\text{m}$ thin wafer Ga_2O_3 technology despite the atypical current–voltage characteristics.
- 3. The failure of ML size diodes at high power densities provides a first insight into operating limits and shows the need for further studies with a larger sample size once available.
- 4. Despite the early stage of development and high on-resistance, power-loop efficiencies with a state-of-the-art Si diode can be exceeded.
- 5. Measurements of the chip temperature validate the considerations in Chapter 5 and show that increasing the chip size reduces the junction temperature but also limits the switching frequency and the power density. Therefore, it is not recommended as the main method of circumventing the low thermal conductivity of Ga_2O_3 .
- 6. It is vital for future device generations to reduce the on-resistance without increasing the capacitive charge. This will especially concern the optimization of electric field management methods, the thickness of the epitaxial layer, and its doping profile.

Chapter 7

Summary and outlook

 β -Ga₂O₃ is a semiconductor material that combines a wider bandgap than that of SiC or GaN with the availability of wafers fabricated by standard melt-growth techniques. Based on these material properties, it could help meet the growing demand for low-cost, high-efficiency power electronics. Motivated by this, the basic electrical, thermal, and switching properties of first non-commercial β -Ga₂O₃ diodes from a planned manufacturing line are studied in this work with the goal of building a bridge from fundamental device-level research to first applications with Ga₂O₃ devices in power electronics. The corresponding guiding questions formulated in Chapter 1 concern characteristic similarities and differences between Ga₂O₃ and SiC diodes, the assessment of the device quality in view of a potential application in power electronics, the feasibility of an implementation in fast-switching power converters despite the low thermal conductivity, and the identification of associated problems but also potential solutions.

First, a low-inductive discrete package based on an AlN ceramic substrate was designed. Bare Ga_2O_3 diode chips with unknown characteristics and SiC reference samples were assembled on these substrates in various ways to enable the required variety of different experiments. Studying the nanosecond short switching transients is an especially challenging task. Therefore, new current sensors are comparatively evaluated. A new magnetic sensor, referred to as IPM sensor, was selected to capture single switching transients. The resistive Mshunt was used for continuous measurements with longer DC periods. This preliminary work allows for a gradual answering of the initial guiding questions through combined experimental and simulative studies.

Several β -Ga₂O₃ diodes exhibit temperature-dependent forward and reverse current-voltage profiles similar to those of commercial SiC diodes. For multiple diodes, however, a transition between different conduction mechanisms is observed: both in forward and reverse direction, the conduction losses first decrease and then increase with rising junction temperature. This seems to be related to the manufacturing process of the dice and a possibly linked influence of trap states rather than intrinsic properties of Ga₂O₃ or packaging issues. The differences are intensified by a variation of material properties, such as the effective donor concentration of the epitaxial layer between different devices of the same type. Spice models that can replicate these varying characteristics in a simple but effective way are created to help design future power converters.

For all Ga_2O_3 diodes, the relative increase in differential on-resistance with rising temperature is lower than that of commercial SiC diodes, which can be traced back to the lower temperaturedependence of the electron mobility in Ga_2O_3 . In combination with a fairly homogeneous junction as indicated by the low change of ideality factor and Schottky barrier height with rising temperature, the Ga_2O_3 samples could, in principle, be suitable for high-temperature applications and allow for the conduct of more application-oriented tests. However, they still suffer from significantly higher specific on-resistances than their SiC counterparts, and the low thermal conductivity of Ga_2O_3 limits the power and current rating with the current 600 µm thick and cathode-side cooled standard devices significantly.

Driven by this concern, methods to increase the current and power rating of Ga_2O_3 diodes were studied. The combination of thermal measurements and simulations shows that the heat dissipation in Ga_2O_3 differs significantly from that in SiC. Although substrate thinning to 200 µm can reduce the junction temperature of Ga_2O_3 diodes effectively, cooling from the junction side is found to be substantially more effective despite the smaller cooling area. However, if not the entire anode area is precisely covered with die attach material, the resulting gaps can lead to a severe increase in the local peak temperature and must be avoided. Underfill materials can alleviate this issue. The results are combined with different scenarios regarding potential improvements of the growth conditions and material properties. This way it is shown that, despite the low thermal conductivity, thinned or junction-side cooled Ga_2O_3 diodes could achieve junction temperatures similar to those of commercial SiC counterparts at the same forward current as a result of their potentially lower conduction losses.

The newest generation of 600 μ m thick and 200 μ m thin Ga₂O₃ cathode-side cooled diodes was tested for the first time in a 400 V to 200 V buck converter with switching frequencies between 100 kHz and 350 kHz. Even 600 μ m thick cathode-side cooled samples can be implemented successfully despite the early stage of development and high thermal resistance. In agreement with the previous considerations, the rise in junction temperature is significantly stronger than that of a commercial SiC reference diode of similar size. Increasing the die size is effective in reducing the junction temperature only to a certain extent, as it also reduces the maximum specific power density and limits the switching frequency. However, despite the currently still high on-resistances, higher power-loop efficiencies than with a state-of-the-art bipolar Si diode can be achieved thanks to the purely unipolar device structure and the associated absence of recombination losses.

Further device-level research must focus on improving the tradeoff between on-resistance and capacitive charge. This especially concerns electric field management strategies to fully exploit the high theoretical breakdown field of Ga_2O_3 also in large-area devices. Although an improvement in the device quality is observed for the newest generation diodes, the variation in the basic device characteristics must be further reduced. Once a larger sample size is available, this is also the basis for reliability studies whose importance is underlined by the presumably temperature-induced failure of two Ga_2O_3 diodes during buck converter application. Apart from Ga_2O_3 as a material in itself, the packaging procedures may need to be reconsidered. This could also entail the development of new underfill materials with high thermal conductivity for junction-side cooled assemblies.

In summary, a device quality sufficient to enable application-oriented research has been achieved. The experimental and simulative results reveal certain limitations, but also point to possible solutions that can help pave the way to Ga_2O_3 power electronics.

A Overview of material properties of different semiconductors

Table A.1: Summary of important material properties of Si, 4H-SiC, wurzite GaN, and β -Ga₂O₃ at room temperature. Note that the given values vary slightly in literature. In case of Ga₂O₃, the properties additionally depend on the crystal direction due to its strong anisotropy. Electrical and thermal properties are taken from [12, 28, 71, 85], Vickers hardness H_v and Young's moduli E_{young} are taken from [105, 107, 108, 229, 230] for Si, 4H-SiC and GaN, and from [102, 231, 104] for β -Ga₂O₃.

| | Si | 4H-SiC | Bulk GaN | β –Ga $_2$ O $_3$ |
|--|------|--------|----------|-------------------------|
| $E_{\rm G}~({\rm eV})$ | 1.1 | 3.3 | 3.4 | 4.6–4.9 |
| $E_{\rm br}~({\rm MV/cm})$ | 0.3 | 2.5 | 3.3 | ${\sim}8$ |
| $\epsilon_{ m s}$ | 11.8 | 9.7 | 9.0 | 10-12 |
| $\mu_{\rm e} ({\rm cm}^2 ({\rm Vs})^{-1})$ | 1400 | 1000 | 1200 | 200 |
| $v_{\rm sat} (10^7 {\rm cm/s})$ | 1.0 | 2.0 | 2.5 | 1.5 |
| $\lambda_{\rm th} ({\rm W} ({\rm mK})^{-1})$ | 150 | 270 | 210 | 10-30 |
| melt-grown wafers? | yes | no | no | yes |
| $H_{\rm v}$ (GPa) | 9-12 | 30 | 10-11 | 6–13 |
| $E_{\rm young}$ (GPa) | 160 | 500 | 290 | 230 |

B List of measurement equipment

Table B.2: The following measurement equipment and tools have been used throughout the entire thesis in various chapters for temperature-dependent electrical characterization, transient thermal characterization, thermal imaging, computer tomography investigations, and laser ablation.

| Туре | Device Name | Purpose |
|--------------------------|----------------------|-------------------------------------|
| power device analyzer | Keysight B1505A | static electrical characterization |
| air heater | ATS-545 Thermostream | temperature control during |
| | | electrical characterization |
| transient thermal tester | Mentor T3Ster | impedance profiles and |
| | | structure functions |
| infrared camera | Optris PI640i | thermal imaging of chip temperature |
| computer tomography | GE Sensing v tome x | cross-sectional inspection |
| | microfocus CT | of dice and packages |
| laser | Trumpf TruMark | laser ablation/ |
| | Station 5000 | removal of mould mass |

Table B.3: The following measurement equipment was used for the dynamic electrical characterizationin Sections 3.4.2, 3.4.3 and 3.4.5. More details about the current sensors under investigationin this chapter are listed in Tab. 3.3.

| Туре | Device Name | Comment |
|---------------------|----------------------------|------------------------------------|
| oscilloscope | LeCroy Waverunner 104Xi | display waveforms |
| | | of active switch |
| oscilloscope | LeCroy Waverunner 44Xi | display waveforms |
| | | of passive switch and load current |
| voltage probe | LeCroy PP006-WR | drain-source voltage |
| voltage probe | LeCroy ADP033 | gate-source voltage |
| current probe | LeCroy CP031 | load current |
| high-voltage source | TopCon Quadro Power Supply | DC link voltage |
| low-voltage source | Toellner TOE 8952-60 | PCB power supply |
| pulse generator | Aim-TTi TGP3152 | gate driver PWM signal |

 Table B.4: The following measurement equipment was used for the dynamic electrical characterization in Section 3.4.6.

| Туре | Device Name | Comment |
|---------------------|----------------------|------------------------|
| oscilloscope | Tektronix MSO58 | display waveforms |
| voltage probe | Tektronix TIVP05 | diode voltage |
| current probe | Tektronix TCP0030A | load current |
| high-voltage source | EA PSI 91000-30 | DC link voltage |
| low-voltage source | Toellner TOE 8952-60 | PCB power supply |
| pulse generator | Aim-TTi TGP3152 | gate driver PWM signal |

Table B.5: The following measurement equipment was used for the dynamic electrical characterization in Chapter 6, i.e., for double pulse tests and buck converter application.

| Туре | Device Name | Comment |
|---------------------|----------------------------|--|
| oscilloscope | LeCroy Waverunner 104Xi | display diode waveforms |
| oscilloscope | LeCroy Waverunner 44Xi | display transistor waveforms |
| voltage probe | PMK PHV1000 | diode voltage |
| current probe | IPM sensor | diode current (double pulse test) |
| current probe | M-shunt | diode current (continuous operation) |
| voltage probe | LeCroy HVD3106A | transistor drain-source voltage |
| voltage probe | LeCroy HVD3106A | transistor gate-source |
| current probe | PEM CWTUM 015/B | transistor drain current |
| current probe | PEM CWTUM 6/B | transistor drain current |
| current probe | LeCroy CP031 | load current |
| data logger | Graphtec GL220E | log temperature with k-type thermocouple |
| high-voltage source | EA PSI 91000-30 | DC link voltage |
| electronic load | EA ELR 9750-66 | control converter output current |
| power analyzer | ZES Zimmer LMG 500-4 | efficiency measurements |
| low-voltage source | GW-Instek PSP-603 | PCB power supply |
| low-voltage source | Delta Elektronika ES 030-5 | fan power supply |
| pulse generator | Aim-TTi TGP3152 | gate driver PWM signal |

C List of used software

Table C.6: Overview of the most important software used for simulation and data analysis throughout this thesis.

| Software | Purpose |
|-----------------------------------|--|
| KiCad 5.1.6 | design of PCBs and discrete packages |
| PTC Creo 4 | construction of device models |
| Simcenter FloEFD 2020.2 | static and transient thermal simulation |
| Ansys Q3D 2020-R1 | simulation of parasitic loop inductance |
| LTSpice XVII | circuit simulation & device modelling |
| Mentor T3Ster-Master V.3.0.0.7109 | post processing of thermal transient results |
| Volume Graphics myVGL 34 | analysis of X-ray images |
| Matlab R2019a, Matlab R2022b | data analysis |
| OriginPro 2020 SR1 | data analysis |
List of Figures

| 2.1 | Buck converter and half-bridge topology. | 6 |
|------|--|----|
| 2.2 | Structure of a PiN and an MPS diode. | 7 |
| 2.3 | Basic structure of a power MOSFET and a GaN HEMT. | 8 |
| 2.4 | Bandstructure of a Schottky diode. | 10 |
| 2.5 | Critical electric field and figures of merit of Si, SiC, GaN, and β -Ga ₂ O ₃ | 12 |
| 2.6 | How the β -Ga ₂ O ₃ crystal structure affects thermal management | 15 |
| 2.7 | Promising vertical β -Ga ₂ O ₃ diode structures | 19 |
| 2.8 | Ga_2O_3 trench diode structures | 21 |
| 2.9 | Structure of β -Ga ₂ O ₃ transistors | 22 |
| 3.1 | Description of the custom ceramic substrate. | 26 |
| 3.2 | Simulated parasitic inductance of the custom package. | 27 |
| 3.3 | Evaluation of static electric properties. | 31 |
| 3.4 | Measurement uncertainties during transient measurements | 33 |
| 3.5 | Circuit schematics and setup to study novel current sensors | 35 |
| 3.6 | Sensor comparison: characteristic quantities measured with coaxial shunt, | |
| | SMD shunt, and M-shunt | 36 |
| 3.7 | Sensor comparison: characteristic quantities measured with coaxial shunt and | |
| | M-shunt | 37 |
| 3.8 | Sensor comparison: characteristic quantities measured with coaxial shunt, | |
| | Rogowski probe, and Infinity sensor. | 38 |
| 3.9 | Sensor comparison: characteristic quantities measured with coaxial shunt, IPM sensor, and IPM and Infinity sensors. | 40 |
| 3.10 | LTspice circuit and simulation results to verify the sensor measurements | 41 |
| 3.11 | Comparison of sensor characteristics: peak current and current slopes | 43 |
| 3.12 | Comparison of sensor characteristics: transient times and current slopes | 44 |
| 3.13 | Experimental setup and turn-off waveforms with SiC-A reference and IPM | |
| | sensor. | 45 |
| 3.14 | Transient thermal measurements: calibration, setup, and temperature response. | 47 |
| 3.15 | Evaluation of transient thermal properties. | 48 |
| 4.1 | Different types of forward current–voltage characteristics above the knee | |
| | voltage. | 52 |
| 4.2 | Forward characteristics below the knee voltage and TE model | 54 |
| 4.3 | Ideality factor, Schottky barrier height, and built-in potential of a trench and a | |
| | planar diode. | 55 |
| 4.4 | Reverse current–voltage characteristics of a trench MOS and a planar diode. | 58 |
| 4.5 | Breakdown characteristics of Ga_2O_3 Schottky diodes | 59 |
| 4.6 | Capacitance–voltage profiles and capacitive charge of a trench and a planar | |
| | diode | 61 |
| 4.7 | Variation in forward characteristics and on-resistance of four TO247 packaged | |
| | diodes from batch B2 | 62 |
| 4.8 | Variation in forward characteristics of TO247 packaged diodes from batch B3. | 63 |

| 4.9 4.10 | Variation of doping concentration for early and new device generations Increase in on-resistance with rising temperature for multiple Ga_2O_3 and SiC | • | 64 |
|-------------|---|-----|-----|
| 4 1 1 | | • | 66 |
| 4.11 | Thermal impedance profiles of a TO247 packaged Ga_2O_3 and a SiC diode. | • | 68 |
| 4.12 | Basic representation of the structure of a diode Spice model. | • | 69 |
| 4.13 | Simulation of Ga_2O_3 diodes with unusual characteristics in Spice | • | 70 |
| 5.1 | Structure, photograph, and X-ray image of the 200 µm thin TO247 packaged samples | | 74 |
| 5.2 | 200 µm thin TO247 packaged samples: temperature response, parasitic drift of the forward voltage, and time constant distribution | • | 76 |
| 5.3 | 200 µm thin TO247 packaged samples: thermal impedance profiles and struc- ture functions with two different thermal interface materials | • | 70 |
| 5.4 | $200 \mu\text{m}$ thin TO247 packaged samples: CT scans of the solder layers between | • | // |
| | die and CuMo spacer, and between CuMo spacer and lead frame | • | 78 |
| 5.5 | $200 \mu\mathrm{m}$ thin TO247 packaged samples: simulation model | • | 79 |
| 5.6 | $200\mu\mathrm{m}$ thin TO247 packaged samples: thermal impedance profiles and struc- | | |
| | ture functions simulated with a calibrated model. | • | 81 |
| 5.7 | $200\mu\mathrm{m}$ thin TO247 packaged samples: simulation of the heat spreading | • | 82 |
| 5.8 | Overview of samples assembled on DPC substrates for thermal studies | • | 83 |
| 5.9 | Ga ₂ O ₃ and SiC diodes on DPC substrates: structure functions and contribution | | |
| | of the different parts to the junction-to-ambient thermal resistance | • | 85 |
| 5.10 | Ga ₂ O ₃ and SiC diodes on DPC substrates: infrared images of the surface | | |
| | temperature at a heating power of 10 W | • | 86 |
| 5.11 | Ga ₂ O ₃ and SiC diodes on DPC substrates: junction temperature versus applied | | |
| | heating power and power density | • | 87 |
| 5.12 | Ga ₂ O ₃ and SiC diodes on DPC substrates: junction temperature versus applied | | |
| | current density and forward current-voltage profiles | • | 88 |
| 5.13 | Ga_2O_3 on DPC substrates: decreasing chip temperature by increasing the chip | | |
| | size | • | 90 |
| 5.14 | Ga_2O_3 and SiC diodes on DPC substrates: simulation model | • | 92 |
| 5.15 | Ga ₂ O ₃ and SiC diodes on DPC substrates: cross-sectional heat temperature | | |
| | distribution in for cathode-side and junction-side cooled Ga_2O_3 diodes | • | 93 |
| 5.16 | Ga ₂ O ₃ and SiC diodes on DPC substrates: simulated junction temperature | | |
| | versus power dissipation. | • | 94 |
| 5.17 | Ga ₂ O ₃ and SiC diodes on DPC substrates: influence of SiC die size, die-attach | | |
| | thickness, and substrate thickness on junction temperature in junction-side cooled setups. | | 96 |
| 5.18 | Ga_2O_3 and SiC diodes on DPC substrates: effect of underfill materials on the | | |
| | junction temperature in JSC assemblies with and without gap | | 97 |
| 5.19 | Ga ₂ O ₃ and SiC diodes on DPC substrates: How the on-resistance affects the | | |
| | assembly strategy. | • | 99 |
| 6.1 | Circuit schematics and photograph of the experimental setup used for double | | |
| | pulse tests and buck converter operation. | . 1 | 104 |
| 6.2 | Current-voltage characteristics and differential on-resistance of the TO- | | |
| | packaged diodes selected for buck converter operation. | . 1 | 106 |
| 6.3 | Junction capacitance, charge, and energy as well as charge \times resistance figure | | |
| | of merit of the TO-packaged diodes selected for buck converter operation | . 1 | 107 |
| | | | |

| 6.4 | Double pulse tests with all TO-packaged samples under investigation at 15 A and 400 V : current and voltage waveforms. | . 108 |
|------|---|-------|
| 6.5 | Double pulse tests and simulation showing the occurrence of two charging phases during diode turn-off. | . 109 |
| 6.6 | Double pulse tests with $600 \mu\text{m}$ thick Ga_2O_3 diodes and varying turn-on gate resistance. | . 110 |
| 6.7 | Influence of diode chip size on the switching waveforms, the power loss, and the switching energy in diode and MOSEET | 111 |
| 6.8 | Double pulse tests with $600 \mu\text{m}$ thick Ga_2O_3 diodes when varying the load | |
| 6.9 | current | . 112 |
| 6 10 | voltage | . 114 |
| 0.10 | sistor temperature, and power-loop efficiency when increasing the output current | 115 |
| 6.11 | Buck converter operation with TO-packaged diodes: power-loop efficiency, case temperature, and transistor temperature when increasing the switching | |
| 6.12 | frequency | . 116 |
| | on-resistance, capacitive charge, and double pulse tests | . 118 |
| 6.13 | Buck converter operation with 200 µm thin diodes on DPC substrates: effi- ciency and surface temperature for different output currents. | . 119 |
| 6.14 | Buck converter operation with $200\mu\text{m}$ thin diodes: surface temperature as a | |
| (15 | function of the absolute and specific power dissipation. | . 121 |
| 0.13 | increasing switching frequency and corresponding thermal images | . 122 |

List of Tables

| 3.1 | Overview of β -Ga ₂ O ₃ devices studied in this thesis | . 28 |
|-----|---|-------|
| 3.2 | Overview of Si and SiC reference devices. | . 29 |
| 3.3 | Overview of current sensors studied in this thesis | . 34 |
| 5.1 | $200 \mu\text{m}$ thin TO247 packaged samples: overview of on-resistance, K-factor, and thermal resistances | 77 |
| 5.2 | $200 \mu\text{m}$ thin TO247 packaged samples: overview of simulation parameters | . // |
| | used to study the transient thermal properties | . 79 |
| 5.3 | Overview of samples assembled on DPC substrates for thermal studies | . 83 |
| 5.4 | Ga ₂ O ₃ and SiC samples on DPC susbstrates: summary of experimental elec- | |
| | trical characteristics and $\gamma_{\rm T}$ values. | . 89 |
| 5.5 | Ga_2O_3 and SiC diodes on DPC substrates: simulation parameters | . 91 |
| 5.6 | Thermal simulation results for various Ga ₂ O ₃ samples with different substrate | |
| | thicknesses in CSC and ideal JSC assembly | . 94 |
| 5.7 | Material parameters used to calculate the theoretical on-resistance of Ga ₂ O ₃ | |
| | diodes in a realistic-optimistic, a moderate, and a pessimistic scenario | . 98 |
| A.1 | Summary of important material properties of Si, 4H-SiC, wurzite GaN, and | |
| | β -Ga ₂ O ₃ | . 125 |
| B.2 | Measurement equipment and tools used throughout the thesis | . 125 |
| B.3 | Measurement equipment used for testing different current sensors | . 126 |
| B.4 | Measurement equipment used to demonstrate fast switching with the DPC | |
| | package | . 126 |
| B.5 | Measurement equipment used to for the dynamic electrical characterization. | . 126 |
| C.6 | List of used software | . 127 |
| | | |

References

References

- [1] International Energy Agency. "Chapter 6 Outlook for electricity". In: *World Energy Outlook 2022*. 2022, pp. 277–323.
- J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "Power Semiconductor Devices—Key Components for Efficient Electrical Energy Conversion Systems". In: *Semiconductor Power Devices*. Cham, Switzerland: Springer, 2018, pp. 1–20. DOI: 10.1007/978-3-319-70917-8_1.
- [3] T. M. Jahns and B. Sarlioglu. "The incredible shrinking motor drive: accelerating the transition to integrated motor drives". In: *IEEE Power Electronics Magazine* 7.3 (2020), pp. 18–27. DOI: 10.1109/MPEL.2020.3011275.
- [4] C. D. Fuentes, S. Kouro, and S. Bernet. "Comparison of 1700-V SiC-MOSFET and Si-IGBT modules under identical test setup conditions". In: *IEEE Transactions on Industry Applications* 55.6 (2019), pp. 7765–7775. DOI: 10.1109/TIA.2019. 2934713.
- [5] X. She, A. Q. Huang, O. Lucia, and B. Ozpineci. "Review of silicon carbide power devices and their applications". In: *IEEE Transactions on Industrial Electronics* 64.10 (2017), pp. 8193–8205. DOI: 10.1109/TIE.2017.2652401.
- [6] R. Ding, Z. Dou, Y. Qi, W. Mei, and G. Liu. "Analysis on characteristic of 3.3-kV full SiC device and railway traction converter design". In: *IET Power Electronics* 15.10 (2022), pp. 978–988. DOI: 10.1049/pel2.12280.
- [7] S. Ozdemir, F. Acar, and U. S. Selamogullari. "Comparison of silicon carbide MOSFET and IGBT based electric vehicle traction inverters". In: *Proc. 5th Intl. Conf. on Electr. Engr. and Inform. (ICEEI).* Denpasar, Indonesia, Aug. 2015, pp. 1–4. DOI: 10.1109/ICEEI.2015.7387215.
- [8] S. B. Reese, T. Remo, J. Green, and A. Zakutayev. "How much will gallium oxide power electronics cost?" In: *Joule* 3.4 (2019), pp. 903–907. DOI: 10.1016/j. joule.2019.01.011.
- B. N. Pushpakaran, A. S. Subburaj, and S. B. Bayne. "Commercial GaN-based power electronic systems: a review". In: *Journal of Electronic Materials* 49.11 (2020), pp. 6247–6262. DOI: 10.1007/s11664-020-08397-z.
- K. Shirabe, M. M. Swamy, J.-K. Kang, M. Hisatsune, Y. Wu, D. Kebort, and J. Honea. "Efficiency comparison between Si-IGBT-based drive and GaN-based drive". In: *IEEE Transactions on Industry Applications* 50.1 (2014), pp. 566–572. DOI: 10.1109/TIA.2013.2290812.
- [11] R. Roy, V. G. Hill, and E. F. Osborn. "Polymorphism of Ga_2O_3 and the system Ga_2O_3 -H₂O". In: *Journal of the American Chemical Society* 74.3 (1952), pp. 719–722. DOI: 10.1021/ja01123a039.

- M. Higashiwaki. "Introduction". In: *Gallium Oxide*. Ed. by M. Higashiwaki and S. Fujita. Springer Series in Materials Science vol. 293. Cham, Siwtzerland: Springer, 2020, pp. 1–12. DOI: 10.1007/978-3-030-37153-1_1.
- [13] Z. Galazka. "Growth of bulk β -Ga₂O₃ single crystals by the Czochralski method". In: *Journal of Applied Physics* 131.3 (2022), Art. no. 031103. DOI: 10.1063/5.0076962.
- [14] Y. Tomm, P. Reiche, D. Klimm, and T. Fukuda. "Czochralski grown Ga₂O₃ crystals". In: *Journal of Crystal Growth* 220.4 (2000), pp. 510–514. DOI: 10.1016/S0022-0248 (00) 00851-4.
- [15] A. Kuramata, K. Koshi, S. Watanabe, Y. Yamaoka, T. Masui, and S. Yamakoshi. "High-quality β -Ga₂O₃ single crystals grown by edge-defined film-fed growth". In: *Japanese Journal of Applied Physics* 55.12 (2016), Art. no. 1202A2. DOI: 10. 7567/JJAP.55.1202A2.
- [16] K. Hoshikawa, E. Ohba, T. Kobayashi, J. Yanagisawa, C. Miyagawa, and Y. Nakamura.
 "Growth of β-Ga₂O₃ single crystals using vertical Bridgman method in ambient air". In: *Journal of Crystal Growth* 447 (2016), pp. 36–41. DOI: 10.1016/j.jcrysgro.2016.04.022.
- [17] N. Ueda, H. Hosono, R. Waseda, and H. Kawazoe. "Synthesis and control of conductivity of ultraviolet transmitting β -Ga₂O₃single crystals". In: *Applied Physics Letters* 70.26 (1997), pp. 3561–3563. DOI: 10.1063/1.119233.
- [18] A. J. Green, J. Speck, G. Xing, P. Moens, F. Allerstam, K. Gumaelius, T. Neyer, A. Arias-Purdue, V. Mehrotra, A. Kuramata, K. Sasaki, S. Watanabe, K. Koshi, J. Blevins, O. Bierwagen, S. Krishnamoorthy, K. Leedy, A. R. Arehart, A. T. Neal, S. Mou, S. A. Ringel, A. Kumar, A. Sharma, K. Ghosh, U. Singisetti, W. Li, K. Chabak, K. Liddy, A. Islam, S. Rajan, S. Graham, S. Choi, Z. Cheng, and M. Higashiwaki. "β-gallium oxide power electronics". In: *APL Materials* 10.2 (2022), Art. no. 029201. DOI: 10.1063/5.0060327.
- [19] T. Ilahi, T. Izhar, S. M. Qaisar, U. T. Shami, M. Zahid, A. Waqar, and A. Alzahrani. "Design and performance analysis of ultra-wide bandgap power devices-based EV fast charger using bi-directional power converters". In: *IEEE Access* 11 (2023), pp. 25285–25297. DOI: 10.1109/ACCESS.2023.3255780.
- [20] M. Higashiwaki, K. Sasaki, A. Kuramata, T. Masui, and S. Yamakoshi. "Gallium oxide (Ga₂O₃) metal-semiconductor field-effect transistors on single-crystal β -Ga₂O₃ (010) substrates". In: *Applied Physics Letters* 100.1 (2012), Art. no. 013504. DOI: 10.1063/1.3674287.
- [21] K. Sasaki, M. Higashiwaki, A. Kuramata, T. Masui, and S. Yamakoshi. "Ga₂O₃ Schottky barrier diodes fabricated by using single-crystal β -Ga₂O₃ (010) substrates". In: *IEEE Electron Device Letters* 34.4 (2013), pp. 493–495. DOI: 10.1109/LED. 2013.2244057.
- [22] P. Dong, J. Zhang, Q. Yan, Z. Liu, P. Ma, H. Zhou, and Y. Hao. "6 kV/3.4 mOhmcm² vertical β -Ga₂O₃ Schottky barrier diode with BV²/R_{on,sp} performance exceeding 1-d unipolar limit of GaN and SiC". In: *IEEE Electron Device Letters* 43.5 (2022), pp. 765–768. DOI: 10.1109/LED.2022.3160366.

- [23] J. Zhang, P. Dong, K. Dang, Y. Zhang, Q. Yan, H. Xiang, J. Su, Z. Liu, M. Si, J. Gao, M. Kong, H. Zhou, and Y. Hao. "Ultra-wide bandgap semiconductor Ga₂O₃ power diodes". In: *Nature communications* 13.1 (2022), Art. no. 3900. DOI: 10.1038/s41467-022-31664-y.
- [24] W. Li, K. Nomoto, Z. Hu, T. Nakamura, D. Jena, and H. G. Xing. "Single and multi-fin normally-off Ga₂O₃ vertical transistors with a breakdown voltage over 2.6 kV". In: *Proc. 65th Intl. Electron Devices Meeting (IEDM)*. San Francisco, USA, Nov. 2019, pp. 12.4.1–12.4.4. DOI: 10.1109/IEDM19573.2019.8993526.
- [25] A. Bhattacharyya, S. Roy, P. Ranga, C. Peterson, and S. Krishnamoorthy. "High-mobility tri-gate β -Ga₂O₃ MESFETs with a power figure of merit over 0.9 GW/cm²". In: *IEEE Electron Device Letters* 43.10 (2022), pp. 1637–1640. DOI: 10.1109/LED.2022.3196305.
- [26] S. Sharma, K. Zeng, S. Saha, and U. Singisetti. "Field-plated lateral Ga₂O₃ MOSFETs with polymer passivation and 8.03 kV breakdown voltage". In: *IEEE Electron Device Letters* 41.6 (2020), pp. 836–839. DOI: 10.1109/LED.2020.2991146.
- [27] N. Ueda, H. Hosono, R. Waseda, and H. Kawazoe. "Anisotropy of electrical and optical properties in β -Ga₂O₃ single crystals". In: *Applied Physics Letters* 71.7 (1997), pp. 933–935. DOI: 10.1063/1.119693.
- [28] H. He, R. Orlando, M. A. Blanco, R. Pandey, E. Amzallag, I. Baraille, and M. Rérat.
 "First-principles study of the structural, electronic, and optical properties of Ga₂O₃ in its monoclinic and hexagonal phases". In: *Physical Review B* 74.19 (2006), Art. no. 195123. DOI: 10.1103/PhysRevB.74.195123.
- [29] Z. Guo, A. Verma, X. Wu, F. Sun, A. Hickman, T. Masui, A. Kuramata, M. Higashiwaki, D. Jena, and T. Luo. "Anisotropic thermal conductivity in single crystal β -gallium oxide". In: *Applied Physics Letters* 106.11 (2015), Art. no. 111909. DOI: 10.1063/1.4916078.
- [30] I. Batarseh and A. Harb. "Non-isolated Switch Mode DC-DC Converters". In: *Power Electronics*. Cham, Switzerland: Springer, 2018, pp. 173–271. DOI: 10.1007/ 978-3-319-68366-9_4.
- [31] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "Packaging of Power Devices". In: *Semiconductor Power Devices*. Cham, Switzerland: Springer, 2018, pp. 427–488. DOI: 10.1007/978-3-319-70917-8_11.
- [32] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "pin Diodes". In: Semiconductor Power Devices. Cham, Switzerland: Springer, 2018, pp. 201–270. DOI: 10.1007/978-3-319-70917-8_5.
- [33] W. Shockley. "The theory of *p-n* junctions in semiconductors and *p-n* junction transistors". In: *Bell System Technical Journal* 28.3 (1949), pp. 435–489. DOI: 10.1002/j.1538-7305.1949.tb03645.x.
- [34] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "Schottky Diodes". In: *Semiconductor Power Devices*. Cham, Switzerland: Springer, 2018, pp. 271–293. DOI: 10.1007/978-3-319-70917-8_6.

- [35] R. Rupp, R. Gerlach, A. Kabakow, R. Schorner, C. Hecht, R. Elpelt, and M. Draghici.
 "Avalanche behaviour and its temperature dependence of commercial SiC MPS diodes: Influence of design and voltage class". In: *Proc. 26th Intl. Symp. Power Semicond. Dev. and IC's (ISPSD)*. Waikoloa, USA, 2014, pp. 67–70. DOI: 10.1109/ISPSD. 2014.6855977.
- [36] R. Singh, S. Ryu, J. W. Palmour, A. R. Hefner, and J. Lai. "1500 V, 4 amp 4H-SiC JBS diodes". In: *Proc. 12th Intl. Symp. Power Semicond. Dev. and IC's (ISPSD)*. Toulouse, France, 2000, pp. 101–104. DOI: 10.1109/ISPSD.2000.856782.
- [37] F. Bjoerk, J. Hancock, M. Treu, R. Rupp, and T. Reimann. "2nd generation 600V SiC Schottky diodes use merged pn/Schottky structure for surge overload protection". In: *Proc. 21st Appl. Power Electron. Conf. and Expo. (APEC)*. Dallas, USA, 2006, pp. 170–173. DOI: 10.1109/APEC.2006.1620534.
- [38] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "MOS Transistors and Field Controlled Wide Bandgap Devices". In: *Semiconductor Power Devices*. Cham, Switzerland: Springer, 2018, pp. 341–390. DOI: 10.1007/978-3-319-70917-8_9.
- [39] M. Meneghini, C. de Santi, I. Abid, M. Buffolo, M. Cioni, R. A. Khadar, L. Nela, N. Zagni, A. Chini, F. Medjdoub, G. Meneghesso, G. Verzellesi, E. Zanoni, and E. Matioli. "GaN-based power devices: physics, reliability, and perspectives". In: *Journal of Applied Physics* 130.18 (2021), Art. no. 181101. DOI: 10.1063/5.0061354.
- [40] T. Nakamura, Y. Nakano, M. Aketa, R. Nakamura, S. Mitani, H. Sakairi, and Y. Yokotsuji. "High performance SiC trench devices with ultra-low Ron". In: *Proc. 2011 Int. Electron Devices Meeting*. Washington, USA, Dec. 2011, pp. 26.5.1–26.5.3. DOI: 10.1109/IEDM.2011.6131619.
- [41] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "IGBTs". In: Semiconductor Power Devices. Cham, Switzerland: Springer, 2018, pp. 391–425. DOI: 10.1007/978-3-319-70917-8_10.
- [42] X. Huang, Q. Li, Z. Liu, and F. C. Lee. "Analytical loss model of high voltage GaN HEMT in cascode configuration". In: *IEEE Transactions on Power Electronics* 29.5 (2014), pp. 2208–2219. DOI: 10.1109/TPEL.2013.2267804.
- [43] W. Saito, Y. Takada, M. Kuraguchi, K. Tsuda, and I. Omura. "Recessed-gate structure approach toward normally off high-voltage AlGaN/GaN HEMT for power electronics applications". In: *IEEE Transactions on Electron Devices* 53.2 (2006), pp. 356–362. DOI: 10.1109/TED.2005.862708.
- [44] K. J. Chen, L. Yuan, M. J. Wang, H. Chen, S. Huang, Q. Zhou, C. Zhou, B. K. Li, and J. N. Wang. "Physics of fluorine plasma ion implantation for GaN normally-off HEMT technology". In: *Proc. IEEE Int. Electron Devices Meeting (IEDM)*. Washington, USA, Dec. 2012, pp. 19.4.1–19.4.4. DOI: 10.1109/IEDM.2011.6131585.
- [45] S. M. Sze and K. K. Ng. "Metal-Semiconductor Contacts". In: *Physics of Semiconductor Devices*. Ed. by S. M. Sze and K. K. Ng. Hoboken, USA: John Wiley & Sons, 2006, pp. 134–196. DOI: 10.1002/9780470068328.ch3.
- [46] K. P. Schoen, J. M. Woodall, J. A. Cooper, and M. R. Melloch. "Design considerations and experimental analysis of high-voltage SiC Schottky barrier rectifiers". In: *IEEE Transactions on Electron Devices* 45.7 (1998), pp. 1595–1604. DOI: 10.1109/16. 701494.

- Y.-H. Wang, Y.-M. Zhang, Y.-M. Zhang, Q.-W. Song, and R.-X. Jia. "Al/Ti/4H-SiC Schottky barrier diodes with inhomogeneous barrier heights". In: *Chinese Physics B* 20.8 (2011), Art. no. 087305. DOI: 10.1088/1674-1056/20/8/087305.
- [48] K. Ueno, K. Shibahara, A. Kobayashi, and H. Fujioka. "Vertical p-type GaN Schottky barrier diodes with nearly ideal thermionic emission characteristics". In: *Applied Physics Letters* 118.2 (2021), Art. no. 022102. DOI: 10.1063/5.0036093.
- [49] S. M. Sze. "MESFET and related devices". In: *Semiconductor Devices Physics and technology*. 2nd Ed. New York, USA: John Wiley & Sons, 2002, pp. 224–253.
- [50] T. Hatakeyama and T. Shinohe. "Reverse characteristics of a 4H-SiC Schottky barrier diode". In: *Materials Science Forum* 389-393 (2002), pp. 1169–1172. DOI: 10. 4028/www.scientific.net/msf.389-393.1169.
- [51] J. Suda, K. Yamaji, Y. Hayashi, T. Kimoto, K. Shimoyama, H. Namita, and S. Nagao. "Nearly ideal current-voltage characteristics of Schottky barrier diodes formed on hydride-vapor-phase-epitaxy-grown GaN free-standing substrates". In: *Applied Physics Express* 3.10 (2010), Art. no. 101003. DOI: 10.1143/APEX.3.101003.
- [52] M. Higashiwaki, K. Konishi, K. Sasaki, K. Goto, K. Nomura, Q. T. Thieu, R. Togashi, H. Murakami, Y. Kumagai, B. Monemar, A. Koukitu, A. Kuramata, and S. Yamakoshi. "Temperature-dependent capacitance–voltage and current–voltage characteristics of Pt/Ga₂O₃ (001) Schottky barrier diodes fabricated on *n*⁻–Ga₂O₃ drift layers grown by halide vapor phase epitaxy". In: *Applied Physics Letters* 108.13 (2016), Art. no. 133503. DOI: 10.1063/1.4945267.
- [53] S. M. Sze and K. K. Ng. "Metal-Insulator-Semiconductor Capacitors". In: *Physics of Semiconductor Devices*. Ed. by S. M. Sze and K. K. Ng. Hoboken, USA: John Wiley & Sons, 2006, pp. 197–240.
- [54] J. Frenkel. "On pre-breakdown phenomena in insulators and electronic semiconductors". In: *Physical Review* 54.8 (1938), pp. 647–648. DOI: 10.1103/ PhysRev.54.647.
- [55] M. P. Houng, Y. H. Wang, and W. J. Chang. "Current transport mechanism in trapped oxides: a generalized trap-assisted tunneling model". In: *Journal of Applied Physics* 86.3 (1999), pp. 1488–1491. DOI: 10.1063/1.370918.
- [56] K. Fu, H. Fu, X. Huang, T.-H. Yang, C.-Y. Cheng, P. R. Peri, H. Chen, J. Montes, C. Yang, J. Zhou, X. Deng, X. Qi, D. J. Smith, S. M. Goodnick, and Y. Zhao. "Reverse leakage analysis for as-grown and regrown vertical GaN-on-GaN Schottky barrier diodes". In: *IEEE Journal of the Electron Devices Society* 8 (2020), pp. 74–83. DOI: 10.1109/JEDS.2020.2963902.
- [57] D. H. Mudiyanselage, D. Wang, and H. Fu. "Ultrawide bandgap vertical β –(Al_xGa_{1-x})₂O₃ Schottky barrier diodes on free-standing β –Ga₂O₃ substrates". In: *Journal of Vacuum Science & Technology A* 41.2 (2023), Art. no. 023201. DOI: 10.1116/6.0002265.
- [58] J. L. Hudgins, G. S. Simin, E. Santi, and M. A. Khan. "An assessment of wide bandgap semiconductors for power devices". In: *IEEE Transactions on Power Electronics* 18.3 (2003), pp. 907–914. DOI: 10.1109/TPEL.2003.810840.
- [59] N. Kaminski and O. Hilt. "SiC and GaN devices wide bandgap is not all the same". In: *IET Circuits, Devices & Systems* 8.3 (2014), pp. 227–236. DOI: 10.1049/iet-cds.2013.0223.

- [60] F. L. L. Nouketcha, Y. Cui, A. Lelis, R. Green, C. Darmody, J. Schuster, and N. Goldsman. "Investigation of wide- and ultrawide-bandgap semiconductors from impact-ionization coefficients". In: *IEEE Transactions on Electron Devices* 67.10 (2020), pp. 3999–4005. DOI: 10.1109/TED.2020.3009622.
- [61] N. Ma, N. Tanen, A. Verma, Z. Guo, T. Luo, H. Xing, and D. Jena. "Intrinsic electron mobility limits in β -Ga₂O₃". In: *Applied Physics Letters* 109.21 (2016), Art. no. 212101. DOI: 10.1063/1.4968550.
- [62] S. Geller. "Crystal structure of β -Ga₂O₃". In: *The Journal of Chemical Physics* 33.3 (1960), pp. 676–684. DOI: 10.1063/1.1731237.
- [63] M. Marezio and J. P. Remeika. "Bond lengths in the α -Ga₂O₃ structure and the highpressure phase of Ga_{2-x}Fe_xO₃". In: *The Journal of Chemical Physics* 46.5 (1967), pp. 1862–1865. DOI: 10.1063/1.1840945.
- [64] L. Ghadbeigi, J. Cooke, G. T. Dang, T. Kawaharamura, T. Yasuoka, R. Sun, P. Ranga, S. Krishnamoorthy, M. A. Scarpulla, and B. Sensale-Rodriguez. "Optical characterization of gallium oxide α and β polymorph thin-films grown on c-plane Sapphire". In: *Journal of Electronic Materials* 50.6 (2021), pp. 2990–2998. DOI: 10.1007/s11664-021-08809-8.
- [65] G. Yang, P. R. Romeo, A. Apostoluk, and B. Vilquin. "First principles study on the lattice thermal conductivity of α -phase Ga₂O₃". In: *Journal of Vacuum Science & Technology A* 40.5 (2022), Art. no. 052801. DOI: 10.1116/6.0001870.
- [66] E. Chikoidze, H. J. von Bardeleben, K. Akaiwa, E. Shigematsu, K. Kaneko, S. Fujita, and Y. Dumont. "Electrical, optical, and magnetic properties of Sn doped α -Ga₂O₃ thin films". In: *Journal of Applied Physics* 120.2 (2016), Art. no. 025109. DOI: 10.1063/1.4958860.
- [67] S. Fujita and K. Kaneko. "Epitaxial growth of corundum-structured wide band gap III-oxide semiconductor thin films". In: *Journal of Crystal Growth* 401 (2014), pp. 588–592. DOI: 10.1016/J.JCRYSGRO.2014.02.032.
- [68] M. Oda, R. Tokuda, H. Kambara, T. Tanikawa, T. Sasaki, and T. Hitora. "Schottky barrier diodes of corundum-structured gallium oxide showing on-resistance of $0.1 \text{ m}\Omega\text{cm}^2$ grown by MIST EPITAXY". In: *Applied Physics Express* 9.2 (2016), Art. no. 021101. DOI: 10.7567/APEX.9.021101.
- [69] Y. J. Jeong, J.-H. Park, M. J. Yeom, I. Kang, J. Y. Yang, H.-Y. Kim, D.-W. Jeon, and G. Yoo. "Heteroepitaxial α -Ga₂O₃ MOSFETs with a 2.3 kV breakdown voltage grown by halide vapor-phase epitaxy". In: *Applied Physics Express* 15.7 (2022), Art. no. 074001. DOI: 10.35848/1882-0786/ac7431.
- [70] J. E. Medvedeva and C. L. Hettiarachchi. "Tuning the properties of complex transparent conducting oxides: role of crystal symmetry, chemical composition, and carrier generation". In: *Physical Review B* 81.12 (2010), Art. no. 125116. DOI: 10.1103/PhysRevB.81.125116.
- [71] C. Janowitz, V. Scherer, M. Mohamed, A. Krapf, H. Dwelk, R. Manzke, Z. Galazka, R. Uecker, K. Irmscher, R. Fornari, M. Michling, D. Schmeißer, J. R. Weber, J. B. Varley, and C. G. van de Walle. "Experimental electronic structure of In₂O₃ and Ga₂O₃". In: *New Journal of Physics* 13.8 (2011), Art. no. 085014. DOI: 10.1088/1367-2630/13/8/085014.

- [72] M. Mohamed, C. Janowitz, I. Unger, R. Manzke, Z. Galazka, R. Uecker, R. Fornari, J. R. Weber, J. B. Varley, and C. G. van de Walle. "The electronic structure of β -Ga₂O₃". In: *Applied Physics Letters* 97.21 (2010), Art. no. 211903. DOI: 10. 1063/1.3521255.
- [73] M. Orita, H. Ohta, M. Hirano, and H. Hosono. "Deep-ultraviolet transparent conductive β -Ga₂O₃ thin films". In: *Applied Physics Letters* 77.25 (2000), pp. 4166–4168. DOI: 10.1063/1.1330559.
- [74] T. Matsumoto, M. Aoki, A. Kinoshita, and T. Aono. "Absorption and reflection of vapor grown single crystal platelets of β -Ga₂O₃". In: *Japanese Journal of Applied Physics* 13.10 (1974), pp. 1578–1582. DOI: 10.1143/jjap.13.1578.
- [75] A. F. M. Anhar Uddin Bhuiyan, Z. Feng, J. M. Johnson, H.-L. Huang, J. Hwang, and H. Zhao. "MOCVD epitaxy of ultrawide bandgap β -(Al_xGa_{1-x})₂O₃ with high-Al composition on (100) β -Ga₂O₃ substrates". In: *Crystal Growth & Design* 20.10 (2020), pp. 6722–6730. DOI: 10.1021/acs.cgd.0c00864.
- [76] J. B. Varley, A. Janotti, C. Franchini, and C. G. van de Walle. "Role of self-trapping in luminescence and p-type conductivity of wide-band-gap oxides". In: *Physical Review B* 85.8 (2012). DOI: 10.1103/PhysRevB.85.081109.
- [77] A. T. Neal, S. Mou, S. Rafique, H. Zhao, E. Ahmadi, J. S. Speck, K. T. Stevens, J. D. Blevins, D. B. Thomson, N. Moser, K. D. Chabak, and G. H. Jessen. "Donors and deep acceptors in β -Ga₂O₃". In: *Applied Physics Letters* 113.6 (2018), Art. no. 062101. DOI: 10.1063/1.5034474.
- [78] K. Goto, K. Konishi, H. Murakami, Y. Kumagai, B. Monemar, M. Higashiwaki, A. Kuramata, and S. Yamakoshi. "Halide vapor phase epitaxy of Si doped β -Ga₂O₃ and its electrical properties". In: *Thin Solid Films* 666 (2018), pp. 182–184. DOI: 10.1016/j.tsf.2018.09.006.
- [79] E. Ahmadi, O. S. Koksaldi, S. W. Kaun, Y. Oshima, D. B. Short, U. K. Mishra, and J. S. Speck. "Ge doping of β -Ga₂O₃ films grown by plasma-assisted molecular beam epitaxy". In: *Applied Physics Express* 10.4 (2017), Art. no. 041102. DOI: 10.7567/APEX.10.041102.
- [80] K. Sasaki, A. Kuramata, T. Masui, E. G. Víllora, K. Shimamura, and S. Yamakoshi. "Device-quality β -Ga₂O₃ epitaxial films fabricated by ozone molecular beam epitaxy". In: *Applied Physics Express* 5.3 (2012), Art. no. 035502. DOI: 10.1143/APEX.5. 035502.
- [81] E. G. Víllora, K. Shimamura, Y. Yoshikawa, T. Ujiie, and K. Aoki. "Electrical conductivity and carrier concentration control in β -Ga₂O₃by Si doping". In: *Applied Physics Letters* 92.20 (2008), Art. no. 202120. DOI: 10.1063/1.2919728.
- [82] Z. Feng, A. F. M. Anhar Uddin Bhuiyan, M. R. Karim, and H. Zhao. "MOCVD homoepitaxy of Si-doped (010) β -Ga₂O₃ thin films with superior transport properties". In: *Applied Physics Letters* 114.25 (2019), Art. no. 250601. DOI: 10.1063/1. 5109678.
- [83] Y. Zhang, A. Neal, Z. Xia, C. Joishi, J. M. Johnson, Y. Zheng, S. Bajaj, M. Brenner, D. Dorsey, K. Chabak, G. Jessen, J. Hwang, S. Mou, J. P. Heremans, and S. Rajan. "Demonstration of high mobility and quantum transport in modulation-doped β-(Al_xGa_{1-x})₂O₃/Ga₂O₃ heterostructures". In: *Applied Physics Letters* 112.17 (2018), Art. no. 173502. DOI: 10.1063/1.5025704.

- [84] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "Semiconductor Properties". In: *Semiconductor Power Devices*. Cham, Switzerland: Springer, 2018, pp. 21–99. DOI: 10.1007/978-3-319-70917-8_2.
- [85] K. Ghosh and U. Singisetti. "Ab initio velocity-field curves in monoclinic β -Ga₂O₃". In: *Journal of Applied Physics* 122.3 (2017), Art. no. 035702. DOI: 10.1063/1. 4986174.
- [86] I. A. Khan and J. A. Cooper. "Measurement of high-field electron transport in silicon carbide". In: *IEEE Transactions on Electron Devices* 47.2 (2000), pp. 269–273. DOI: 10.1109/16.822266.
- [87] Z. Galazka, K. Irmscher, R. Uecker, R. Bertram, M. Pietsch, A. Kwasniewski, M. Naumann, T. Schulz, R. Schewski, D. Klimm, and M. Bickermann. "On the bulk β -Ga₂O₃ single crystals grown by the Czochralski method". In: *Journal of Crystal Growth* 404 (2014), pp. 184–191. DOI: 10.1016/j.jcrysgro.2014.07.021.
- [88] M. Slomski, N. Blumenschein, P. P. Paskov, J. F. Muth, and T. Paskova. "Anisotropic thermal conductivity of β -Ga₂O₃ at elevated temperatures: effect of Sn and Fe dopants". In: *Journal of Applied Physics* 121.23 (2017), Art. no. 235104. DOI: 10.1063/1.4986478.
- [89] M. Handwerg, R. Mitdank, Z. Galazka, and S. F. Fischer. "Temperature-dependent thermal conductivity and diffusivity of a Mg-doped insulating β -Ga₂O₃ single crystal along [100], [010] and [001]". In: *Semiconductor Science and Technology* 31.12 (2016), Art. no. 125006. DOI: 10.1088/0268-1242/31/12/125006.
- [90] J. Munshi, A. Roy, S. Hansen, C. E. Ekuma, and G. Balasubramanian. "Effect of vacancy defects on the thermal transport of β -Ga₂O₃". In: *Molecular Simulation* 47.12 (2021), pp. 1017–1021. DOI: 10.1080/08927022.2021.1941943.
- [91] S. J. Pearton, F. Ren, M. Tadjer, and J. Kim. "Perspective: Ga₂O₃ for ultra-high power rectifiers and MOSFETS". In: *Journal of Applied Physics* 124.22 (2018), Art. no. 220901. DOI: 10.1063/1.5062841.
- [92] D. Vaca, M. Barry, L. Yates, N. Nepal, D. S. Katzer, B. P. Downey, V. Wheeler, L. Nyakiti, D. J. Meyer, S. Graham, and S. Kumar. "Measurements and numerical calculations of thermal conductivity to evaluate the quality of β -gallium oxide thin films grown on sapphire and silicon carbide by molecular beam epitaxy". In: *Applied Physics Letters* 121.4 (2022), Art. no. 042107. DOI: 10.1063/5.0098048.
- [93] M. A. Mastro, C. R. Eddy, M. J. Tadjer, J. K. Hite, J. Kim, and S. J. Pearton. "Assessment of the (010) β -Ga₂O₃ surface and substrate specification". In: *Journal* of Vacuum Science & Technology A 39.1 (2021), Art. no. 013408. DOI: 10.1116/ 6.0000725.
- [94] B. Chatterjee, A. Jayawardena, E. Heller, D. W. Snyder, S. Dhar, and S. Choi. "Thermal characterization of gallium oxide Schottky barrier diodes". In: *The Review of scientific instruments* 89.11 (2018), Art. no. 114903. DOI: 10.1063/1.5053621.
- [95] W. Xu, T. You, Y. Wang, Z. Shen, K. Liu, L. Zhang, H. Sun, R. Qian, Z. An, F. Mu, T. Suga, G. Han, X. Ou, Y. Hao, and X. Wang. "Efficient thermal dissipation in wafer-scale heterogeneous integration of single-crystalline β–Ga₂O₃ thin film on SiC". In: *Fundamental Research* 1.6 (2021), pp. 691–696. DOI: 10.1016/j.fmre.2021.11.003.

- [96] H. Gong, F. Zhou, X. Yu, W. Xu, F.-f. Ren, S. Gu, H. Lu, J. Ye, and R. Zhang. "70-μm-body Ga₂O₃ Schottky barrier diode with 1.48 K/W thermal resistance, 59 A surge current and 98.9% conversion efficiency". In: *IEEE Electron Device Letters* 43.5 (2022), pp. 773–776. DOI: 10.1109/LED.2022.3162393.
- [97] B. Wang, M. Xiao, J. Knoll, C. Buttay, K. Sasaki, G.-Q. Lu, C. Dimarino, and Y. Zhang. "Low thermal resistance (0.5 K/W) Ga₂O₃ Schottky rectifiers with double-side packaging". In: *IEEE Electron Device Letters* 42.8 (2021), pp. 1132–1135. DOI: 10.1109/LED.2021.3089035.
- [98] F. Orlandi, F. Mezzadri, G. Calestani, F. Boschi, and R. Fornari. "Thermal expansion coefficients of β -Ga₂O₃single crystals". In: *Applied Physics Express* 8.11 (2015), Art. no. 111101. DOI: 10.7567/APEX.8.111101.
- [99] E. G. Víllora, K. Shimamura, T. Ujiie, and K. Aoki. "Electrical conductivity and lattice expansion of β -Ga₂O₃below room temperature". In: *Applied Physics Letters* 92.20 (2008), Art. no. 202118. DOI: 10.1063/1.2910770.
- [100] M. Leszczynski, T. Suski, H. Teisseyre, P. Perlin, I. Grzegory, J. Jun, S. Porowski, and T. D. Moustakas. "Thermal expansion of gallium nitride". In: *Journal of Applied Physics* 76.8 (1994), pp. 4909–4911. DOI: 10.1063/1.357273.
- [101] M. Nakabayashi, T. Fujimoto, M. Katsuno, and N. Ohtani. "Precise determination of thermal expansion coefficients observed in 4H-SiC single crystals". In: *Materials Science Forum* 527-529 (2006), pp. 699–702. DOI: 10.4028/www.scientific. net/MSF.527-529.699.
- [102] W. Mu, Z. Jia, Y. Yin, Q. Hu, Y. Li, B. Wu, J. Zhang, and X. Tao. "High quality crystal growth and anisotropic physical characterization of β -Ga₂O₃ single crystals grown by EFG method". In: *Journal of Alloys and Compounds* 714 (2017), pp. 453–458. DOI: 10.1016/j.jallcom.2017.04.185.
- [103] V. I. Nikolaev, V. Maslov, S. I. Stepanov, A. I. Pechnikov, V. Krymov, I. P. Nikitina, L. I. Guzilova, V. E. Bougrov, and A. E. Romanov. "Growth and characterization of β–Ga₂O₃ crystals". In: *Journal of Crystal Growth* 457 (2017), pp. 132–136. DOI: 10.1016/j.jcrysgro.2016.05.049.
- [104] H. Yamaguchi, S. Watanabe, Y. Yamaoka, K. Koshi, and A. Kuramata. "Mechanical properties and dislocation dynamics in β -Ga₂O₃". In: *Japanese Journal of Applied Physics* 61.4 (2022), Art. no. 045506. DOI: 10.35848/1347-4065/ac5adb.
- [105] K. Yuan, Y. Sun, Y. Lu, X. Liang, D. Tian, X. Ma, and D. Yang. "Comparison on mechanical properties of heavily phosphorus- and arsenic-doped Czochralski silicon wafers". In: *AIP Advances* 8.4 (2018), Art. no. 045301. DOI: 10.1063/1. 5025516.
- [106] X.-J. Ning, N. Huvey, and P. Pirouz. "Dislocation cores and hardness polarity of 4H-SiC". In: *Journal of the American Ceramic Society* 80.7 (1997), pp. 1645–1652. DOI: 10.1111/j.1151-2916.1997.tb03033.x.
- [107] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "Reliability and Reliability Testing". In: *Semiconductor Power Devices*. Cham, Switzerland: Springer, 2018, pp. 489–582. DOI: 10.1007/978-3-319-70917-8_12.

- [108] T. Matsumoto, S. I. Nishizawa, and S. Yamasaki. "Calculation of lattice constant of 4H-SiC as a function of impurity concentration". In: *Materials Science Forum* 645-648 (2010), pp. 247–250. DOI: 10.4028/www.scientific.net/MSF.645– 648.247.
- [109] P. J. Wellmann. "Review of SiC crystal growth technology". In: Semiconductor Science and Technology 33.10 (2018), Art. no. 103001. DOI: 10.1088/1361-6641/aad831.
- [110] Z. Galazka, R. Uecker, D. Klimm, K. Irmscher, M. Naumann, M. Pietsch, A. Kwasniewski, R. Bertram, S. Ganschow, and M. Bickermann. "Scaling-up of bulk β -Ga₂O₃ single crystals by the Czochralski method". In: *ECS Journal of Solid State Science and Technology* 6.2 (2017), Q3007–Q3011. DOI: 10.1149/2.0021702jss.
- [111] A. Kuramata, K. Koshi, S. Watanabe, Y. Yamaoka, T. Masui, and S. Yamakoshi. "Bulk crystal growth of Ga₂O₃". In: *Proc. Oxide-based Materials and Devices IX (SPIE 10533)*. San Francisco, USA, 2018, pp. 1–13. DOI: 10.1117/12.2301405.
- [112] K. Hoshikawa, T. Kobayashi, Y. Matsuki, and E. Ohba. "2-inch diameter (100) β -Ga₂O₃ crystal growth by the vertical Bridgman technique in a resistance heating furnace in ambient air". In: *Journal of Crystal Growth* 545 (2020), Art. no. 125724. DOI: 10.1016/j.jcrysgro.2020.125724.
- [113] E. G. Víllora, K. Shimamura, Y. Yoshikawa, K. Aoki, and N. Ichinose. "Large-size β -Ga₂O₃single crystals and wafers". In: *Journal of Crystal Growth* 270.3-4 (2004), pp. 420–426. DOI: 10.1016/j.jcrysgro.2004.06.027.
- [114] H. Murakami, K. Nomura, K. Goto, K. Sasaki, K. Kawara, Q. T. Thieu, R. Togashi, Y. Kumagai, M. Higashiwaki, A. Kuramata, S. Yamakoshi, B. Monemar, and A. Koukitu. "Homoepitaxial growth of β -Ga₂O₃ layers by halide vapor phase epitaxy". In: *Applied Physics Express* 8.1 (2015), Art. no. 015503. DOI: 10.7567/APEX.8.015503.
- [115] G. Wagner, M. Baldini, D. Gogova, M. Schmidbauer, R. Schewski, M. Albrecht, Z. Galazka, D. Klimm, and R. Fornari. "Homoepitaxial growth of β -Ga₂O₃ layers by metal-organic vapor phase epitaxy". In: *physica status solidi (a)* 211.1 (2014), pp. 27–33. DOI: 10.1002/PSSA.201330092.
- [116] S. Rafique, L. Han, M. J. Tadjer, J. A. Freitas, N. A. Mahadik, and H. Zhao. "Homoepitaxial growth of β -Ga₂O₃ thin films by low pressure chemical vapor deposition". In: *Applied Physics Letters* 108.18 (2016), Art. no. 182105. DOI: 10.1063/1. 4948944.
- [117] K. D. Leedy, K. D. Chabak, V. Vasilyev, D. C. Look, J. J. Boeckl, J. L. Brown, S. E. Tetlak, A. J. Green, N. A. Moser, A. Crespo, D. B. Thomson, R. C. Fitch, J. P. McCandless, and G. H. Jessen. "Highly conductive homoepitaxial Si-doped Ga₂O₃ films on (010) β -Ga₂O₃ by pulsed laser deposition". In: *Applied Physics Letters* 111.1 (2017), Art. no. 012103. DOI: 10.1063/1.4991363.
- [118] S. Mobtakeri, Y. Akaltun, A. Özer, M. Kılıç, E. Ş. Tüzemen, and E. Gür. "Gallium oxide films deposition by RF magnetron sputtering; a detailed analysis on the effects of deposition pressure and sputtering power and annealing". In: *Ceramics International* 47.2 (2021), pp. 1721–1727. DOI: 10.1016/j.ceramint.2020.08.289.
- [119] S.-d. Lee, K. Kaneko, and S. Fujita. "Homoepitaxial growth of beta gallium oxide films by mist chemical vapor deposition". In: *Japanese Journal of Applied Physics* 55.12 (2016), Art. no. 1202B8. DOI: 10.7567/JJAP.55.1202B8.

- [120] S. J. Pearton, J. Yang, P. H. Cary, F. Ren, J. Kim, M. J. Tadjer, and M. A. Mastro. "A review of Ga₂O₃ materials, processing, and devices". In: *Applied Physics Reviews* 5.1 (2018), Art. no. 011301. DOI: 10.1063/1.5006941.
- [121] H. Okumura. "Growth of double-barrier β -(Al_xGa_{1-x})₂O₃/Ga₂O₃ structure and heavily Sn-doped Ga₂O₃ layers using molecular-beam epitaxy". In: *Japanese Journal of Applied Physics* 59.7 (2020), Art. no. 075503. DOI: 10.35848/1347-4065/ ab9a8b.
- [122] F. Alema, B. Hertog, A. Osinsky, P. Mukhopadhyay, M. Toporkov, and W. V. Schoenfeld. "Fast growth rate of epitaxial β–Ga₂O₃ by close coupled showerhead MOCVD". In: *Journal of Crystal Growth* 475 (2017), pp. 77–82. DOI: 10.1016/j.jcrysgro. 2017.06.001.
- [123] J. H. Leach, K. Udwary, J. Rumsey, G. Dodson, H. Splawn, and K. R. Evans. "Halide vapor phase epitaxial growth of β -Ga₂O₃ and α -Ga₂O₃ films". In: *APL Materials* 7.2 (2019), Art. no. 022504. DOI: 10.1063/1.5055680.
- [124] V. I. Nikolaev, A. I. Pechnikov, S. I. Stepanov, I. P. Nikitina, A. N. Smirnov, A. V. Chikiryaka, S. S. Sharofidinov, V. E. Bougrov, and A. E. Romanov. "Epitaxial growth of (-201) β -Ga₂O₃ on (0001) sapphire substrates by halide vapour phase epitaxy". In: *Materials Science in Semiconductor Processing* 47 (2016), pp. 16–19. DOI: 10.1016/j.mssp.2016.02.008.
- [125] E. Ahmadi and Y. Oshima. "Materials issues and devices of α and β -Ga₂O₃". In: Journal of Applied Physics 126.16 (2019), Art. no. 160901. DOI: 10.1063/1. 5123213.
- [126] K. Konishi, K. Goto, R. Togashi, H. Murakami, M. Higashiwaki, A. Kuramata, S. Yamakoshi, B. Monemar, and Y. Kumagai. "Comparison of O₂ and H₂O as oxygen source for homoepitaxial growth of β–Ga₂O₃ layers by halide vapor phase epitaxy". In: *Journal of Crystal Growth* 492 (2018), pp. 39–44. DOI: 10.1016/j.jcrysgro. 2018.04.009.
- [127] J. B. Varley, J. R. Weber, A. Janotti, and C. G. van de Walle. "Oxygen vacancies and donor impurities in β -Ga₂O₃". In: *Applied Physics Letters* 97.14 (2010), Art. no. 142106. DOI: 10.1063/1.3499306.
- [128] A. M. Cowley and S. M. Sze. "Surface states and barrier height of metal-semiconductor systems". In: *Journal of Applied Physics* 36.10 (1965), pp. 3212–3220. DOI: 10.1063/1.1702952.
- [129] Y. Yao, R. F. Davis, and L. M. Porter. "Investigation of different metals as ohmic contacts to β -Ga₂O₃: comparison and analysis of electrical behavior, morphology, and other physical properties". In: *Journal of Electronic Materials* 46.4 (2017), pp. 2053–2060. DOI: 10.1007/s11664-016-5121-1.
- [130] Y. Yao, R. Gangireddy, J. Kim, K. K. Das, R. F. Davis, and L. M. Porter. "Electrical behavior of β–Ga₂O₃ Schottky diodes with different Schottky metals". In: *Journal of Vacuum Science & Technology B, Nanotechnology and Microelectronics: Materials, Processing, Measurement, and Phenomena* 35.3 (2017), Art. no. 03D113. DOI: 10.1116/1.4980042.
- [131] N. Allen, M. Xiao, X. Yan, K. Sasaki, M. J. Tadjer, J. Ma, R. Zhang, H. Wang, and Y. Zhang. "Vertical Ga₂O₃ Schottky barrier diodes with small-angle beveled field plates: a Baliga's figure-of-merit of 0.6 GW/cm²". In: *IEEE Electron Device Letters* 40.9 (2019), pp. 1399–1402. DOI: 10.1109/LED.2019.2931697.

- [132] H. H. Gong, X. X. Yu, Y. Xu, X. H. Chen, Y. Kuang, Y. J. Lv, Y. Yang, F.-F. Ren, Z. H. Feng, S. L. Gu, Y. D. Zheng, R. Zhang, and J. D. Ye. " β –Ga₂O₃ vertical heterojunction barrier Schottky diodes terminated with p-NiO field limiting rings". In: *Applied Physics Letters* 118.20 (2021), Art. no. 202102. DOI: 10.1063/5.0050919.
- [133] F. Zhou, H. H. Gong, Z. P. Wang, W. Z. Xu, X. X. Yu, Y. Yang, F.-F. Ren, S. L. Gu, R. Zhang, Y. D. Zheng, H. Lu, and J. D. Ye. "Over 1.8 GW/cm² beveledmesa NiO/ β -Ga₂O₃ heterojunction diode with 800 V/10 A nanosecond switching capability". In: *Applied Physics Letters* 119.26 (2021), Art. no. 262103. DOI: 10.1063/5.0071280.
- [134] X. Ji, C. Lu, Z. Yan, L. Shan, X. Yan, J. Wang, J. Yue, X. Qi, Z. Liu, W. Tang, and P. Li. "A review of gallium oxide-based power Schottky barrier diodes". In: *Journal of Physics D: Applied Physics* 55.44 (2022), Art. no. 443002. DOI: 10.1088/1361-6463/ac855c.
- [135] W. Li, Z. Hu, K. Nomoto, Z. Zhang, J.-Y. Hsu, Q. T. Thieu, K. Sasaki, A. Kuramata, D. Jena, and H. G. Xing. "1230 V β -Ga₂O₃ trench Schottky barrier diodes with an ultra-low leakage current of <1 μ A/cm²". In: *Applied Physics Letters* 113.20 (2018), Art. no. 202101. DOI: 10.1063/1.5052368.
- [136] A. Bhattacharyya, S. Roy, P. Ranga, D. Shoemaker, Y. Song, J. S. Lundh, S. Choi, and S. Krishnamoorthy. "130 mA mm⁻¹ β -Ga₂O₃ metal semiconductor field effect transistor with low-temperature metalorganic vapor phase epitaxy-regrown ohmic contacts". In: *Applied Physics Express* 14.7 (2021), Art. no. 076502. DOI: 10. 35848/1882-0786/ac07ef.
- [137] M. Mohamed, K. Irmscher, C. Janowitz, Z. Galazka, R. Manzke, and R. Fornari. "Schottky barrier height of Au on the transparent semiconducting oxide β -Ga₂O₃". In: *Applied Physics Letters* 101.13 (2012), Art. no. 132106. DOI: 10.1063/1. 4755770.
- [138] Z. Hu, Q. Feng, Z. Feng, Y. Cai, Y. Shen, G. Yan, X. Lu, C. Zhang, H. Zhou, J. Zhang, and Y. Hao. "Experimental and theoretical studies of Mo/Au Schottky contact on mechanically exfoliated β -Ga₂O₃ thin film". In: *Nanoscale research letters* 14.1 (2019), Art. no. 2. DOI: 10.1186/s11671-018-2837-2.
- [139] Z. Hu, H. Zhou, Q. Feng, J. Zhang, C. Zhang, K. Dang, Y. Cai, Z. Feng, Y. Gao, X. Kang, and Y. Hao. "Field-plated lateral β -Ga₂O₃ Schottky barrier diode with high reverse blocking voltage of more than 3 kV and high DC power figure-of-merit of 500 MW/cm²". In: *IEEE Electron Device Letters* 39.10 (2018), pp. 1564–1567. DOI: 10.1109/LED.2018.2868444.
- [140] M. Higashiwaki, A. Kuramata, H. Murakami, and Y. Kumagai. "State-of-the-art technologies of gallium oxide power devices". In: *Journal of Physics D: Applied Physics* 50.33 (2017), Art. no. 333002. DOI: 10.1088/1361-6463/aa7aff.
- [141] Q. He, W. Mu, H. Dong, S. Long, Z. Jia, H. Lv, Q. Liu, M. Tang, X. Tao, and M. Liu. "Schottky barrier diode based on β -Ga₂O₃ (100) single crystal substrate and its temperature-dependent electrical characteristics". In: *Applied Physics Letters* 110.9 (2017), Art. no. 093503. DOI: 10.1063/1.4977766.
- [142] X. Lu, X. Zhang, H. Jiang, X. Zou, K. M. Lau, and G. Wang. "Vertical β -Ga₂O₃ Schottky barrier diodes with enhanced breakdown voltage and high switching performance". In: *physica status solidi* (*a*) 211 (2019), Art. no. 1900497. DOI: 10.1002/pssa.201900497.

- [143] B. Wang, M. Xiao, X. Yan, H. Y. Wong, J. Ma, K. Sasaki, H. Wang, and Y. Zhang.
 "High-voltage vertical Ga₂O₃ power rectifiers operational at high temperatures up to 600 K". In: *Applied Physics Letters* 115.26 (2019), Art. no. 263503. DOI: 10.1063/1.5132818.
- [144] Q. He, W. Hao, X. Zhou, Y. Li, K. Zhou, C. Chen, W. Xiong, G. Jian, G. Xu, X. Zhao, X. Wu, J. Zhu, and S. Long. "Over 1 GW/cm² vertical Ga₂O₃ Schottky barrier diodes without edge termination". In: *IEEE Electron Device Letters* 43.2 (2022), pp. 264–267. DOI: 10.1109/LED.2021.3133866.
- [145] M. Xiao, B. Wang, J. Liu, R. Zhang, Z. Zhang, C. Ding, S. Lu, K. Sasaki, G.-Q. Lu, C. Buttay, and Y. Zhang. "Packaged Ga₂O₃ Schottky rectifiers with over 60 A surge current capability". In: *IEEE Transactions on Power Electronics* 36.8 (2021), pp. 8565–8569. DOI: 10.1109/TPEL.2021.3049966.
- [146] Y. Wang, S. Cai, M. Liu, Y. Lv, S. Long, X. Zhou, X. Song, S. Liang, T. Han, X. Tan, and Z. Feng. "High-voltage (201) β -Ga₂O₃ vertical Schottky barrier diode with thermally-oxidized termination". In: *IEEE Electron Device Letters* 41.1 (2020), pp. 131–134. DOI: 10.1109/LED.2019.2956016.
- [147] J. Yang, S. Ahn, F. Ren, S. J. Pearton, S. Jang, J. Kim, and A. Kuramata. "High reverse breakdown voltage Schottky rectifiers without edge termination on Ga₂O₃". In: *Applied Physics Letters* 110.19 (2017), Art. no. 192101. DOI: 10.1063/1.4983203.
- [148] J. Yang, S. Ahn, F. Ren, S. J. Pearton, S. Jang, and A. Kuramata. "High breakdown voltage (-201) β–Ga₂O₃ Schottky rectifiers". In: *IEEE Electron Device Letters* 38.7 (2017), pp. 906–909. DOI: 10.1109/LED.2017.2703609.
- [149] C.-H. Lin, Y. Yuda, M. H. Wong, M. Sato, N. Takekawa, K. Konishi, T. Watahiki, M. Yamamuka, H. Murakami, Y. Kumagai, and M. Higashiwaki. "Vertical Ga₂O₃ Schottky barrier diodes with guard ring formed by nitrogen-ion implantation". In: *IEEE Electron Device Letters* 40.9 (2019), pp. 1487–1490. DOI: 10.1109/LED. 2019.2927790.
- [150] K. Konishi, K. Goto, H. Murakami, Y. Kumagai, A. Kuramata, S. Yamakoshi, and M. Higashiwaki. "1-kV vertical Ga₂O₃ field-plated Schottky barrier diodes". In: *Applied Physics Letters* 110.10 (2017), Art. no. 103506. DOI: 10.1063/1.4977857.
- [151] R. Sharma, M. Xian, M. E. Law, M. Tadjer, F. Ren, and S. J. Pearton. "Design and implementation of floating field ring edge termination on vertical geometry β –Ga₂O₃ rectifiers". In: *Journal of Vacuum Science & Technology A* 38.6 (2020), Art. no. 063414. DOI: 10.1116/6.0000693.
- [152] W. Li, K. Nomoto, Z. Hu, D. Jena, and H. G. Xing. "Field-plated Ga₂O₃ trench Schottky barrier diodes with a BV²/R_{on,sp} of up to 0.95 GW/cm²". In: *IEEE Electron Device Letters* 41.1 (2020), pp. 107–110. DOI: 10.1109/led.2019.2953559.
- [153] H. Zhou, Q. Feng, J. Ning, C. Zhang, P. Ma, Y. Hao, Q. Yan, J. Zhang, Y. Lv, Z. Liu, Y. Zhang, K. Dang, P. Dong, and Z. Feng. "High-performance vertical β -Ga₂O₃ Schottky barrier diode with implanted edge termination". In: *IEEE Electron Device Letters* 40.11 (2019), pp. 1788–1791. DOI: 10.1109/LED.2019.2939788.
- [154] S. Roy, A. Bhattacharyya, and S. Krishnamoorthy. "Design of a β -Ga₂O₃ Schottky barrier diode with p-type III-Nitride guard ring for enhanced breakdown". In: *IEEE Transactions on Electron Devices* 67.11 (2020), pp. 4842–4848. DOI: 10.1109/TED.2020.3025268.

- [155] F. Zhou, H. Gong, W. Xu, X. Yu, Y. Xu, Y. Yang, F.-f. Ren, S. Gu, Y. Zheng, R. Zhang, J. Ye, and H. Lu. "1.95-kV beveled-mesa NiO/β–Ga₂O₃ heterojunction diode with 98.5% conversion efficiency and over million-times overvoltage ruggedness". In: *IEEE Transactions on Power Electronics* 37.2 (2022), pp. 1223–1227. DOI: 10.1109/TPEL.2021.3108780.
- [156] J. Yang, F. Ren, Y.-T. Chen, Y.-T. Liao, C.-W. Chang, J. Lin, M. J. Tadjer, S. J. Pearton, and A. Kuramata. "Dynamic switching characteristics of 1 A forward current β -Ga₂O₃ rectifiers". In: *IEEE Journal of the Electron Devices Society* 7 (2019), pp. 57–61. DOI: 10.1109/JEDS.2018.2877495.
- [157] J. Yang, C. Fares, F. Ren, Y.-T. Chen, Y.-T. Liao, C.-W. Chang, J. Lin, M. Tadjer, D. J. Smith, S. J. Pearton, and A. Kuramata. "Switching behavior and forward bias degradation of 700V, 0.2A, β–Ga₂O₃ vertical geometry rectifiers". In: *ECS Journal of Solid State Science and Technology* 8.7 (2019), pp. Q3028–Q3033. DOI: 10.1149/2.0061907jss.
- [158] Y. Wei, X. Luo, Y. Wang, J. Lu, Z. Jiang, J. Wei, Y. Lv, and Z. Feng. "Experimental study on static and dynamic characteristics of Ga₂O₃ Schottky barrier diodes with compound termination". In: *IEEE Transactions on Power Electronics* 36.10 (2021), pp. 10976–10980. DOI: 10.1109/TPEL.2021.3069918.
- [159] K. Sasaki, D. Wakimoto, Q. T. Thieu, Y. Koishikawa, A. Kuramata, M. Higashiwaki, and S. Yamakoshi. "First demonstration of Ga₂O₃ trench MOS-type Schottky barrier diodes". In: *IEEE Electron Device Letters* 38.6 (2017), pp. 783–785. DOI: 10.1109/LED.2017.2696986.
- [160] W. Li, K. Nomoto, Z. Hu, D. Jena, and H. G. Xing. "Guiding principles for trench Schottky barrier diodes based on ultrawide bandgap semiconductors: a case study in Ga₂O₃". In: *IEEE Transactions on Electron Devices* 67.10 (2020), pp. 3938–3947. DOI: 10.1109/TED.2020.3003292.
- [161] F. Otsuka, H. Miyamoto, A. Takatsuka, S. Kunori, K. Sasaki, and A. Kuramata. "Largesize $(1.7 \times 1.7 \text{ mm}^2) \beta$ –Ga₂O₃ field-plated trench MOS-type Schottky barrier diodes with 1.2 kV breakdown voltage and 10⁹ high on/off current ratio". In: *Applied Physics Express* 15.1 (2022), Art. no. 016501. DOI: 10.35848/1882-0786/ac4080.
- [162] J. A. Appels and H. Vaes. "High voltage thin layer devices (RESURF devices)". In: *Proc. 25th Intl. Electron Devices Meeting (IEDM)*. Washington, USA, 1979, pp. 238–241. DOI: 10.1109/IEDM.1979.189589.
- [163] W. Li, K. Nomoto, Z. Hu, D. Jena, and H. G. Xing. "Fin-channel orientation dependence of forward conduction in kV-class Ga₂O₃ trench Schottky barrier diodes". In: *Applied Physics Express* 12.6 (2019), Art. no. 061007. DOI: 10.7567/1882-0786/ab206c.
- [164] N. K. Kalarickal, Z. Feng, A. F. M. Anhar Uddin Bhuiyan, Z. Xia, W. Moore, J. F. McGlone, A. R. Arehart, S. A. Ringel, H. Zhao, and S. Rajan. "Electrostatic engineering using extreme permittivity materials for ultra-wide bandgap semiconductor transistors". In: *IEEE Transactions on Electron Devices* 68.1 (2021), pp. 29–35. DOI: 10.1109/TED.2020.3037271.
- [165] Z. Jian, S. Mohanty, and E. Ahmadi. "Temperature-dependent current-voltage characteristics of β -Ga₂O₃ trench Schottky barrier diodes". In: *Applied Physics Letters* 116.15 (2020), Art. no. 152104. DOI: 10.1063/5.0002520.

- [166] W. Tang, X. Zhang, T. He, Y. Ma, B. Feng, X. Wei, G. He, S. Zhang, X. Huo, Y. Cai, S. Ding, X. Zhang, and B. Zhang. "Temperature-dependent electrical characteristics of β -Ga₂O₃ trench Schottky barrier diodes via self-reactive etching". In: *Journal of Physics D: Applied Physics* 54.42 (2021), Art. no. 425104. DOI: 10.1088/1361-6463/ac1290.
- [167] H.-H. Wan, J.-S. Li, C.-C. Chiang, X. Xia, F. Ren, H. N. Masten, J. S. Lundh, J. A. Spencer, F. Alema, A. Osinsky, A. G. Jacobs, K. Hobart, M. J. Tadjer, and S. J. Pearton. "NiO/β-(Al_xGa_{1-x})₂O₃/Ga₂O₃ heterojunction lateral rectifiers with reverse breakdown voltage >7 kV". In: *Journal of Vacuum Science & Technology A* 41.3 (2023), Art. no. 032701. DOI: 10.1116/6.0002393.
- [168] S. Roy, A. Bhattacharyya, C. Peterson, and S. Krishnamoorthy. " β -Ga₂O₃ lateral high-permittivity dielectric superjunction Schottky barrier diode with 1.34 GW/cm² power figure of merit". In: *IEEE Electron Device Letters* 43.12 (2022), pp. 2037–2040. DOI: 10.1109/LED.2022.3216302.
- [169] M. H. Wong, H. Murakami, Y. Kumagai, and M. Higashiwaki. "Enhancement-mode β -Ga₂O₃ current aperture vertical MOSFETs with N-ion-implanted blocker". In: *IEEE Electron Device Letters* 41.2 (2020), pp. 296–299. DOI: 10.1109/LED. 2019.2962657.
- [170] Z. Hu, K. Nomoto, W. Li, Z. Zhang, N. Tanen, Q. T. Thieu, K. Sasaki, A. Kuramata, T. Nakamura, D. Jena, and H. G. Xing. "Breakdown mechanism in 1 kA/cm² and 960 V E-mode β -Ga₂O₃ vertical transistors". In: *Applied Physics Letters* 113.12 (2018), Art. no. 122103. DOI: 10.1063/1.5038105.
- [171] Z. Hu, K. Nomoto, W. Li, N. Tanen, K. Sasaki, A. Kuramata, T. Nakamura, D. Jena, and H. G. Xing. "Enhancement-mode Ga₂O₃ vertical transistors with breakdown voltage >1 kV". In: *IEEE Electron Device Letters* 39.6 (2018), pp. 869–872. DOI: 10.1109/LED.2018.2830184.
- [172] Y. Ma, X. Zhou, W. Tang, X. Zhang, G. Xu, L. Zhang, T. Chen, S. Dai, C. Bian, B. Li, Z. Zeng, and S. Long. "702.3 Acm⁻²/10.4 mΩcm² β–Ga₂O₃ U-shape trench gate MOSFET with N-ion implantation". In: *IEEE Electron Device Letters* 44.3 (2023), pp. 384–387. DOI: 10.1109/LED.2023.3235777.
- [173] J. Böcker, K. Tetzner, S. Heucke, O. Hilt, E. Bahat-Treidel, S. Dieckerhoff, and J. Würfl. "Dispersion effects in on-state resistance of lateral Ga_2O_3 MOSFETs at 300 V switching". In: *Electronics Letters* 56.16 (2020), pp. 838–840. DOI: 10.1049/el.2020.1286.
- [174] C. Kuring, K. Tetzner, A. Popp, S. Heucke, O. Hilt, S. B. Anooz, J. Wurfl, and S. Dieckerhoff. "Switching behavior and dynamic on-resistance of lateral β -Ga₂O₃ MOSFETs up to 400 V". In: *Proc. IEEE 8th Workshop on Wide Bandgap Power Dev. and Appl. (WiPDA*). Redondo Beach, USA, Nov. 2021, pp. 52–57. DOI: 10.1109/WiPDA49284.2021.9645079.
- [175] N. Chasserio, S. Guillemet-Fritsch, T. Lebey, and S. Dagdag. "Ceramic substrates for high-temperature electronic integration". In: *Journal of Electronic Materials* 38.1 (2009), pp. 164–174. DOI: 10.1007/s11664-008-0571-8.
- [176] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "Appendix D: Thermal Parameters of Important Materials in Packaging Technology". In: *Semiconductor Power Devices*. Cham, Switzerland: Springer, 2018, p. 703. DOI: 10.1007/978-3-319-70917-8.

- [177] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "Appendix E: Electric Parameters of Important Materials in Packaging Technology". In: *Semiconductor Power Devices*. Cham, Switzerland: Springer, 2018, p. 705. DOI: 10.1007/978-3-319-70917-8.
- [178] M. C. Caponet, F. Profumo, R. W. de Doncker, and A. Tenconi. "Low stray inductance bus bar design and construction for good EMC performance in power electronic circuits". In: *IEEE Transactions on Power Electronics* 17.2 (2002), pp. 225–231. DOI: 10.1109/63.988833.
- [179] G. Engelmann. "Reducing device stress and switching losses using active gate drivers and improved switching cell design". Ph.D. Thesis. Aachen, Germany: RWTH Aachen University, 2018.
- [180] T. Liu, T. T. Y. Wong, and Z. J. Shen. "A new characterization technique for extracting parasitic inductances of SiC power MOSFETs in discrete and module packages based on two-port S-parameters measurement". In: *IEEE Transactions on Power Electronics* 33.11 (2018), pp. 9819–9833. DOI: 10.1109/TPEL.2017.2789240.
- [181] P. T. Vianco. "An overview of surface finishes and their role in printed circuit board solderability and solder joint performance". In: *Circuit World* 25.1 (1999), pp. 6–24. DOI: 10.1108/03056129910244518.
- [182] R. N. Lumley, T. B. Sercombe, and G. B. Schaffer. "Surface oxide and the role of magnesium during the sintering of aluminum". In: *Metallurgical and Materials Transactions A* 30.10 (1999), pp. 457–463. DOI: 10.1007/s11661-999-0335y.
- [183] S. M. Sze. "MOSFET and related devices". In: *Semiconductor Devices Physics and technology*. 2nd Ed. New York, USA: John Wiley & Sons, 2002, pp. 224–253.
- [184] A. Fiedler, R. Schewski, Z. Galazka, and K. Irmscher. "Static dielectric constant of β -Ga₂O₃ perpendicular to the principal planes (100), (010), and (001)". In: *ECS Journal of Solid State Science and Technology* 8.7 (2019), pp. Q3083–Q3085. DOI: 10.1149/2.0201907jss.
- [185] K. Schon. "Übertragungsverhalten linearer Systeme, Faltung und Entfaltung". In: *Hochspannungsmesstechnik*. Wiesbaden, Germany: Springer, 2016, pp. 275–315.
 DOI: 10.1007/978-3-658-15178-2_9.
- [186] S. Sandler. "Faster-switching GaN: presenting a number of interesting measurement challenges". In: *IEEE Power Electronics Magazine* 2.2 (2015), pp. 24–31. DOI: 10.1109/MPEL.2015.2420232.
- [187] M. Adelmund, Bödeker Christian, and N. Kaminski. "Optimisation of shunt resistors for fast transients". In: Proc. Intl. Exhib. and Conf. for Power Electron., Intell. Mot., Renew. Energy & Energy Mgmt. (PCIM Europe). Nuremberg, Germany, 2016, pp. 1232–1239.
- [188] W. Zhang, Z. Zhang, and F. Wang. "Review and bandwidth measurement of coaxial shunt resistors for wide-bandgap devices dynamic characterization". In: *Proc. 11th Annu. Energy Conv. Cong. and Expo. (ECCE)*. Baltimore, USA, March 2019, pp. 3259–3264. DOI: 10.1109/ECCE.2019.8912750.

- [189] H. Lutzen, J. Müller, V. Polezhaev, T. Huesgen, and N. Kaminski. "Reducing the impact of Skin effect induced measurement errors in M-shunts by deliberate field coupling". In: *Proc. 24th Europ. Conf. Power Electron. Appl. (EPE ECCE)*. Hanover, Germany, 2022, pp. 1–11.
- [190] T. Wickramasinghe, B. Allard, C. Buttay, C. Joubert, C. Martin, J.-F. Mogniotte, H. Morel, P. Bevilacqua, T.-L. Le, and S. Azzopardi. "A study on shunt resistor-based current measurements for fast switching GaN devices". In: *Proc. 45th Annu. Conf. of the IEEE Indust. Electron. Soc. (IECON).* Lisbon, Portugal, 2019, pp. 1573–1578. DOI: 10.1109/IECON.2019.8927490.
- [191] A. Sewergin, A. H. Wienhausen, S. Blasius, and R. W. de Doncker. "Control hardware and suitable current sensors for fast switching SiC DC-DC converters". In: Proc. Intl. Exhib. and Conf. for Power Electron., Intell. Mot., Renew. Energy & Energy Mgmt. (PCIM Europe). Nuremberg, Germany, 2019, pp. 2012–2016.
- [192] H. Lutzen, K. Mitsui, D. Silber, K. Wada, and N. Kaminski. "Optimisation and proof of concept studies for the M-shunt structure applied to printed circuit boards". In: *Proc. 11th Int. Conf. Integr. Power Electron. Syst. (CIPS).* Berlin, Germany, 2020, pp. 114–119.
- [193] H. Lutzen, V. Polezhaev, K. B. Rawal, M. K. Ahmmed, T. Huesgen, and N. Kaminski. "Temperature compensated M-shunts for fast transient and low inductive current measurements". In: *Proc. 12th Intl. Conf. Integr. Power Electron. Syst. (CIPS)*. Berlin, Germany, 2022, pp. 493–499.
- [194] J. Wang, M. H. Hedayati, D. Liu, S.-E. Adami, H. C. P. Dymond, J. J. O. Dalton, and B. H. Stark. "Infinity sensor: temperature sensing in GaN power devices using peak di/dt". In: *Proc. 10th Annu. Energy Convers. Cong. and Expo. (ECCE)*. Portland, USA, 2018, pp. 884–890. DOI: 10.1109/ECCE.2018.8558287.
- [195] Semiconductor devices—discrete devices part 9: insulated-gate bipolar transistors (IGBTs). IEC 60747-9:2007. Geneva, CH: International Electrotechnical Comission (IEC), Sept. 1, 2007.
- [196] J. Lautner and B. Piepenbreier. "Analysis of GaN HEMT switching behavior". In: *Proc. 9th Intl. Conf. on Power Electron. (ICPE-ECCE Asia).* Seoul, Korea, 2015, pp. 567–574. DOI: 10.1109/ICPE.2015.7167840.
- [197] Transient Dual Interface Test Method for the Measurement of the Thermal Resistance Junction-to-Case of Semiconductor Devices with Heat Flow Through a Single Path. JESD51-14. USA: JEDEC Solid State Technology Association, Nov. 1, 2010.
- [198] U. Scheuermann and R. Schmidt. "Investigations on the V_{CE}(T)-method to determine the junction temperature by using the chip itself as sensor". In: *Proc. Intl. Exhib.* and Conf. for Power Electron., Intell. Mot., Renew. Energy & Energy Mgmt. (PCIM Europe). Nuremberg, Germany, 2009, pp. 802–807.
- [199] V. Székely and T. van Bien. "Fine structure of heat flow path in semiconductor devices: A measurement and identification method". In: *Solid-State Electronics* 31.9 (1988), pp. 1363–1368. DOI: 10.1016/0038–1101 (88) 90099–8.
- [200] V. Szekely. "Identification of RC networks by deconvolution: chances and limits". In: *IEEE Transactions on Circuits and Systems I: Fundamental Theory and Applications* 45.3 (1998), pp. 244–258. DOI: 10.1109/81.662698.

- [201] D. Schweitzer, H. Pape, and L. Chen. "Transient measurement of the junction-to-case thermal resistance using structure functions: chances and limits". In: *Proc. 24th Annu. IEEE Semiond. Therm. Meas. and Mngmt. Symp. (STHERM).* San Jose, USA, 2008, pp. 191–197. DOI: 10.1109/STHERM.2008.4509389.
- [202] M. Adamowicz, S. Giziewski, J. Pietryka, and Z. Krzeminski. "Performance comparison of SiC Schottky diodes and silicon ultra fast recovery diodes". In: *Proc. 7th Intl. Conf.-Workshop Compat. and Power Electron. (CPE).* Tallinn, Estonia, 2011, pp. 144–149. DOI: 10.1109/CPE.2011.5942222.
- [203] T. R. McNutt, A. R. Hefner, H. A. Mantooth, J. Duliere, D. W. Berning, and R. Singh. "Silicon carbide PiN and merged PiN Schottky power diode models implemented in the Saber circuit simulator". In: *IEEE Transactions on Power Electronics* 19.3 (2004), pp. 573–581. DOI: 10.1109/TPEL.2004.826420.
- [204] F. A. Padovani and R. Stratton. "Field and thermionic-field emission in Schottky barriers". In: *Solid-State Electronics* 9.7 (1966), pp. 695–707. DOI: 10.1016/ 0038-1101(66)90097-9.
- [205] J. H. Werner and H. H. Güttler. "Barrier inhomogeneities at Schottky contacts". In: Journal of Applied Physics 69.3 (1991), pp. 1522–1533. DOI: 10.1063/1. 347243.
- [206] W. Mönch. "Barrier heights of real Schottky contacts explained by metal-induced gap states and lateral inhomogeneities". In: *Journal of Vacuum Science & Technology B: Microelectronics and Nanometer Structures* 17.4 (1999), pp. 1867–1876. DOI: 10.1116/1.590839.
- [207] M. Grundmann. "Diodes". In: *The Physics of Semiconductors*. Ed. by M. Grundmann. Graduate Texts in Physics. Cham: Springer International Publishing, 2016, pp. 583–667. DOI: 10.1007/978-3-319-23880-7_21.
- [208] M. Rebien, W. Henrion, M. Hong, J. P. Mannaerts, and M. Fleischer. "Optical properties of gallium oxide thin films". In: *Applied Physics Letters* 81.2 (2002), pp. 250–252. DOI: 10.1063/1.1491613.
- [209] M. Schubert, R. Korlacki, S. Knight, T. Hofmann, S. Schöche, V. Darakchieva, E. Janzén, B. Monemar, D. Gogova, Q.-T. Thieu, R. Togashi, H. Murakami, Y. Kumagai, K. Goto, A. Kuramata, S. Yamakoshi, and M. Higashiwaki. "Anisotropy, phonon modes, and free charge carrier parameters in monoclinic β -gallium oxide single crystals". In: *Physical Review B* 93.12 (2016). DOI: 10.1103/PhysRevB.93. 125209.
- [210] A. Jayawardena, A. C. Ahyi, and S. Dhar. "Analysis of temperature dependent forward characteristics of Ni/ $(\overline{2}01) \beta$ -Ga₂O₃ Schottky diodes". In: *Semiconductor Science and Technology* 31.11 (2016), Art. no. 115002. DOI: 10.1088/0268-1242/31/11/115002.
- [211] D. Peters, K. O. Dohnke, C. Hecht, and D. Stephani. "1700 V SiC Schottky diodes scaled to 25 A". In: *Materials Science Forum* 353-356 (2001), pp. 675–678. DOI: 10.4028/www.scientific.net/MSF.353-356.675.
- [212] T. Moule, S. Dalcanale, A. S. Kumar, M. J. Uren, W. Li, K. Nomoto, D. Jena, H. G. Xing, and M. Kuball. "Breakdown mechanisms in β -Ga₂O₃ trench-MOS Schottkybarrier diodes". In: *IEEE Transactions on Electron Devices* 69.1 (2022), pp. 75–81. DOI: 10.1109/TED.2021.3130861.

- [213] C. Buttay, C. Raynaud, H. Morel, G. Civrac, M.-L. Locatelli, and F. Morel. "Thermal stability of silicon carbide power diodes". In: *IEEE Transactions on Electron Devices* 59.3 (2012), pp. 761–769. DOI: 10.1109/TED.2011.2181390.
- [214] J. Lutz, H. Schlangenotto, U. Scheuermann, and R. de Doncker. "Modeling Parameters of Carrier Mobilities in Si and 4H-SiC". In: *Semiconductor Power Devices*. Cham, Switzerland: Springer, 2018, pp. 689–690. DOI: 10.1007/978-3-319-70917-8.
- [215] L. Cheng, J.-Y. Yang, and W. Zheng. "Bandgap, mobility, dielectric constant, and Baliga's figure of merit of 4H-SiC, GaN, and β -Ga₂O₃ from 300 to 620 K". In: *ACS Applied Electronic Materials* 4.8 (2022), pp. 4140–4145. DOI: 10.1021/acsaelm.2c00766.
- [216] A. Fiedler. " β -Ga₂O₃ Layers: Results and Discussion". In: *Electrical and optical characterization of* β -Ga₂O₃. Ph.D. Thesis. Berlin, Germany: Humboldt-Universität zu Berlin, 2020. DOI: 10.18452/20958.
- [217] A. E. Islam, N. P. Sepelak, K. J. Liddy, R. Kahler, D. M. Dryden, J. Williams, H. Lee, K. Gann, A. Popp, K. D. Leedy, N. S. Hendricks, J. L. Brown, E. R. Heller, W. Wang, W. Zhu, M. O. Thompson, K. D. Chabak, and A. J. Green. "500 °C operation of β -Ga₂O₃ field-effect transistors". In: *Applied Physics Letters* 121.24 (2022), Art. no. 243501. DOI: 10.1063/5.0113744.
- [218] "Properties of WCu, MoCu, and Cu/MoCu/Cu High-performance Heat Sink Materials and Manufacturing Technologies". In: *Advanced Thermal Management Materials*. Ed. by G. Jiang, L. Diao, and K. Kuang. New York, USA: Springer New York, 2013, pp. 73–87. DOI: 10.1007/978-1-4614-1963-1_5.
- [219] A. Vladimirescu. The SPICE book. New York, NY: Wiley, 1994.
- [220] Krzysztof Goecki, Janusz Zarebski, Damian Bisewski, and Jacek Dabrowski. "Nonlinear compact thermal model of SiC power semiconductor devices". In: Proc. 17th Intl. Conf. Mixed Design Integr. Circ. Syst. (MIXDES). Wroclaw, Poland, 2010, pp. 365–370.
- [221] J. Zarebski, K. Gorecki, and J. Dabrowski. "Modeling SiC MPS diodes". In: *Proc.* 20th Intl. Conf. on Microelectron. (ICM). Sharjah, UAE, 2008, pp. 192–195. DOI: 10.1109/ICM.2008.5393829.
- [222] M. R. Marks, Z. Hassan, and K. Y. Cheong. "Characterization methods for ultrathin wafer and die quality: a review". In: *IEEE Transactions on Components, Packaging* and Manufacturing Technology 4.12 (2014), pp. 2042–2057. DOI: 10.1109/ TCPMT.2014.2363570.
- [223] B. Albano, B. Wang, Y. Zhang, and C. Dimarino. "Electro-thermal device-package co-design for ultra-wide bandgap gallium oxide power devices". In: *Proc. IEEE Energy Conv. Cong. and Expo. (ECCE)*. 2022 IEEE Energy Conversion Congress and Exposition (ECCE). Detroit, USA, Oct. 2022, pp. 1–7. DOI: 10.1109/ ECCE50734.2022.9948059.
- [224] C. Bödeker, E. Ayerbe, and N. Kaminski. "Impact of a Kelvin source connection on discrete high power SiC-MOSFETs". In: *Materials Science Forum* 924 (2018), pp. 723–726. DOI: 10.4028/www.scientific.net/msf.924.723.

- [225] F. C. Lee, P. Barbosa, P. Xu, J. Zhang, B. Yang, and F. Canales. "Topologies and design considerations for distributed power system applications". In: *Proceedings of the IEEE* 89.6 (2001), pp. 939–950. DOI: 10.1109/5.931492.
- [226] Y. Jang and M. M. Jovanovic. "Light-load efficiency optimization method". In: *IEEE Transactions on Power Electronics* 25.1 (2010), pp. 67–74. DOI: 10.1109/TPEL. 2009.2024419.
- [227] Z. Tong, G. Zulauf, J. Xu, J. D. Plummer, and J. Rivas-Davila. "Output capacitance loss characterization of silicon carbide Schottky diodes". In: *IEEE Journal of Emerging and Selected Topics in Power Electronics* 7.2 (2019), pp. 865–878. DOI: 10.1109/JESTPE.2019.2904290.
- [228] C. N. M. Ho, F. Canales, A. Coccia, and M. Laitinen. "A circuit-level analytical study on switching behaviors of SiC diode at basic cell for power converters". In: *Proc.* 2008 IEEE Ind. Appl. Soc. Annu. Meeting. Edmonton, Canada, Oct. 2008, pp. 1–8. DOI: 10.1109/08IAS.2008.367.
- [229] I. Yonenaga, T. Hoshi, and A. Usui. "High temperature hardness of bulk single crystal GaN". In: *MRS Internet Journal of Nitride Semiconductor Research* 5.S1 (2000), pp. 343–348. DOI: 10.1557/S1092578300004488.
- [230] A. Ben Amar, M. Faucher, V. Brandli, Y. Cordier, and D. Théron. "Young modulus extraction of epitaxial heterostructure AlGaN/GaN for MEMS application". In: *physica status solidi (a)* 211.7 (2014), pp. 1655–1659. DOI: 10.1002/pssa. 201330339.

Online sources

- [231] Tamura Corp. Single-crystal gallium oxide (Ga₂O₃) substrates. Document number SCS-C-000002-D. URL: https://www.tamura-ss.co.jp/en/gao/pdf/ 150210.pdf (visited on 05/28/2023).
- [232] Novel Crystal Technology, Inc. World's first successful epitaxial deposition of gallium oxide on a 6-inch wafer using the HVPE method. Press release, 01. March 2022. URL: https://www.novelcrystal.co.jp/eng/2022/1107/ (visited on 05/28/2023).
- [233] Novel Crystal Technology, Inc. *World's first inverted gallium oxide DI-MOS transistor*. Press release, 27. September 2022. URL: https://www.novelcrystal.co. jp/eng/2022/1324/ (visited on 05/28/2023).
- [234] ANSYS, Inc. Ansys Q3D Extractor. Version 2020-R1. URL: https://www.ansys.com/de-de/products/electronics/ansys-q3d-extractor (visited on 05/28/2023).
- [235] CREE. CPW2-0600S010 Silicon Carbide Schottky Diode Chip Z-Rec Rectifier. Datasheet Rev. B. Feb. 2017. URL: https://assets.wolfspeed.com/ uploads/2020/12/CPW2-0650S010.pdf (visited on 05/28/2023).
- [236] STMicroelectronics. STPSC16H065A 650 V power Schottky silicon carbide rectifier. Datasheet DS12768 Rev. 1. Oct. 2018. URL: https://www.st.com/ resource/en/datasheet/stpsc16h065a.pdf (visited on 05/28/2023).
- [237] IXYS. DHG20C600PB Sonic Fast Recovery Diode. Datasheet (preliminary). Feb. 2020. URL: https://ixapps.ixys.com/DataSheet/DHG20C600PB. pdf (visited on 05/28/2023).

- [238] Subhas chandra Bose, J. V., I. Imrie, H. Ostmann, and P. Ingram. Sonic A New Generation of Fast Recovery Diodes. IXYS. URL: https://ixapps.ixys. com/DataSheet/4592a0fc-7250-493a-a3e4-c5c1c69dc02e.pdf (visited on 05/28/2023).
- [239] Keysight Technologies Japan. Keysight Technologies B1505A Power Device Analyzer / Curve Tracer. Manual Part Number B1505-90000. Edition 15, Feb. 2021. URL: https://www.keysight.com/us/en/assets/9018-01849/usermanuals/9018-01849.pdf?success=true (visited on 05/28/2023).
- [240] Infineon. IDK04G65C5 650 V SiC Schottky Diode. Datasheet Rev. 2.1. Aug. 2017. URL: https://www.infineon.com/dgdl/Infineon-IDK04G65C5-DS-v02_01-EN.pdf?fileId=db3a304342e8be2c0142eb938e7500a4 (visited on 05/28/2023).
- [241] TM Research Products, Inc. SDN-414-05 current viewing resistor: datasheet specifications. URL: https://www.tandmresearch.com/index.php?page=products (visited on 05/28/2023).
- [242] Fraunhofer Institute and ipmdesign. *High Speed Current Measurement for Ultra Low Inductance Power Modules*. URL: http://www.ipmdesign.de/ (visited on 05/28/2023).
- [243] Power Electronic Measurements Ltd. CWT Ultra mini: datasheet specifications. Feb. 2020. URL: https://www.pemuk.com/products/cwt-currentprobe/cwt-ultra-mini.aspx (visited on 05/28/2023).
- [244] GaN Systems, Inc. Application brief 08: GaN switching loss simulation using LTspice. 20. July 2021. URL: https://gansystems.com/wp-content/uploads/ 2018/05/GN008-GaN_Switching_Loss_Simulation_LTspice_ 20180523.pdf (visited on 05/28/2023).
- [245] Mentor Graphics Corp. Thermal Transient Results Post-Processing Tool. Version V 3.0.0.7109. URL: https://plm.sw.siemens.com/en-US/simcenter/ physical-testing/t3ster/ (visited on 05/28/2023).
- [246] Novel Crystal Technology, Inc. *High-quality 100-mm beta-type gallium oxide epitaxial wafers have been developed.* Press release, 16. June 2021. URL: https: //www.novelcrystal.co.jp/eng/2021/887/ (visited on 05/28/2023).
- [247] Novel Crystal Technology, Inc. Third-generation gallium-oxide 100-mm epitaxial wafer with ten times fewer killer defects. Press release, 14. March 2022. URL: https://www.novelcrystal.co.jp/eng/2022/1162/ (visited on 05/28/2023).
- [248] Siemens AG. "Parametric Study Examples". In: *Simcenter FLOEFD for Creo Tutorial for Software Version 2021.2* (July 2021), pp. 493–524.

Student work supervised by the author

[249] T. Maier. Konzepte zur Charakterisierung neuartiger breitbandiger Halbleiter. Project Thesis Report. Baden-Wuerttemberg Cooperative State University Ravensburg (Friedrichshafen, Germany) and ZF Friedrichshafen AG (Friedrichshafen, Germany). 04/2021 -09/2021.

- [250] J. Bücheler. Characterization and Modeling of Semiconductor Power Devices. Internship semester. University of Applied Sciences Ravensburg-Weingarten (Ravensburg, Germany) and ZF Friedrichshafen AG (Friedrichshafen, Germany). 09/2021–02/2022.
- [251] S. Heydt. Characterization and Modeling of Semiconductor Power Devices. Internship semester. University of Applied Sciences Ravensburg-Weingarten (Ravensburg, Germany) and ZF Friedrichshafen AG (Friedrichshafen, Germany). 03/2022–08/2022.
- [252] P. Bulander. Simulation und Entwicklung neuartiger Ga₂O₃ Halbleiter in der Leistungselektronik. Internship Semester Report. University of Applied Sciences Ravensburg-Weingarten (Ravensburg, Germany) and ZF Friedrichshafen AG (Friedrichshafen, Germany). 08/2022–01/2023.

Publications of the author

- [253] F. Wilhelmi, S. Kunori, K. Sasaki, A. Kuramata, Y. Komatsu, and A. Lindemann. "Temperature-dependent electrical characteristics of a β–Ga₂O₃ Schottky barrier diode". In: *Proc. Intl. Exhib. and Conf. for Power Electron., Intell. Mot., Renew. Energy and Energy Mgmt. Digit. Days (PCIM Europe)*. Nuremberg, Germany, 2021, pp. 1112–1118.
- [254] F. Wilhelmi, S. Kunori, K. Sasaki, A. Kuramata, Y. Komatsu, and A. Lindemann. "Packaged β -Ga₂O₃ trench MOS Schottky diode with nearly ideal junction properties". In: *IEEE Transactions on Power Electronics* 37.4 (2022), pp. 3737–3742. DOI: 10.1109/TPEL.2021.3122902.
- [255] F. Wilhelmi, Y. Komatsu, S. Yamaguchi, Y. Uchida, R. Nemoto, and A. Lindemann. "Packaged β -Ga₂O₃ Schottky diodes with reduced thermal resistance by substrate thinning to 200 µm". In: *Proc. 12th Intl. Conf. Integr. Power Electron. Syst. (CIPS)*. Berlin, Germany, 2022, pp. 180–185.
- [256] F. Wilhelmi, Y. Komatsu, S. Yamaguchi, Y. Uchida, R. Nemoto, and A. Lindemann.
 "Effect of substrate thinning and junction-side cooling on thermal properties of Ga₂O₃ diodes". In: *Proc. Int. Conf. Elect. Packag. (ICEP)*. Sapporo, Japan, 2022, pp. 145–146. DOI: 10.23919/ICEP55381.2022.9795473.
- [257] F. Wilhelmi, A. Schmid, and A. Lindemann. "Assessment of state-of-the-art current sensors for fast switching". In: *Proc. Intl. Exhib. and Conf. for Power Electron.*, *Intell. Mot., Renew. Energy & Energy Mgmt. (PCIM Europe)*. Nuremberg, Germany, 2022, pp. 1136–1145. DOI: 10.30420/565822158.
- [258] F. Wilhelmi, S. Heydt, Y. Uchida, S. Yamaguchi, S. Kunori, Y. Komatsu, and A. Lindemann. "Electrothermal Spice model for β -Ga₂O₃ Schottky barrier diodes". In: *Proc. 4th Intl. Workshop on Gallium Oxide and Related Materials*. Nagano, Japan, 2022, pp. 316–317.
- [259] F. Wilhelmi, Y. Komatsu, S. Yamaguchi, Y. Uchida, T. Kase, S. Kunori, and A. Lindemann. "Improving the heat dissipation and current rating of Ga₂O₃ Schottky diodes by substrate thinning and junction-side cooling". In: *IEEE Transactions on Power Electronics* 38.6 (2023), pp. 7107–7117. DOI: 10.1109/TPEL.2023. 3250026.
- [260] F. Wilhelmi, Y. Komatsu, S. Yamaguchi, Y. Uchida, T. Kase, S. Kunori, and A. Lindemann. "Switching properties of 600 V Ga₂O₃ diodes with different chip sizes and thicknesses". In: *IEEE Transactions on Power Electronics* 38.7 (2023), pp. 8406–8418. DOI: 10.1109/TPEL.2023.3260023.

Res Electricae Magdeburgenses



MAgdeburger FOrum zur Elektrotechnik

- MAFO-01 ORTHS, A.: Multikriterielle, optimale Planung von Verteilungsnetzen im liberalisierten Energiemarkt unter Verwendung von spieltheoretischen Verfahren, 2003. ISBN 3-929757-57-5.
- MAFO-02 PURMANN, M.: Optimierung des Betriebsverhaltens von PEM-Brennstoffzellen unter Berücksichtigung von elektrischem und Gesamtwirkungsgrad bei unterschiedlichen Lastanforderungen und Betriebsparametern, 2004. ISBN 3-929757-63-X.
- MAFO-03 AL-HAMID, M.: Extraktion von höheren Moden in TEM-Wellenleitern, 2004. ISBN 3-929757-64-8.
- MAFO-04 HAASE, H., J. NITSCH, and T. STEINMETZ: *Transmission-Line Super Theory A new Approach to an Effective Calculation of Electromagnetic Interference*, 2004. ISBN 3-929757-67-2.
- **MAFO-05** BACHRY, A.: Power Quality Studies in Distribution Systems Involving Spectral Decomposition, 2004. ISBN 3-929757-68-0.
- **MAFO-06** STYCZYNSKI, Z. A.: Power Network and Renewables A Scientific Report, 2004. ISBN 3-929757-69-9.
- **MAFO-07** BLUME, E.: Numerische Analyse der Kopplung linearer Antennen innerhalb eines Resonators, 2004. ISBN 3-929757-71-0.
- MAFO-08 HANDSCHIN, E. and Z. A. STYCZYNSKI: Power System Application of the Modern Battery Storage, 2004. ISBN 3-929757-75-3.
- MAFO-09 HAASE, H.: Full-Wave Field Interactions of Nonuniform Transmission Lines, 2005. ISBN 3-929757-78-8.
- **MAFO-10** NITSCH, D.: Die Wirkung eingekoppelter ultrabreitbandiger elektromagnetischer Impulse auf komplexe elektronische Systeme, 2005. ISBN 3-929757-79-6.
- MAFO-11 HADZI-KOSTOVA, B.: Protection Concepts in Distribution Networks with Decentralized Energy Resources, 2005. ISBN 3-929757-84-2.
- **MAFO-12** STEINMETZ, T.: Ungleichförmige und zufällig geführte Mehrfachleitungen in komplexen technischen Systemen, 2006. Nummerierung geändert, ISBN 3-929757-98-2.
- MAFO-13 STYCZYNSKI, Z. and J. HAUBROCK: Influence of Distributed and Renewable Generation on Power System Security – Proceedings of the CRIS Workshop 2006, 2006. ISBN 3-929757-99-0.
- MAFO-14 HEIDECK, G.: Ein autonomes Brennstoffzellensystem: Optimierungsansätze, 2006. ISBN 3-929757-94-X.
- MAFO-15 STYCZYNSKI, Z. und H.-D. MUSIKOWSKI: Dresdener Kreis Elektroenergieversorgung 7, 2006. ISBN 3-929757-85-0.
- **MAFO-16** GRONWALD, F.: Antenna Theory in Resonating Systems derived from Fundamental Electromagnetism, 2007. ISBN 3-929757-93-1.
- MAFO-17 KRAUTHÄUSER, H.G.: Grundlagen und Anwendungen von Modenverwirbelungskammern, 2007. ISBN 978-3-929757-43-9.

- MAFO-18 DZIENIS, C.: Ersatzmodelle nichtlinearer Lasten in elektrischen Verteilungsnetzen, 2007. ISBN 978-3-929757-07-1.
- MAFO-19 STYCZYNSKI, Z. and J. HAUBROCK: Renewable and Dispersed Power Generation in Power Systems, 2007. ISBN 978-3-929757-44-6.
- **MAFO-20** HAUBROCK, J.: Parametrierung elektrischer Äquivalentschaltbilder von PEM-Brennstoffzellen, 2008. ISBN 978-3-940961-02-0.
- MAFO-21 ANGELOV, A. N.: Rechnergestütztes Lernen im Bereich der Regenerativen Energien (Ausgewählte Aspekte), 2008. ISBN 978-3-940961-03-7.
- **MAFO-22** KOMARNICKI, P.: Anwendung hochgenauer, synchroner Messungen zur Verbesserung des Betriebs von Verteilungsnetzen, 2008. ISBN 978-3-940961-04-4.
- MAFO-23 ROGGATZ, C.: Trainingssimulator für die Führung von elektrischen Systemen mit dezentralen Energieeinspeisungen – Trainingsszenarien und Umsetzung, 2008. ISBN 978-3-940961-05-1.
- MAFO-24 RUDION, K.: Aggregated Modelling of Wind Farms, 2008. ISBN 978-3-940961-14-3.
- MAFO-25 GANJAVI, M. R.: Protection System Coordination Using Expert System, 2008. ISBN 978-3-940961-15-0.
- MAFO-26 BOFINGER, S.: Energieversorgungsnetze mit einem hohen Anteil an photovoltaischer Solarenergie: Standortbestimmung, Solarstromprognose, Netzintegration, 2008. ISBN 978-3-940961-25-9.
- MAFO-27 STYCZYNSKI, Z. and P. KOMARNICKI: Distributed and Renewable Power Generation, 2008. ISBN 978-3-940961-26-6.
- **MAFO-28** KOCHETOV, S. V.: Time- and Frequency-Domain Modeling of Passive Interconnection Structures in Field and Circuit Analysis, 2008. ISBN 978-3-940961-27-3.
- MAFO-29 MAGDOWSKI, M.: Entwicklung und Validierung eines Werkzeugs zur Berechnung der elektromagnetischen Einkopplung von stochastischen Feldern in Leitungsstrukturen, 2008. ISBN 978-3-940961-28-0.
- **MAFO-30** SONNEMANN, F.: Elektromagnetische Effekte an elektrischen Zündmitteln (EED) mit angeschlossener Zündkreiselektronik (ZKE) bei impulsförmiger, breitbandiger Bestrahlung, 2009. ISBN 978-3-940961-32-7.
- **MAFO-31** SMIEJA, T.: Multikriterielle Planung interregionaler Elektrizitätsnetze im liberalisierten Energiemarkt, 2009. ISBN 978-3-940961-35-8.
- MAFO-32 HEYDE, C. O.: Dynamic Voltage Security Assessment for On-Line Control Room Application, 2010. ISBN 978-3-940961-40-2.
- MAFO-33 STYCZYNSKI, Z. A. and N. I. VOROPAI (EDITORS): Renewable Energy Systems Fundamentals, Technologies, Techniques and Economics, 2010. ISBN 978-3-940961-42-6.
- MAFO-34 Styczynski, Z. A. & N. I. Voropai (Editors): Renewable Energy Systems Fundamentals, Technologies, Techniques and Economics (Russian Version), 2010. ISBN 978-3-940961-44-0.
- MAFO-35 STYCZYNSKI, Z. A. and A. LINDEMANN (EDITORS): Integration of Renewable Energies into the Grid / Proceedings of the Power & Energy Student Summit 2010, 2010. ISBN 978-3-940961-47-1.

- MAFO-36 STYCZYNSKI, Z. A. und H.-D. MUSIKOWSKI (EDITORS): Dresdener Kreis Elektroenergieversorgung 11, 2010. ISBN 978-3-940961-51-8.
- **MAFO-37** GURBIEL, M. A.: Definition and Testing of a Digital Interface of a Power Substation, 2011. ISBN 978-3-940961-54-9.
- MAFO-38 LOMBARDI, P.: Multi Criteria Optimization of an Autonomous Virtual Power Plant, 2011. ISBN 978-3-940961-55-6.
- MAFO-39 POWALKO, M.: Beobachtbarkeit eines elektrischen Verteilungsnetzes Ein Beitrag zum Smart Grid, 2011. ISBN 978-3-940961-62-4.
- MAFO-40 STYCZYNSKI, Z. A., K. RUDION, and C. NGUYEN-MAU (EDITORIAL BOARD): *Power System Dynamic Security Assessment*, 2011. ISBN 978-3-940961-61-7.
- **MAFO-41** KÄBISCH, M.: Optimale Dimensionierung und Betriebsführung einer brennstoffzellenbasierten Auxiliary Power Unit im Fahrzeug, 2011. ISBN 978-3-940961-67-9.
- MAFO-42 STYCZYNSKI, Z. A. und N. I. VOROPAI (EDITORS): Special Issue Grant 220 Russian Federation SSmart Grid for Efficient Energy Power System for the Future", Proceedings Volume I, 2012. ISBN 978-3-940961-74-7.
- MAFO-43 STYCZYNSKI, Z. A., P. KOMARNICKI und A. NAUMANN (EDITORS): Abschlussbericht Harz.ErneuerbareEnergien-mobility, 2012. ISBN 978-3-940961-71-6.
- **MAFO-44** HEUER, M.: Diagnosetool für stationär betriebene PEM-Brennstoffzellensysteme, 2012. ISBN 978-3-940961-77-8.
- MAFO-45 STÖTZER, M.: Demand Side Integration in elektrischen Verteilnetzen Potenzialanalyse und Bewertung, 2012. ISBN 978-3-940961-78-5.
- **MAFO-46** MAGDOWSKI, M.: Vergleich der Einkopplung deterministischer und statistischer elektromagnetischer Felder in Leitungen, 2012. ISBN 978-3-940961-75-4.
- MAFO-47 NAUMANN, A.: Leitwarte im Smart Grid, 2012. ISBN 978-3-940961-81-5.
- MAFO-48 RUDION, K.: Offshore Power System Planning Selected Aspects, 2012. ISBN 978-3-940961-82-2.
- MAFO-49 NGUYEN-MAU, C.: Electric Power System Stability Enhancement by Voltage Source Converter based High Voltage Direct Current Technology, 2012. ISBN 978-3-940961-84-6.
- MAFO-50 GUO, H.: Measurement-Based Load Modeling for Smart Grid Planning, 2012. ISBN 978-3-940961-86-0.
- MAFO-51 STYCZYNSKI, Z. A.: Proceedings No. 2 in the Scope of Mega Grant Baikal, 2013. ISBN 978-3-940961-95-2.
- MAFO-52 STYCZYNSKI, Z. A.: Proceedings No. 3 in the Scope of Mega Grant Baikal, 2013. ISBN 978-3-940961-98-3.
- MAFO-53 WENGE, C.: Optimaler Betrieb von mobilen Speichern im Smart Grid. -Mobilitätsleitwarte-, 2013. ISBN 978-3-944722-01-6.
- **MAFO-54** Röhrig, C.: Smart Distribution Planung unter Berücksichtigung von residualen Lasten, 2014. ISBN 978-3-944722-06-1.
- MAFO-55 MIDDELSTÄDT, F.: Research of SEM Poles of Complex Wire Structures, 2014. ISBN 978-3-944722-07-8.

- **MAFO-56** STYCZYNSKI, Z. A. (EDITOR): Power Network and Renewables A Scientific Report 15 Years Research, 2014. ISBN 978-3-944722-08-5.
- MAFO-57 SOLONINA, N. N., V. S. STEPANOV, and K. V. SUSLOV: Information technology in *intelligent power networks*, 2014. ISBN 978-3-944722-13-9.
- MAFO-58 KREBS, R.: Fundamentals of Power System Protection, 2014. ISBN 978-3-944722-15-3.
- MAFO-59 MOSKALENKO, N.: Optimal Dynamic Energy Management System in Smart Homes, 2014. ISBN 978-3-944722-16-0.
- **MAFO-60** HAUER, I.: Optimale Last- und Erzeugungsanpassung bei kritischen Netzzuständen Algorithmen und deren Bewertung, 2014. ISBN 978-3-944722-18-4.
- MAFO-61 RICHTER, M.: Dresdner Kreis Elektroenergieversorgung 15, Begleitband zum Workshop 2014, 2014. ISBN 978-3-944722-21-4.
- MAFO-62 STYCZYNSKI, Z. A., A. RICHTER und P. KÜHNE: Second ELECON Workshop Begleitband zum Workshop 2014, 2014. ISBN 978-3-944722-23-8.
- **MAFO-63** BERNSTEIN, P.: Modellgestützte optimale Betriebsführung von PEM-Brennstoffzellen für autonome Anlagen, 2015. ISBN 978-3-944722-24-5.
- **MAFO-64** RABE, S.: Betrieb einer Zweipunkt-Offshore-HGÜ-Verbindung Modelluntersuchungen, 2015. ISBN 978-3-944722-31-3.
- MAFO-65 ARENDARSKI, B.: Reliability Assessment of Smart Grids, 2015. ISBN 978-3-944722-32-0.
- **MAFO-66** PAN, X.: Numerisches EMV-Simulationsverfahren zur Berechnung der Störaussendung elektrischer Antriebssysteme in Hybridfahrzeugen, 2016. ISBN 978-3-944722-34-4.
- MAFO-67 RICHTER, M.: PMU-basierte Zustandsabschätzung in Smart Distribution, 2016. ISBN 978-3-944722-43-6.
- MAFO-68 BIELCHEV, I.: Adaptiver Distanzschutz im Standard IEC 61850, 2016. ISBN 978-3-944722-45-0.
- **MAFO-69** HASSAN, A.: Modeling of Single and Double-Shielded Cables for EMC Applications, 2016. ISBN 978-3-944722-41-2.
- **MAFO-70** LIU, X.: Control of Voltage Source Converter Based High Voltage Direct Current Transmission Systems for Grid Code Compliance, 2016. ISBN 978-3-944722-46-7.
- MAFO-71 KAISER, M.: Fusion of Interventional Ultrasound & X-ray, 2016. ISBN 978-3-944722-51-1.
- **MAFO-72** GRIEGER, F.: Ein Beitrag zur Bestimmung der Zuverlässigkeit von Leistungshalbleiterbauelementen unter Berücksichtigung der Anwendung, 2016. ISBN 978-3-944722-52-8.
- MAFO-73 L1, M.: Towards a Robust Electromagnetic Tracking System for Use in Medical Applications, 2018. ISBN 978-3-944722-66-5.
- MAFO-74 KLABUNDE, C., J. DANCKER, N. GAST, T. SCHRÖTER, F. SCHULZ, J. ROSSBERG und A. RICHTER: Statusbericht der Otto-von-Guericke-Universität Magdeburg zum Verbundprojekt: Intelligentes Multi-Energie-System (SmartMES), 2018. ISBN 978-3-944722-69-6.

- **MAFO-75** SCHRÖTER, T.: 19. Dresdner Kreis Elektroenergieversorgung, Begleitband zum Workshop 2018 in Magdeburg, 2018. ISBN 978-3-944722-79-5.
- MAFO-76 KLABUNDE, C., J. DANCKER, N. GAST, T. SCHRÖTER, F. SCHULZ und J. ROSS-BERG: Intelligentes Multi-Energie-System (SmartMES) – Statusbericht der Otto-von-Guericke-Universität Magdeburg zum Verbundprojekt, 2. Statusseminar 04. April 2019 in Magdeburg, 2019. ISBN 987-3-944722-80-1.
- MAFO-77 HELM, S., J. DANCKER, M. FRITSCH und T. SCHRÖTER: Power and Energy Student Summit 2019, 09.–11. Juli, 2019. ISBN 978-3-944722-84-9.
- MAFO-78 CHYCHYKINA, I.: Comparison of Different Redispatch Optimization Strategies, 2019. ISBN 978-3-944722-89-4.
- MAFO-79 BALISCHEWSKI, S.: Multifunktionaler Einsatz von Batteriespeichern in elektrischen Verteilnetzen, 2020. ISBN 978-3-944722-92-4.
- **MAFO-80** PETZOLD, J.: Analytische Beschreibung der Kopplung elektromagnetischer Felder durch Aperturen in Resonatoren, 2020. ISBN 978-3-944722-91-7.
- MAFO-81 MIDDELSTÄDT, L.: Transiente Effekte in leistungselektronischen Schaltungen mit schnellschaltenden Leistungshalbleitern unter besonderer Berücksichtigung der elektromagnetischen Verträglichkeit, 2020. ISBN 978-3-944722-95-5.
- MAFO-82 LIU, Y.: Contribution to Improve the EMI Performance of Electrical Drive Systems in Vehicles With Special Consideration of Power Semiconductor Modules, 2021. ISBN 978-3-948749-01-9.
- MAFO-83 GERLACH, A.: Regelung von direktangetriebenen elektrischen Maschinen für Verbrennungsmotoren, 2021. ISBN 978-3-948749-03-3.
- MAFO-84 ZHANG, Y.: Analysis and Control of Resonances in HVDC Connected DFIG-Based Offshore Wind Farms, 2021. ISBN 978-3-948749-05-7.
- **MAFO-85** PANNICKE, E.: Empfangsspulen für bildgeführte Eingriffe mittels Magnetresonanztomographie, 2021. ISBN 978-3-948749-12-5.
- **MAFO-86** KASPER, J.: Analysis of the Stochastic Electromagnetic Field Coupling to Single and Multiconductor Transmission Line Structures, 2021. ISBN 978-3-948749-13-2.
- MAFO-87 HAUER, I.: Abschlussbericht zum Projekt InKola: Infrastrukturkopplung Platzierung und Betrieb von Ladestationen aus Verkehrs- und Energienetzsicht, 2022. ISBN 978-3-948749-15-6.
- MAFO-88 RAYA, M.: Circuit Models of Shielded Single and Multiconductor Cables for EMC Analyses, 2022. ISBN 978-3-948749-16-3.
- **MAFO-89** PRIBAHSNIK, F.: GaN-Specific Mechanical Phenomena and Their Influence on Reliability in Power HEMT Operation, 2022. ISBN 978-3-948749-17-0.
- **MAFO-90** WILLMANN, B.: Elektromagnetische Umweltverträglichkeit eines Elektrofahrzeugs mit kontaktlosem Ladesystem, 2022. ISBN 978-3-948749-19-4.
- MAFO-91 DANCKER, J.: Sensitivity Factors for Integrated Energy Systems: A Joined Quasi-Steady-State Approach, 2022. ISBN 978-3-948749-24-8.
- **MAFO-92** TAYYAB, M.: Holistic Approach for Microgrid Planning and Operation for E-Mobility Infrastructure Under Consideration of Multi-Type Uncertainties, 2022. ISBN 978-3-948749-25-5.

- MAFO-93 RICHTER, A.: Virtuelle Kraftwerke im Verteilnetz Systemstützender Betrieb im wirtschaftlichen Kontext – Eine gesamtheitliche Betrachtung virtueller Kraftwerke, 2022. ISBN 978-3-948749-28-6.
- **MAFO-94** WOLDU, T.: Modeling and Simulation of Power System Dynamics for Studying the Impacts of Increasing Wind Power in a Weak Grid System, 2022. ISBN 978-3-948749-31-6.
- **MAFO-95** KÖNNEKE, N.: Entwurf eines Testbetts für Assistenzsysteme in der Netz- und Systemführung, 2023. ISBN 978-3-948749-32-3.
- MAFO-96 GLENDE, E. und M. GEBHARDT: 23. Dresdner Kreis Elektroenergieversorgung, Begleitband zum Workshop 2022 in Magdeburg, 2022. ISBN 978-3-948749-33-0.
- **MAFO-97** KEMPIAK, C.: Lastwechselmethoden für Siliziumkarbid-MOSFETs unter Berücksichtigung von deren Schwellspannungsinstabilität, 2023. ISBN 978-3-948749-38-5.
- MAFO-98 FRITSCH, M.: Measurement of Partial Discharges on Power Cables: A Step Towards Successful Online Monitoring, 2024. ISBN 978-3-948749-43-9.
- MAFO-99 MIDDELSTÄDT, F.: Electromagnetic Field Coupling to Thin Wires Above a Ground, 2024. ISBN 978-3-948749-43-9.
- **MAFO-100** HELM, S.: Optimiertes Lademanagement von E-Kfz zur Spannungsstabilisierung im Niederspannungsnetz, 2024. ISBN 978-3-948749-48-4.
- **MAFO-101** WILHELMI, F.: On the characteristic electrical and thermal properties of beta-phase gallium oxide (β -Ga₂O₃) Schottky diodes in view of their potential application in power electronics, 2024. ISBN 978-3-948749-49-1.